

Systems Improved Numerical Differencing Analyzer Engineering-Program Manual

T. Ishimoto and L. C. Fink

June 1971

NASA Contract 9-10435

Prepared for: National Aeronautics and Space Administration Manned Spacecraft Center Under Contract NASA 9-10435

Prepared by:

V Fink

1. U. F1

Approved by:

RALZOUTE.

R. G. Payne, Program Manager

Approved by:

Bevans, Manager J

Heat Transfer & Thermodynamics Department

Approved by:

2. X. L. t

R. L. Dotts NASA Technical Monitor NASA Manned Spacecraft Center

ACKNOWLEDGEMENT

This engineering-program manual includes the efforts of several individuals who in no small way contributed to the publication of this document. Many thanks are due to:

J. D. Gaski, who as the prime mover behind the SINDAS (SINDA, CINDA-3G and CINDA, provided valuable counseling on the inner workings of the program, expecially the execution routines. Many of the computational features in these routines are original with him; it is through his cooperative assistance that the writing of this document became a reasonable venture.

<u>R. L. Dotts</u>, who must be given special mention not because of his position as NASA/MSC technical monitor but because of his sincere interest in the use of SINDA and his willingness to provide assistance whenever and wherever possible.

<u>Mrs. Dorothy Gramlich</u>, who typed this manuscript with a cheerful, professional attitude and technique that made this phase of the program a more pleasant one.

TABLE OF CONTENTS

	ACKNOWLEDGEMENT	Page 111
	PREFACE	xiii
1.	NOMENCLATURE AND MNEMONICS	1-1
1.1	Nomenclature	11
1.2	Mnemonics	1-2
1.2.1	Control Constants	88
1.2.2	Numerical Solution Routines	1-4
1.2.3	Options	. 0
1.2.4	Routines and Subroutines of Preprocessor	1-5
1.2.5	Others	1-7
2.	BACKGROUND ON SINDA	2-1
3.	GENERAL SINDA PROGRAM DESCRIPTION	3-1
3.1	SINDA Operating System	11
3.2	Use of Lumped-Parameter Concept	
3.3	Pseudo-Compute Sequence (PCS)	3-2
3.3.1	Long Pseudo-Compute Sequence (LPCS)	3-5
3.3.2	Short Pseudo-Compute Sequence (SPCS)	3-6
3.3.3	Second Long Pseudo-Compute Sequence (LPCS2)	3-8
3.3.4	Pseudo-Compute Sequence One (PCS1) and Pseudo-Compute Sequence Two (PCS2)	
3.4	Data Logistics	39
3.4.1	Relative Numbers	.11
3.4.2	Storage Requirements and Dynamic Storage Allocation	21
3.5	Order of Computation	3-10
4.	PREPROCESSOR	4-1
4.1	General Description	91
4.2	Description of Subroutines	4-5
4.2.1	Routine Name: SINDA	4-6
4.2.2	Routine Name: PREPRØ	4-7
4.2.3	Subroutine Name: ALPINT	4-8
4.2.4	Subroutine Name: BLKCRD	4-9
4.2.5	Subroutine Name CØDERD	4-10

.

		Page
4.2.6	Subroutine Name: CØNVRT	4-11
4.2.7	Subroutine Name: DATARD	4-12
4.2.8	Subroutine Name: ERRMES	4-13
4.2.9	Subroutine Name: FINDRM	4-14
4.2.10	Subroutine Name: GENLNK	4-15
4.2.11	Subroutine Name: GENUK	4-16
4.2.12	Subroutine Name: INCØRE	4-17
4.2.13	Subroutine Name: MXTØFN	4-18
4.2.14	Subroutine Name: NØDEDA	4-19
4.2.15	Subroutine Name: PCS2	4-20
4.2.16	Subroutine Name: PRESUB	4-21
4.2.17	Subroutine Name: PSEUDØ	4-22
4.2.18	Subroutine Name: QDATA	4-23
4.2.19	Subroutine Name: RELACT	4-24
4.2.20	Subroutine Name: SEARCH	4-25
4.2.21	Subroutine Name: SETFMT	4-26
4.2.22	Subroutine Name: SINDA4	4-27
4.2.23	Subroutine Name: SKIP	4-28
4.2.24	Subroutine Name: SPLIT	4-29
4.2.25	Subroutine Name: SQUEEZ	4-30
4.2.26	Subroutine Name: STFFB	4-31
4.2.27	Subroutine Name: TYPCHK	4-32
4.2.28	Subroutine Name: WRTDTA	4-33
4.2.29	Subroutine Name: WRTPMT	4-34
4.2.30	Subroutine Name: WRTBLK	4-35
4.3	Labeled Common Variables	4-36
4.3.1	Labeled Common Map	**
4.3.2	Definition of Labeled Common Variables	 4-38
4.3.3	Dynamic Storage Structure	4-42
4.4	SINDA "Tapes" and Their Formats	4-46
4.4.1	LB3D - Program Data "Tape"	-11
4.4.2	LB4P - Program FØRTRAN "Tape"	4-47
4.4.3	INTERN - Preprocessor Scratch "Tape"	n
4.4.4	LUT1 - Dictionary "Tape"	11 _.
4.4.5	LUT3 - Parameter Runs "Tape"	4-48
	•	

.

.

.

V

			Page
	4.5	Overlay Structure	4-49
	4.6	Structure of Pseudo-Compute Sequences	4-51
	4.6.1	Descriptions	· 11
	4.6.2	Structure of PCS1	11
	4.6.3	Structure of PCS2	4-53
	4.7	Other Information	tt.
	4.7.1	Subroutine Lengths	11
	4.7.2	Maximum Thermal Problem Size and Maximum Data Value Size	11 s. •
5	ia	REVIEW OF LUMPED PARAMETER EQUATIONS AND BASIC NUMERICAL SOLUTIONS	5-1
	5.1	Lumped-Parameter Representation	5-2
	5.1.1	Some Thoughts on Lumped-Parameter Errors	5-3
	5.1.2	Lumped-Parameter Equations	5-5
	5.2	Basic Finite Difference Formulations	5-6
	5.2.1	Forward Finite Difference Explicit Method	5-8
3	5.2.2	Implicit Finite Difference Method	5-9
2	5.2.3	Steady State	5-11
•	5.2.4	Some Comments	, 91
6	•	SINDA NUMERICAL SOLUTION ROUTINES	6-1
	6.1	Objective and Presentation Format	11
	6.2	General Computational Procedure and Features	6-2
	6.2.1	Order of Computation	11
	6.2.1.1	Finite Difference Algorithm	H.
	6.2.1.2	Updating of Optionally Specified Properties	6-5
	6.2.2	Operations Blocks	u,
	6.2.2.1	EXECUTION Operations Block	6-9
	6.2.2.2	VARIABLES 1 Operations Block	11
	6.2.2.3	VARIABLES 2 Operations Block	11
	6.2.2.4	ØUTPUT CALL Operations Block	6-10
	6.2.3	Control Constants	11
	6.2.3.1	Alphabetical Listing and Brief Description of Control Constants	11
	6.2.3.2	User Specified and Optionally User Specified Control Constants	6-16

p	a	g	ė
4	a	Ö	-

.

6.2.4	Time-Step Calculations	6-22
6.2.5	Computation of Temperatures	6-25
6.2.5.1	Transient Explicit Routines	IJ
6.2.5.2	Transient Implicit Routines	6-29
6.2.5.3	Steady State Routines	6-34
6.2.6	Radiation Damping	6-37
6.2.7	Acceleration of Convergence by Extrapolation Technique	6-39
6.2.7.1	Extrapolation Technique	н
6.2.7.2	Programming Considerations	6-41
6.2.7.3	Routines Using Acceleration of Convergence	"
6.2.7.4	Comment on Acceleration of Convergence	11
6.2.8	Other Characteristics of the SINDA Numerical Solution Routines	6-42
6.2.8.1	Units	
6.2.8.2	General Comments on Computational Features	11
6.3	Transient Explicit Solution Routines	6-43
6.3.1	Subroutines: CNFRWD and CNFRDL	6-45
6.3.4.1	General Comments	38
6.3.4.2	Finite Difference Approximation and Computational Algorithm	6-46
6.3.1.3	Comments on the Computational Procedure	6-47
6.3.1.4	Control Constants	6-48
6.3.1.5	Error and Other Messages	6-49
6.3.2	Subroutine: CNFAST	6-55
6.3.2.1	General Comments	.11
6.3.2.2	Finite Difference Approximation and Computational Algorithm	11
6.3.2.3	Comments on the Computational Procedure	6-56
6.3.2.4	Control Constants	11
6.3.2.5	Error and Other Messages	6-57
6.3.3	Subroutine: CNEXPN	6-60
6.3.3.1	General Comments	11 -
6.3.3.2	Finite Difference Approximation and Computational Algorithm	ŦŦ .,
6.3.3.3	Comments on the Computational Procedure	6-61

		Page
6.3.3.4	Control Constants	6-62
6.3.3.5	Error and Other Messages	Û,
6.3.4	Subroutine: CNDUFR	6-66
6.3.4.1	General Comments	U.
6.3.4.2	Finite Difference Approximation and Computational Algorithm	11
6.3.4.3	Comments on the Computational Procedure	6-68
6.3.4.4	Control Constants	Ť.
6.3.4.5	Error and Other Messages	6-69
6.3.5	Subroutine: CNQUIK	6-72
6.3.5.1	Géneral Comments	Ų,
6.3.5.2	Finite Difference Approximation and Computational Algorithm	77
6.3.5.3	Comments on the Computational Procedure	6-73
6.3.5.4	Control Constants	6-74
6.3.5.5	Error and Other Messages	11
6.4	Transient Implicit Solution Routines	6-78
6.4.1	Subroutine: CNBACK	6-80
6.4.1.1	General Comments	88
6.4.1.2	Finite Difference Approximation and Computational Algorithm	28
6.4.1.3	Comments on the Computational Procedure	,1
6.4.1.4	Control Constants	6-82
6.4.1.5	Error and Other Messages	6-83
6.4.2	Subroutine: CNFWBK	6-86
6.4.2.1	General Comments	Ĥ
6.4.2.2	Finite Difference Approximation and Computational Algorithm	11
6.4.2.3	Comments on the Computational Procedure	6-88
6.4.2.4	Control Constants	6-89
6.4.2.5	Error and Other Messages	"
6.4.3	Subroutine: CNVARB	6-93
6.4.3.1	General Comments	11
6.4.3.2	Finite Difference Approximation and Computational Algorithm	
6.4.3.3	Comments on the Computational Procedure	6-95

)

Sector Sector

			Page
6.4.3.4	Control Constants		6-95
6.4.3.5	Error and Other Messages		11
6.5	Steady State Routines		6-98
6.5.1	Subroutine: CINDSS		6-101
6.5.1.1	General Comments		U
6.5.1.2	Finite Difference Approxim Algorithm	ation and Computational	6-102
6.5.1.3	Comments on the Computation	nal Procedure	6-103
6.5.1.4	Control Constants		6-104
6.5.1.5	Error and Other Messages		
6.5.2	Subroutine: CINDSL	•	6-107
6.5.2.1	General Comments	1000 1000 1000 1000	28
6.5.2.2	Finite Difference Approxim Algorithm	ation and Computational	17
6.5.2.3	Comments on the Computatio	nal Procedure	6-109
6.5.2.4	Control Constants		ET.
6.5.2.5	Error and Other Messages		11
6.5.3	Subroutine: CINDSM		6-113
6.5.3.1	General Comments		H .
6.5.3.2	Finite Difference Approxim Algorithm	ation and Computational	11
6.5.3.3	Comments on the Computatio	nal Procedure	6-114
6.5.3.4	Control Constants		6-115
6.5.3.5	Error and Other Messages		
7.	REFERENCES		
	APPENDICES		
Α.	Computer Listings of SINDA Routines	Explicit Solution	
	CNFRWD		A-2
	CNFRDL		A-13
	CNFAST		A-24
	CNEXPN		A-33
	CNDUFR		A-44
	CNQUIK		A-55

. .

)

•

жř.

بر کر ہے

ix

Computer Listings of SINDA Implicit S Routines	Solution
CNBACK	B-2
CNFWBK	B-14
CNVARB	B-26
Computer Listings of SINDA Steady Sta Routines	ate Solution
CINDSS	C-2
CINDSL	C-11
CINDSM	C-20

~

Page

FIGURES

Figure No.		Page
3.3-1	Thermal Circuit for a One-Dimensional System	3-4
4.1-1	SINDA Preprocessor - Major Logic and Interface with the User Specified Problem	4-2
6.2-1	General Order of Computation for Numerical Solution Routines	6-3
6.2-2	Computation Pattern for Numerically Solving Explicit and Implicit Finite Difference Algorithm	6–4
6.2-3	Evaluation of Nonlinear Capacitance, Source or Conductance	6-26
6.2-4	Method of Extrapolation to Accelerate Convergence	6-40
6.3-1	QSUM and GSUM for "Block" Diffusion Node Temperature Calculation, CNFRWD and CNFRDL	6-52
6.3-2	Calculation of Arithmetic-Node Temperature by "Successive Point" Iteration	6-53
6.3-3	Functional Flow Chart for CNFRWD and CNFRDL	6-54
6.3-4	Functional Flow Chart for CNFAST	6-59
6.3-5	Functional Flow Chart for CNEXPN	6-65
6.3-6	Functional Flow Chart for CNDUFR	6-71
6.3-7	Functional Flow Chart for CNQUIK	6-77
6.4-1	Functional Flow Chart for CNBACK	6-85
6.4-2	Functional Flow Chart for CNFWBK	6-92
6.4-3	Functional Flow Chart for CNVARB	6-98
6.5-1	Functional Flow Chart for CINDSS	6-106
6.5-2	Functional Flow Chart for CINDSL	6-112
6.5-3	Functional Flow Chart for CINDSM	6-118

xi

•...•.

. 5 .≛., ÷

TABLES
and an arrest day had be

Table No.		Page
3.2-1	Tabular Identification of Conductor and Adjoining Node Numbers	3-5
3.3-1	Example of Conductor Connections	3-7
3.3-2	Long Pseudo-Compute Sequence (LPCS) for the Example of Table (3.3-1)	87
3.3-3	Short Pseudo-Compute Sequence (SPCS) for the Example of Table (3.3-1)	12
4.1-1	Terminology used in Description of SINDA Preprocessor Routines	4-2
6.2-1	Optionally Specified Capacitance Expressions	6-6
6.2-2	" Impressed Source Expressions	TŤ.
6.2-3	" Coefficient Expressions	6-7
6.2-4	Definition of Symbols for Tables 6.2-1 through 6.2-3	6-8
6.2-5	Characteristics of User Specified Control Constants	6-11
6.3-1	Basic Computational Steps for CNFRWD and CNFRDL	6-51
6.3-2	Basic Computational Steps for CNFAST	6-58
6.3-3	Basic Computational Steps for CNEXPN	6-64
6.3-4	Basic Computational Steps for CNDUFR	6-70
6.3-5	Basic Computational Steps for CNQUIK	6-76
6.4-1	Basic Computational Steps for CNBACK	6-84
6.4-2	Basic Computational Steps for CNFWBK	6-91
6.4-3	Basic Computational Steps for CNVARB	6-97
6.5-1	Basic Computational Steps for CINDSS	6–105
6.5-2	Basic Computational Steps for CINDSL	6-111
6.5-3	Basic Computational Steps for CINDSM	6-116

 \mathbf{i}

1

•

PREFACE

The present SINDA computer program has evolved from the CINDA-3G program, which in turn evolved from the CINDA program, etc. With each major program revision an updated user manual was generated, but a more in-depth presentation of programming considerations and the theoretical development of the numerous subroutines were not generated. This SINDA program manual represents a preliminary effort to fill some of the existing void by describing the program structure, by identifying the major functions of each processor routine with a functional flow chart, and by a more in-depth mathematical description of the numerical solution subroutines. It is not the intent of this engineering-program manual, however, to provide sufficient detailed information for a user to make modifications and/or additions to the existing subprograms. 1.

- NOMENCLATURE AND MNEMONICS
- 1.1 <u>Nomenclature</u>

^a ij	= k(A/l) _{ij} , conduction coefficient between nodes i
A	= array
Α	= area
(A/l) _{ii}	= effective ratio of cross-sectional area to distance
	between nodes i and j.
b.,,	= radiation factor between nodes i and j (composed of
L'T	radiation interchange factor and area)
° C _i	= capacity of ith node
c	= C $/\Lambda t$, capacity of ith node divided by time-step
1 100	i = 0 (allows certain fraction of "old" temperature
<u>du</u>	- I - DA (allows certain flaction of old temperature
	to be included as part of temperature change for
	current time-step, refer to Section 6.2.5.1)
DN	\equiv DAMPA (user control constant, refer to Sections 6.2.5.1
	and 6.2.3.2)
F,F1,F2	= multiplying factors, either user constants or literals,
	refer to Tables 6.2-1, 6.2-2 and 6.2-3).
G _{ij}	$= a_{ij} + \sigma b_{ij} (T_i^2 + T_j^2) (T_i + T_j)$
Gk	= a or ob ij, conduction or radiation coefficient.
k	= thermal conductivity
L	= a literal multiplying factor
N	= number of variable temperature nodes (NNA + NND)
NNA	= number of arithmetic-nodes
NND	= number of diffusion nodes
R	= resistance
P	= total number of nodes
P.	= impressed heat load into the ith node
t	time
Λt	= time-step
t	= $(TIME0 + TIMEN)/2.0$, mean time
т Т	= temperature ($^{\circ}F_{\circ}$ $^{\circ}P$)
- 	$= (T \pm T)/2 0$ mean homeosition (2)
¹ m	= $(1 + 1)/2.0$, mean temperature ("F or "R)

1 - 1 1<

and the second second

al also a secondar

يني **سو** بو ^{ار} و

	x,y	∞ coordinate								
	α	= (k/C) , thermal diffusivity								
	C.	$= \int_{j=1}^{p} G_{ij}/C_{i}, \text{ refer to equation 6.3-7}$								
	β = factor that ranges from 0 to 1/2 (refer to equation 6									
	$\beta' = 2\beta$ (used in subroutine CNVARB)									
	³ ij	= radiation interchange factor including inter-								
	reflections between nodes i and j.									
	$\sigma = \text{Stefan-Boltzmann constant (.1714 x 10-8 Btu/hr °R4ft2)}$									
	T n	= $\frac{\Delta t_n}{\Delta t_{n-1} + \Delta t_n}$, weighting factor (equation 6.3-13)								
$(A^{i}:t_{m})$	Interpolat	ted value of array A using t_m as the independent variable								
(A ¹ :T ₁)	11	n n's n's n's n's n's n's n's n's n's n'								
(A ⁱ :T _i)	11									
$(A^{i}:T_{m})$	11	n de la companya de								
(A ^p :T _i)	Polynomial	1 n n n n n n n n n n								
(A ^p :T _i)	51	и и в и т Т								
(A ^P :T _m)	13									
(A ^b :T,,t) Interpo	olated value of the bivariate array A using T_i and t_m								
U	as inde	ependent variables.								
(A ^b :T _m ,t) Interpo	olated value of the bivariate array A using $\mathtt{T}_{\mathtt{m}}$ and $\mathtt{t}_{\mathtt{m}}$								
	as inde	ependent variables.								
Subscript	S									
	1	= ith node								
	J 4 4	= between nodes 1 and 1								
$i_n = undating of ith temperature source at time-$										
step n.										
	i,k	= updating of ith temperature, etc. at kth iteration.								
	ij,n	= updating of coefficient between nodes i and j at								
time-step n										
	ij,k	= updating of coefficient between nodes i and j at								
		kth iteration								
	m	= mean $1 - 2$								
	4. sj	2<								
		y a gar gar sa a sa baa baa baa baa a sa a sa a s	en bra							

1.2 <u>Mnemonics</u>

.

1.2.1 <u>Control constants</u> (refer to Sections 6.2.3.1 and 6.2.3.2)
ARLXCA = allowable arithmetic node relaxation temperature change
ARLXCC = calculated maximum arithmetic node relaxation temperature change
ATMPCA = allowable arithmetic node temperature change
ATMPCC = calculated maximum arithmetic node temperature change
BACKUP = back switch
BALENG = specified system energy balance
CSGFAC = time-step factor
$CSGMAX = maximum value of C_{i}/\Sigma G_{ij}$
$CSGMIN = minimum value of C_{i} / \Sigma G_{ij}$
CSGRAL = allowable range between CSGMIN and CSGMAX
DAMPA = arithmetic node damping factor
DAMPD = diffusion node damping factor
DRLXCA = allowable diffusion node relaxation temperature change
DRLXCC = calculated diffusion node relaxation temperature change
DTIMEH = allowable maximum time-step
DTIMEI = specified time-step for implicit solutions
DTIMEL = allowable minimum time-step
DTIMEU = contains computed time-step
DTMPCA = allowable diffusion node temperature change
DTMPCC = calculated maximum diffusion node temperature change
ENGBAL = calculated system energy balance
ITEST = contains dummy integer constant
JTEST = contains dummy integer constant
KTEST = contains dummy integer constant
LAXFAC = number of iterations for linearized lumped parameter system,
CINDSM only.
LINECT = line counter location for program output
$L\phi\phi$ PCT = contains number of iterations performed
NØCØPY = contains no copy switch for matrix users
$NL\phi\phi P$ = number of specified iteration loops
<pre>ØPEITR = output each iteration switch</pre>
PAGECT = page counter location for program output

1 - 3 3< RTEST = contains dummy floating point constants STEST = contains dummy floating point constants TIMEM = (TIME # + TIME N)/2.0, mean time for computational interval TIMEN = TIMEN + DTIMEU, new time at the end of computational interval TIMEND = problem stop time TIMEØ = old time at the start of the computational interval TTEST = contains dummy floating point constants UTEST = contains dummy floating point constants VTEST = contains dummy floating point constants 1.2.2 Numerical Solution Routines (refer to Sections 6.3 - 6.5) CINDSS = steady state routine, refer to Section 6.5.1 CINDSL = steady state routine, refer to Section 6.5.2 CINDSM = steady state routine, refer to Section 6.5.3 CNBACK = implicit routine, refer to Section 6.4.1 CNDUFR = explicit routine, refer to Section 6.3.4 CNEXPN = explicit routine, refer to Section 6.3.3CNFAST = explicit routine, refer to Section 6.3.2CNFRDL = explicit routine, refer to Section 6.3.1 CNFWBK = implicit routine, refer to Section 6.4.2 CNFWRD = explicit routine, refer to Section 6.3.1 CNQUIK = explicit routine, refer to Section 6.3.5 CNVARB = implicit routine, refer to Section 6.4.3 1.2.3 Options (used in Tables 6.2-1 - 6.2-3) BIV = Bivariate Interpolation Variable DIT = Double Interpolation with Time as variable **DIV =** Double Interpolation Variable **DPV = Double Polynomial Variable** DTV = Double interpolation with Time and Temperature as Variables SIT = Single Interpolation with Time as variable SIV = Single Interpolation Variable SPV = Single Polynomial Variable

> 1 - 4 A<

and a second second

1.2.4 Routines and Subroutines of Preprocessor

SINDA = routine that specifies overlay of preprocessor to system allocator.

- PREPRØ = main routine for preprocessor; initializes counters and FØRTRAN
 logical units; sets length of dynamic storage array and controls
 major logic.
- ALPINT = subroutine that accepts an integer in alphanumeric format and converts it to integer format; determines relative number of this actual number and converts it back to alphanumeric format.
- BLKCRD = subroutine that formats the five generated FØRTRAN routines (SINDA, EXECTN, VARBL1, VARBL2, and ØUTCAL) in 507 word blocks.
- CØDERD = Subroutine that reads and checks the block header cards for the data blocks.
- CØNVRT = subroutine that converts Hollerith data to integer data.
- DATARD = subroutine that scans the data block card images under an A format and determines appropriate format to reread the card images.
- ERRMES = subroutine that prints most of the error messages generated within the data blocks.
- FINDRM = subroutine that moves the data in the dynamic storage array either up or down by 100 words.
- GENLNK = subroutine that generates the driver (FØRTRAN routine named SINDA) for the user's program.
- GENUK = subroutine that generates user constants.
- INCØRE = subroutine that reads data into the dynamic storage array for the parameter-runs option.
- MXTØFN = subroutine that processes data for the "m" option (converts card images from mixed FØRTRAN/SINDA notation to FØRTRAN notation.
- NØDEDA = subroutine that processes data for node and conductor data blocks.
- PCS2 = subroutine that packs the FØRTRAN addresses for the array and constants locations required by the second pseudo-compute sequence.
- PRESUB = subroutine that reads and checks the block header cards for the operations blocks and generates the non-executable FØRTRAN cards for each of the operations blocks via a call to BLKCRD.

1 - 5

- QDATA = subroutine that checks and processes all data input in the source data block.
- RELACT = subroutine that finds the relative node numbers from the actual node number; computes the FORTRAN address for arrays and user constants from the actual number.
- SEARCH = subroutine that retains a relative number for nodes, conductors, user constants, and arrays, given the actual number.
- SETFMT = subroutine that processes the card for the "new format" option; that is, it sets up the format for data cards as specified by the cards with an "N" in column one.
- SINDA4 = subroutine that reads and processes the user input cards from the operations blocks.
- SKIP = subroutine that is used when a problem is RECALLED; it positions the tape to the proper problem as specified on the first card of the data deck.
- SPLIT = subroutine that reads the data from the RECALL tape and splits
 the RECALL information onto the proper data "tape" and the
 dictionary "tape."
- SQUEEZ = subroutine that compresses the specified data groups in the dynamic storage array.
- STFFB = subroutine that fills out a card image in array KBLK with Hollerith blanks.
- TYPCHK = subroutine that checks the input from the data blocks for the correct type (integer, floating point, or alphanumeric.
- WRTDTA = subroutine that writes the program data "tape" in the format required by INPUTT or INPUTG.
- WRTPMT = subroutine that writes the required data for parameter runs on the parameter "runs" "tape" and the dictionary "tape."
- WRTBLK = subroutine that writes the 507 word blocks contained in array KBLK on the program FØRTRAN "tape."

1.2.5 Others .

× -

•.

1.2.5		<u>Others</u>
SINDA	, f	Systems Improved Numerical Differencing Analyzer
LPCS	=	Long Pseudo-Compute Sequence
SPCS	#	Short Pseudo-Compute Sequence
LPCS2	W	Second Long Pseudo-Compute Sequence
PCS1	H	Pseudo-Compute Sequence One
PCS2	æ.	Pseudo-Compute Sequence Two
TSUM	-	elapsed time from last printout
TPRINT	=	time of last printout.

•

1 - 7 7<

2. BACKGROUND ON SINDA

The original CINDA^{*1} (Chrysler Improved Numerical Differencing Analyzer) computer program was developed by the Thermodynamics Section of the Aerospace Physics Branch of Chrysler Corporation Space Division at NASA Michoud Assembly Facility and was coded in FORTRAN-II and FAP for the IEM-7094 computers. CINDA was the product of an intensive analytical, engineering and programming effort that surveyed numerous thermal analyzertype programs and studied several in-depth. The foundation for CINDA was the storage and addressing of information required only for the network solution and the systems features which allowed the re-utilization of core storage area and brought into core only those instructions necessary for the solution of a particular problem. A systems compiler computer program that automatically optimized the utilization of computer core space was developed. This meant the generation of an integrated operation of relative addressing, packing features, peripheral tape storage units and overlay features.

CINDA evolved into CINDA-3G² which was developed by the same group that generated CINDA with a major portion of the work done under contract NASA/MSC NAS9-7043. CINDA-3G represented (essentially) a complete rework of CINDA in order to take advantage of the improved systems software and machine speeds of the 3rd generation computers. CINDA was unsuitable for standard operation on third generation computers since it was virtually a self-contained program having its own Update, Monitor and Compiler. On the other hand, CINDA-3G consisted of a preprocessor (written in FORTRAN) which accepted the user input data and the block data input. The user input data was converted into advanced FORTRAN language subroutines and block data input was passed onto the system FORTRAN Compiler. This required a double pass on data where previously only one was required, but the increased speed and improved software of the third generation machines more than compensated for the double pass.

SINDA³ (Systems Improved Numerical Differencing Analyzer) was developed by the Heat Transfer and Thermodynamics Department of TRW Systems Group, Redondo Beach. Most of the improvements and subroutine additions to CINDA-3G was done as part of the NASA/MSC contract NAS 9-8389, * Superscript numbers refer to the literature cited in the Reference Section.

2 - 1

8<

entitled, "Development of Digital Computer program for Thermal Network Correction." Programming and systems integration were directed to the UNIVAC-1108 computer.

SINDA relied quite heavily on CINDA-3G and data deck compatibility was rigorously followed; as a result, CINDA-3G data decks should, in the main, be directly operational on the SINDA program although some differences exist. For example, properties are updated before VARIABLES 1 call in CINDA-3G, whereas the properties are updated within the numerical solution routines after VARIABLES 1 call in SINDA. The primary differences between SINDA and CINDA-3G are: (1) elimination wherever possible of assembly language coding; (2) increased mnemonic options to aid the program user in data input; (3) inclusion of a second pseudo computer sequence for evaluation of nonlinear network elements; and (4) additional subroutines such as STEP (sensitivity analysis) and KALØBS-KALFIL (Kalman filtering). Most of the changes and additions to CINDA-3G were required in order to integrate the thermal network correction subroutine package into the existing CINDA-3G program.

During the development of SINDA a number of useful improvements became apparent. As a result, modifications to SINDA were made a part of the NASA/MSC contract NASA 9-10435 entitled, "Development of an Advanced SINDA Thermal Analyzer System." These changes that include a variable input format, simplified parameter runs, and generated user constants were made by the same group that developed SINDA. These improvements are reported in an updated SINDA users manual.⁴

> 2 - 2 9<

3. GENERAL SINDA PROGRAM DESCRIPTION

3.1 SINDA Operating System

SINDA is more like an operating system rather than applications SINDA is programmed as a preprocessor in order to accommodate program. the desired operations relative to overlay features, data packing, dynamic storage allocation and subroutine library file, but yet be This preprocessor operates in an integral fashwritten in FØRTRAN. ion with a library of numerous and varied subroutines,³,⁴ which may be called in any desired sequence but yet operate in an integrated manner. The preprocessor reads the input data, assigns relative numbers, packs this information, forms a pseudo-compute sequence(s) (which will be described briefly in a later paragraph of this section and is described in more detail in Section 4, called Preprocessor), and writes the operations blocks on a peripheral unit as FØRTRAN source language with all of the data values dimensioned exactly in labeled common. In turn, controls are shifted to the system FØRTRAN compiler which compiles the constructed subroutines and enters execution. The FØRTRAN allocator has access to the SINDA subroutine library and loads only those subroutines called by the problem being processed.

As a result of this type of systems operation SINDA is extremely dependent upon the systems software. However, once the program is operational on a particular computer, the user-prepared problem data deck can be confined to the control cards and deck set-up requirements at a particular installation.

It should be recognized that the use of a preprocessor provides a computer with a large capability and considerable flexibility, but because of the numerous options that are generally offered, user instructions are more difficult than other thermal analyzer-type programs which have less flexibility.

3.2 Use of Lumped-Parameter Concept

Use of SINDA is based on a lumped parameter representation of a physical system. This means that SINDA does not solve a set of partial differential equations that represents a distributive system, but rather SINDA numerically solves a set of ordinary (and in general) nonlinear

10< 3-1

differential equations that represent a lumped parameter system. The procedure for the formulation and the numerical solutions of the lumped parameter equations are reported extensively in literature and basic considerations are presented in Section 5. For the discussion to follow on the pseudo compute sequence it is convenient to indicate a general set of ordinary <u>linear</u> differential heat balance equations,

$$\frac{dT_{i}}{dt} = \frac{1}{C_{i}} \left[q_{i} + \sum_{j=1}^{p} a_{ij} (T_{j} - T_{i}) \right]$$
(3.2-1)

$$i = 1, 2, \dots, N \text{ (number of variable temperatures)}$$

$$T_{j} = \text{ constant}, N < j \leq p$$

$$C_{i} = \text{ the ith nodal capacity}$$

$$q_{i} = \text{ the heat load into node i (impressed)}$$

$$a_{ij} = \text{ the conduction coefficient between nodes i and j } \left[= k \left(\frac{A}{k} \right)_{ij} \right]$$

Suppose an implicit numerical method as discussed in Section 5.2.2 of this manual is chosen; the implicit finite difference form becomes after letting,

$$dT_i/dt \approx (T_{i,n+1} - T_{i,n})/\Delta t, T_j = T_{j,n+1} \text{ and } T_i = T_{i,n+1},$$

$$\frac{C_{i}}{\Delta t} (T_{i,n+1} - T_{i,n}) = q_{i} + \sum_{j=1}^{p} a_{ij} (T_{j,n+1} - T_{i,n+1})$$
(3.2-2)

where,

where,

$$T_{i,n} = \text{temperature at time point } t_n$$

 $T_{i,n+1} = \text{temperature at time point } t_{n+\Delta t}$
 $\Delta t = \text{time-step}$

Rearrangement of equation (3.2-2) yields,

t = time

$$(\overline{C}_{i} + \sum_{\substack{j=1\\j\neq i}}^{p} a_{ij}) T_{i,n+1} - \sum_{\substack{j=1\\j\neq i}}^{N} a_{ij} T_{j,n+1} = q_{i} + \overline{C}_{i} T_{i,n} + \sum_{\substack{j=N+1\\j=N+1}}^{p} a_{ij} T_{j,n} \quad (3.2-3)$$

$$i = 1, 2, \dots, N$$

$$T_{j,n} = \text{constant}, N < j \le p$$

where,
$$C_i = C_i / \Delta t$$
, average capacity of node i over Δt time-step

3.3 Pseudo-Compute Sequence (PCS)

A pseudo-compute sequence as generated by the SINDA preprocessor

is a list of numbers that indicates the position of required data values in various arrays such as conductance, temperature and capacitance. This meaning will become clearer by formulating equation (3.2-3) in a matrix form. The matrix formulation is straightforward since temperatures at time-step n+1 are the unknowns and terms on the right side of equation (3.2-3) represent the forcing function. Let us expand equation (3.2-3) to show this,

$$(\overline{C}_{1} + \sum_{j=1}^{p} a_{1j})T_{1,n+1} - a_{12}T_{2,n+1}, \dots, -a_{1N}T_{N,n+1} = q_{1} + \overline{C}_{1}T_{1,n} + \sum_{j=N+1}^{p} a_{1j}T_{j,n}$$

- $a_{21}T_{1,n+1} + (\overline{C}_{2} + \sum_{j=1}^{p} a_{2j})T_{2,n+1}, \dots, -a_{2N}T_{N,n+1} = q_{2} + \overline{C}_{2}T_{2,n} + \sum_{j=N+1}^{p} a_{2j}T_{j,n}$

 $-a_{N1} T_{1,n+1} - a_{N2} T_{2,n+1}, \dots, (\overline{C}_{N1} + \sum_{j=1}^{p} a_{Nj}) T_{N,n+1} = q_{N} + \overline{C}_{N} T_{N,n} + \sum_{j=N+1}^{p} a_{Nj} T_{j,n}$ Thus the matrix form of equation (3.2-3) becomes,

$$[\beta] \{T'\} = \{Q\}$$
(3.3-1)

where,

$$\beta = \begin{pmatrix} p \\ j=2 \\ j\neq 1 \end{pmatrix}, \quad -a_{12} , \dots, -a_{1N} \\ -a_{21} , p \\ j \neq 2 \\ \vdots \\ j\neq 2 \\ -a_{N1} , -a_{N2} , \dots, p \\ j=1 \\ \vdots \\ j\neq 2 \\ -a_{N1} , -a_{N2} , \dots, p \\ j=1 \\ j\neq 2 \\ j\neq 2 \\ -a_{N1} \end{pmatrix}$$

$$T' = \begin{pmatrix} T_{1,n+1} \\ T_{2,n+1} \\ \vdots \\ T_{N,n+1} \end{pmatrix}, \quad (3.3-3) ; \{Q\} = \begin{pmatrix} q_1 + \overline{c}_1 T_{1,n} + p \\ q_2 + \overline{c}_2 T_{2,n} + p \\ j=N+1 \\ \vdots \\ q_N + \overline{c}_N T_{N,n} + p \\ j=N+1 \\ -a_{Nj} T_{j,n} \end{pmatrix}$$

$$(3.3-4)$$

The matrix represented by equation (3.3-2) <u>appears</u> to be a full matrix (very small number of elements that are zero), but in reality most

of the off-diagonal elements are zero. Thus, if equation (3.2-3) was to be solved by a matrix inversion technique, all elements including zeros must be stored. Since the number of elements varies as N^2 (N is the number of nodes), the required number of data locations would vary as N^2 and the computer time required for matrix inversion would be proportional to N^3 .

The explicit and iterative implicit numerical methods (refer to Section 5) of solving equation (3.2-1) lend themselves for optimizing the data storage area required and for reducing the solution time. If the conductors are numbered and related to the appropriate adjoining nodes as indicated in Table (3.2-1), retention of adjoining node number for each conductor provides a means of identifying element position in the coefficient matrix. This can be seen by considering the one-dimensional heat conduction example pictured in Figure (3.3-1).



Figure 3.3-1. Thermal Circuit for a One-Dimensional System

The set of equations associated with the problem of Figure (3.3-1) may be readily expressed as,

$$\begin{bmatrix} (\overline{C}_{1}+G_{1}), & -G_{1}, & 0, & 0, & 0 \\ -G_{1}, & (\overline{C}_{2}+G_{1}+G_{2}), & -G_{2}, & 0, & 0 \\ 0, & -G_{2}, & (\overline{C}_{3}+G_{2}+G_{3}), & -G_{3}, & 0 \\ 0, & 0, & -G_{3}, & (\overline{C}_{4}+G_{3}+G_{4}), & -G_{4} \\ 0, & 0, & 0, & 0, & -G_{4}, & (\overline{C}_{5}+G_{4}) \end{bmatrix} \begin{pmatrix} T_{1} \\ T_{2} \\ T_{3} \\ T_{4} \\ T_{5} \end{pmatrix} = \begin{pmatrix} Q_{1} \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \end{pmatrix}$$
(3.3-5)

By comparing the element position of equation (3.3-5) with the tabular identification in Table (3.2-1) it is seen that elements with zero values need not be stored. The main diagonal term is never zero and is a composite of capacitance and off-diagonal conductors.

3 - 4 13<

Conductance Number	ith Node	Adjoining Node Number	Comment
G₿	N#	N#	
1	1	2	G1 is conductor #1 between nodes 1 and 2.
1	2	1	G1 is conductor #1 between nodes 2 and 1.
2	2	3	G2 is conductor #2 between nodes 2 and 3.
•	•	÷ •	
•		• :	
4	5	4	G4 is conductor #4 between nodes 5 and 4.

Table (3.2-1)Tabular Identification of Conductor
and Adjoining Node Numbers

It is of interest to note that the use of a pseudo-compute sequence is only one of a number of ways to store data efficiently. For example, TRW TAP⁵ does not employ a pseudo-compute sequence because of other user requirements. However, from a data storage standpoint, it appears that the use of a pseudo-compute sequence utilizes computer core most efficiently.

More than one pseudo-compute sequence is formed by SINDA. Both a so-called long (LPCS) and a so-called short (SPCS) pseudo-compute sequence as used in CINDA-3G² are formed and in addition a second long pseudo-compute (LPCS2) required for thermal network correction is also formed in SINDA. A detailed discussion of these pseudo-compute sequences will be presented in Section 4.6, but is of interest here to indicate the characteristics of these "sequences."

3.3.1 Long Pseudo-Compute Sequence (LPCS)

A long pseudo-compute sequence identifies the position and value of all off-diagonal elements of the coefficient matrix. This is done by operating on adjoining node numbers which have been assigned relative node numbers by the preprocessor. Since nodal temperatures are calculated sequentially in ascending numerical order, the conductor and adjoining node number are searched until node one is found with the conductor number and the other adjoining node number stored in a single core location. In addition, several indicators are stored in this single core location. These

3 - 5

14<

indicators are : (1) var C (indicates the input of a capacitor as a variable); (2) var G (indicates the input of a conductor as a variable); (3) rad (indicates the input of a radiation conductance); (4) Q (indicates the input of a source in the source data block); (5) one-way (indicates the input of a one-way conductor); and (6) last G (indicates the last conductor to a particular node). Order of indicator storage is indicated in Table (3.3-2).

Search is continued until all node-one's have been located and characteristics processed. The procedure is repeated for all node-two's and so forth sequentially until all nodes have been processed. The important consideration of a LPCS is the encounter of each conductor of the coefficient matrix twice. Formation of a pseudo-compute sequence for the example shown in Table (3.3-1) is given in Table (3.3-2). A pseudocompute sequence starts with node one and advances the node number by one each time a last conductor indicator (last G) is passed. The conductor and node numbers identify the position of the conductor value in an array of conductor values and the position of the temperature, capacitor and source values in arrays of temperature, capacitor and source values respectively.

A long pseudo-compute sequence is well-suited for "successive point" iteration (refer to Section 5.2.2 for a discussion of this) of the implicit finite difference equations because all elements of the coefficient matrix are identified. Thus, when a row of the coefficient matrix is processed and a new value of temperature obtained, the new temperature can then be used in the calculation procedure of succeeding rows.

3.3.2 Short Pseudo-Compute Sequence (SPCS)

The short pseudo-compute sequence identifies each conductor only once and since the coefficient matrix (equation 3.3-1) is symmetrical, all sparsity and off-diagonal elements of the coefficient matrix are accounted for. The node being processed and the adjoining node number reveal temperature- and source-value locations. The short pseudo-compute sequence for the example in Table (3.3-1) is formed in Table (3.3-3). By placing a minus sign on the initially encountered other-adjoining nodes, these nodes are not recognized on a second encounter. A short pseudo-compute sequence

3 - 6

Conductor No.	Adjoining Node	Numbers
G#	N#	N#
1	1	2
2	1	3
3	1	4
4	2	3
5	2	4
6	3 S	4

Table (3.3-1) Example of Conductor Connections

Table (3.3-2) Long Pseudo-Compute Sequence (LPCS) for the Example of Table (3.3-1)

Node No. Searched	Last G	var C	var G	rad	Q	G#	One- way	Node # Stored
1		ي جاني				1		2
1						2		3
1	1					3		4
2						1		1
2						4		3
2	1					5		4
3						2		1
3						4		2
3	1					6		4
4						5		2
4	1					6		3

Table (3.3-3) Short Pseudo-Compute Sequence (SPCS) for the Example of Table (3.3-1)

Node No.	Last	var	var	rad	Q	G#	One-	Node #
Searched	G	G	G				way	Stored
1			·			1	•	2
1						2		3
1	1					3		4
2						4		3
2	1					5		4
3	.1					6		4
4	1					0		0
				3 - 7				
		an a	16<	a na tanàna amin'ny taona 2008–2014.				

is well-suited for explicit numerical solutions methods which calculate the energy flow through the conductor, add it to the source location of the node being processed and subtract it from the source location for the adjoining node. The SPCS can be used for implicit methods of solution but the "block" iterative procedure (refer to Section 5.2.2 for a discussion of this) must be used since succeeding rows of conductor and adjoining node numbers do not contain the necessary element information.

3.3.3 Second Long Pseudo-Compute Sequence (LPCS2)

The second long pseudo-compute sequence (LPCS2) as a user input option flags a non-linear conductor between two diffusion nodes twice; LPCS flags the non-linear conductor only one. LPCS2 is required for the thermal network correction of a sparse network by the use of subroutine KAFIL (refer to Reference 3 or 6).

3.3.4 <u>Pseudo-Compute Sequence One (PCS 1) and Pseudo-Compute Sequence</u> <u>Two (PCS 2)</u>

PCS 1 and PCS 2 are not user options but are fixed internally. The contents of PCS 1 and PCS 2 are governed by the user input of LPCS, SPCS or LPCS2). PCS 1 contains two relative addresses (conductor and adjoining node locations), two non-linear type indicators, and an impressed source indicator. Indicators are keyed through a simple counter to a second pseudo-compute sequence (PCS 2) which contains integer addresses or relative constant and array starting locations necessary for evaluation of temperature varying coefficients and time varying coefficients for sources. When the input data contain literal values in SIV type calls, the preprocessor stores the values as extended user constants and supplies the relative constant address to the second pseudo-compute sequence. Detailed discussion on PCS 1 and PCS 2 is presented in Section 4.6.

> 3 - 8 17<

1

3.4 Data Logistics

3.4.1 <u>Relative Numbers</u>

Both the long and short pseudo-compute sequences require the storage of only the finite values in the coefficient matrix, thereby taking advantage of matrix sparsity. If the short pseudo-compute sequence is used, the advantage of symmetry is accounted for. Conductors with the same constant value may share the same conductor number and value. The storage efficiency of the pseudo-compute sequences requires the sequential numbering of the nodes and the conductors. Since the numbering of thermal math-models is arbitrary and not sequential, the SINDA program assigns relative numbers. (starting from one, sequential and ascending) to the actual numbers of the incoming node data, conductor data, constants data and array data in the order received. Thus, numbers not used in the actual numbering system are neither identified nor required.

3.4.2 Storage Requirements and Dynamic Storage Allocation

All numerical solution subroutines require three locations for each diffusion node data (temperature, capacitance and source), two locations for each arithmetic node data (temperature and source), one location for each boundary data (temperature) and one location for each conductor value. In addition intermediate data storage ranging from zero to three locations per node may be required for the storage of temperatures and temperature differences; acceleration of convergence (refer to Section 6.2.7) used in the implicit and steady state routines (except CINDSS) requires three locations. Storage requirements for conductances depends upon the problem. For example, each internal diffusion and arithmetic node of a three-dimensional conduction system with rectangularnodalization will be connected with only three being unique; thus, each diffusion node (or arithmetic node) in a three-dimensional conduction system requires from six to nine storage locations for data values (temperature, capacitance, source, three conductors and up to three intermediate locations). Now each of the conductors for the short pseudo-compute sequence requires a single core location that contains two integer values (conductor and adjoining node numbers) and six indicators (refer to Section 3.3.1 for description). Each of the conductors between variable temperatures for the

> 3 - 9 ⊈8<

long pseudo-compute sequence requires two core locations since the conductors are used twice during the computational process. This means that each internal node of a three-dimensional conduction system will require six <u>data addressing locations</u> for the long pseudo-compute sequence and, on the average, three data addressing locations for the short pseudocompute sequence.

Thus for a three-dimensional conduction system (no radiation), the number of required core locations per node can vary from nine (temperature, capacitance, source, three unique conductors and three data addressing locations) to fifteen (temperature, capacitance, source, six conductors and six data addressing locations) exclusive of the second pseudo-compute sequence which is required for variable coefficients, capacitance and sources.

The user must allocate an array of data locations which is to be used for intermediate data storage and initialize the array start and length indicators. Each subroutine that requires intermediate storage area has access to this array and the start and length indicators. During a subroutine execution a check on the sufficiency of space is made and start and length indicator are updated. If a subroutine calls upon another subroutine that requires intermediate storage, the called subroutine repeats the check and update procedure. Whenever any subroutine terminates its operation, the start and length indicators are returned to their entry values. This process is termed "Dynamic Storage Allocation" and allows subroutines to share a common working area.

3.5 Order of Computation

A network data deck consists of four data blocks (node, conductor, constants, and array), one optional data block (source) and four operations blocks which are preprocessed by the preprocessor and passed on to the system FØRTRAN compiler. Non-network problems require no node or conductor data blocks. The operations blocks are named EXECUTIØN, VARIABLES 1, VARIABLES 2 and OUTPUT CALLS: the SINDA preprocessor constructs these blocks into individual subroutines with the entry names EXECTN, VARBL1, VARBL2 and ØUTCAL, respectively. After a successful FØRTRAN compilation, control is passed to the EXECTN subroutine. This means that the order of computation depends on the sequence of subroutine calls placed in the EXECUTIØN block

3 - 10

by the program user. No other operations blocks are performed unless called upon by the user either directly by name or indirectly through a subroutine call. The numerical solution subroutines described in Section 6 internally call upon VARBL1, VARBL2 and ØUTCAL; The internal order of computation for these routines is similar with the primary difference being the numerical solution method. A general flow diagram of the numerical solution routines, as well as a detailed description of each is presented in Section 6.

3 - 11

20 <

4. PREPROCESSOR

4.1 General Description

The SINDA preprocessor reads and analyzes the user input deck and from this information constructs a program tailored to the user's requirements.

The rationale for a preprocessor is flexibility and speed. Flexibility is achieved by providing the user with a library of routines to solve problems, manipulate data, and print selected values. In addition, the user may insert non-SINDA routines into the constructed program. Speed (defined here as minimal execution time) is achieved by structuring the data in an efficient manner.

The SINDA preprocessor consists of thirty routines with seven overlay links. All of the routines are written in FØRTRAN except for one assembly language routine which writes a "tape" in a format acceptable to the FØRTRAN compiler. These routines provide the user with a number of major options in the type of problem to be solved and the form of the data to be used. Henceforth these major options are designated as "major logic" of the preprocessor. See Figure 4.1-1 for a flow chart of the major logic of the preprocessor and its interface with the user program.

The major logic consists of the five following options: (1) NASA MSC EDIT feature; (2) RECALL option; (3) generation of a THERMAL problem; (4) generation of a GENERAL problem; (5) and PARAMETER RUNS option. The primary features of each item of the major logic is discussed below.

- (1) EDIT feature: The first card of the deck is checked for the user request of the EDIT feature. If the EDIT feature is requested the input "tape" is changed from the system input "tape" to the EDIT "tape" and control is transferred to subroutine EDIT for processing. On return a branch is made to the THERMAL or GENERAL section as specified by the data on the EDIT "tape." If the EDIT feature has not been requested, the check for RECALL is made.
- (2) RECALL option: The first card of the deck is checked for user request of the RECALL option. If the RECALL option



Figure 4.1-1. SINDA Preprocessor - Major Logic and Interface with the User Specified Problem

4 - 2

. N. S

22<

and the second second second second second
is requested, control is transferred to subroutine SPLIT for processing. On return a branch is made to the PARAMETER RUNS section. If the RECALL option has not been requested, the second card of the deck is checked for the type of problem, THERMAL or GENERAL.

- (3) THERMAL problem: The type of pseudo-compute sequence requested is noted, the title block is read, the data blocks are read and processed, the pseudo-compute sequence is formed, the driver for the user program (SINDA) is written on "tape," the operations blocks are read and processed and their FØRTRAN equivalents are written on "tape," and finally a check is made for the user requests of the PARAMETER RUNS option.
- (4) GENERAL problem: This section is identical to a THERMAL problem except that only constants data and array data of the data blocks are read and processed; a pseudo-compute sequence is not formed.
- (5) PARAMETER RUNS option: A check is made for the user request of the PARAMETER RUNS option. If the PARAMETER RUNS option is requested, the appropriate data blocks are read and processed. If not, the preprocessor is terminated.

Description of SINDA preprocessor routines is presented in the sections to follow. Terminology used in the description is listed and defined in Table 4.1-1.

4 - 3

23<

· . -

Table (4.1-1) Terminology Used in Description of SINDA Preprocessor Routines

- DATA BLOCKS: The five user input blocks which contain data rather than instructions; these DATA BLOCKS are NODE DATA, CONDUCTOR DATA, CONSTANTS DATA, ARRAY DATA and the optional blocks SOURCE DATA.
- (2) OPERATIONS BLOCKS: The four user input blocks which contain instructions on problem solution, as opposed to data contained in the DATA BLOCKS. These OPERATIONS BLOCKS are EXECUTION, VARIABLES 1, VARIABLES 2 and ØUTPUT CALLS.
- (3) Non-fatal error: An error that does not terminate the preprocessor immediately. That is, the preprocessor will continue scanning the remaining cards of the input deck for errors. However, the user program will not be executed.
- (4) Fatal error: An error that terminates the run immediately.
- (5) N/A: Means not applicable.
- (6) "TAPE": The term "tape" in quotes is used to signify any external storage device. That is, any piece of computer hardware, excluding the central processor, on which data can be stored and retrieved. The three most familiar examples are: magnetic tape, drum and disk.
- (7) Dictionary: A list of the actual SINDA numbers in relative order. For example, the actual node number corresponding to the kth relative node number is the kth item of the node number dictionary.
- (8) Data group: A data group composed of the pertinent information extracted from a particular data block. For example, the two groups derived from the constants data are: the user constants numbers and the user constants values.
- (9) Bit manipulation: Terminology that implies the ability to store and access information within a computer word. This capability is also called packing and unpacking.
- (10) Routine: A general term used to describe any program element.
- (11) Subroutine: A special type of program element that is callable from a routine.
- (12) Fixed constants: The term used in the preprocessor for control constants.

24<

4.2 Description of Subroutines

Sections 4.2.1 through 4.2.30 below describe the 30 routines of the SINDA preprocessor. The descriptions are based on the UNIVAC 1108 computer under the EXEC II operating system; it should be understood, however, that much of the information is machine-dependent and is dependent upon the facilities operating system. Note that the element named SINDA (Section 4.2.1) and the element named PREPRØ (Section 4.2.2) are not subroutines in the technical sense of the word; hence, these two elements are referred to by the more general term "routine."

Each element of the preprocessor is described by the following eleven subtitles:

- (1) SUBROUTINE NAME this specifies the name of the element.
- (2) PROGRAMMING LANGUAGE This may be FØRTRAN, ASM, or MAP. FØRTRAN implies FØRTRAN V, ASM stands for assembly language (sometimes called SLEUTH II) and MAP is a special language which defines the overlay structure.
- (3) PURPOSE This gives a brief statement of the functional capabilities of the element.
- (4) RESTRICTIONS This gives an indication of where the input parameters come from, the form of the input parameters and the placement of the output parameters.
- (5) "TAPES" USED This represents a list of each FØRTRAN logical unit referenced within this element.
- (6) SPECIAL FEATURES This specifies programming features that are unique to a particular machine.
- (7) OTHER SUBROUTINES CALLED This represents a list of the external references.
- (8) CALLING SEQUENCE This gives a list of the subroutine arguments, if any, and a brief discussion of their use.
- (9) ERROR PROCEDURES This discusses the steps taken when an error is encountered.
- (10) STORAGE REQUIRED This gives the octal and decimal storage required for this element.

25<

(11) LABELED COMMON - This represents a list of each labeled common name used in this element.

4.2.1 ROUTINE NAME: SINDA

PROGRAMMING LANGUAGE: MAP

<u>PURPOSE</u>: This routine specifies the overlay structure of the preprocessor to the system allocator (loader).

RESTRICTIONS: N/A

"TAPES" USED: N/A

SPECIAL FEATURES: N/A

OTHER SUBROUTINES USED: N/A

CALLING SEQUENCE: N/A

ERROR PROCEDURES: N/A

STORAGE REQUIRED: N/A

LABELED COMMON: N/A



4.2.2 ROUTINE NAME: PREPRØ

PROGRAMMING LANGUAGE: FØRTRAN

<u>PURPOSE</u>: This routine is the main routine (i.e., the driver) for the preprocessor. It initializes the counters and FORTRAN logical units, sets the length of the dynamic storage array, and controls the major logic. The major logic includes: (1) the EDIT feature (NASA MSC only); (2) the RECALL of a stored problem; (3) setup of a new user problem; (4) and preprocessor termination procedures.

RESTRICTIONS: N/A

"TAPES" USED:

System input "tape"	NIN
System output "tape"	NØUT
Problem data "tape"	LB3D
Problem FORTRAN "tape"	LB4P
Dictionary "tape"	LUTL
Parameter runs "tape"	LUT3
Recall "tape"	LUT7
Internal scratch "tape"	INTERN

<u>SPECIAL FEATURES</u>: System error termination - the problem data unit (LB3D) and the problem FORTRAN unit (LB4P) are flagged to stop before the data scan begins in the event that a system error terminates the preprocessor prematurely. The reason the problem data unit is flagged to stop is that for a RECALL problem the problem FORTRAN unit must not be written on.

OTHER SUBROUTINES USED: CØDERD, GENLNK, PRESUB, PSEUDØ, SINDA4, SPLIT and WRTBLK.

CALLING SEQUENCE: N/A

ERROR PROCEDURES: The error termination procedures are controlled by three flags named ERDATA, PRØGRM, and ENDRUN. The three flags are in the labeled common block named DATA. ERDATA is used to flag <u>non-fatal</u> errors encountered while reading the data blocks, while PRØGRM performs the same function for the operations blocks. See Section 4.7.2.

STORAGE REQUIRED: 443 octal words = 291 decimal words. See Section 4.7.1. LABELED COMMON: BUCKET, CRDBLK, DATA, LØGIC, PLØGIC, and TAPE.

4 - 7

4.2.3 SUBROUTINE NAME: ALPINT

<u>PURPOSE</u>: This subroutine accepts an integer in the alphanumeric format nAl, and converts it to integer format, determines the relative number of this actual number, and converts the relative number back to an alphanumeric format of the form mAl.

<u>RESTRICTIONS</u>: The input and output is transmitted via the labeled common block named CIMAGE (see Section 4.3). The input must consist exclusively of the ten decimal digits.

"TAPES" USED: System output "tape" NOUT

SPECIAL FEATURES: None

OTHER SUBROUTINES USED: SEARCH

CALLING SEQUENCE: ALPINT(KLET, IST, IEND, J)

KLET	is an integer variable that indicates which dictionary
	is to be used for converting actual to relative.
IST	is the starting location of the alphanumeric integer.
IEND	is the ending location of the alphanumeric integer.
J	points to the last location + 1 of the converted integer

ERROR PROCEDURES: In the event that a given actual number has no relative number in the dictionary list, an error message will be issued and the relative operations blocks error flag (PRØGRM) will be set to 1.0. <u>STORAGE REQUIRED</u>: 665 octal words = 437 decimal words. See Section 4.7.1. <u>LABELED COMMON</u>: BUCKET, CIMAGE, DATA PØINT, and TAPE.

4.2.4 SUBROUTINE NAME: BLKCRD

PROGRAMMING LANGUAGE: FØRTRAN

<u>PURPOSE</u>: This subroutine formats the five generated FØRTRAN routines (SINDA, EXECTN, VARBL1, VARBL2, and ØUTCAL) in 507 word blocks, which are acceptable to the FØRTRAN compiler. This information is stored in labeled common block CRDBLK, array KBLK. A complete discussion of the required tape format is found in UNIVAC 1108, EXEC II, Programmers Reference Manual, UP-4058 C, Appendix D.4 entitled, Program Elements on Magnetic Tape (via CUR).

<u>RESTRICTIONS</u>: The input is Hollerith card images with a 14A6 format. It is transmitted either through the array IMAGE in labeled common CRDBLK, or through "tape" INTERN.

"TAPES" USED: Internal scratch "tape" INTERN

SPECIAL FEATURES: None

OTHER SUBROUTINES USED: STFFB and WRTBLK

CALLING SEQUENCE: BLKCRD

ERROR PROCEDURES: none

STORAGE REQUIRED: 753 octal words = 491 decimal words. See Section 4.7.1.

LABELED COMMON: CRDBLK and TAPE.

4.2.5 SUBROUTINE NAME: CØDERD

PROGRAMMING LANGUAGE: FØRTRAN

<u>PURPOSE</u>: This subroutine reads and checks the block header cards for the data blocks. It also performs the following functions: (1) the second data card of the deck is checked for a thermal or general problem, and if it is a thermal problem the type of pseudo-compute sequence specified is noted; (2) the title block is read and processed; (3) the actual array and constant numbers from the automated options are converted into FØRTRAN addresses; and (4) the parameter run block header cards are read and checked.

RESTRICTIONS: None

"TAPES USED:System input "tape"NINSystem output "tape"NØUTFØRTRAN V reread30

SPECIAL FEATURES: The FØRTRAN V intrinsic function FLD is used for bit manipulation.

CALLING SEQUENCE: CØDERD

ERROR PROCEDURES: In general, the errors checked for in this subroutine are of the fatal type; for example, data blocks out of order. The result of a fatal error is that the fatal error flag (ENDRUN) is set to 1.0 and control is returned to PREPRO for immediate termination.

STORAGE REQUIRED: 3213 octal words = 1675 words decimal. See Section 4.7.1.

LABELED COMMON: BUCKET, DATA, LØGIC, PLØGIC, and TAPE.

4.2.6 SUBROUTINE NAME: CØNVRT

PROGRAMMING LANGUAGE: FØRTRAN

PURPOSE: This subroutine converts Hollerith data to integer data.

<u>RESTRICTIONS</u>: The Hollerith data must be contained in one word and consist of only the ten decimal digits.

"TAPES" USED: None

<u>SPECIAL FEATURES</u>: The FØRTRAN V intrinsic function FLD is used for bit manipulation.

OTHER SUBROUTINES USED: ERRMES

CALLING SEQUENCE: CØNVRT(IST, IEND, ITEMP, CRDERR)

- IST is the pointer to the first bit of the first character.
- IEND is the pointer to the first bit of the last character.
- ITEMP is the word containing the Hollerith data on entry and the integer number on return.
- CRDERR is a logical error flag which is set true if an error is encountered during the conversion.

ERROR PROCEDURES: If a non-integer is encountered, an error message is printed and CRDERR is set to true.

STORAGE REQUIRED: 150 octal words = 104 decimal words. See Section 4.7.1.

LABELED COMMON: None

4 - 11

4.2.7 SUBROUTINE NAME: DATARD

PROGRAMMING LANGUAGE: FØRTRAN

<u>PURPOSE</u>: This subroutine scans the data block card images under an A format and determines the appropriate format (of the form Fn, In, or An) to reread the card image. The card images are then reread under the generated format. In addition, the constants data block and the array data block are processed.

RESTRICTIONS: None

"TAPES" USED:System input "tape"NINSystem output "tape"NØUTFØRTRAN V reread30

SPECIAL FEATURES: The FORTRAN V intrinsic function FLD is used for bit manipulation.

OTHER SUBROUTINES USED: ERRMES, FINDRM, GENUK, NØDEDA (and its entry point CØNDDA), SETFMT, SQUEEZ, and TYPCHK.

CALLING SEQUENCE: DATARD

ERROR PROCEDURES: All errors checked for in this subroutine are non-fatal. An error message is printed either internally or from subroutine ERRMES and the data blocks error flag (ERDATA) is set to 1.0.

STORAGE REQUIRED: 5344 octal words = 2788 decimal words. See Section 4.7.1. LABELED COMMON: BUCKET, CHECKD, DATA, FLAGS, LØGIC, PLØGIC, PØINT, and TAPE.

4 - 12

365

a in the contraction of the later

4.2.8 SUBROUTINE NAME: ERRMES

PROGRAMMING LANGUAGE: FØRTRAN

<u>PURPOSE</u>: This subroutine prints most of the error messages that can be generated within the data blocks.

RESTRICTIONS: None

"TAPES" USED: System output "tape" NØUT

SPECIAL FEATURES: None

II

J

K

OTHER SUBROUTINES USED: System subroutine EXIT

CALLING SEQUENCE: ERRMES (JUMP, I, J, K)

JUMP is an integer that points to the appropriate error message via a computed GØ TØ statement

are data words which allow a maximum of three printed

data words per error message.

ERROR PROCEDURES: If the number of error messages printed exceeds 199, the preprocessor is terminated by a call to EXIT.

STORAGE REQUIRED: 2166 octal words = 1142 decimal words. See Section 4.7.1. LABELED COMMON: DATA and TAPE



ليهم سريعة أداد المراجب بالمراجب

4.2.9 SUBROUTINE NAME: FINDRM

PROGRAMMING LANGUAGE: FØRTRAN

<u>PURPOSE</u>: This subroutine moves the data in the dynamic storage array either up or down by 100 words. In the process it may delete certain groups of data that are no longer needed.

RESTRICTIONS: None

"TAPES" USED: System output "tape" NØUT

SPECIAL FEATURES: None

OTHER SUBROUTINES USED: SQUEEZ, and system subroutine EXIT.

CALLING SEQUENCE: FINDRM(LØCNØ,M)

LØCNØ is a pointer to a portion of the dynamic storage array where the data group that needs more room resides.
M is the address where the next data value is to be stored.

ERROR PROCEDURES: If the dynamic storage array is full, an error message is printed and the preprocessor is terminated via CALL EXIT.

STORAGE REQUIRED: 407 octal words = 263 decimal words. See Section 4.7.1. LABELED COMMON: BUCKET LØGIC PØINT, and TAPE.

4 - 14

4.2.10 SUBROUTINE NAME: GENLNK

PROGRAMMING LANGUAGE: FØRTRAN

<u>PURPOSE</u>: This subroutine generates the driver, FORTRAN routine name and SINDA, for the user's program.

- All

Sugar Har Sugar Bernard Barrena La

RESTRICTIONS: None

"TAPES" USED: Internal Scratch "tape" INTERN

SPECIAL FEATURES: None

OTHER SUBROUTINES USED: BLKCRD

CALLING SEQUENCE: GENLNK

ERROR PROCEDURES: None

LABELED COMMON: CRDBLK, DATA, LØGIC, PLØGIC, and TAPE.

4 - 15 35<

4.2.11 SUBROUTINE NAME: GENUK

PROGRAMMING LANGUAGE: FØRTRAN

PURPOSE: This subroutine is used to generate user constants.

<u>RESTRICTIONS</u>: The input data is taken from array TEMP in labeled common CHECKD and the output data (i.e., the generated user constants) is put into array B in labeled common BUCKET.

"TAPES" USED: None

SPECIAL FEATURES: None

OTHER SUBROUTINES USED: ERRMES, FINDRM, and TYPCHK.

CALLING SEQUENCE: GENUK(IWRDS)

IWRDSis the number of words to be processed in array TEMP.ERROR PROCEDURES:The input data is checked and if an error is found,control is transferred to subroutine ERRMES.

STORAGE REQUIRED: 451 octal words = 297 decimal words. See Section 4.7.1. LABELED COMMON: BUCKET, CHECKD, DATA, and PØINT.



an the River Stars

4.2.12 <u>SUBRCUTINE NAME</u>: INCØRE

PROGRAMMING LANGUAGE: FØRTRAN

<u>PURPOSE</u>: This subroutine reads data into the dynamic storage array for the parameter runs option.

RESTRICTIONS: None

"TAPES" USED: Dictionary "tape" LUT1 Parameter runs "tape" LUT3

SPECIAL FEATURES: None

OTHER SUBROUTINES USED: None

CALLING SEQUENCE: INCØRE(ITEST)

ITEST is an integer flag which determines the data group group to be read.

ERROR PROCEDURES: None

STORAGE REQUIRED: 600 octal words = 384 decimal words. See Section 4.7.1. LABELED COMMON: BUCKET, DATA, LØGIC, PLØGIC, PØINT, and TAPE.

4 - 17

4.2.13 SUBROUTINE NAME: MXTØFN

PROGRAMMING LANGUAGE: FØRTRAN

<u>PURPOSE</u>: This subroutine processes the data for the "M" option. That is, it converts card images from mixed F ϕ RTRAN/SINDA notation to F ϕ RTRAN notation.

<u>RESTRICTIONS</u>: The input (array IHØLL) and output (array JHØLL) are both in labeled common CIMAGE and they are both in an 80Al format. The FØRTRAN from array JHØLL is copied to array IMAGE under a 14A6 format for processing by the FØRTRAN compiler.

"TAPES" USED: None

SPECIAL FEATURES: The FORTRAN V intrinsic function FLD is used for bit manipulation.

OTHER SUBROUTINES USED: ALPINT and BLKCRD

CALLING SEQUENCE: MXTØFN

ERROR PROCEDURES: None

STORAGE REQUIRED: 522 octal words = 338 decimal words. See Section 4.7.1. LABELED COMMON: CIMAGE and CRDBLK



4.2.14 SUBROUTINE NAME: NØDEDA

PROGRAMMING LANGUAGE FØRTRAN

<u>PURPOSE</u>: This subroutine processes the data for the node and conductor data blocks.

<u>RESTRICTIONS</u>: The input is received via labeled common CHECKD: array TEMP and the processed data are stored in the dynamic storage array.

"TAPES" USED: None

SPECIAL FEATURES: This subroutine has a second entry point named CØNDDA. Also, the FØRTRAN V intrinsic function FLD is used for bit manipulation.

OTHER SUBROUTINES USED: ERRMES, FINDRM, RELACT, and TYPCHK

GALLING SEQUENCE: NØDEDA(JUMP, IWRDS)

or CØNDDA (JUMP, IWRDS)

JUMP is a flag which indicates which code option (columns 8, 9, and 10 of the data card) the user has selected.
IWRDS is the number of data values in array TEMP to be

processed.

ERROR PROCEDURES: If an error is detected while scanning the input data, control is transferred to subroutine ERRMES.

STORAGE REQUIRED: 7030 octal words 3608 decimal words. See Section 4.7.1. LABELED COMMON: BUCKET, CHECKD, DATA, FLAGS, and PØINT.

4.2.15 SUBROUTINE NAME: PCS2

PROGRAMMING LANGUAGE: FØRTRAN

<u>PURPOSE</u>: This subroutine packs the FØRTRAN addresses for the array and constants locations required by the pseudo-compute sequence.

FESTRICTIONS: None

"TAPES" USED: None

SPECIAL FEATURES: The FØRTRAN V intrinsic function FLD is used for bit manipulation.

OTHER SUBROUTINES'USED: None

CALLING SEQUENCE: PCS2 (IB, IPCS, LITA)

- IB is the word in the dynamic storage array where the addresses are found.
- IPCS is the word into which the addresses are packed.
- LITA is a flag that is set to 1 if the array address was input as a literal and therefore has been added to the constants data.

ERROR PROCEDURES: None

STORAGE REQUIRED: 54 octal words = 44 decimal words. See Section 4.7.1.

LABELED COMMON: None

4 - 20

4.2.16 SUBROUTINE NAME: PRESUB

PROGRAMMING LANGUAGE: FØRTRAN

<u>PURPOSE</u>: This subroutine reads and checks the block header cards for the operations blocks and generates the non-executable FØRTRAN cards for each of the operations blocks via a call to BLKCRD.

RESTRICTIONS: None

"TAPES" USED: System input "tape" NIN System output "tape" NØUT

SPECIAL FEATURES: None

OTHER SUBROUTINES USED: BLKCRD

CALLING SEQUENCE: PRESUB(N)

N

is an integer from 1 to 4 which indicates which operations block is being processed.

ERROR PROCEDURES: If the card read is not the correct block header card, an error message is printed and the fatal error flag is set to 1.0.

STORAGE REQUIRED: 200 octal words = 128 decimal words. See Section 4.7.1. LABELED COMMON: CRDBLK, DATA, LØGIC, and TAPE.



4.2.17 SUBROUTINE NAME: PSEUDØ

PROGRAMMING LANGUAGE: FØRTRAN

<u>PURPOSE</u>: This subroutine forms the first and second pseudo-compute sequence. See Section 4.6.

<u>RESTRICTIONS</u>: The necessary input is extracted from the dynamic storage array and the output (the two pseudo-compute sequences) is placed in the dynamic storage array.

"TAPES" USED: NØRT

<u>SPECIAL FEATURES</u>: The FØRTRAN V intrinsic function FLD is used for bit manipulation.

OTHER SUBROUTINES USED: FINDRM, PCS2 and WRTDTA.

CALLING SEQUENCE: PSEUDØ

ERROR PROCEDURES: If an error is encountered while forming the pseudocompute sequences, an error message will be printed and the non-fatal error flag (ERDATA) is set to 1.0.

STORAGE REQUIRED: 2244 octal words = 1188 decimal words. See Section 4.7.1. LABELED COMMON: BUCKET, DATA, LØGIC, PLØGIC, and TAPE.



4.2.18 SUBROUTINE NAME: QDATA

PROGRAMMING LANGUAGE: FØRTRAN

<u>PURPOSE</u>: This subroutine checks and processes all data input in the source data block.

<u>RESTRICTIONS</u>: The input is received from the calling sequence and labeled common CHECKD. The processed data is placed in the dynamic storage array.

"TAPES" USED: None

SPECIAL FEATURES: The FØRTRAN V intrinsic function FLD is used for bit manipulation.

OTHER SUBROUTINES USED: ERRMES, FINDRM, RELACT, and TYPCHK.

CALLING SEQUENCE: QDATA(CØDE, IWRDS)

CØDE is the three letter option from columns 8, 9, and 10 of the data card.

IWRDS is the number of words in array TEMP to be processed.

ERROR PROCEDURES: If an error is encountered, control is transferred to subroutine ERRMES.

STORAGE REQUIRED: 1655 octal words = 941 decimal words. See Section 4.7.1. LABELED COMMON: BUCKET, CHECKD, and PØINT.

이는 사이가 한 것에서 가지 않는 것이 없는 것이다.

na serve de la prime de la composition La composition de la c La composition de la c

4 - 23

4.2.19 SUBROUTINE NAME: RELACT

PROGRAMMING LANGUAGE: FØRTRAN

<u>PURPOSE</u>: This subroutine finds the relative node number from the actual node number. In addition, it computes the FØRTRAN address for arrays and user constants from the actual number.

<u>RESTRICTIONS</u>: This subroutine is used in conjunction with the data blocks.

"TAPES" USED: None

SPECIAL FEATURES: The FØRTRAN V intrinsic function FLD is used for bit manipulation.

OTHER SUBROUTINE USED: ERRMES

CALLING SEQUENCE: RELACT (K, MM, J, JJ)

- K determines the path through the program via a computedGØ TØ statement.
- MM is the actual number on entry, and the FØRTRAN address on return.

J are print variables for subroutine ERRMES.

ERROR PROCEDURES: In the event an error is encountered, control is transferred to subroutine ERRMES.

STORAGE REQUIRED: 311 octal words = 201 decimal words. See Section 4.7.1. LABELED COMMON: BUCKET, DATA, and PØINT.

> 4 - 24 A4<

4.2.20 SUBROUTINE NAME: SEARCH

PROGRAMMING LANGUAGE: FØRTRAN

<u>PURPOSE</u>: This subroutine returns a relative number for nodes, conductors, user constants, and arrays, given the actual number.

<u>RESTRICTIONS</u>: This subroutine is used in conjunction with the operations blocks.

"TAPES" USED: None

<u>SPECIAL FEATURES</u>: The FØRTRAN V intrinsic function FLD is used for bit manipulation.

OTHER SUBROUTINES USED: None

CALLING SEQUENCE: SEARCH(N, IA, NDIM, LØC)

- N is the actual number.
- IA is the first word of the dictionary of actual numbers to be searched.

NDIM is the number of words of IA to be searched.

LØC is the relative number returned to the calling program.

ERROR PROCEDURES: If the input actual number is not found in the dictionary, LØC is set to zero.

STORAGE REQUIRED: 101 octal words = 65 decimal words. See Section 4.7.1. LABELED COMMON: None

4 - 25

4.2.21 SUBROUTINE NAME: SETFMT

PROGRAMMING LANGUAGE: FØRTRAN

<u>PURPOSE</u>: This subroutine processes the cards for the "new format" option; that is, it sets up the format for data cards as specified by the cards with a N in column one.

<u>RESTRICTIONS</u>: The input/output array is passed through the calling sequence argument.

"TAPES" USED: FØRTRAN V reread 30

SPECIAL FEATURES: None

OTHER SUBROUTINES USED: None

CALLING SEQUENCE: SETFMT (JUMP, B)

JUMP is an integer flag that determines the path through the code.

is an array which contains the card images.

ERROR PROCEDURES: None

В

STORAGE REQUIRED: 221 octal words = 145 decimal words. See Section 4.7.1.

LABELED COMMON: None

4.2.22 SUBROUTINE NAME: SINDA4

PROGRAMMING LANGUAGE: FØRTRAN

<u>PURPOSE</u>: This subroutine reads and processes the user input cards from the operations blocks.

RESTRICTIONS: None

'TAPES" USED:	System input "tap	e"	MIN
	System output."ta	pe"	NØUT
	Internal scratch	"tape"	INTERN
	FØRTRAN V reread		30

SPECIAL FEATURES: None

OTHER SUBROUTINES USED: BLKCRD, MXTØFN, and SEARCH.

CALLING SEQUENCE: SINDA4 (NAME)

NAME is an integer flag that tells the subroutine which operations block is being processed.

ERROR PROCEDURES: In the event an error is encountered while processing the operations blocks, an error message is printed and the error flag PROGRM is set to 1.0.

STORAGE REQUIRED: 2372 octal words = 1274 decimal words. See Section 4.7.1. LABELED COMMON: BUCKET, CIMAGE, CRDBLK, DATA, LØGIC, PLØGIC, PØINT, and TAPE.

4 - 27

17<

and the second second

4.2.23 SUBROUTINE NAME: SKIP

FROGRAMMING LANGUAGE: FØRTRAN

<u>PURPOSE</u>: This subroutine is used when a problem is being RECALLed. It positions the tape to the proper problem as specified on the first card of the data deck.

RESTRICTIONS: The data is read from tape R. There is no output.

"TAPES" USED: RECALL "tape" LUT7

SPECIAL FEATURES: None

OTHER SUBROUTINES USED: None

CALLING SEQUENCE: SKIP

ERROR PROCEDURES: None

STORAGE REQUIRED: 324 octal words = 212 decimal words. See Section 4.7.1.

LABELED COMMON: TAPE

4 - 28 18<

4.2.24 SUBROUTINE NAME: SPLIT

PROGRAMMING LANGUAGE: FØRTRAN

PURPOSE: This subroutine reads the data from the RECALL tape and splits the RECALL information onto the program data "tape" (LB3D) and the dictionary "tape" (LUT1).

<u>RESTRICTIONS</u>: The input is from the RECALL "tape" and the output is placed on the program data "tape," the dictionary "tape," and the parameter runs "tape."

"TAPES" USED:	RECALL "tape"	LUT7
	Program data "tape"	LB3D
÷	Dictionary "tape"	LUT1
	Parameter runs "tape"	LUT3

SPECIAL FEATURES: None

OTHER SUBROUTINES CALLED: SKIP

CALLING SEQUENCE: SPLIT(ID)

ID is the RECALL name punched in the first card of the data deck.

ERROR PROCEDURES: None

STORAGE REQUIRED: 746 octal words = 486 decimal words. See Section 4.7.1. LABELED COMMON: BUCKET, DATA, and TAPE.

4.2.25 SUBROUTINE NAME: SQUEEZ

PROGRAMMING LANGUAGE: FØRTRAN

<u>PURPOSE</u>: This subroutine compresses the specified data groups in the dynamic storage array. The compression is accomplished by placing the data groups sequentially in the dynamic storage array.

RESTRICTIONS: None

"TAPES" USED: None

SPECIAL FEATURES: None

OTHER SUBROUTINES USED: None

CALLING SEQUENCE: SQUEEZ(IST, IEND)

IST is the data group number where the compression is to start.

IEND is the last data group number for this compression.

ERROR PROCEDURES: None

STORAGE REQUIRED: 115 octal words = 77 decimal words. See Section 4.7.1. LABELED COMMON: BUCKET and PØINT

4.2.26 SUBROUTINE NAME: STFFB

PROGRAMMING LANGUAGE: FØRTRAN

<u>PURPOSE</u>: This subroutine fills out a card image in array KBLK with Hollerith blanks.

<u>RESTRICTIONS</u>: The pointers to the words to set to blank are in the calling sequence, and the array containing the card images is in labeled common CRDBLK.

"TAPES" USED: None

SPECIAL FEATURES: None

OTHER SUBROUTINES USED: None

CALLING SEQUENCE: STFFB(I,J)

I is the first work in KBLK to set to blank.

J is the last word in KBLK to set to blank.

ERROR PROCEDURES: None

STORAGE REQUIRED: 41 octal words = 33 decimal words. See Section 4.7.1.

LABELED COMMON: CRDBLK.

4 - 31

4.2.27 SUBROUTINE NAME: TYPCHK

PROGRAMMING LANGUAGE: FØRTRAN

<u>PURPOSE</u>: This subroutine checks the input from the data blocks for the correct type; type means integer, floating point, or alphanumeric. Also, it regulates the conversion of the A's and K's for the automated options via a call to CØNVRT.

<u>RESTRICTIONS</u>: The input and output are transferred through the calling sequence arguments and labeled common CHECKD.

"TAPES" USED: None

SPECIAL FEATURES: The FØRTRAN V intrinsic function FLD is used for bit manipulation.

OTHER SUBROUTINES USED: CØNVRT and ERRMES

CALLING SEQUENCE: TYPCHK(JUMP, IERR, J)

JUMP	indicates what type the word should be.
IERR	tells subroutine ERRMES which error message to print
	if the word is not of the type indicated by JUMP.
J	is a pointer to the word type in array KFLFX.

ERROR PROCEDURES: If a word is not of the proper type control is transferred to subroutine ERRMES to print an error message and the logical flag CRDERR is set to true.

STORAGE REQUIRED: 233 octal words = 155 decimal words. See Section 4.7.1. LABELED COMMON: CHECKD

4.2.28 SUBROUTINE NAME: WRTDTA

PROGRAMMING LANGUAGE: FØRTRAN

PURPOSE: This subroutine writes the program data "tape" in the format required by INPUTT or INPUTG.

<u>RESTRICTIONS</u>: The data to be written on "tape" is found in the dynamic storage array.

"TAPES" USED: Program data "tape" LB3D

SPECIAL FEATURES: None

OTHER SUBROUTINES USED: None

CALLING SEQUENCE: WRTDTA(JUMP)

JUMP is an integer flag that indicates which data block to write and what format to use.

ERROR PROCEDURES: None

STORAGE REQUIRED: 645 octal words 421 decimal words. See Section 4.7.1.

4 - 33

53<

LABELED COMMON: BUCKET, DATA, LØGIC, PLØGIC, PØINT, and TAPE

4.2.29 SUBROUTINE NAME: WRTPMT

PROGRAMMING LANGUAGE: FØRTRAN

<u>PURPOSE</u>: This subroutine writes the data that is needed for parameter runs on the parameter runs "tape" and writes the dictionary "tape."

<u>RESTRICTIONS</u>: The information that is written on the "tapes" is found in the dynamic storage array.

"TAPES" USED: Dictionary "tape" LUT1 Parameter runs "tape" LUT3

SPECIAL FEATURES: None

OTHER SUBROUTINES USED: None

CALLING SEQUENCE: WRTPMT(JUMP)

JUMP is an integer flag that indicates to WRTPMT which set of information to write.

ERROR PROCEDURES: None

STORAGE REQUIRED: 401 octal words = 257 decimal words. See Section 4.7.1. LABELED COMMON: BUCKET, DATA, LØGIC, PØINT, and TAPE.



4.2.30 SUBROUTINE NAME: WRTBLK

PROGRAMMING LANGUAGE: Assembly Language

<u>PURPOSE</u>: This subroutine writes the 507 word blocks contained in array KBLK on the program FØRTRAN "tape."

RESTRICTIONS: None

"TAPES" USED: Program FØRTRAN "tape" LB4P

SPECIAL FEATURES: None

OTHER SUBROUTINES USED: None

CALLING SEQUENCE: WRTBLK

ERROR PROCEDURES: None

STORAGE REQUIRED: 14 octal words = 12 decimal words. See Section 4.7.1.

4 - 35

55<

LABELED COMMON: CRDBLK

4.3 LABELED CØMMØN VARIABLES

The SINDA preprocessor uses nine labeled common blocks to pass data and flags between the various subroutines. Labeled common names, in alphabetical order, are:

BUCKET	CHECKED	CIMAGE
CRDBLK	DATA	FLAGS
LØGIC	PLØGIC	PØINT

Note that the UNIVAC 1108 version does not utilize blank common. The two sections that follow give: 1) a map of the labeled common usage by subroutine name and by overlay link; 2) a definition of the variables used within each labeled common block; and 3) dynamic storage structure.

4.3.1 Labeled Common Map

The map below gives the labeled common name, a list of the overlay links that use it by link number and a list of the routines that use it.

LABELED CØMMØN NAME	ØVERLAY LINK NAMES	RØ N	UTINE AMES
BUCKET	0, 1, 2, 4, 5	ALPINT	CØDERD
		DATARD	FINDRM
		GENUK	incøre
		NØDEDA	PREPRØ
		PSEUDØ	QDATA
		RELACT	SINDA4
		SPLIT	SQUEEZ
		WRTDTA	WRTPMT
CHECKD	1	DATARD	GENUK
		NØDEDA	QDATA
		ТҮРСНК	
CIMAGE	4	ALPINT	MXTØFN
		SINDA4	
CRDBLK	0, 3, 4	BLKCRD	GENLNK
		MXTØFN	PREPRØ
		PRESUB	SINDA4
		STFFB	WRTBLK

LABELED CØMMØN NAME	ØVERLAY LINK NAMES	RØUTINE NAMES
DATA	0, 1, 2, 3, 4, 5	ALPINT CØDERD
		DATARD ERRMES
		GENLINK GENUK
		incøre nødeda
		PREPRØ PRESUB
		PSEUDØ RELACT
		SINDA4 SPLIT
•		WRTDTA WRTPMT
FLAGS	ĺ	DATARD NØDEDA
LØGIC	0, 1, 2	CØDERD DATARD
•		FINDRM INCØRE
		PREPRØ PSEUDØ
		WRTDTA WRTPMT
PLØGIC	0, 1, 3, 4	CØDERD DATARD
		GENLNK INCØRE
		PREPRØ SINDA4
		WRTDTA
PØINT	0, 1, 2, 4	ALPINT CØDERD
		DATARD FINDRM
		GENUK INCØRE
		NØDEDA PREPRØ
		PSEUDØ QDATA
		RELACT SINDA4
		SQUEEZ WRTDTA
		WRTPMT
TAPE	0, 1, 2, 3, 4, 5	ALPINT BLKCRD
	,	CØDERD DATARD
		ERRMES FINDRM
		GENLNK INCØRE
		PREPRØ PRESUB
	ی د. ۲	PSEUDØ SINDA4
		SKIP SPLIT
		WRTDTA WRTPMT

- 4.3.2 Definition of Labeled Common Variables
 - (1) Labeled common name BUCKET.

BUCKET is the dynamic storage array (see Section 4.3.3).

(2) Labeled common name CHECKD

CHECKD is used to temporarily store and check the user's input data.

VARIABLE NAME	DESCRIPTION
TEMP (35) or ITEMP	A temporary storage array, which contains the user's input data as read from the data cards.
XGEN(35) or IGEN	A temporary storage array used to store a copy of TEMP when the user is generating data.
KFLFX(35)	An indicator used to check the data array which contains one of the following numbers:
	<pre>l = floating point number 0 = integer number -l = Hollerith word</pre>
CRDERR	A logical flag: Set true if and only if an error was found on the data card now being processed.

(3) Labeled common name CIMAGE

CIMAGE is used to store and manipulate Hollerith card images for the 'M' option.

VARIABLE NAME	DESCRIPTION
IHØLL(80)	The input card image, read under an 80Al format.
JHØLL(160)	The constructed card image, also an 80A1 format.
(4) Labeled common name CRDBLK

CRDBLK is used to construct the five generated FØRTRAN routines.

VARIABLE NAME	DESCRIPTION
LSTART	A logical flag that signals the start of a new routine if set to true.
LECARD	A logical flag that signals the end of a routine.
LCØPY	A logical flag that tells the program to copy the card image (14A6) found in IMAGE to the next available slot in KBLK.
NW	Is a counter whose value is the next available word in KBLK.
KBLK(507)	An array that contains FØRTRAN card images of the generated routines.
IMAGE(14)	An array that contains one card image to be copied into KBLK.

(5) Labeled common name DATA

DATA is used to store the counters that indicate (to the program) how many of each data type has been encountered. In addition, it contains three error flags.

VARIABLE NAME	DESCRIPTION
NND	The number of diffusion nodes.
NNA	The number of arithmetic nodes.
NNB	The number of boundary nodes.
NNT	The total number of nodes.
NGL	The number of linear conductors.
NGR	The number of radiation conductors.
NGT	The total number of conductors.
NÙC	The number of user constants.
NEC1	The number of added constants from automated options in the node data block.

NEC2 The number of added constants from the automated options in the conductor data block. NCT The total number of constants. LENA The total number of words used in the array data block. ERDATA The non-fatal error flag for the data blocks. ERDATA \neq 0.0 means an error has been found. PRØGRAM The non-fatal error flag for the operations blocks. PRØGRAM \neq 0.0 indicates an error condition. ENDRUN The fatal error flag for the preprocessor. ENDRUN \neq 0.0 signals the program to terminate immediately. LSEQ1 The length of the first pseudo-compute sequence.

LSEQ2 The length of the second pseudo-compute sequence.

LØNG A logical flag set to true if the user is requesting the long pseudo-compute sequence.

(6) Labeled common name FLAGS

FLAGS contains three flags that are used to go to the proper block of coding in subroutine NØDEDA.

VARIABLE NAME	DESCRIPTION	
LEAP	Used with the GEN option.	
NØNLIN	Flags a set of multiply connect ductors as radiation, if set t	ted con-
INDX	Determines path when multiply connected conductors require more than one data card.	

(7) Labeled common name LØGIC

LØGIC contains a number of logical flags and the fifty fixed constants.

	ti da	
	VARIABLE NAME	DESCRIPTION
	LNØDE	Set to true if any node data was processed.
	LCØND	Set to true if any conductor data was processed.
	lcønst	Set to true if any user constants were processed
	LARRAY	Set to true if any array data was processed.
	LPRINT	Debug print flag, set to true if there is an asterisk in column 80 of the BCD 3 THERMAL/GENERAL card.
	KBRNCH	An integer that specifies which data block is being processed.
or	FIXC(50) IFIXC	The array that contains the fixed (control) constants.
	KTPRNT	Optional print flag for a list of relative versus actual user constant numbers. Set time if there is an asterisk in column 80 of the BCD 3CØNSTANTS DATA card.
	AYPRNT	Optional print flag for a list of actual array numbers versus FØRTRAN address. Set true if there is an asterisk in column 80 of the BCD 3ARRAY DATA card.
	GENERL	Set true for a general problem.
	LQ	Set true if any data was processed from the source data block.
(8)	Labeled common na	me PLØGIC

 \mathbf{or}

 $PL \ensuremath{\texttt{GIC}}$ contains a number of logical flags that are used in conjunction with parameter runs.

VARIABLE NAME	DESCRIPTION
PARINT	Set true for initial parameters run.
PARFIN	Set true for final parameters run.
PNØDE	Set true if node data was processed.
PCØND	Set true if conductor data was processed.
PCØNST	Set true if user constants data was processed.

PARRAY	Set true if array data was processed.
PTITLE	Set true if a new title was input.
PCHGID	Contains the alphanumeric word INITIAL or FINAL to be used as the run identification on "tape" LB3D.

(9) Labeled common name PØINT

P¢INT is used in conjunction with dynamic storage array BUCKET. See Section 4.3.3.

4.3.3 Dynamic Storage Structure

Dynamic storage represents one of the techniques of maximizing problem size with a computer with finite core. In dynamic storage each data set is placed sequentially into <u>one</u> array end-to-end. This eliminates the wasted core inherent with the traditional system of dimensioning each variable at some fixed length. However, the price paid for the additional core is the extra time required to compute the address of a variable.

The SINDA preprocessor used three arrays to store and address the data sets. The data sets are stored in an array named B, or IB, or BB. This array resides in labeled common BUCKET. The length at which B can be dimensioned depends on the system that the computer facility uses. At NASA MSC approximately 30,000 words are allocated to B. In addition, in labeled common PØINT there are two arrays named LØC and LEN, each dimensioned at 20. LØC (I) contains the starting location in B for the Ith data set and LEN (I) contains the length of the Ith data set.

The information below gives, in detail, the contents of the dynamic storage array for each data block as it exists just after the data block has been processed.

(1) Node data block

data set 1:

bit 1,	automated option flag		
bit 2,	Q from SØURCE DATA flag		
bits 16-35,	actual node number		

data	set 2:	
	bits 0-35,	temperature value
data	set 3:	
	bits 0-35,	capacitance value
data	sot h.	
uata	bite $0-5$	non-linear canacitance type
	bit 6	literal error flag
, 7 [°]	bite $7-20$	actual array number
	bit 21	literal constant flog
	bite 22_35	Actual constant number
	DICS 22-33,	actual constant number
data	set 5:	
	bits 0-35,	literals encountered in 4.
Sourc	ce data block	
data	set 2 (first word	of group):
	bits 0-5,	source option type
	bits 6-20,	relative node number
	bits 21-35,	not used
data	set 2 (second word	d of group):
	bits 0-5,	not used
	bit 6,	literal array flag
	bits 7-20,	actual array number
	bit 21,	literal constant flag
	bits 22-35,	actual constant number
data	set 3:	
	bits 0-35,	literals encountered in 2
Condu	uctor data block	
data	set 6:	
	bits 0-35.	actual conductor number
data	set /:	
•	Dit U,	multi connections flag
	Dit 1,	radiation flag
	DIT 2,	automated option flag
	data data data data data data data data	<pre>data set 2: bits 0-35, data set 3: bits 0-35, data set 4: bits 0-5 bit 6, bits 7-20, bit 21, bits 22-35, data set 5: bits 0-35, Source data block data set 2 (first word bits 0-5, bits 6-20, bits 21-35, data set 2 (second word bits 0-5, bits 0-5, bits 0-5, bit 6, bits 7-20, bit 21, bits 22-35, data set 3: bits 0-35, Conductor data block data set 6: bits 0-35, data set 7: bit 0, bit 1, bit 2,</pre>

(2)

(3)

bits 3-5,	not used
bit 6,	1 way flag for NA
bits 7, 20,	relative node number NA
bit 21	1 way flag for NB
bits 22-35,	relative node number NB

data set 8:

bits 0-35,

conductance value

data set 9:

bits 0-5, conductor option type bit 6, literal array flag bits 7-20, actual array number bit 21, literal constant flag bits 22-35, actual constant number

data set 10: bits 0-35,

literals encountered in 9

(4) Constants data block

data set 11: bits 0-35

bits 0-35, actual constant number

data set 12: bits 0-35,

constant value

(5) Array data block

data set 13:

bits 0-35,

bits 0-35,

data set 14:

array length

actual array number

data set 15: bits 0-35,

array value

(6) Pseudo-compute sequences

data	set	16	(1st pseudo-compute sequence):	
	bit	0,	last conductor flag	
	bit	1,	automated capacitance f	1ag
	bit	2,	automated conductance f	lag

radiation conductance flag
Q from source block flag
relative conductor number
1 way conductor flag
relative adjoining node number

data set 17 (second pseudo-compute sequence):
 bits 0-4, automated option type
 bit 5, not used
 bits 6-21, FØRTRAN address for array
 bit 22, not used
 bits 23-35, relative constant number

The bit numbering convention above conforms to the UNIVAC standard notation, where each 36 bit word is numbered 0 through 35 from left to right. Each of the 1 bit flags above is querried in the following manner: 0 means NO, and 1 means YES. If the literal array flag or the literal constant flag is set to 1, then the bits immediately to the right of the flag do not contain the actual array or constant number. Instead, they contain a pointer to the next data set where the literal value is stored. In those data sets that store information for the automated options it is sometimes necessary to use more than one word per option. When this is the case, the automated option type (bits 0-5) is set to 0.

4.4 SINDA "Tapes" and Their Formats

The SINDA program in its normal operating mode utilizes six "tapes." Five of these "tapes" are assigned by the program and the remaining one contains the program; it is assigned via control cards. The store and recall options require one additional "tape" each and the NASA edit feature requires two additional "tapes." The following paragraphs contain information on the five normal SINDA "tapes."

4.4.1 LB3D - Program Data "Tape"

This "tape" is set up by the preprocessor (WRTDTA) and read by INPUTT, for a thermal problem, or INPUTG, for a general problem, just prior to performing the instructions of the execution block. The contents of this unit are:

- Problem identification.
 WRITE(LB3D)RUNID
- (2) Title information (20 words).
 WRITE(LB3D)(TITLE(I), I=1, 20)
- (3) The number of: diffusion nodes, arithmetic nodes, and total nodes; followed by a temperature value for each node; then a capacitance value for each diffusion, if any. WRITE(LB3D)NND,NNA,NNT,(T(I),I=1,NNT) IF(NND.GT.0)WRITE(LB3D)(C(I),I=1,NND)
- (4) The total number of conductors followed by a conductor value for each one.

WRITE(LB3D)NGT, (G(I), I=1, NGT)

(5) The total number of user constants are followed by the 50 control constant values; then the user constants values, if any.

WRITE(LB3D)NCT, (FIXC(I), I=1, 50)
IF(NCT.GT.0)WRITE(LBJD)(K(I), I=1, NCT)

- (6) The total number of arrays and the overall length of the array data; then the array values, if any. WRITE(LB3D)NAT,LENA
 - IF (LENA.GT.O)WRITE(A(I), I=1, LENA)

(7) The lengths of the first and second pseudo-compute sequences, followed by the data for the first pseudo-compute sequence; then the data for the second pseudo-compute sequence, if any. WRITE(LE3D)LSEQ1,LSEQ2, (P1(I), I=1,LSEQ1) IF(LSEQ2.GT.0)WRITE(LE3D) (P2(I), I=1,LSEQ2)

Note that (3), (4), and (7) above apply only to a thermal problem.

4.4.2 LB4P - Program FØRTRAN "Tape"

This "tape" is especially formatted in 507 word blocks as required by the FØRTRAN compiler. Where:

- WORD 1 on the first block of each routine contains the name of the routine.
 - WORD 2 contains the integer number of card images in the block.

WORDS 3 - 506 contain the card images

- WORD 507 is set to +0 except on the last block of each routine where it is set to -0.
- 4.4.3 INTERN Preprocessor Scratch "Tape"

Generally INTERN is used to pass card images to subroutine BLKCRD under a 14A6 format.

4.4.4 LUT1 - Dictionary "Tape"

This "tape" contains a list of the actual SINDA numbers in a relative order. That is, the actual node number corresponding to the kth relative node number is the kth item of the node number dictionary. The format of this "tape" is:

 The total number of nodes, followed by an actual node number for each node.

WRITE(LUT1)NNT, (NN(I), I=1, NNT)

(2) The total number of conductors, followed by the list of actual conductor numbers.

WRITE(LUT1)NGT, (NG(I), I=1, NGT)

4 - 47

a ta Alexandre San Alexandre a completion and

- (3) The number of user constants, the total number of constants, followed by a list of the actual constant numbers. WRITE(LUT1)NUC,NCT, (NK(I), I=1,NCT)
- (4) The total number of arrays followed by a list of the actual array numbers, then the total number of arrays followed by a list of the length of each array.

WRITE(LUT1)NAT, (NA(I), I=1, NAT) WRITE(LUT1)NAT, (LA(I), I=1, NAT)

4.4.5 LUT3 - Parameter Runs "Tape"

This "tape" contains some data from the original problem. It is required by the initial parameters capability. The format of "tape" LUT3 is:

- (1) The original title.
 WRITE(LUT3)(TITLE(I), I=1,20)
- (2) A list of original temperature and capacitance values.
- WRITE(LUT3)NND, (T(I), I=1, NNT) IF(NND.GT.0)WRITE(LUT3)(C(I), I=1, NND)
- (3) A list of the original conductor values. WRITE(LUT3)(G(I), I=1, NGT)
- (4) Lists of the original fixed and user constants.
 WRITE(LUT3)NUC, NCT, (FIXC(I), I=1,50)
 IF(NCT.GT.0)WRITE(LUT3)(K(I), I=1, NCT)
- (5) The original array values. WRITE(LUT3)NAT, LENA IF(LENA.GT.0)/RITE(LUT3)(A(I), I=1, LENA)

4.5 Overlay Structure

• .

The SINDA preprocessor has an overlay structure composed of a main link (designated LINKO below) which is always in core and five sublinks (designated LINK1, LINK2, LINK3, LINK4, LINK5, and LINK6 below) which overlay one another as they are brought into core.



Note that the first subroutine listed above in each of the sublinks serves as the driver for that sublink and it is also the subroutine called from PREPRØ.

4 - 49

مصغو معمد للبركان بالالالا لأنجاه

Another approach to overlay specification is to think of each link as a functional unit, hence the graph below.



4 - 50

Z.

 \overline{i}_{i}

4.6 Structure of Pseudo-Compute Sequences

4.6.1 Descriptions

The use and structure of the two pseudo-compute sequences generated by the SINDA preprocessor appear to be rather confusing and mysterious. The term "pseudo" itself leads to immediate interpretation difficulties. Suppose that an element G_k between nodes i and j is to be identified for the ith node; by specifying explicitly i,j, and k the element is completely defined. On the other hand, SINDA explicitly specifies j and k but i is implicit in the DØ-LØØP; hence, the description pseudo-compute sequence (PCS) arises. Confusion also arises from the lack of information regarding the need for the (PCS) and the difficulties in reading the packed information. In short, the PCS as used in SINDA is simply two lists of relative numbers which are ordered in a specific manner. The two lists of relative numbers form the heart of the PCS, although other information pertinent to the computation must also be considered.

The PCS is necessary because the data as input by the user does not lend itself efficiently to the computational capabilities of FØRTRAN. As a result, the preprocessor scans the user input data and places the relative numbers (FØRTRAN addresses) into an array in the order in which the data will be used at a later time by the user selected numerical solution routine.

Packing of the data is a technique that conserves computer storage by placing two or more pieces of information in one computer word. This allows the user to execute a larger problem than the one that can be accommodated if the traditional one computer word for each piece of information approach. The penalty for this larger problem capability is an increase in execution time required for the extraction of information each time it is used.

4.6.2 Structure of PCS1

The first PCS, designated PCS1, contains the following information:

bit 0 last G for this node flag bit 1 automated C option flag bit 2 automated G option flag bit 3 radiation G flag bit 4 Q from source data flag bits 5-20 relative G number bit 21 1 way G flag bits 22-35 relative adjoining node number

The 36 bits of each computer word are numbered 0 through 35 from left to right. All of the 1 bit flags are set such that 0 means NØ and 1 means YES.

PCS1 is stored in an array named NSQ1 and is ordered by relative node number. That is, for relative node number 1 the conductor data is scanned and each time a conductor connected to node number 1 is encountered the PCS1 information is stored in NSQ1. When all of the conductor data has been scanned, the 0 bit of the latest word of PCS1 information is set to 1. This process is repeated for relative node numbers 2, 3, etc., until all diffusion and arithmetic nodes have been processed. PCS1 will be formed as either long or short as specified by the user on the BCD 3THERMAL card. This option is applied to diffusion nodes only since arithmetic nodes are always formed under the long option. The difference between the long PCS and the short PCS is that in the long PCS, each conductor will be listed twice, whereas in the short PCS each conductor will be listed once. This assumes the conductor connects two diffusion nodes. If one or both of the nodes is arithmetic, then the conductor will be listed twice, and if one of the nodes is a boundary the conductor will only be listed once. For example, given conductor number k which connects diffusion nodes i to j, where i < j. The long PCS would contain the k, j information for the processing of node i, and the k, i information for the processing of node j; whereas, the short PCS would only contain the k,j information. The short PCS thus has the advantage of requiring less computer storage than the long PCS, but a block iterative method (refer to Section 5.2.2) must be used; in general, the short PCS requires more iterations to converge than the successive point iterative (refer to Section 5.2.2) method which requires the long PCS.

4.6.3 Structure of PCS2

The second PCS is designated PCS2. The following information is stored whenever bit 1, bit 2, or bit 4 of PCS1 is set to one.

bits 0-4	automated option code
bit 5	rot used
bits 6-21	FØRTRAN address of the array or relative
	constant number.
bit 22	not used
bits 23-35	relative constant number

If the automated option is a doublet type, like DIV, and therefore requires two words to store the information, the automated option code on the second word is set to zero. In the event that more than one of the flag bits (bits 1, 2, or 4) of PCS1 is set to one, then the following order is imposed on PCS2: the capacitance information is stored first, the source block information second and finally the conductor information.

The PCS2 information is stored in array named NSQ2. This array is the same under the long PCS1 or the short PCS1 since automated conductors are only flagged on their first encounter.

4.7 Other Information

This section contains miscellaneous information that may be of interest to the user.

4.7.1 Subroutine Lengths

The storage required by a particular routine will vary depending on the type of computer and the system being used. The routine lengths given in Section 4.2 are based on compiler listings made on 23 January 1971 at Jacobi Computation Center.* The machine is a UNIVAC 1108 with a highly modified system. The numbers represent the sum of the computer storage for computer instructions, constants, and simple variables.

4.7.2 <u>Maximum Thermal Problem Size and Maximum Data Value Size</u>

A short formula for estimating the maximum thermal problem size that can be run on SINDA, and a list of the maximum size of the various data values is given below.

* Now called Computation and Systems Corporation, Los Angeles, California.

Estimation of Maximum Problem Size

NNT + 3*NGT + NCT + $4*NA\emptyset \leq LENBKT$

where,

NNT	is the total number of nodes.
NGT	is the total number of conductors.
NCT	is the total number of constants (user
	constants plus literals from automated options).
naø	is the number of automated options specified.
- LENBKT	is the length of the dynamic storage array as
	set in routine PREPRØ.

Maximum Size of Data Values

Actual node number			
Core storage	2 ³³ -1		
Print out	999,999		
Relative node number			
Core storage	16,383		
Temperature			
Core storage	$\pm 10^{38}$		
Capacitance	30		
Core Storage	$\pm 10^{30}$		
Relative conductor number	<u> </u>		
Core storage	2 ⁵⁵ -1		
Actual user constants number			
Core storage	32,767		
Automated options	16,383		
Relative user constants number			
Core storage	32,767		
Automated options	8,191		

User constant values	
Integer	$\frac{1}{2} 2^{35} - 1$
Floating point	$\pm 10^{38}$
Alphanumeric	6 characters
Actual array number	
Core storage	2 ³⁵ -1
Automated options	16,383
Print out	99,999
Relative array number	
Core storage	2 ³⁵ -1
Automated options	65,535
Print out	99,999
Array values	
Integer	$\pm 2^{35} - 1$
Floating point	$\pm 10^{38}$
Alphanumeric	6 characters

Note that some of the maxima, such as the relative conductor number of 2^{35} -1, are strictly academic since the dynamic storage array is considerably smaller than the indicated maximum data value size.

*)

5.

REVIEW OF LUMPED PARAMETER EQUATIONS AND BASIC NUMERICAL SOLUTIONS

The use of SINDA as mentioned in a previous section is based on a lumped parameter representation of a physical system.⁷ Thus SINDA solves numerically a set of ordinary (in general nonlinear) differential equations that represent the transient behavior of a lumped parameter system or a set of nonlinear algebraic equations representing steady state conditions. Numerous numerical solution techniques are reported in literature; a few of these are listed in the Reference Section.⁸⁻²⁴ These numerical methods are based on finite difference algorithms as opposed to finite element methods which have received considerable attention recently.²⁵⁻²⁹ For problems that are generally encountered in spacecraft thermal design, use of the finite element method appears to be inappropriate because of the nonlinearity presented with radiation heat transfer and because of complex geometric configurations.

Variations of the basic finite difference algorithms are numerous because no single numerical solution technique is optimum for all the endless types of thermal problems that can be encountered. Furthermore, because of the nonlinearity of the problems, a specific set of criterions to indicate solution accuracy and stability is not available and does not appear to be forthcoming. As a result, the user is placed in a rather awkward and confused position of not knowing which subroutine to use if a choice is available. Some thermal analyzer-type computer programs allow no choice, as a result, user decision is not necessary. SINDA represents a computer program at the other extreme of user decision flexibility by providing a number of numerical solution methods.

The intent of this Section 5 is to review and formulate the basic numerical solution methods with the presentation (from an engineering standpoint) of the characteristics of each SINDA numerical solution routine deferred to Section 6. In addition to place the use of SINDA in a proper perspective relative to accurate temperature prediction of a physical system, difficulties associated with lumped-parameter representation are discussed here.

5.1 Lumped Parameter Representation

Reduction of a distributive (physical) system to a lumped system which can be represented as an equivalent thermal network is a rather important phase of thermal analysis. From a temperature accuracy standpoint lumping (or nodalization) of the physical system may be far more important than a numerical solution technique that is used in a computer program. The latter is often given undue attention with apparent ignorance of other error sources which may be far more important. A general discussion on lumped parameter representation is not intended for presentation here since the subject material is extensively covered in technical literature, but it is convenient for continuity to indicate basic considerations.

For simple geometries and linear problems, it is rather straightforward to solve the partial differential equations of the type,

	<u>9</u> 9	$\frac{T}{t} =$	$\alpha \nabla^2 \mathbf{T} + \mathbf{Q}$			(5.1-1)
where,	a.	-	thermal diffusivity	(k/C)		
	Ŧ	-	temperature			
	Q	=	source			
	\bar{v}^2	-	$\frac{\partial^2}{\partial x^2} + \frac{\partial^2}{\partial y^2}$	(two	dimensional)	

Numerous analytical solutions of (5.1-1) for different types of boundary conditions and geometries are available.³⁰,³¹ Finite difference algorithms formed directly from the partial differential equations are also abundantly reported in literature.¹²,¹⁴ These finite difference formulations were generally developed for well-defined geometries and symmetrical discretization. For these problems, the so-called nodal connections or resistances are immediately available and, in general, automatically generated by the computer program. Thus, the need for a lumped parameter representation does not exist. For these types of problems, inaccuracies due to truncation and solution stability are specifically established.

For complex geometries and nonlinear problems such as those that include thermal radiation exchange, analytical solutions of thermal problems are limited.³²,³⁴ As a result, it is a common practice because of practical considerations to nodalize a physical system directly with-

5 - 2

out undue consideration of inaccuracies. Thus, a user merely represents the heat flow between two connecting nodes by using the basic network building block,

$$I_{ij} = (T_i - T_j)/R_{ij}$$
 (5.1-2)

where,

 R_{ij} represents an effective resistance between adjoining nodes i and j.

It should be particularly noted here that SINDA employs the concept of conductance in lieu of resistance which is common with most network-type computer programs. Thus the heat flow is represented as:

$$q_{ij} = G_{ij}(T_i - T_j)$$
 (5.1-3)

where, G_{ii} is the conductance between node i and node j.

The proper value of G_{ij} (or R_{ij}) for an arbitrary nodalization is (and should be) of concern to the user but because of the multitude of variables that must be considered, any discussion here would be incomplete. An excellent article on asymmetrical finite difference networks is presented in Reference 35.

5.1.1 Some Thoughts on Lumped Parameter Errors

Reduction of a physical system to a topological model consisting of a network with resistors and capacitors requires considerable engineering judgment. More often than not, nodal size of a model is governed by budget and schedule constraints. As a result, the discrete areas larger than desired are often used. This does not necessarily mean, however, that the use of a large number of nodes will always yield realistic results since the uncertainties of the input parameters can be appreciable.³⁶

Spatial truncation errors are controlled by selecting the grid size so that nonlinear temperature distributions lie within required accuracy by linear interpolation between nodal points, and that variation of temperaturedependent properties over the volume of each node is within required limits of the average values determined for the nodal point. The assumption of linear temperature distribution assumed for the lumped-parameter equations (equation 5.1-6) leads to a spatial truncation error of the order $O(\Delta x^2)$ only if all nodes are symetrically located.²⁷ If a non-uniform grid is used, the accuracy of computation is only $O(\Delta x)$. Spatial truncation errors are thus inherent in the mathematical model and beyond user control once inputted into

SINDA. For a two-dimensional problem with symmetrical grids, the spatial truncation error can be expressed for typical explicit and implicit methods 25,^{12,27}

$$E = -\frac{(\Delta x)^2}{12} \frac{\partial^4 T}{\partial x^4} - \frac{(\Delta y)^2}{12} \frac{\partial^4 T}{\partial y^4} + O(\Delta x^4)$$
 (5.1-4)

Temperature distribution other than linear can also be formulated;³⁸ however, most thermal analyzer-type computer programs such as TRUMP³⁹ and including SINDA are based on the linear assumption.

Time truncation errors are directly dependent upon the time-step since the error for the typical explicit and implicit method is,

$$E = -\frac{\Delta t}{2} \frac{\partial^2 T}{\partial t^2}$$
(5.1-5)

Normally the time step is dependent upon a particular criterion chosen by the user. A more detailed discussion on user control of the time step will be given for each numerical solution routine within the SINDA subroutine library.^{3,4}

Another approximation error which is due to discretizing is the assumption of constant radiosity for the discrete areas. Inaccuracies can be expected to affect the level and distribution of temperature. The analysis of thermal radiation exchange has received considerable attention in recent years because of its importance in spacecraft thermal design. 40, 52 The influence of non-uniform local heat flux on overall heat transfer between a gray differential area parallel to a gray infinite plane is examined in Reference 43; the assumption of uniform local heat flux appears to be reasonable for this geometry and for the evaluation of the overall heat flux calculations. A method of analysis suitable for engineering applications is developed in Reference 50 for computing local radiant flux and local temperature of opaque surfaces in a space environment. A study evaluating the validity of commonly used simplified methods of radiant heat transfer analysis is reported in Reference 48. A study directed at improving the understanding and prediction of orbiting spacecraft thermal performance is presented in References 46 and 49. A method presented in Reference 51 provides a means of evaluating the uncertainties associated with thermal radiation exchange. For an excellent status review (as of 1969) on radiation exchange between surfaces and in enclosures, the reader should consult Reference 52.

The above discussion merely serves to indicate that considerable care must be given when nodalizing a physical system and that the numerical evaluation of the finite difference equations must be considered from the total temperature error context. This means that user attention to a given numerical solution must be placed in a proper perspective.

5.1.2 Lumped-Parameter Equations

Using the network building block as expressed by equation (5.1-3) the lumped parameter system is identified as a set of ordinary non-linear differential equations by taking a heat balance as an ith node,

$$\frac{dT_{i}}{dt} = \frac{1}{C_{i}} \left[q_{i} + \sum_{j=1}^{p} a_{ij} (T_{j} - T_{i}) + \sum_{j=1}^{p} \sigma b_{ij} (T_{j}^{4} - T_{i}^{4}) \right]$$
(5.1-6)

i = 1,2,...,N (number of variable temperatures T_j = constant, N < j $\leq p$

where, C_i = the ith nodal capacity which may be a function of temperature

- q = the heat into node i and may be a function of time and temperature (impressed)
- a_j = the conduction coefficient between nodes i and j; it
 may be a function of time and temperature

bij = the radiation coefficient between nodes i and j; it
may be a function of time and temperature

 σ = Stefan-Boltzmann constant

Coefficients a_{ij} and b_{ij} are SINDA input quantities with the temperature factor of equation (5.1-6) calculated internally by the program. Both a_{ij} and b_{ij} may be variables. Conductance updating is a subject for discussion in a later paragraph. The user requirement to input the coefficients, a_{ij} and b_{ij} , provides considerable program flexibility, but at the same time user generation of these input quantities presents, in some instances, rather difficult engineering judgment decisions.

Radiation coefficient b_{ij} is, in essence, a radiation interchange factor, 3_{ij} , 3-55 (also known as script F) between nodes i and j. Generation of this quantity analytically can be quite difficult and inaccurate. A number of methods and computer programs (see, for example,

Reference 56) are available for evaluating the shape factors which represent an important part of determining script F. A direct generation of script F is normally through the use of the Monte Carlo technique,⁴⁹ but a recent development utilizes a matrix formulation for determining the script F in an enclosure containing surfaces with arbitrary emission and reflection characteristics.^{57,58} An experimental technique is reported in Reference 59.

5.2 Basic Finite Difference Formulations

The various numerical solution techniques differ in the finite difference formulations for the time-derivative (refer to equation 5.1-6); since the thermal equation is of the parabolic type, the transient heat transfer problems are of the initial value type. This means that at some time point, $t = n\Delta t$ (n is the number of time steps, Δt) all values of T_i are known. Thus,

$$T_{i,n+1} = T_{i,n} + \begin{bmatrix} dT_{i,n} \\ dt \end{bmatrix}_{t_{n,n+1}} \Delta t$$
 (5.2-1)

where $t_{n,n+1}$ represents the time interval between $t = n\Delta t$ and $t = (n+1)\Delta t$

It is apparent that the selection of the proper value to $(dT_{i,n}/dt)t_{n,n+1}$ cannot be explicit and its selection identifies one numerical method from another. Although many finite difference formulations of the parabolic differential equation are available, two general classifications are commonly denoted as explicit or implicit. These numerical methods are well-documented in literature; the reader should refer to Reference 12 for a comprehensive discussion on various finite difference approximations. Explicit methods also discussed in References 14, 17, 19 and 20, among others, are step-by-step in time and equations.

Explicit methods include:

(1) Forward-difference explicit approximation¹², ¹⁴

This is an Euler method that computes temperatures in a step-by-step fashion. The requirement of stability places an upper limit on the time increment. SINDA subroutines CNFRWD, CNFRDL and CNFAST fall within this

category. CNFAST is a modified version of CNFRWD which allows the user to specify the minimum time step to be taken. Refer to Sections 6.3.1 and 6.3.2 for details.

(2) Dufort-Frankel approximation⁹, ¹², ¹⁷

The Dufort and Frankel finite difference formulation is a three level formula that appears to be unconditionally stable. SINDA subroutine CNDUFU uses the Dufort-Frankel finite difference algorithm (refer to Section 6.3.4 for a detailed discussion).

(3) Exponential approximation 1, 17

The exponential approximation is found by integrating the heat balance equations after making linear and constant coefficient assumptions. This method is unconditionally stable for linear systems but may be unstable for some types of nonlinear problems. SINDA subroutine CNEXPN employs this method and is discussed at length in Section 6.3.3.

(4) Alternating direction approximations¹⁷

This technique employs two formulations, one on odd time levels and the other on even time levels and is unconditionally stable.

The implicit finite difference formulations require a simultaneous computational procedure. In addition to Reference 12, implicit methods are also discussed in References 8, 10, 17, and 20, among others. Implicit methods include:

(1) Backward difference implicit approximation¹²

The backward difference weights only the flux terms at $t = (n+1)\Delta t$. As a result, the method is stable for all values of Δt . SINDA subroutine CNBACK employs this method and is detailed in Section 6.4.1.

(2) Crank-Nicolson approximation⁸

The Crank-Nicolson method uses the arithmetic average of the heat flux at the two time levels, $t = n\Delta t$ and

 $t = (n+1)\Delta t$. The method is unconditionally stable. SINDA CNFWBK uses this method and is discussed in Section 6.4.2.

Steady state analysis also requires an implicit method of solution. SINDA steady state subroutines are called CINDSS, CINDSL and CINDSM which are detailed in Sections 6.5.1, 6.5.2, and 6.5.3.

5.2.1 Forward Finite Difference Explicit Method

By replacing the first derivative of temperature with respect to time, dT/dt, with the forward first difference quotient, equation (5.1-6) becomes,

$$c_{\underline{i}} \frac{(T_{\underline{i},\underline{n+1}} - T_{\underline{i},\underline{n}})}{\Delta t} = q_{\underline{i}} - \sum_{\underline{j=1}}^{p} a_{\underline{i}\underline{j}} (T_{\underline{j},\underline{n}} - T_{\underline{i},\underline{n}}) + \sum_{\underline{j=1}}^{p} \sigma b_{\underline{i}\underline{j}} (T_{\underline{j},\underline{n}}^{4} - T_{\underline{i},\underline{n}}^{4}) \quad (5.2-1)$$

$$t = n\Delta t$$

$$i = 1, 2, \dots, N$$

$$T_{\underline{j},\underline{n}} = \text{constant}, N \leq \underline{j} \leq p$$

where, the second subscript on T represents the time level such that

$$T_{i,n} = T_i (n\Delta t)$$

Equation (5.1-6) is represented in the form expressed by equation (5.1-3) by letting,

$$G_{ij} = a_{ij} + \sigma b_{ij} (T_i^2 + T_j^2) (T_i + T_j)$$
 (5.2-2)

It is interesting to note that the finite difference form of (5.1-6) (and thus 5.2-1) represents a second central-difference quotient of $\nabla^2 T$ (refer to (5.1-1).

The computational procedure for the forward difference formulation is rather straightforward since only a single unknown temperature at each time step, $T = n\Delta t$ for each equation is present. Note that the averaging of dT_i/dt) assigns a weighting factor to the heat flux terms only (terms on the right side of equation (5.1-6) at $t = n\Delta t$). Along with the computational simplicity, however, is the stability constraint which places an upper limit on the time increment, Δt , that can be used in the numerical procedure. The stability criterion for the explicit finite difference method

is (for the most limiting node), 12, 14

$$\Delta t < C_{i} / \sum_{j=1}^{p} G_{ij}$$
 (5.2-3)
 $i = 1, 2, ..., N$

A modified stability criterion that allows for a larger time step which results in a conditionally stable temperature for the most limiting node is reported in Reference 23. Since the stability criterion will govern the maximum time step that can be used, it is thus particularly important that a user gives some attention to those factors that compose the condition of stability when nodalizing a physical system.

In the discussion presented so far, arithmetic nodes (nodes with no heat capacity) have not been mentioned. Normally, the computational procedure treats arithmetic nodes separately from the diffusion nodes; arithmetic-node temperatures are solved implicitly. Detailed discussion on the general procedure will be presented in a later paragraph as well as in Section 6 which discusses the various SINDA numerical solution routines.

5.2.2 Implicit Finite Difference Method

The implicit difference equations can be constructed for heat transfer problems in many ways (see, for example, References 12 and 20).

Replacement of equation (5.1-6) with the backward time difference yields,

$$C_{i} \frac{(T_{i,n+1} - T_{i,n})}{\Delta t} = q_{i} + \sum_{j=1}^{p} a_{ij} (T_{j,n+1} - T_{i,n+1}) + \sum_{j=1}^{p} \sigma b_{ij} (T_{j,n+1}^{4} - T_{i,n+1}^{4})$$
(5.2-4)
$$i = 1, 2, \dots, N$$

$$T_{j,n+1} = \text{constant}, N < j \le p$$

 $T_{i,n} = T_i (n\Delta t)$

The computational procedure for the backward difference formulation must necessarily be re-iterative because of the need to solve a set of simultaneous non-linear equations.

In view of the importance of iteration techniques (such as method of

successive approximation), it may be of interest to formulate equation (5.2-4) into an interative form. If we let $C_i/\Delta t \equiv \overline{C_i}$, use equation (5.2-2) in equation (5.2-4) and solve the resultant expression for $T_{i,n+1}$, this yields the recurrent equation for a given time increment, Δt , and time-step, n,

$$T_{i,k+1} = \frac{\overline{C}_{i,k} T_{i,k} + \sum_{j=1}^{k} G_{ij,k} T_{j,k} + q_{i,k}}{\overline{C}_{i,k} + \sum_{j=1}^{p} G_{ij,k}}$$
(5.2-5)
 $i = 1, 2, ..., N$

where,
$$\overline{C}_{i,k} = C_{i,k}/\Delta t$$

 $G_{ij,k} = a_{ij,k} + \sigma b_{ij,k} (T_{j,k}^2 + T_{i,k}^2)(T_{j,k} + T_{i,k})$ (5.2-6)
 $T_{j,k} = \text{constant}, N < j \le p$
 $k = \text{kth iteration (note that } \overline{C}_{i,k}, q_{i,k}, a_{ij} \text{ and } b_{ij} \text{ are shown}$

to be updated every iteration; SINDA routines update these quantities once each time-step)

The iterative pattern is initiated by assuming "old" temperatures $(T_{i,k} \text{ and } T_{i,k})$ on the right side of equation (5.2-5) to evaluate a "new" set of temperatures $(T_{i, k+1})$ on the left side of the equation (5.2-5); this single set of calculations represents an iteration. By replacing all of the "old" temperatures $(T_{i,k})$ on the right side of equation (5.2-5) with the just calculated "new" set of temperatures $(T_{i,k+1})$, a second iteration can be made. The iteration procedure is continued until a termination criterion such as the number of iterations or the maximum absolute difference between $T_{i,k}$ and $T_{i,k+1}$ is less than some prespecified value has been satisfied. It should be noted that G_{ij} , C_{i} and q_{i} are shown to be updated every iteration. This iterative process is termed "block" iteration since the "old" temperatures on the right side are replaced in a "block" (a set of temperatures) fashion with the "new" temperatures.

Another iterative technique is to utilize on the right side of equation (5.2-5) each "new" temperature as soon as it is calculated. This iterative method is termed "successive point" iteration and appears to yield solutions about 25% faster than the "block" iteration method.

Equation (5.2-5) can be expressed in a "successive point" form as follows:

$$T_{i,k+1} = \frac{\overline{C}_{i,k} T_{i,k} + \sum_{j=1}^{i} G_{ij,k} T_{j,k+1} + \sum_{j=i+1}^{p} G_{ij,k} T_{j,k} + q_{i,k}}{\overline{C}_{i} + \sum_{j=1}^{p} G_{ij,k}}$$
(5.2-7)
where,
$$G_{ij,k} = a_{ij,k} + \sigma b_{ij,k} (T_{j,\ell}^{2} + T_{i,k}^{2}) (T_{j,\ell} + T_{i,k})$$
($\ell = k \text{ if } j \ge i \text{ and } \ell = k+1 \text{ if } j < i$)
$$T_{j,k} = \text{constant, } N < j \le p$$
k = kth iteration (note that $\overline{C}_{i,k} q_{i,k}, a_{ij}$, and b_{ij} are shown to be updated every iteration)

The iterative method as used in SINDA follows a fixed, predetermined sequence of operations in contrast with a relaxation procedure which is also one of successive approximations but is not processed out in a predetermined sequence. The relaxation procedure seeks and operates on the node with the maximum temperature difference between the "old" and the "new." From a programming standpoint, the search operation requires as much computational time as the temperature calculation itself.

5.2.3 Steady State Method

Standard steady state equations follows directly from equation (5.2-5) for block iteration or from equation (5.2-7) for successive point iteration by letting $\overline{C_i} = 0$ in these equations. The comments made in Section 5.2.2 are equally applicable here.

5.2.4 Some Comments

The finite difference expressions presented in this Section 5.2 represent standard formulations and thus do not show computational techniques and artifices which are used, some more or some less, in all programs. SINDA numerical solution routines contain many computational features (many original with J. D. Gaski) which enhance problem solution. The various computational aspects of the numerical solution methods as used in the SINDA routines are discussed in rather lengthy detail in Section 6.

6. SINDA NUMERICAL SOLUTION ROUTINES

6.1 Objective and Presentation Format

SINDA has available to the user a number of numerical solution routines which employ various numerical methods. A brief description of these routines with the required SINDA input quantities and format are contained in the SINDA users manual;³,⁴ a general review of numerical methods was presented in Section 5. Unfortunately, the brief description is not sufficient for a casual SINDA user to make a selection decision from among several routines that are available and for a serious user to fully understand the computational procedure as well as to understand the role of the various control constants that are employed in each routine.

It is the intent of this section to fill wherever possible and practical the description void that presently exists with the numerical solution routines by detailing the characterisitcs of each. It is not the intent here to provide sufficient detailed information for a user to make modifications and/or additions to the existing subprograms, but rather to provide information that will aid the user in assessing the various numerical solution routines and in evaluating the numerical results.

Each of the numerical solution routines is detailed from a theoretical as well as from a computational standpoint. Control constants and their role are described and indicated in a step-by-step verbal flow computational procedure. Details of many of the numerous computational checks have purposely been omitted because of the complex interactions. Minute details of each routine can be obtained only from the individual computer listings; a computer listing of each of the SINDA numerical solution routines is presented in Appendices A, B and C. General computational procedure and features that apply to most, if not all of the SINDA numerical solution routines are assembled in a single section (6.2) in order to eliminate undue repetition. The description of each routine is heavily dependent upon, and coupled to, the general description of Section 6.2. The routines have been categorized as steady state or transient with the latter subcategorized as explicit or implicit in order to allow for an orderly presentation as well as to simplify future additions.

6.2 General Computational Procedure and Features

Each of the SINDA numerical solution routines employs a particular finite difference approximation of the lumped parameter heat balance equations. In spite of the uniqueness of each routine, portions of the computational procedure used in each are similar. Also, many of the routines have identical features such as the acceleration of convergence and the use of control constants. As a result, it is convenient to place in this section repetitious material. In some instances material presented here is repeated in the discussion of a particular numerical solution routine.

6.2.1 Order of Computation

It was reported in Section 3.5 that the order of computation depends on the sequence of subroutine calls placed in the EXECUTIØN block by the program user. No other operations block is performed unless called upon by the user either directly by name or indirectly from subroutines which internally call upon them. Numerical solution subroutines internally call upon operations blocks VARIABLES 1, VARIABLES 2, and ØUTPUT CALLS. The internal order of computation for these numerical solution routines is similar with the primary difference between one routine and another being the finite difference approximation employed in a particular routine. A flow diagram indicating the general order of computation for the numerical solution routines is depicted in Figure 6.2-1.

6.2.1.1 Finite Difference Algorithm

Although each of the SINDA numerical solution routines employs a particular finite difference approximation which is detailed for each numerical solution routine, the computational pattern is similar. Within the box depicted as SFDA in Figure 6.2-1, solution of the finite difference algorithm occurs. The computational sequence for transient solutions follows one of two patterns: (1) one for explicit finite difference methods; and (2) one for implicit finite difference methods; steady state solutions follow closely the implicit pattern. Both numerical flow pictures are depicted in Figure 6.2-2; details within the flow pictures are different for each routine and are described separately under the individual SINDA numerical solution routines (refer to Sections 6.3 - 6.5).



Figure (6.2-1) General Order of Computation for Numerical Solution Routines



* For CINDA -3G users, it should be noted that the updating of properties occurs within the numerical solution routine after VARIABLES 1 call. CINDA-3G evaluates the variable properties <u>before VARIABLES 1 call</u>.

> Figure 6.2-2. Numerical Computational Pattern for Explicit and Finite Difference Algorithms

6.2.1.2 Updating of Optionally Specified Properties

Optionally specified properties are defined here as those items which result in pointers being set in the second pseudo compute sequence (refer to Section 3.3.4). The term optional refers to mnemonic options that are available for different types of variable properties.^{3, 4} The properties are updated in all SINDA numerical solution routines the same way. This definition is used here in lieu of stating that optionally specified properties are time and/or temperature varying properties since source data may be specified to be constant. The pointers are set by one or more of the following user input quantities:

- (1) All capacitances, C_i, specified as f(T) or f(t,T) in NODE DATA BLOCK:
- (2) All data, q,, entered in the SOURCE DATA BLOCK:
- (3) All coefficients, G_k, specified as f(T) or f(t,T) in CONDUCTOR DATA BLOCK. It should be noted here that the term coefficient as used here requires amplification. The conductance, G_{ij}, may be for conduction or for radiation; that is,

$$G_{ij} \equiv G_{k} = a_{ij} \text{ (for conduction conductance)}$$

$$G_{ij} = \sigma b_{ij} (T_{i}^{2} + T_{j}^{2})(T_{i} + T_{j}) \text{ (for radiation conductance)}$$

$$= G_{k} (T_{i}^{2} + T_{j}^{2})(T_{i} + T_{j})$$

Thus, note that the calculated conduction conductance G_{ij} is identical to the updated G_k , whereas for the calculated radiation conductance only $\sigma_{b_{ij}}$ is equivalent to the updated G_k .

The type of optional properties is identified by the integer stored in the first six bits of the second pseudo compute sequence which indicates to the program which option is in effect. Optional property types are listed and described for the three categories of input quantities in Table 6.2-1 for capacitance, Table 6.2-2 for impressed source, and Table 6.2-3 for coefficients with the definition of symbols listed in Table 6.2-4.

6.2.2 Operations Blocks

In a previous paragraph, it was mentioned that the sequence of subroutine calls placed in the EXECUTION block by the user determines the

TABLE 6.2-1 OPTIONALLY SPECIFIED CAPACITANCE EXPRESSIONS

Option	Type	Expression	
SIV	1	$C_i = F(A^i:T_i)$	
DIV	2	$C_{i} = F1(A_{1}^{i}:T_{i}) + F2(A_{2}^{i}:T_{i})$	
DIV	3	$C_{i} = F1(L) + F2(A^{i}:T_{i})$	
DIV	4	$C_{i} = F1(A^{i}:T_{i}) + F2(L)$	
SPV	5	$C_i = F(A^p:T_i)$	
DPV	6	$C_{i} = F1(A_{1}^{p}:T_{i}) + F2(A_{2}^{p}:T_{i})$	
DPV	7	$C_{i} = F1(L) + F2(A^{p}:T_{i})$	
DPV	8	$C_{i} = F1(A^{p}:T_{i}) + F2(L)$	
BIV	9	$C_i = F(A^b:T_i, t_m)$,
			1

Notation: Refer to Table 6.2-4.

TABLE 6.2-2 OPTIONALLY SPECIFIED IMPRESSED SOURCE EXPRESSIONS

Option	Туре	Expression
blank	1	$q_i = q_i + F$
SIV	2	$q_i = q_i + F(A^i:T_i)$
SIT	3	$q_i = q_i + F(A^i:t_m)$
DIT	4	$q_{i} = q_{i} + F1(A_{1}^{i}:t_{m}) + F2(A_{2}^{i}:t_{m})$
DIT	5	$q_i = q_i + F1(L) + F2(A^i:t_m)$
DIT	6	$q_{i} = q_{i} + F1(A^{i}:t_{m}) + F2(L)$
DTV	7	$q_{i} = q_{i} + F1(A_{1}^{i}:t_{m}) + F2(A_{2}^{i}:T_{i})$
DTV	8	$q_{i} = q_{i} + F1(L) + F2(A^{i}:T_{i})$
DTV	9	$q_i = q_i + F1(A^i:t_m) + F2(L)$

Notation: Refer to Table 6.2-4.

Table 6.2-3.		Optionally Specified Coefficient Expressions for Conduction and Radiation		
Options		Туре	Expression	
	SIV		1	$G_{k} = F(A^{i}:T_{m})$
1	SIV		2	$G_k = F(A^i:T_i)$
	DIV	(conduction)	3	$G_k = 1.0/[1.0/F1(A_1^i:T_i) + 1.0/F2(A_2^i:T_j)]$
		(radiation)		$G_{k} = [F1(A_{1}^{i}:T_{i})][F2(A_{2}^{i}:T_{j})]$
	DIV	(conduction)	4	$G_k = 1.0/[1.0/F1(L) + 1.0/F2(A^i:T_j)]$
		(radiation)		$C_{k} = [F1(L)][F2(A^{1}:T_{j})]$
	DIV	(conduction)	5	$G_k = 1.0/[1.0/F1(A^{i}:T_i) + 1.0/F2(L)]$
		(radiation)		$G_{k} = [F1(A^{i}:T_{i})][F2(L)]$
	SPV		6	$G_k = F(A^p;T_m)$
	SPV		7	$G_k = F(A^p:T_i)$
	DPV	(conduction)	8	$G_k = 1.0/[1.0/F1(A_1^p:T_i) + 1.0/F2(A_2^p:T_j)]$
		(radiation)		$G_{k} = [F1(A_{1}^{p}:T_{i})][F2(A_{2}^{p}:T_{j})]$
	DPV	(conduction)	9	$G_k = 1.0/[1.0/F1(L) + 1.0/F2(A^p:T_j)]$
		(radiation)		$G_{k} = [F1(L)][F2(A^{p}:T_{j})]$
	DPV	(conduction)	10	$G_k = 1.0/[1.0/F1(A^P:T_i) + 1.0/F2(L)]$
		(radiation)		$G_{k} = [F1(A^{p}:T_{i})][F2(L)]$
	BIV		11	$G_k = F(A^b:T_m, t_m)$
	SIV		12	$G_k = F(A^i:T_j)$
	SPV		13	$G_k = F(A^p:T_j)$

Notation: Refer to Table 6.2-4; note $G_k \equiv \sigma b_{ij}$ (for radiation) $\equiv a_{ij}$ (for conduction) •

					•
Table 6.2-4.	Definition of	Symbols	for Tables	6.2-1 -	- 6.2-3
	•				

Symbols	Definition
c _i	Capacitance of ith node.
F, F1, F2	Multiplying factors, either user constants or literal
G _k (=a _{ij})	Conduction coefficient.
G _k (=ob _{ij})	Radiation coefficient.
L	A literal multiplying factor.
q _i	Heat load into the ith node. (impressed)
∆t	Time-step
t _m	Mean time, (TIMEØ + TIMEN)/2.0
Tm	Mean temperature, $(T_i + T_j)/2.0$
(A ¹ :t _m)	Interpolated value of array A using t_m as the independent variable.
(A ¹ :T ₁)	Interpolated value of array A using T as the independent variable.
$(A^b:T_i, t_m)$	Interpolated value of the bivariate array A using T_i and t_m as independent variables.
(A ^b :T _m , t _m)	Interpolated value of the bivariate array A using T_m and t_m as independent variables.

Mnemonic Options

BIV	Bivariate Interpolation Variable
DIT	Double Interpolation with Time as variable
DIV	Double Interpolation Variable
DPV	Double Polynomial Variable
DTV	Double interpolation with Time and Temperature as Variables
SIT	Single Interpolation with Time as variable
SIV	Single Interpolation Variable
SPV	Single Polynomial Variable

Subscripts

1	Indicates	the	ith node.
, t	Indicates	the	jth node.
2	Indicates	two	(array).
order of computation. Operations blocks number four, EXECUTION, VARIABLES 1, VARIABLES 2, and OUTPUT CALLS. These operations blocks are described in the SINDA Users Manual^{3, 4} but their role insofar as the numerical solution routines are concerned may be of particular interest.

6.2.2.1 EXECUTION Operations Block

The EXECUTION operations block provides the user considerable flexibility in the use of SINDA calls and FØRTRAN operations. Combinations of SINDA calls and FØRTRAN operations are innumerable since the user is actually programming. Now all instructions contained in the VARIABLES 1, VARIABLES 2 and ØUTPUT CALLS are performed each iteration or on the output call interval. Thus, if an operation being performed in VARIABLES 1 utilizes and generates non-changing constants, the operation should be placed in the EXECUTION block (prior to the numerical solution call) so that it will be performed only once and thus eliminate repetitious nonchanging calculations. Operations of this type are conveniently performed in the EXECUTION operations block. Note, however that a constant impressed source should be placed in the optional source data block for SINDA and VARIABLES 1 block for CINDA-3G.

6.2.2.2 VARIABLES 1 Operations Block

The VARIABLES 1 operations block provides the user with a means of specifying at a point in the computational sequence, as shown in Figure 6.2-1, the evaluation of nonlinear network elements, coefficients and boundary values not considered by the various mmemonic codes utilized for node, conductor and source data. It is seen from Figure 6.2-1 that VARIABLES 1 operations occur just prior to entering the numerical solution phase in order to define the network completely.

6.2.2.3 VARIABLES 2 Operations Block

VARIABLES 2 operations are post-solution operations in contrast to the VARIABLES 1 operations which are pre-solution operations as shown in Figure 6.2-1. VARIABLES 2 provides the user with a means to examine the characteristics of the numerical solution and make corrections. For example, the heat flow from one node to another can be evaluated or a temperature(s) determined without material phase change can be corrected to account for the phase change by using the VARIABLES 2 operations block.

6 - 9

÷.,

6.2.2.4 OUTPUT CALL Operations Block

The OUTPUT CALL operations block provides the user with a means of calling any desired subroutine with the operation performed on the output interval. In addition to various subroutines for printing output, several plotting subroutines are available.^{3,4}

6.2.3 Control Constants

Control constants number forty-nine and have alphanumeric names. Control constant values are communicated through program common to specific subroutines which require them. Whenever possible, control constant values not specified are set internally to acceptable values. If a required control constant value is not specified, an appropriate error message is printed and the program terminated. Each of the SINDA numerical solution routines employs a number of control constants which fall under the categories as: (1) user specified; (2) optionally user specified; (3) internally set by program; and (4) dummy. These control constants are listed alphabetically with a brief description of each in Section 6.2.3.1 followed by a detailed description of user specified control constants in Section 6.2.3.2; nominal values of these control constants that must be specified or are optionally specified for each SINDA numerical solution routine are indicated in Table 6.2-5. Specification of these control constants is detailed under the discussion of each SINDA numerical solution routine.

6.2.3.1 Alphabetical Listing and Brief Description of Control Constants

ARLXCA (control constant 19)

Maximum arithmetic node relaxation temperature change allowed between iterations; this check occurs after each iteration. Specification is required for the implicit and steady state routines (except CINDSM) and if not specified an error message is printed if the number of arithmetic nodes is greater than zero. Specification is not required for explicit routines and if not specified, ARLXCA is set to 1.E+8.

ARLXCC (control constant 30)

Maximum arithmetic node relaxation temperature change calculated by program; ARLXCC < ARLXCA check is made.

. Table 4.2-5. Characteristics of Last Specified Control Constants

1 Automatic matrix 1 and transform 1 and t							51	NDA Nume	rical So	Iut ion 1	out inc.	Ĩ	Ī	molicit.	Ī
Manuscle	0	onst ant.	_	CHADGE	eady stat	CHNDSY	CNFRMD	CNFRDI	CNFAST	CNENTR	CNDUFR	CNQU1K	CNBACK	CNFWBK	CNVARB
$ \ \ \ \ \ \ \ \ \ $		Number		CONTO	TERVID									1	ţ
1 1 Alloughe artinecta node tenparature - - - - - - - - - 1.164	K	19	Allowable artthmetic node relaxation temperature change	**	*	۱ ,	1.E+3	1.1.48	1.1.48	1.1.48	1.E+8	1.E+8	:	1	K
11 actual politich - - - 0.0 0 0 0 0<	×	н	Allowable arithmetic node temperature change		1	't	:	=	ı	:	± '	2	1.E+8	1.E+8	1.E+8
0 3) System energy balance - 10	d.	12	Backup switch	ı	1	ı	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
C 4 Time-utery factor for explicit - - 1.0 1	:0	33	System energy balance	1	4	*	ı	ı	a.	ı	•	э,	1	ł	ŀ
9 Arthmetic-node damping factor 1.0	ų	4	Time-step factor for explicit routines	ł	а	1	1.0	1.0	- ,	1.0	1.0	1.0		•	•
10 Diffusion-node damping factor 1.0	_	6	Arlthmetic-node damping factor	1.0	1.0	•	1.0	1.0	1.0	1.0	1.0	1.0	1.0	1.0	1.0
A 26 Allovable diffusion-rode relaxation *** </td <td></td> <td>10</td> <td>Diffusion-node damping factor</td> <td>1.0</td> <td>1.0</td> <td>1.0</td> <td>i</td> <td>1</td> <td>I,</td> <td>ł</td> <td>١</td> <td>١</td> <td>1.0</td> <td>1.0</td> <td>0.1</td>		10	Diffusion-node damping factor	1.0	1.0	1.0	i	1	I,	ł	١	١	1.0	1.0	0.1
R 8 Maximum time-step allowed - - 1.E+8	N.	26	Allowable diffusion-node relaxation temperature change		***	***	,	1	4	.t	•	ı.	***	***	***
1 22 Specified time-step for implicit -	EH	. 80	Maximum time-step allowed	۱	ı.	ĩ	1.6+8	1.E+8	1.E+B	1.E+8	1.E+8	1.E+8	1.E+8	1.E+8	1.6+8
1 21 Minimum time-step allowed - - 0.0 0.0 0.0 0.0 -	13	22	Specified time-step for implicit routines	ı	•	ŀ	ŗ	ł	n.	,	·	•	*	*	*
A Allovable diffusion-node temperature - - - 1.E+8	EL	21	Minimum time-step allowed	•	ť	.,•	0.0	0.0	*	0.0	0.0	0.0	·	•	۰.
C 49 Number of iterations for linearized -	V	Q	Allowable diffusion-node temperature change	•	•	•	1.E+8	1.E+8	,	1.E+8	1.E+8	1.E+8	1,648	1.E+8	1.E+8
7 3 Number of desired iteration loops * * 1 1 1 1 1 *	VC	7	Number of iterations for linearized system	a.	ji.		1	4	ı	۱	1	ı	۰.	i [`]	ı
11 13 Time interval for activating - - - * * * * * * * *	6 .	S	Number of desired iteration loops	*	•	*	1	1	1	-	1	1	*	•	*
D 3 Problem stop-time -	1	18	Time interval for activating OUTPUT CALLS	,	ł.		*	×	*	*	*	*	*	*	*
13 01d time of problem start time - - - 0.0	QN	'n	Problem stop-time	'n	,	a	****	***	****	***	***	****	****	***	****
Pseudo Compute Sequence SPCS LPCS LPCS SPCS SPCS SPCS SPCS LPCS LPCS sic storage requirements NND 3(NND) 3(NND) 3(NND) 3(NND) 3(NND) 3(NND) sic storage requirements NND 2(NND) 3(NND) 3(NND) 3(NND) 3(NND) +2(NNA) +3(NA) +NNA +NNA +NNA +NNA +NNA +NNA +NNA +NNA +NNA +NNA +NNA +NNA +NNA +NNA +NNA	6	13	Old time of problem start time	•	1	•	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
nic storege requirements NND 2 (NND) 3 (ND) 3 (NND) 3			Pseudo Compute Sequence	SPCS	LPCS	LPCS	SPCS	LPCS	SPCS	SPCS	SPCS	SPCS	LPCS	LPCS	SOAT
+2 (NNA) +3 (NNA) +NNA +NNA +NNA +NNA +NNA +NNA +NNA	nic .	storage	requirements	QNN	2 (UND)	3 (1111)	UNN	QNN	QNN	DNN	2 (NND)	2 (NND)	(UND) S	(DNN) E	3 (UND)
+NUE +NNB +NNB					+2 (NNA)	(NNN) E+	+NNA	ANN +		VNN+	+NNA	ANN +	ANN+	VNN+	VNN+
						+NGT							+NNB	ENN+	+NNB

cified. r in i . Notation:

ont II und sheritist.	ii ii vhen	я 	" TIMEND S TIML& for Imp					
r ressage printo	•	= =	=					the first and field
th error	-	н н	•	des.		•		102 500
termination wi		=	ء (arithmetic-no	diffusion .	" houndarv	Ler at conduct	the tree to be the
L'un	Ŧ	=	-	number of	-	3	total nu	tailt
= 9			-	tie t	:	:	:	
Indicates "	Ŧ	=		is t	.=	=	=	

-

.

ATMPCA (control constant 11)

Maximum arithmetic node temperature change allowed between time steps for transient routines; the check occurs after the specified number of iterations. If not specified or if specified to be \leq 0.0, ATMPCA is set to 1.E+8.

- ATMPCC (control constant 15) Maximum arithmetic temperature change calculated by program; ARMPCC < ATMPCA check is made.
- BACKUP (control constant 12)

Backup switch that is checked after VARIABLES 1 and VARIABLES 2 calls. Initialized at zero. If specified to be non-zero, the completed time step is erased and repeated.

BALENG (control constant 33)

A user specified system energy balance to be maintained; this control constant is presently used only in CINDSM. If not specified, an error message will be printed.

CSGFAC (control constant 4)

Time step factor for explicit routines except CNFAST. If not specified or if specified to be less than 1.0, CSGFAC is set internally to 1.0.

CSGMAX (control constant 23)

Maximum value of $C_{i}/\Sigma G_{ij}$; this value aids in the checkout of the thermal network and is calculated only by the output subroutines, CSGDMP and RCDUMP.

CSGMIN (control constant 17)

Minimum value of $C_i/\Sigma G_{ij}$; this value is used to limit the computational time step for explicit methods of solution. If CSGMIN is calculated to be ≤ 0.0 , an error message is printed.

CSGRAL (control constant 24)

Allowable range between CSGMIN and CSGMAX: this control constant is not presently used but is included for future considerations.

DAMPA (control constant 9)

Arithmetic-node damping factor for all numerical solution routines; if not specified, or if specified to be ≤ 0.0 , DAMPA is set to 1.0. (Refer to equation 6.2-6.)

DAMPD (control constant 10)

Diffusion node damping factor for implicit and steady state routines; if not specified or is specified to be ≤ 0.0 , DAMPD is set to 1.0. (Refer to equation 6.2-20.)

DRLXCA (control constant 26)

Maximum diffusion node relaxation temperature change allowed between iterations for implicit and steady state routines; this check occurs after each iteration. If not specified an error message will be printed when the number of diffusion nodes is greater than zero.

DRLXCC (control constant 27)

Maximum diffusion node relaxation temperature change calculated by the program; DRLXCA < DRLXCC check is made.

DTIMEH (control constant 8)

Maximum time step allowed; applies to transient routines. If not specified or if specified to be ≤ 0.0 , DTIMEH is set to 1.E-8.

DTIMEI (control constant 22)

Specified time step for implicit solutions; if not specified, an error message will be printed and the "run" terminated.

DTIMEL (control constant 21)

Minimum time step allowed for explicit routines. If not specified for CNFAST, an error message will be printed and the "run" terminated. If DTIMEU is less than DTIMEL the routines will

terminate with an error message, except for CNFAST which will do a steady state solution on the offending node. For all routines DTIMEL is initially set at 0.0 internally.

DTIMEU (control constant 2)

Contains time step used in computational procedure.

DTMPCA (control constant 6)

Maximum diffusion node temperature change allowed between time steps for transient routines. If not specified or if specified to be ≤ 0.0 , DTMPCA is set to 1.E+8.

DTMPCC (control constant 15)

Maximum diffusion node temperature change calculated by program; DTMPCA < DTMPCC check is made.

ENGBAL (control constant 32)

Calculated energy balance of the system; presently used only in CINDSM.

LAXFAC (control constant 49)

Specified number of iterations to be performed on a linearized system with no updating of elements during a set of LAXFAC iterations for CINDSM only; if not specified, an error message is printed and the "run" terminated.

LINECT (control constant 28)

A line counter location for program output (integer).

LØØPCT (control constant 20)

Contains number of iterations performed (integer).

NØCØPY (control constant 34)

Contains the no copy switch for matrix users.

NLØØP (control constant 5)

Number of specified iteration loops. Must be specified for the steady state and implicit routines; if not specified, an

error message is printed and the "run" is terminated. Optional specification for solution of the arithmetic nodes in the explicit routines; if not specified, NLØØP is set to integer 1.

ØPEITR (control constant 7)

Output each iteration if ØPEITR is specified to be non-zero; if not specified, ØPEITR is set at zero. May be switched on and off during a run.

ØUTPUT (control constant 18)

Time interval for activating ØUTPUT CALLS of transient routines; if not specified, error message is printed and the "run" terminated. May be addressed by user and modified during a run in VARIABLES 2. Can be used in steady state routines for a series of steady state solutions.

PAGECT (control constant 29)

A page counter location for program output (integer).

TIMEM (control constant 14)

Mean time for a computation interval; $TIMEM = \frac{TIME\emptyset + TIMEN}{2.0}$.

TIMEN (control constant 1)

New time at the end of the computational interval. TIMEN = $TIME\emptyset + DTIMEU$.

TIMEND (control constant 3)

Problem stop time for transient analysis. Must be > TIMEØ for all routines; if not, an error message is printed and "run" terminated. May be addressed by the user and modified during a run.

TIMEØ (control constant 13)

Old time at the start of the computational interval. Also used as the problem start time and may be negative; if not specified, TIMEØ is set at zero.

ITEST, JTEST, KTEST, LTEST, MTEST (control constants 39, 40, 41, 42 and 43, respectively)

Contain dummy integer constants.

RTEST, STEST, TTEST, UTEST, VTEST (control constants 44, 45, 46, 47, and 48, respectively)

Contain dummy floating point constants.

(Control constant 31)

Problem type indicator, 0 = THERMAL SPCS, 1 = THERMAL LPCS, 2 = GENERAL.

(Control constant 35)

Contains relative node number of CSGMIN.

(Control constant 36)

Contains relative node number of DTMPCC.

(Control constant 37)

Contains relative node number of ARLXCC.

(Control constant 38)

Contains relative node number of ATMPCC.

6.2.3.2 User Specified and Optionally User Specified Control Constants

The availablity of control constants which must be specified or which can optionally be specified provides the user with considerable flexibility to alter the computational criteria and hence the calculated temperatures. On the other hand, this flexibility presents the user with the problem of imputting control constant values if the nominal values are not suitable. An attempt will be made here to provide some guidelines on control constant values based on rather limited data presently available, but it should be recognized that suitable values to be used are dependent on the problem to be solved and often a trade-off must be made between accuracy and computational time. This normally can be obtained only through the use of the numerical solution routines.

ARLXCA (Allowable Arithmetic Node Relaxation Temperature Change)

This control constant must be specified for the implicit routines if any arithmetic node is present and for the steady state routines except CINDSM. For the explicit solution routines, ARLXCA may be optionally specified; if not specified ARLXCA is set to 1.E+8. ARLXCA represents a maximum temperature change convergence criterion for the arithmetic nodes; ARLXCA is checked each iterative step. It is used in conjunction with control constant NLOØP. Satisfaction of either NLØØP or ARLXCA during any iterative step terminates the arithmetic node temperatures calculation for that time-step with computation proceeding on to the next one. Typically, an ARLXCA value is 0.01, but its value is dependent upon the magnitude of expected temperatures. The 0.01 value tries for 5th digit accuracy for temperatures in the hundreds. An ARLXCA value of 0.0001 would try for seventh digit accuracy. Since the computer will not yield 8 digit accuracy, an ARLXCA value < .0001 will always result in NLØØP iterations being performed.

ATMPCA (Allowable Arithmetic Node Temperature Change)

This control constant may be optionally specified by the user for the implicit routines and for the explicit routines except CNFAST. If not specified, ATMPCA is internally set at 1.E+8. ATMPCA represents an allowable arithmetic-node temperature change criterion between one timestep and another with the calculated temperature change stored in control constant ATMPCC. If the maximum arithmetic-node temperature change is greater than ATMPCA, the time-step, Δt , is shortened to,

$\Delta t = .95 * \Delta t$ (ATMPCA/ATMPCC)

and the arithmetic-node and diffusion-node temperatures re-set to former values. The computational procedure is repeated with the smaller time-step. Specification of ATMPCA prevents a rapid temperature change between timesteps with the value to be specified dependent upon the problem. Thus, the user should estimate the number of time-steps and the range of the temperature to arrive at a reasonable value. For typical spacecraft-type thermal problems an ATMPCA of about 10°F is typical.

BACKUP (Backup Switch)

Control constant BACKUP provides the SINDA user with the means to utilize any thermal numerical solution subroutine as a predictor program. All of the numerical solution subroutines set control constant BACKUP to zero, just prior to the call on VARIABLES 2. Then immediately after the return from VARIABLES 2. a nonzero check on BACKUP is made. If BACKUP is nonzero, all temperature calculations for the just completed time-step are eliminated, the old temperatures (temperatures calculated at the previous time-step) are placed in the temperature locations and the control is routed to the start of the computational sequence.

ي دي. دريا ميغر و ايو

It should be noted that the user must provide the necessary check and criterion in VARIABLES 2 if the iteration is to be repeated. Thus, if the iteration is to be repeated, BACKUP must be nonzero and a criterion that can be met in subsequent passes established. For example, the criterion may require the correction of a parameter used by the network solution. Further, if other calls in VARIABLES 2 are not to be performed FØRTRAN instructions must be generated to bypass these calls.

It should be noted that BACKUP is sometimes checked after VARIABLES 1. However, for the present this use should be ignored since BACKUP check after VARIABLES 1 is planned for future additions of special boundary calculation subroutines.

BALENG (User Specified System Energy Balance)

This control constant is presently used in the steady state routine CINDSM but not in the other SINDA numerical solution routines. BALENG must be specified, otherwise the "run" is terminated with an error message printout; the value of BALENG is a criterion that represents an acceptable net energy balance (energy in minus energy out) of the system in the calculation of steady state temperatures. A value for BALENG depends upon the magnitude of system energy under consideration. As a guideline 1/2% of the total energy into the system (including heat flow from the boundary) is a reasonable value.

CSGFAC (Time Step Factor)

This control constant may be optionally specified by the user

for the explicit routines except CNFAST and it provides the user with some control on the compute time-step as indicated in Section 6.2.4. If CSGFAC is not specified or is specified to be less than one by the user, it is internally set at 1.0. For subroutines CNFRWD and CNFRDL which are conditionally stable CSGFAC is a divisor; a value of CSGFAC greater than one is used to obtain higher accuracy. For subroutines CNEXPN, CNDUFR and CNQUIK, which are unconditionally stable, CSGFAC is a multiplier (refer to page 6-24); a value of CSGFAC greater than one is used to decrease the computational time. A question may be raised, why a value of CSGFAC less than one is not allowed for CNEXPN, CNDUFR and CNQUIK? The reason for this is that it is more accurate to use CNFRWD (or CNFRDL) if a smaller time-step than the one associated with CSGFAC equal to one is desired.

DAMPA (Damping Factor for Arithmetic Nodes)

This control constant may be optionally specified for all of the SINDA numerical solution routines; if not specified, of if specified to be ≤ 0.0 , DAMPA is set to 1.0. In the development of the finite difference expressions as reported in technical literature, little (if any) mention is made about the so-called damping factor. The damping factor does nothing more than to allow a certain fraction (1.0 - DAMPA) of the "old" temperature (temperature at the previous time-step or iteration) to be included as part of the temperature change for the current time-step or iteration. The value to be used is dependent upon the problem and to some extent upon the routine. Typically, a value of 0.6 is used but a value as small as 0.01 has been used with CINDSL for a thermal radiation-dominated problem. In general, a choice for DAMPA becomes a trial and error procedure. DAMPA is used only with arithmetic nodes (refer to equation 6.2-6).

DAMPD (Diffusion Node Damping Factor)

This control constant may be optionally specified for the implicit and steady state routines; if not specified or if specified to be \leq 0.0, DAMPD is set to 1.0. DAMPD serves the same purpose for the diffusion nodes as DAMPA provides for the arithmetic nodes (refer to equation 6.2-21).

DRLXCA (Allowable Diffusion-Node Relaxation Temperature Change)

This control constant must be specified for the implicit routines and for the steady state routines except CINDSM. DRLXCA serves the same

purpose for the diffusion-nodes as control constant ARLXCA does for the crithmetic nodes. Thus, the discussion on ARLXCA equally holds true for DKLXCA. It may be asked, why ARLXCA and DRLXCA? The reason for this is that it provides greater computational flexibility.

DTIMEH (Maximum Time-Step Allowed)

This control constant may be optionally specified for the explicit and the implicit routines. DTIMEH represents the maximum time-step allowed during the computational process. One use of DTIMEH is the prevention of a single large and a single small computational time-step during an output interval by specifying DTIMEH as a fraction of the output interval. If DTIMEH is not specified, DTIMEH is set to 1.0E+8.

DTIMEI (Specified Time-Step for Implicit Routines)

This control constant must be specified for the implicit routines and is not used by the other routines. If not specified, the "run" terminates with an error message printout. DTIMEI represents a specified time-step and is arbitrary, but the governing criterion should be minimum computational time with satisfactory temperature accuracy. This means that DTIMEI should be specified in conjunction with control constant NLØØP which represents the maximum number of computational iterations allowed during each time-step. Since each iterative calculation is essentially equivalent to a time-step calculation, DTIMEI should be normally greater than NLØØP*CSGMIN, where CSGMIN is the time-step used in the explicit routines. If savings in computational time cannot be met with the same accuracy by using the implicit routines, it is more reasonable to use the explicit routines.

DTIMEL (Minimum Time-Step Allowed)

This control constant must be specified for subroutine CNFAST and is optional for other explicit solution routines. If not specified for CNFAST, the "run" terminates with an error message printout. DTIMEL represents the minimum time-step allowed; for all the explicit routines except CNFAST, if the calculated time-step is less than DTIMEL, the "run" terminates with an error message printout. For subroutine CNFAST, if the calculated time-step of any node, as expressed by $C_i/\Sigma G_{ij}$ and stored in CSGMIN, is less than DTIMEL, the temperature of the nodes not satisfying DTIMEL are calculated

$$6 - 20$$

using the steady state equations without computational iterations (refer to Section 6.3.3 for details on the CNFAST routine). The purpose of this control constant for CNFAST is to shorten the computational time; the danger in its use is that with a large DTIMEI. a large number of diffusion nodes will receive the steady state equations without iterations. As a result, the temperature inaccuracies can be expected to be large.

DTMPCA (Allowable Diffusion Node Temperature Change)

This control constant may be optionally specified by the user for the implicit routines and for the explicit routines except CNFAST. DTMPCA represents a diffusion-node temperature change criterion between one timestep and another. If the maximum diffusion-node temperature change which is stored in DTMPCC is greater than DTMPCA, the time-step is shortened to,

 $\Delta t = .95 * \Delta t$ (DTMPCA/DTMPCC)

and the diffusion-node and arithmetic-node temperatures re-set to former values. The computational procedure is repeated with the smaller timestep. DTMPCA serves the same purpose for the diffusion nodes as control constant DRLXCA provides the arithmetic nodes.

LAXFAC (Number of Iterations for Linearized Lumped Parameter System)

LAXFAC is used only in the steady state routine CINDSM and represents the number of iterations to be performed on a linear lumped parameter system with no updating of elements during a set of LAXFAC iterations. The system elements are re-evaluated for the new set of temperatures and in turn temperatures are recalculated for another set of LAXFAC iterations with a more severe relaxation criterion. The number of iterations will not exceed control constant NLØØP which represents the total number of iterations. NLØØP will not be met only if relaxation criteria are met during an iterative loop and between iterative loops and if the system energy balance as stored in BALENG is satisfied (refer to Section 6.5.3 for details).

NLØØP (Number of Iteration Loops)

This control constant must be specified for the implicit and the steady state routines; if not specified, the "run" terminates with an error message printout. NLØØP may be optionally specified for the explicit routines since it is used for the arithmetic nodes; if not specified, NLØØP is set to 1. The value of NLØØP to be used depends upon the problem

to be solved. For a steady state problem it is not unusual to have NLØØP equal to several hundred, whereas for a transient problem the implicit routines NLØØP should be specified as discussed for control constant DTIMEI. In general, a trial and error procedure is required to arrive at a suitable value of NLØØP.

OUTPUT (Time Interval for Activating ØUTPUT CALLS)

This control constant must be specified for all numerical solution routines except steady state routines since the first time-step used is generally set to ØUTPUT. The input value is left to the judgment of the user. Normally, the output interval is gauged by the length of the run and the expected temperature response characteristics. As a "rule-of-thumb" the output interval lies between CSGMIN and CSGMAX, with ØUTPUT being several times larger than CSGMIN. The values of CSGMIN and CSGMAX can be obtained from the output subroutines CSGDMP and RCDUMP.³,⁴ Subroutines CSGDMP and RCDUMP are designed to aid in the checkout of thermal problem data decks and should be used before making a transient computer rum.

TIMEND (Problem Stop Time)

The use of this control constant is self-explanatory. For the subroutines as they are presently coded, TIMEND must be specified as larger than TIMEØ, otherwise an error message is printed and the "run" terminated. For the explicit routines, if TIMEND is not larger than TIMEØ a time-step of zero will result and the "TIME STEP TOO SMALL" error message will be printed. The implicit routines will print the error message, "TRANSIENT TIME NOT SPECIFIED." If a solution is to be terminated by the use of a criteria, but the run is not to be terminated, this can be accommodated by setting TIMEND=TIMEØ when the criteria is met.

TIMEØ ("Old" Time or Problem Start Time)

This control constant represents the "old" time or the problem start time for the transient routines. If not specified, TIMEØ is set to 0.0. An important consideration in the use of TIMEØ is that TIMEØ may be set to negative.

6.2.4 <u>Time-Step Calculations</u>

Each numerical solution routine requires the use of a time-step that depends upon many considerations, such as the output interval, the end

of the problem time, the stability criterion for explicit routines, etc. In spite of the unique solution procedure of each of the numerical solution routines, the overall time-step calculation procedure for the transient routines is essentially identical. The numerous time-step checks, as well as the selection of the time-step, are indicated below (for definition of control constants refer to Section 6.2.3):

- (1) Check that elapsed time, t, does not exceed problem end time.
 - If: TIMEØ + ØUTPUT > TIMEND
 - Set: ØUTPUT = TIMEND TIMEØ

TIMEØ is the old time ØUTPUT is the output time interval TIMEND is the problem stop time

(2) Set initial time-step, Δt, which is stored in DTIMEU (control constant for time-step). The initial time step for the SINDA numerical routines is as follows:

Numerical	Routines	Initial Time-Step
EXPLICIT	CNFRWD	ØUTPUT
EXPLICIT	CNFRDL	ØUTPUT
EXPLICIT	CNEXPN	ØUTPUT
EXPLICIT	CNDUFR	ØUTPUT
EXPLICIT	CNQUIK	ØUTPUT
EXPLICIT	CNFAST	DTIMEL (minimum time-step allowed)
IMPLICIT	CNBACK	DTIMEI (specified time-step)
IMPLICIT	CNFWBK	DTIMEI
IMPLICIT	CNVARB	DTIMEI

- (3) Check ∆t (stored in DTIMEU) against maximum allowable time-step.
 If: DTIMEU > DTIMEH
 Set: DTIMEU = DTIMEH
- (4) Check sum of elapsed time since last printout, TSUM, and timestep, DTIMEU, against ØUTPUT.

If: TSUM + DTIMEU > ØUTPUT

- **Set:** $\Delta t = \emptyset UTPUT TSUM$
- If: $TSUM + \Delta t < \phi UTPUT$

and if: $TSUM + 2(\Delta t) > ØUTPUT$

Set: $\Delta t = 1/2$ (OUTPUT - TSUM)

(5) Store

Set: DTIMEU = Δt

(6) Check DTIMEU against minimum allowable time-step.

If: DTIMEU < DTIMEL

Result: An error message is printed and the "run" terminated except for CNFAST, CNBACK, CNFWBK and CNVARB.

(7) Set new time (TIMEN)

Set: TIMEN = TPRINT + TSUM + Δt

TPRINT is the time of the last printout. TSUM is the time from the last printout.

(8) Set mean time (TIMEM)

Set: TIMEM = 1/2 (TIMEN + TIMEØ)

(9) Calculate (or specify) time-step.

The calculated (or specified) time-step for the SINDA numerical routines is as follows:

Numerical Routines	Calculated Time-Step
EXPLICIT CNFRWD	0.95 * CSGMIN/CSGFAC
EXPLICIT CNFRDL	0.95 * CSGMIN/CSGFAC
EXPLICIT CNEXPN	0.95 * CSGMIN * CSGFAC
EXPLICIT CNDUFR	0.95 * CSGMIN * CSGFAC
EXPLICIT CNQUIK	0.95 * CSGMIN * CSGFAC
EXPLICIT CNFAST	larger of CSGMIN or DTIMEL
IMPLICIT CNBACK	DTIMEI
IMPLICIT CNFWBK	DTIMEI
IMPLICIT CNVARB	DTIMEI
-	

CSGMIN = $C_i / \Sigma G_{ii}$ (minimum value, i = 1,2,...,NND)

where: C, is the capacitance of the ith node

G_i is the conductance from node i to node j

CSGFAC is the time-step factor (see above).

(10) It should be recognized that individual routines may have slight variations to the time-step calculations.

6.2.5 Computation of Temperatures

The actual calculation of temperatures, be it for diffusion nodes or for arithmetic nodes, represents the end result of a long computational procedure with many checks and criteria. Nevertheless, if one confines the discussion to the DØ loops of nodal types, a rather compact but general computational pattern becomes apparent. More details are presented in the individual sections describing each numerical solution routine. (Sections 6.3 - 6.5)

6.2.5.1 Transient Explicit Routines

For the explicit routines the diffusion and arithmetic nodes are treated separately. Diffusion-node temperatures are calculated explicitly, whereas the arithmetic-node temperatures are computed implicitly. This means that at each time-step an iterative loop is set-up for the arithmetic nodes; none is required for the diffusion nodes.

Diffusion-Node Temperatures

Calculation of the diffusion-node temperatures follows the VARIABLES 1 call; the computational pattern is:

 $D\phi - L\phi\phi P$ (I = 1, NND) on the diffusion nodes is established.

The functions associated with the variable capacitance C_i , the variable impressed source q_i , and the variable coefficients G_k (a_{ij} for conduction and $\sigma_{b_{ij}}$ for radiation), between diffusion-diffusion and diffusion-arithmetic nodes are updated at the beginning of each time-step. These functional types are described in Section 6.2.1.2 and the computational pattern is indicated in the flow chart of Figure 6.2-3.

Using the updated C_i , q_i and G_k , the branch heat flow sum, Q_{si} , and conductance sum X_i , are calculated (refer for example to flow chart of Figure 6.3-1).

ş.



* Variable capacitance (c_i), impressed source (q_i), or variable coefficient (G_k).

Figure 6.2-3. Evaluation of Nonlinear Capacitance, Source or Conductance

$$Q_{si} = \sum_{j=1}^{p} G_{ij,n} (T_{j,n} - T_{i,n}) + q_{i,n}$$
(6.2-1)

$$\mathbf{x}_{i} = \sum_{j=1}^{p} \mathbf{G}_{ij,n}$$
(6.2-2)

where, p = total number of nodes; n = time-step old $C_i, q_i, a_{ij}, b_{ij} = \text{optionally specified (refer to Table 6.2-1 - 6.2-4)}$ $G_{ij,n} = a_{ij,n} + ob_{ij,n} (T_{j,n}^2 + T_{i,n}^2)(T_{j,n} + T_{i,n})$

Stability criterion $C_i / \sum_{j=1}^{p} G_{ij,n}$ is computed and the smallest value is stored in control constant CSGMIN. If CSGMIN ≤ 0.0 , an error message is printed and the "run" terminated.

Diffusion-node temperatures are calculated by using the appropriate finite difference expression associated with each routine. These routines and algorithms are identified as:

CNFRWD and CNFRDL (Section 6.3.1), uses standard forwarddifference algorithm.

CNFAST (Section 6.3.2), uses a modified CNFRWD computational procedure to decrease the computational time.

- CNEXPN (Section 6.3.3), uses the exponential prediction method.
- CNDUFR (section 6.3.4), uses DuFort-Frankel method.
- CNQUIK (Section 6.3.5), uses half DuFort-Frankel and half exponential prediction metnod.

Symbolically, the expression for the diffusion-node temperatures may be written as,

$$T_{i,n+1} = T_{i,n} + \frac{\Delta t \ Q_{si}}{C_i}$$
 (6.2-3)

Except for CNFAST the maximum diffusion-node temperature change which is stored in DTMPCC is checked against the allowable diffusion node temperature change which may be specified by the user via the control constant DTMPCA (if not specified DTMPCA = 1.0E+8). If DTMPCA is not satisfied, the time-step is decreased to,

 $\Delta t = .95 * \Delta t$ (DTMPCA/DTMPCC)

and all temperatures re-set to former values. The computational procedure is repeated with the smaller time-step. CNFAST does not allow for the recalculation of diffusion-node temperatures.

Arithmetic-Node Temperatures

Calculation of the arithmetic-node temperatures always follows the computation of the diffusion-node temperatures and uses "successive point" iteration. The computational pattern is as follows:

Arithmetic-node damping factors DN and DD are established.

- DN = DAMPA (optionally specified user constant, if not specified DAMPA = 1.0; factor for the current time-step temperature change)
- DD = 1.0 DN (factor that allows a certain fraction of the "old" temperature to be included as part of the temperature change for the current time-step)

Iterative DØ-LØØP (K=1,NLØØP) is established (NLØØP is the number of iterations specified by the user, if not specified, NLØØP = 1).

DØ-LØØP (I=NND, NND + NNA) for the arithmetic nodes is established.

Impressed source q_i and coefficient G_k (a ij for conduction and σb_{ij} for radiation) are updated once for each time-step.

Using the updated G_k and q_i , the branch heat flow sum Q_{si} and the conductance sum X_i are calculated (refer to flow chart of Figure 6.3-2).

$$Q_{si} = \sum_{j=1}^{p} G_{ij,n} (T_{j,k} - T_{i,k})$$
(6.2-4)

$$\mathbf{X}_{i} = \sum_{j=1}^{\Sigma} \mathbf{G}_{ij,n}$$
(6.2-5)

Arithmetic node temperatures are calculated for each iterative loop by using the following "successive point" expression, which is employed in all of the routines,

$$T_{i,k+1} = DD^{*}T_{i,k} + DN^{*} \left(\frac{q_{i,n} + \sum_{j=1}^{i} G_{ij,n} T_{j,k+1} + \sum_{j=i+1}^{p} G_{ij,n} T_{j,k}}{\sum_{j=1}^{p} G_{ij,n}} \right) (6.2-6)$$
where, $i = (NND+1)$, $(NND+2)$,..., $(NND + NNA)$

$$T_{j,k} = \text{constant}, (NND + NNA) < j \le p$$

$$p = \text{total number of nodes}$$

$$T_{i,k} = \text{temperature at kth iteration}$$

$$G_{ij,n} = a_{ij,n} + \sigma b_{ij,n} (T_{j,\ell}^{2} + T_{i,k}^{2}) (T_{j,\ell} + T_{i,k})$$

$$(\ell = k \text{ if } j \ge i \text{ and } \ell = k+1 \text{ if } j < i)$$

$$(a_{ij,n} \text{ and } b_{ij,n} \text{ mean updating at time-step, n)}$$

$$q_{i}, a_{ij}, b_{ij} = \text{optionally specified (refer to Tables 6.2-1 - 6.2-4)}$$

 $DN \equiv DAMPA$ (arithmetic node damping factor) DD = 1.0 - DN

The maximum arithmetic-node relaxation temperature change is calculated and checked against the allowable arithmetic-node relaxation temperature change which may be specified via the control constant ARLXCA. This relaxation convergence check is made during each iterative step calculation and is used in conjunction with control constant NLØØP. Satisfaction of either ARLXCA or NLØØP during any iterative step terminates the arithmeticnode temperature calculation.

For each time step, except for CNFAST, the maximum arithmetic-node temperature change which is stored in control constant ATMPCC is checked against the allowable arithmetic-node temperature change which may be specified via the control constant ATMPCA (if not specified, ATMPCA = 1.0E+8). If ATMPCA is not satisfied, the time-step is decreased to,

 $\Delta t = .95 * \Delta t (ATMPCA/ATMPCC)$

and all temperatures re-set to former values. The computational procedure is repeated with the smaller time-step. CNFAST does not allow for recalculation of arithmetic-node temperatures.

6.2.5.2 <u>Transient Implicit Routines</u>

Both diffusion-node and arithmetic-node temperatures are calculated by "successive point" iteration. Although these calculations are performed cn the same iterative pass, diffusion node temperatures are evaluated on its own computational loop using a specified algorithm associated with a particular implicit routine. Calculation of the arithmetic-node temperatures is also done on its own computational loop and is identical in all the implicit routines. As a matter of fact, arithmetic-node temperatures are calculated in the same manner in all the SINDA numerical solution routines. Use of a separate computational loop for the diffusion nodes permits the extrapolation of diffusion-node temperatures provided acceleration of convergence criterion is met (refer to Section 6.2.7).

Diffusion-Node Temperatures

In order to facilitate the discussion to follow on the computational procedure, it is convenient to examine the forward-backward finite difference expression.¹³

$$C_{i} \frac{(T_{i,k+1} - T_{i,n})}{\Delta t} = \beta T_{forward} + (1 - \beta) T_{backward}$$
(6.2-7)
 $\beta = factor with range 0 \le \beta \le 1/2$

where:

$$T_{\text{forward}} = q_{i,n} + \sum_{j=1}^{p} a_{ij,n} (T_{j,n} - T_{i,n}) + \sum_{j=1}^{p} \sigma b_{ij,n} (T_{j,n}^{4} - T_{i,n}^{4}) (6.2-8)$$

$$T_{\text{backward}} = q_{i,n} + \sum_{j+1}^{p} a_{ij,n} (T_{j,k+1} - T_{i,k+1}) + \sum_{j=1}^{p} \sigma b_{ij} (T_{j,k+1}^{4} - T_{i,k+1}^{4}) (6.2-9)$$

$$i = 1, 2, \dots, N$$

$$T_{j,n}; T_{j,k+1} = \text{constant}, N < j \le p$$

n = nth time-step; k = kth iteration within a given time-step. $C_{i}, q_{i}, a_{ij}, b_{ij} = optionally specified (refer to Tables 6.2-1 -- 6.2-4)$

Any value of β less than one yields an implicit set of equations which must be solved simultaneously. For values of β less than or equal to one-half equation (6.2-7) represents an unconditionally stable set of equations, whereas values of β greater than one-half yields a set of equations with conditional stability.

The standard implicit algorithm used in subroutine CNBACK follows directly from equation (6.2-7) by letting $\beta = 0$, whereas the Crank-Nicolson method used in subroutine CNFWBK follows by letting $\beta = 1/2$. Subroutine

CNVARB uses a variable factor which is based upon the ratio of CSGMIN/DTIMEU; this ratio is internally calculated in CNVARB (refer to Section 6.4.3.2). In order to simplify the presentation, the following notation is used.

For CNBACK ($\beta = 0$):

$$Q_{i} = q_{i,n} + \overline{C}_{i,n} T_{i,n}$$
 (6.2-10)

$$Q_{sum} = Q_{i} + \sum_{j=1}^{i} G_{ij,n} T_{j,k+1} + \sum_{j=i+1}^{p} G_{ij,n} T_{j,k}$$
(6.2-11)

$$G_{sum} = \overline{C}_{i,n} + \sum_{j=1}^{p} a_{ij,n}$$
(6.2-12)

$$G_{ij,n} = a_{ij,n} + \sigma b_{ij,n} T_{j,l}^{3}$$
 (6.2-13)

 $(l = k, if j \ge i and l = k+1, if j < i)$

$$(q_{j})_{ave} = \frac{1}{2} \sum_{j=1}^{p} \sigma b_{ij,n} [(T_{i,k}^{4}) + (T_{i,k}^{4})_{2}], average heat loss (6.2-14)$$

from ith node, called radiation damping (refer to Section 6.2.6 for details)

= 0, if radiation is not present

For CNFWBK ($\beta = \frac{1}{2}$) (note equation (6.2-7) is multiplied by 2):

$$Q_{i} = 2q_{i,n} + 2\overline{C}_{i,n} T_{i,n} + \sum_{j=1}^{p} a_{ij,n} (T_{j,n} - T_{i,n}) + \sum_{j=1}^{p} \sigma b_{ij,n} (T_{j,n}^{4} - T_{i,n}^{4})$$
(6.2-15)

Q_{sum} = same as equation (6.2-11)

 $G_{sum} = 2\overline{C}_{i,n} + \sum_{j=1}^{p} a_{ij,n}$ (6.2-16) $G_{ij,n} = same as equation (6.2-13)$ $(q_i)_{ave} = same as equation (6.2-14)$

For CNVARB (variable β ') (note that equation (6.2-7) is multiplied by 2, so that β ' = 2 β now ranges, $0 \le \beta$ ' ≤ 1.0):

$$Q_{i} = 2q_{i,n} + 2\overline{C}_{i,n} T_{i,n} + \beta \left(\sum_{j=1}^{p} a_{ij,n} (T_{j,n} - T_{i,n}) + \sum_{j=1}^{p} \sigma b_{ij,n} (T_{j,n}^{4} - T_{i,n}^{4}) \right) (6.2-17)$$

$$Q_{sum} = Q_{i} + (2.0 - \beta') \begin{pmatrix} I \\ \Sigma \\ j=1 \end{pmatrix} = \begin{pmatrix} p \\ j, n \\ j, k+1 \end{pmatrix} + \begin{pmatrix} p \\ \Sigma \\ j=i+1 \end{pmatrix} = \begin{pmatrix} p \\ ij, n \\ j, k+1 \end{pmatrix}$$
(6.2-18)

$$G_{sum} = 2 \overline{C}_{i,n} + (2.0 - \beta') \sum_{j=1}^{p} a_{ij,n}$$
 (6.2-19)

 $G_{ij,n}$ = same as equation (6.2-13)

$$(q_i)_{ave} = \frac{2.0 - \beta'}{2} \sum_{j=1}^{p} \sigma b_{ij,n} [(T_{i,k}^4) + (T_{i,k}^4)_2]$$
, average heat (6.2-20)

loss from ith node, called radiation damping (refer to Section 6.2.6 for details)

= 0, if radiation is not present

i = 1, 2, ..., N

 β = 2.0*CSGMIN/DTIMEU (range allowed, $0 \le \beta' \le 1.0$)

 $T_{j,n}$; $T_{j,k}$ = constant, $N < j \le p$ (p is the total number of nodes)

n = nth time-step; k = kth iteration

 C_i , q_i , a_{ij} , b_{ij} = may be optionally specified (refer to Tables 6.2-1 - 6.2-4) $\overline{C}_{i,n} = C_{i,n}/\Delta t$

Calculation of the diffusion-node temperatures follows VARIABLES 1 call; the computational pattern is:

Iterative DØ-LØØP (kl=1,NLOOP) for the total nodal system is established. First Iterative Loop:

DO-LOOP (I=1,NND) on diffusion nodes is established.

The functions associated with the variable capacitance C_i , the variable impressed source q_i , and the variable coefficients G_k (a_{ij} for conduction and ob for radiation) between diffusion-diffusion and diffusion-arithmetic nodes are updated once for each time-step. These functional types are described in Section 6.2.1.2 and the computational pattern is indicated in the flow chart of Figure 6.2-3.

All known quantities (those evaluated at time-step n) are summed and are identified by the symbol Q_i (equations 6.2-10, 6.2-15 and 6.2-17). CSGMIN is evaluated.

Radiation damping is used; average radiation heat loss, $(q_i)_{ave}$, from the ith node is evaluated (refer to Section 6.2.6).

For CNVARB, $\beta' = 2.0 \times CSGMIN/DTIMEU$ is calculated.

The diffusion-node temperatures are calculated by "successive point" iteration (actually CNBACK and CNFWBK have slightly different first iterative pattern than CNVARB but the difference is not significant).

DD = 1.0 - DN

For CNVARB, the diffusion-node relaxation temperature change is calculated; maximum value is stored in DRLXCC.

Second and Succeeding Iterative Loops:

With the iterative loops after the first, those quantities C_i , q_i , and G_k which were updated during the first iteration are held constant.

Diffusion-node temperatures are found by using equation (6.2-21).

The diffusion-node relaxation temperature change is calculated and the maximum value stored in DRLXCC.

Check of DRLXCC against DRLXCA (allowable maximum diffusion-node relaxation temperature change) is made after the arithmetic-node temperature calculations.

Each third iteration, a check on solution convergence is made; if convergence is occurring linear extrapolation to accelerate convergence is made (refer to Section 6.2.7).

Arithmetic-Node Temperatures (if any)

During the first iterative loop the impressed source q_i and coefficient G_k (a for conduction and σ_{ij} for radiation) between arithmeticarithmetic nodes are updated once each time-step. On every loop, arithmeticnode temperatures are calculated using "successive point" iteration. The finite difference algorithm is presented in Section 6.2.5.1 (equation 6.2-6).

The arithmetic-node relaxation temperature change is calculated and the maximum is stored in ARLXCC.

During Each Iterative Loop After the First

Both DRLXCC and ARLXCC are checked against DRLXCA and ARLXCA, respectively. If both DRLXCA and ARLXCA are satisfied, the iteration ceases.

If LØØPCT equals NLØØP the message "RELAXATION CRITERIA NOT MET" is printed.

Both the calculated maximum diffusion-node and arithmetic-node temperature change (stored in DTMPCC and ATMPCC, respectively) are checked against the corresponding allowable temperature change stored in DTMPCA and ATMPCA. If DTMPCA is not satisfied, the time-step is decreased to,

 $\Delta t = .95 \star \Delta t (DTMPCA/DTMPCC)$

and all temperatures re-set to former values. The computational procedure is repeated with the smaller time-step.

If ATMPCA is not satisfied, the time-step is decreased to,

 $\Delta t = .95 \star \Delta t (ATMPCA/ATMPCC)$

and all temperatures re-set to former values. The computational procedure is repeated with the smaller time-step.

6.2.5.3 Steady State Routines

Diffusion nodes and arithmetic nodes are treated separately in CINDSS and CINDSL even though from a physical standpoint a distinction between diffusion nodes (nodes with capacitance) and arithmetic nodes (nodes with no capacitance) doesn't exist. Thus, the set of control constants for the diffusion nodes and another set of control constants for arithmetic nodes are similar to those used in the transient routines. No distinction in the type of nodes is made in CINDSM.

The computational procedure to be discussed applies only to CINDSS and CINDSL: CINDSM is considerably different (refer to Section 6.5.3).

Diffusion-Node Temperatures (nodes specified with capacitance even though the problem is steady state)

An iterative $D\emptyset$ -L $\emptyset\emptyset$ P (K1=1,NL $\emptyset\emptyset$ P) is established.

Within this iterative loop a DØ-LØØP (I=1,NND) on the diffusion nodes is made. The functions associated with the impressed source q_i and the variable coefficients G_k (a ij for conduction and σ_{b} for radiation) between diffusion-diffusion and diffusion-arithmetic nodes are updated each iteration.

Diffusion-node temperatures are calculated using "block" iteration for CINDSS and "successive point" iteration for CINDSL.

"Block" iteration (CINDSS):

$$T_{i,k+1} = DD*T_{i,k} + DN* \frac{\left(q_{i,k} + \sum_{j=1}^{p} G_{ij,k} T_{j,k}\right)}{\sum_{j=1}^{p} G_{ij,k}}$$
(6.2-22)

$$G_{ij,k} = a_{ij,k} + \sigma b_{ij,k} (T_{j,k}^2 + T_{i,k}^2) (T_{j,k} + T_{i,k})$$

$$DN = DAMPD (diffusion-node damping factor)$$

$$DD = 1.0 - DN$$

$$i = 1,2,...,NND (number of diffusion nodes)$$

$$k = kth iteration; p = total number of nodes$$

$$q_{i},a_{ij},b_{ij} = optionally specified to Tables (6.2-1 - 6.2-4)$$

$$T_{j,k} = constant, (NND + NNA) < j \le p (NNA is the number of arithmetic nodes$$

"Successive point" iteration (CINDSL):

$$T_{i,k+1} = DD*T_{i,k} + DN* \frac{\begin{pmatrix} q_{i,k} + \stackrel{i}{\stackrel{\Sigma}{j=1}} G_{ij,k} T_{j,k+1} + \stackrel{p}{\stackrel{\Sigma}{\stackrel{\sigma}{j=i+1}} G_{ij,k} T_{j,k} \end{pmatrix}}{\stackrel{p}{\stackrel{\Sigma}{j=1}} G_{ij,k}} (6.2-23)$$

$$G_{ij,k} = a_{ij,k} + \sigma b_{ij,k} (T_{j,\ell}^2 + T_{i,k}^2) (T_{j,\ell} + T_{i,k}) (\ell = k \text{ if } j \ge i \text{ and } \ell = k+1 \text{ if } j < i)$$

$$DN = DAMPD$$

$$DD = 1.0 - DN$$

i = 1, 2, ..., (NND + NNA) k = kth iteration; p = total number of nodes q_i, a_{ij}, b_{ij} = optionally specified to Tables (6.2-1 - 6.2-4) $T_{j,k}$ = constant, (NND + NNA) < j < p (NNA is the total number of arithmetic nodes)

Diffusion-node relaxation temperature change is calculated and the maximum is stored in DRLXCC.

Arithmetic-Node Temperatures (nodes specified with no capacitance)

Within this iterative $D\emptyset - L\emptyset\emptyset P$ a $D\emptyset - L\emptyset\emptyset P$ (I=NND+1, NND + NNA) is established.

The functions associated with impressed source q_i and variable coefficients G_k (a for conduction and b for radiation) between arithmetic-arithmetic nodes are updated each iteration.

Arithmetic-node temperatures are calculated using "successive point" iteration.

$$T_{i,k+1} = AD^{*}T_{i,k} + AN^{*} \frac{\begin{pmatrix} q_{i,k} + \sum_{j=1}^{p} G_{ij,k} T_{j,k+1} + \sum_{j=i+1}^{p} G_{ij,k} T_{j,k} \end{pmatrix}}{\sum_{j=1}^{p} G_{ij,k}} (6.2-24)$$

$$G_{ij,k} = a_{ij,k} + \sigma b_{ij,k} (T_{j,l}^{2} + T_{i,k}^{2}) (T_{j,l} + T_{i,l}) (T_{j,l} + T_{i,l}) (l = k \text{ if } j \ge 1 \text{ and } l = k+1 \text{ if } j < 1)$$

$$AN = DAMPA \text{ (arithmetic-node damping factor)}$$

$$AD = 1.0 - AN$$

$$i = (NND+1), (NND+2), \dots, (NND + NNA) \text{ (number of arithmetic nodes)}$$

$$k = k \text{ th iteration}$$

$$p = \text{ total number of nodes}$$

$$T_{j,k} = \text{ constant, (NND + NNA) < j \le p}$$

The arithmetic-node relaxation temperature change is calculated and the maximum value is stored in ARLXCC.

6 - 36

During Each Iterative Loop

Both DRLXCC and ARLXCC are checked against DRLXCA and ARLXCA, respectively. If both relaxation criteria, DRLXCA and ARLXCA, are satisfied, the iteration ceases.

If both relaxation criteria, DRLXCA and ARLXCA, are not met with NLØØP iterations, the message "ITERATION COUNT EXCEEDED, NLØØP = " is printed.

Energy balance of the system is calculated and is stored in control constant ENGBAL.

6.2.6 Radiation Damping

Radiation damping denotes an averaging of radiation heat loss technique used to prevent or minimize large temperature oscillations. This method is currently employed in only the implicit routines. The technique which is original with J. D. Gaski is based upon practical and computational considerations. Solution of numerous problems without large temperature oscillations indicates the effectiveness of the approach.

The radiation averaging technique is relatively simple conceptually and rather easily incorporated in the numerical solution routines. The computational pattern is such that the diffusion nodes are encountered sequentially. Let the encountered node be the ith node. A check is made for the presence of a radiation coefficient, $G_k = \sigma b_{ij}$, to the ith node. If one or more radiation connections is present, the radiation heat <u>loss</u>, $(q_i)_{rl}$, from the ith node is calculated based upon the previous temperature $T_{i,k}$.

$$(q_i)_{rl} = \sum_{j=1}^{\Sigma} \sigma b_{ij,n} T^4_{i,k}$$
 (6.2-24)

where,

j = all radiation connections to node i n = nth time-step k = kth iteration Using $(q_i)_{r1}$, a second temperature $(T_{i,k})_2$, is found as follows: $(T_{i,k})_2 = [Q_{sum} - (q_i)_{r1}]/G_{sum}$ (6.2-25)

where,

$$Q_{sum} = \overline{C}_{i} T_{i,n} + q_{i,n} + \sum_{j=1}^{i} a_{ij,n} T_{j,k+1} + \sum_{j=i+1}^{p} a_{ij,n} T_{j,k} + \sum_{j=i+1}^{i} \sigma b_{ij,n} T_{j,k} + \sum_{j=i+1}^{p} \sigma b_{ij,n} T_{j,k}^{4}$$
(6.2-26)

$$G_{sum} = \overline{C}_{i} + \sum_{j=1}^{p} a_{ij,n}$$
(6.2-27)

Note that in the evaluation of $(T_{i,k})_2$, the damping factor DAMPD is not used. Note further that G_{sum} does not contain $\Sigma ob_{ij,n} T_{i,k}^3$ since it is accounted for in the radiation loss term, $(q_i)_{r1}$.

Now a second radiation heat loss based on $(T_{i,k})_2$ is found,

$$(q_i)_{r2} = \sum_{j=1}^{\infty} \sigma b_{ij,n} (T_{i,k})_2^4$$
 (6.2-28)

Equations (6.2-24) and (6.2-28) are then averaged,

$$(q_i)_{ave} = [(q_i)_{r1} + (q_i)_{r2}]/2.0$$
 (6.2-29)

This average radiation heat loss from an ith node is used in the diffusion-node finite difference algorithm as follows,

$$T_{i,k+1} = DD* T_{i,k} + DN* \frac{(Q_{sum} - (q_i)_{ave})}{G_{sum}}$$
 (6.2-30)

where, DN = DAMPD DD = 1.0 - DN $(q_i)_{ave} = average radiation heat loss (equation 6.2-29)$ $G_{sum} = \overline{C}_i + \sum_j a_{ij,n}$ $Q_{sum} = of the form shown by equation (6.2-26). The actual$ expression depends upon algorithm. Equation (6.2-26)is for the standard implicit method.

The reason behind the use of $(q_i)_{ave}$ is that if the initial temperature $T_{i,k}$ is too large, the heat loss from the ith node, $(q_i)_{r1}$ would then be too large. As a result the evaluation of $(T_{i,k})_2$ with $(q_i)_{r1}$ would yield a temperature that is too low. Thus, the averaging of of $(q_i)_{r1}$ and $(q_i)_{r2}$ would be much closer to the true heat loss from the

ith node. If $T_{i,k}$ is too small then $(T_{i,k})_2$ would be too large; the averaging scheme still holds true.

6.2.7 Acceleration of Convergence by Extrapolation Technique

Several of the SINDA numerical solution routines use an extrapolation technique to accelerate convergence of the iterative procedure. The extrapolation technique is used in the implicit routines CNBACK, CNFWBK, and CNVARB for the iterative temperature solution of the diffusion nodes, but is not used for the iterative temperature solutions of the arithmetic nodes. The extrapolation method is also used in the steady state routines CINDSL and CINDSM for the iterative temperature solution of both the diffusion and the arithmetic nodes.

6.2.7.1 Extrapolation Technique

The extrapolation is based on a zero temperature difference condition which is defined to be a point where the temperature change of a particular node over two successive iterations is zero. The governing equations are developed as follows:

Consider the temperatures of an ith node at three successive iterations as shown in Figure 6.2-4a. Let these temperatures, which are assumed to be successively decreasing (or increasing), be denoted as,

where,

k is the present iteration

k-1 is the previous iteration

k-2 is two iterations before the kth iteration

By taking the differences,

$$\Delta T_{i,k-1} = T_{i,k-2} - T_{i,k-1}$$

 $\Delta T_{i,k} = T_{i,k-1} - T_{i,k}$

and plotting these temperature differences as a function of iterations, the iterative point of zero temperature difference can be found by linear extrapolation as shown in Figure 6.2-4b. The corresponding expression for the line is found by using the point, $\Delta T_{i,k}$ at I = k and the slope, $(\Delta T_{i,k} - \Delta T_{i,k-1}) / (k-(k-1))$, to yield,



No. of Iterations, I

Figure 6.2-4a. Temperature (ith) vs. No. of Iterations



No. of Iterations, I

Figure 6.2-4b. Temperature Difference vs. No. of Iterations





Figure 6.2-4c. Extrapolation of Temperature (ith) to New Value Figure 6.2-4. Method of Extrapolation to Accelerate Convergence

$$\Delta T_{i,I} = \Delta T_{i,k} + (\Delta T_{i,k} - \Delta T_{i,k-1})(I - k)$$
 (6.2-31)

where, I = iterations

Since at the zero temperature difference condition, $\Delta T_{i,I} = 0$, the expression for the extrapolated iterations, $K_{c} = (K - k)$, is found to be,

$$K_{e} = -\Delta T_{i,k} / (\Delta T_{i,k} - \Delta T_{i,k-1})$$
(6.2-32)

Now, by extrapolating the line established by the temperatures, $T_{i,k-1}$ and $T_{i,k}$ to the line I = K, as shown in Figure 6.2-4c, the extrapolated temperature $T_{i,K}$ is found. The expression is readily found to be,

$$T_{i,I} = T_{i,k} + (T_{i,k} - T_{i,k-1})(I - k)$$
 (6.2-33)

Since I = K and $K - k = K_o$, equation (6.2-33) becomes,

$$T_{i,K} = T_{i,k} (1 + K_e) - K_e T_{i,k-1}$$
 (6.2-34)

6.2.7.2 Programming Considerations

Each applicable node is tested at the completion of each third iteration to determine if the extrapolation method should be applied. If K_c is calculated to be less than or equal to zero, extrapolation is neglected since the error function is diverging. If K_e is calculated to be greater than zero, a new temperature is calculated based on equation (6.2-34); however, to avoid problems associated with a nearly-zero slope of the line representing the temperature difference vs. number of iterations relationship (Figure 6.2-4b), K_e , is set to a number K_m ; otherwise, K_e could be a very large number. For the implicit routines, CNBACK, CNFWBK, and CNVARB, $K_m = 10$. For the steady state routine CINDSL $K_m = 8$ and for steady state routine CINDSM a criterion based upon the maximum temperature is used.

6.2.7.3 Routines Using Acceleration of Convergence

SINDA numerical solution routines that employ the acceleration of convergence features are:

CINDSL, CINDSM	Steady state routines
CNBACK, CNFWBK, CNVARB	Transient implicit routines

6.2.7.4 Comment on Acceleration of Convergence

Neither an extensive study on the value of the acceleration convergence feature has been made, nor has one been reported, but the

ديرغ

limited results presently available indicate that for the steady state routine CINDSL, the number of iterations is reduced approximately 20%. Results are not available for the implicit routines.

A study of the acceleration of convergence feature is made difficult because the method is not a user option in the applicable SINDA numerical solution routines. Thus, the user must be sufficiently versed with the routines in order to delete the acceleration of convergence feature.

6.2.8 Other Characteristics of the SINDA Numerical Solution Routines 6.2.8.1 Units

SINDA, as presently coded, requires that the temperatures must be specified in degrees Fahrenheit (°F) since the conversion factor to obtain degrees absolute is internally set at 460.0. This means that the units must be consistent with °F (or °R). The execution routines as presently coded do not permit the use of other units.

6.2.8.2 General Comments on Computational Features

Many of the computational features such as radiation damping are original with J. D. Gaski. No theoretical proofs are offered since a practical "gut-feel" development was often used in lieu of a sophisticated mathematical approach; the features, in general, appear to meet the intended objectives. It should be particularly noted that the numerical solution routines are computationally similar; within a particular numerical solution class explicit, implicit or steady state, the computational similarity is even more pronounced. Yet on the other hand, similarity of patterns are broken for no particular reason other than the programmer's whim. 6.3 Transient Explicit Solution Routines

SINDA explicit solution routine number six. These are identified as follows:

- CNFRWD Conditionally stable explicit forward difference. Requires short pseudo-compute sequence (SPCS).
- CNFRDL Identical to CNFRWD except that the long pseudocompute sequence (LPCS) is required.
- CNFAST Modified CNFRWD for accelerated forward differencing. Requires short pseudo-compute sequence (SPCS).
- CNEXPN Unconditionally stable explicit differencing using exponential prediction. Requires short pseudo-compute sequence (SPCS).
 - CNDUFR Unconditionally stable explicit differencing using DuFort-Frankel method. Requires short pseudo-compute sequence (SPCS).
 - CNQUIK Unconditionally stable explicit differencing using a combination of half CNEXPN and half CNDUFR. Requires short pseudo-compute sequence (SPCS).

A detailed description of each explicit routine is presented on the pages to follow with heavy reliance upon the general description of Section 6.2. A brief description of these routines is summarized first.

<u>CNFRWD</u> uses an explicit forward differencing algorithm and requires the short pseudo-compute sequence (SPCS). The explicit method is characterized by computational simplicity and stability limitations. Since the allowable time-step is governed by the smallest time constant of the network, care must be given in reducing the physical system to a reasonable lumped-parameter model. Arithmetic-node temperatures are calculated by "successive point" iteration.

<u>CNFRDL</u> is identical to CNFRWD except that CNFRDL requires the long pseudo-compute sequence instead of the short pseudo compute sequence. CNFRDL requires slightly less solution time than CNFRWD but the difference is not significant; CNFRDL does require more core storage, however.

<u>CNFAST</u> represents a modified CNFRWD with the modifications intended to decrease the computational time. A user specified control constant DTIMEL which contains the minimum time-step allowed is used as a criterion for isolating those diffusion nodes that are to receive the steady state calculations. A large pocket of internally converted diffusion nodes can present considerable accuracy problems.

<u>CNEXPN</u> uses an unconditionally stable explicit method with the intent to reduce computational time at the expense of temperature accuracy. If accuracy is an important consideration, another routine such as CNFRWD would be a better choice. As a note of interest, CNEXPN solutions tend to lag in time the true solutions.

<u>CNDUFR</u> uses the unconditionally stable DuFort-Frankel method with the intent to reduce computational time by using time-steps greater than those allowed with the conditionally stable explicit methods. Again accuracy may be compromised. CNDUFR solutions tend to lead in time the true solutions.

<u>CNQUIK</u> uses half CNEXPN and half CNDUFR. Why? Since CNEXPN solutions tend to lag in time and CNDUFR solutions tend to lead in time, a combination may yield better solutions. Preliminary results indicate that CNQUIK solutions are more accurate than either CNEXPN or CNDUFR for the same computational time.
6.3.1 Subroutines: CNFRWD and CNFRDL

6.3.1.1 General Comments

Subroutines CNFRWD and CNFRDL are numerical solution routines that use the forward finite difference explicit approximation^{12, 14} of the parabolic differential equation. CNFRWD and CNFRDL are identical except that CNFRDL requires the short pseudo compute sequence (SPCS) whereas CNFRDL requires the long pseudo compute sequence (LPCS). The need for both routines becomes apparent when it is understood that if a steady state numerical solution routine is followed by a transient numerical solution routine, both routines must have consistent PCS (LPCS or SPCS). As a note of interest, each arithmetic node receives the long pseudo compute sequence (LPCS) but this is done internally by the program.

The forward finite difference explicit method as used in CNFRWD and CNFRDL is the conventional Euler method that neither provides a check on the accuracy nor does it provide any scheme of correction once the temperature values are calculated except for the arithmetic nodes which are reiterated NLØØP-times. The explicit method is characterized by computational simplicity and stability limitations with the temperature error at any time point being of the order Δt , $O(\Delta t)$, provided the stability criterion is satisfied. For a rapidly changing boundary condition, such as a heat source, there is no assurance that the calculated temperatures are accurate during the transient period, particularly near the start of the transient, even though the stability criterion is satisfied. Since the allowable time step is governed by the smallest time constant of the network, care must be given in reducing the physical system to a lumpedparameter model. Nonlinearity due to the presence of thermal radiation exchange or temperature-time varying coefficients can lead to numerical solution difficulties; the presence of arithmetic nodes can also present difficulties. These routines offer a number of control constants many of which can be optionally specified by the user to affect the numerical results.

Even with the experience gained through the use of these routines, no realistic criteria can be stated except for the qualitative guidelines indicated above. It is thus recommended that the user becomes familiar with various control constants and their role. The presentation

to follow is intended to provide the instructional information.

6.3.1.2 Finite Difference Approximation and Computational Algorithm

The forward finite difference explicit formulation of the lumped parameter heat balance equations was presented in Section 5.2.1. For convenience, the expression is repeated here.

$$C_{i} \frac{(T_{i,n+1} - T_{i,n})}{\Delta t} = q_{i,n} - \sum_{j=1}^{p} a_{ij} (T_{j,n} - T_{i,n}) + \sum_{j=1}^{p} \sigma b_{ij} (T_{j,n}^{4} - T_{i,n}^{4})$$

(From equation 5.2-1 of Section 5.2.1)

By letting $G_{ij,n} = a_{ij,n} + \sigma b_{ij,n} (T_{j,n}^2 + T_{i,n}^2)(T_{j,n} + T_{i,n})$, equation (5.2-1) becomes,

$$C_{i} \frac{(T_{i,n+1} - T_{i,n})}{\Delta t} = q_{i,n} + \sum_{j=1}^{p} G_{ij,n} (T_{j,n} - T_{i,n})$$
(6.3-1)

The algorithm as used in the subroutines for the diffusion nodes and for the arithmetic nodes may be expressed as follows.

Diffusion Nodes

$$T_{i,n+1} = T_{i,n} + \frac{\Delta t}{C_i} \left[q_{i,n} + \sum_{j=1}^{p} G_{ij,n} (T_{j,n} - T_{i,n}) \right]$$
(6.3-2)

where,

, n = nth time-step

 $\Delta t = time-step$ (refer to Section 6.2.4)

i = 1,2,...,NND (number of diffusion nodes)

Arithmetic Nodes (if any)

$$T_{i,k+1} = DD* T_{i,k} + DN* \frac{\begin{pmatrix} q_{i,n} + \sum_{j=1}^{i} G_{ij,n} T_{j,k+1} + \sum_{j=i+1}^{p} G_{ij,n} T_{j,k} \end{pmatrix}}{\sum_{\substack{j=1 \\ j=1}}^{p} G_{ij,n}}$$
(6.3-3)

where, k = kth iteration loop; i = (NND + 1), (NND + 2),...,(NND + NNA) $C_{i},q_{i},a_{ij},b_{ij}$ = optionally specified (refer to Tables 6.2-1 - 6.2-4) $T_{j,k}$ = constant, (NND + NNA) < j < p (NNA is the number of arithmetic nodes and p is the total number of nodes) DN = DAMPA (arithmetic node damping factor, refer to Section 6.2.3.2) DD = 1.0 - DN $G_{ij,n} = a_{ij,n} + \sigma b_{ij,n} (T_{j,\ell}^2 + T_{i,k}^2)(T_{j,\ell} + T_{i,k})$

 $(l = k, if j \ge i and l = k+1, if j < i)$

6.3.1.3 Comments on the Computational Procedure

The important steps of the computational procedure used in subroutines CNFRWD and CNFRDL are indicated in Table 6.3-1. For a detailed step-by-step computational description, the user must examine the computer listings for CNFRWD and CNFRDL presented in Appendix A, but some general computational details are given in Section 6.2.5.1. Both CNFRWD and CNFRDL use essentially the same computational steps with the difference occurring in the calculation of the diffusion-node temperatures as shown in the flow chart of Figure 6.3-1; a flow chart for the calculation of the arithmetic-node temperatures is shown in Figure 6.3-2. A functional flow chart of CNFRWD and CNFRDL is shown in Figure 6.3-3. The difference between CNFRWD and CNFRDL is due to the use of the short pseudo-compute sequence (SPCS) by CNFRWD and the use of the long pseudo-compute sequence (LPCS) by CNFRDL.

All diffusion-node temperatures are calculated by a two-pass operation prior to the calculation of the arithmetic node temperatures. On the first pass the pseudo-compute sequence for the diffusion nodes is addressed and the heat flow is calculated and the direction determined for each conductor encountered; the appropriate heat flow and conductance summations are performed. Refer to Section 6.2.5.1 for more details on the computational procedure.

The stability criterion of each diffusion node is calculated and the minimum value is placed in control constant CSGMIN. The time-step used (stored in control constant DTIMEU) is calculated as 95% of CSGMIN divided by control constant CSGFAC which is set at 1.0 unless specified larger by the user. A "look ahead" feature is used when DTIMEU is calculated. If one time-step will pass the output time point the time-step is set to lie on the output time point; if two time-steps will pass the output time point, the time-step is set so that the end of the two time-steps will lie on the output time point. DTIMEU is checked against both DTIMEH and DTIMEL. If DTIMEU exceeds DTIMEH, DTIMEU is set equal to DTIMEH, and if DTIMEU is less than DTIMEL, the "run" is terminated. DTIMEL is internally set to zero if not specified and DTIMEH is set to 1.0E+8 if not specified. The maximum diffusion node temperature change over a time-step is placed in control constant DTMPCC and is checked against the allowable diffusion node temperature change stored in the optionally user specified control constant DTMPCA which is not specified is set to 1.0E+8. If DTMPCC is larger than DTMPCA, DTIMEU is shortened and the calculations repeated. Refer to Section 6.2.4 for detailed procedure on time-step calculation.

The user may iterate the arithmetic node calculations during a time-step by specifying control constant NLØØP and adjust the solution by the use of ARLXCA. The maximum arithmetic node temperature change over an iteration is placed in control constant ARLXCC and is checked against the arithmetic node temperature change criterion stored in ARLXCA. Satisfaction of either NLØØP or ARLXCA terminates the iterative process for that time-step. If the arithmetic node iteration count exceeds NLØØP the results are retained and computation proceeds without user notification. The maximum arithmetic node temperature change over the time-step is stored in control constant ATMPCC and is checked against the allowable temperature change stored in ATMPCA. If larger, the time-step is shortened and the calculation repeated. The user may also specify the control constant DAMPA in order to dampen possible oscillation due to nonlinearities.

6.3.1.4 <u>Control Constants</u>

Control constants ØUTPUT and TIMEND (> TIMEØ) must be specified as indicated in Table 6.2-5 and described in Section 6.2.3.2; otherwise the "run" will terminate with an error message. The function of optionally

specified control constants ARLXCA, ATMPCA, BACKUP, CSGFAC, DAMPA, DTIMEH, DTIMEL, DTMPCA, NLØØP, and TIMEØ is described in Section 6.2.3.2. Note particularly that TIMEØ may be set negative and that NLØØP is set to one if not specified.

6.3.1.5 Error and Other Messages

If control constants ØUTPUT and TIMEND are not specified, the following error message will be printed for each,

ØUTPUT	"NØ 🤅	Ø ØUTPUT		INTERVAL"	
TIMEND	"TM	E STEP	TØØ	SMALL"	

The reason for the TIMEND error message is that a direct check on TIMEND is not made; the resultant error message just happens to be a quirk in the coding.

If the short pseudo-compute sequence SPCS is not specified, the error message will be,

"CNFRWD REQUIRES SHORT PSEUDØ-COMPUTE SEQUENCE"

If the long pseudo-compute LPCS is not specified, the error message will be,

"CNFRDL REQUIRES LONG PSEUDO-COMPUTE SEQUENCE"

If the dynamic storage allocation is not sufficient (NDIM < (NND + NNA)), the message will be,

" LØCATIONS AVAILABLE"

Note that the <u>number printed will be negative</u> indicating the additional storage locations required.

If the time-step used is less than the time-step allowed (DTIMEL) which may be optionally specified by the user, the message will be,

"TIME STEP TØØ SMALL"

If CSGMIN < 0, the message printed will be,

"CSGMIN ZERØ or NEGATIVE"

Checks on the control constants, the pseudo-compute sequence and the dynamic storage allocation are made in the following sequence with the "run" terminating if a single check is not satisfied, ØUTPUT, pseudo-compute sequence, dynamic storage locations

It should be particularly noted that <u>no message</u> is printed if ARLXCA is not satisfied with NL $\phi\phi$ P iterations; ARLXCA and NL $\phi\phi$ P are optionally specified control constants.

Table 6.3-1. Basic Computational Steps for CNFRWD and CNFRDL

- Specification of control constants (all control constants are pre-set to zero). Control constants ØUTPUT and TIMEND must be specified. SPCS is required for CNFRWD and LPCS for CNFRDL. (Refer to Table 6.2-4 for pominal values and Section 6.2.3.2 for description.)
- Sufficiency check on dynamic storage. Requirements = NND + NNA (NND = diffusion nodes and NNA = arithmetic nodes).
- 3. Setting and/or calculation of time-step, Δt . (Refer to Section 6.2.4 for detailed procedure.)
- 4. Setting of source and diffusion node dynamic storage locations at zero.
- 5. Calling of VARIABLES 1. (Refer to Section 6.2.2.2.)
- 6. Checking of BACKUP. (Refer to Section 6.2.3.2.)
- 7. Calculation of diffusion-node temperatures. (Refer to Section 6.2.5.1 for description and to flow chart of Figure 6.3-1.)

Diffusion-node temperatures are calculated by using: (refer to Section 6.3.1.2.)

$$T_{i,n+1} = T_{i,n} + \Delta T_{i,n}$$

where, $\Delta T_{i,n} = \frac{\Delta t}{C_{i,m}} \left[q_{i,n} + \sum_{j=1}^{p} G_{ij,n} (T_{j,n} - T_{i,n}) \right]$

- 8. Erasure of all temperature calculations for latest time-step if allowable temperature change criterion DTMPCA is not satisfied and recalculation of temperatures with reduced time-step.
- 9. Calculation of arithmetic-node temperatures; if the number of iterations equals NLØØP the temperatures are retained without user modification.
 (Refer to Section 6.2.5.1 for description and to flow chart of Figure 6.3.2)
- Erasure of all temperature calculations for latest time-step if allowable temperature change criterion ATMPCA is not satisfied and recalculation of temperatures with reduced time-step.
- 11. Setting of BACKUP to 0.0 and the calling of VARIABLES 2.

If BACKUP is nonzero, temperatures are re-set to former values and the computational procedure repeated.

- 12. Advancing of time, checking of time to print, and the printing at the output interval.
- 13. Calling of ØUTPUT CALLS.
- 14. Checking for problem end-time stored in user specified control constant TIMEND.



Figure 6.3-1. QSUM and GSUM for "Block" Diffusion-Node Temperature Calculation, CNFRWD and CNFRDL





Figure 6.3-3. Functional Flow Chart for CNFRWD and CNFRDL

6.3.2 Subroutine: CNFAST

6.3.2.1 General Comments

Subroutine CNFAST, which requires the short pseudo compute sequence (SPCS) represents a modified CNFRWD with the modifications intended to decrease the computational time. Use of CNFAST requires a user specification of control constant DTIMEL which represents the minimum time-step allowed in addition to control constant ØUTPUT. With minimum computational time and adequate temperature values as the objective, the computational procedure is simplified. A number of checks on control constants are eliminated and temperature nodes with CSGMIN less than the allowable timestep, DTIMEL, are calculated using the steady state equations.

Although experience on the use of CNFAST is rather limited at this time, it is clear that the user specified DTIMEL should be sufficiently small that only a small number of the diffusion nodes should receive the steady state equations. These steady state equations are computed only once during a time-step and thus are not treated computationally the same as the other user-specified arithmetic nodes. A large pocket of internally converted diffusion nodes would lead to large temperature inaccuracies.

6.3.2.2 Finite Difference Approximation and Computational Algorithm

The finite difference expressions for CNFAST are the same as those indicated in Section 6.3.1.2 for subroutines CNFRWD and CNFRDL, but the application of these equations in the computation procedure is different.

Diffusion Nodes

If the user specified control constant, DTIMEL, which represents the maximum time-step allowed as specified by the user is less than or equal to CSGMIN, the diffusion node temperature is calculated as,

$$T_{i,n+1} = T_{i,n} + \frac{\Delta t}{C_i} \left[q_{i,n} + \sum_{j=1}^{p} G_{ij,n} (T_{j,n} - T_{i,n}) \right]$$
 6.3-4)

where, $\Delta t = time-step$ (refer to Section 6.2.4); n = nth time-step $C_{i}, q_{i}, a_{ij}, b_{ij} = optionally specified (refer to Tables 6.2-1 - 6.2-4)$ i = 1, 2, ..., NND (of diffusion nodes with DTIMEL \leq CSGMIN) $T_{j,n} = constant, (NND + NNA) < j \leq p$ (NNA is the number of arithmetic nodes and p is the total number of nodes) $G_{ij,n} = a_{ij,n} + \sigma b_{ij,n} (T_{j,n}^2 + T_{i,n}^2) (T_{j,n} + T_{i,n})$ If DTIMEL > CSGMIN, the time-step is set at DTIMEL and the iiffusion node temperature calculated with no iterations as,

$$T_{i,n+1} = \left(\frac{q_{i,n} + \sum_{j=1}^{p} G_{ij,n} (T_{j,n} - T_{i,n})}{\sum_{\substack{j=1 \\ j=1}}^{p} G_{ij,n}}\right)$$
(6.3-5)

where, n means the nth time-step

i = 1,2,...,NND (number of diffusion nodes with DTIMEL > CSGMIN)

Arithmetic Nodes (if any)

The arithmetic-node temperatures are calculated in the same manner as in CNFRWD (Section 6.3.1.2) or refer to Section 5.2.3 for finite difference algorithm.

6.3.2.3 <u>Comments on the Computational Procedure</u>

The important steps of the computational procedure used in subroutine CNFAST are indicated in Table 6.3-2 and a functional flow chart is shown in Figure 6.3-4. For a detailed computational description, the user should examine the computer listing for CNFAST in Appendix A, but some general computational details are presented in Section 6.2.5.1. The computational procedure is similar to the one used in CNFRWD with the major difference being the use of DTIMEL which represents the user specified minimum time-step allowed. The time-step calculations stored in DTIMEU proceed exactly as in CNFRWD until the check with DTIMEL is made. If DTIMEU (CSGMIN of a node) \geq DTIMEL, the diffusion node temperature calculation is identical to CNFRWD. If DTIMEU (CSGMIN of a node) < DTIMEL, the diffusion node receives the steady state calculation.

Control constants DTMPCA which contains the allowable diffusionnode temperature change and ATMPCA which contains the arithmetic-node temperature change are not checked in CNFAST. Thus time-steps are not shortened and temperature calculations repeated. The remainder of the computational procedure follows those of CNFRWD (Section 6.3.1.3).

6.3.2.4 Control Constants

Control constants DTIMEL, ØUTPUT and TIMEND (>TIMEØ) must be specified as indicated in Table 6.2-5 and described in Section 6.2.3.2;

otherwise the "run" will terminate with an error message. The function of optionally specified control constants ARLXCA, BACKUP, DAMPA, DTIMEH, NLØØP and TIMEØ is described in Section 6.2.3.2. As mentioned before in a previous paragraph, the user should take considerable amount of caution in specifying DTIMEL in order to prevent large pockets of nodes that receive the steady state equation without reiteration. Note also that TIMEØ may be set negative and that NLØØP is set to one if not specified.

6.3.2.5 Error and Other Messages

If control constants DTIMEL, ØUTPUT and TIMEND are not specified, the following error message will be printed for each,

DTIMEL	"NØ DTIMEL"
ØUTPUT	"NØ ØUTPUT INTERVAL"
TIMEND	no message

A direct check on TIMEND is not made; an indirect message is printed for the other explicit routines but is not output for CNFAST.

If the short pseudo-compute sequence SPCS is not specified, the error message will be,

"CNFAST REQUIRES SHORT PSEUDO-COMPUTE SEQUENCE"

If the dynamic storage allocation is not sufficient (NDIM < NND), the message will be,

" LOCATIONS AVAILABLE"

Note that the <u>number printed will be negative</u> indicating the additional storage locations required.

If CSGMIN \leq 0, the message printed will be,

"C/SK ZERØ or NEGATIVE"

13

Checks on the control constants, the pseudo-compute sequence and the dynamic storage allocation are made in the following sequence, with the run terminating if a single check is not satisfied,

ØUTPUT, DTIMEL, pseudo-compute sequence, and dynamic storage locations.

It should be particularly noted that <u>no message</u> is printed if ARLXCA is not satisfied with NLØØP iterations; ARLXCA and NLØØP are optionally specified control constants.

Table 6.3-2. Basic Computational Steps for CNFAST

- 1. Specification of control constants. Control constants DTIMEL, ØUTPUT and TIMEND must be specified. SPCS is required for CNFAST. (Refer to Table 6.2-5 for values and Section 6.2.3.2 for descriptions.)
- Sufficiency check on dynamic storage. Requirements = NND (NND = diffusion nodes).
- 3. Setting and/or calculation of time-step, Δt . (Refer to Section 6.2.4 for detailed procedure.)

Note that initial time-step equal DTIMEL and subsequent time-step is the larger of CSGMIN or DTIMEL.

- 4. Setting of source and diffusion node dynamic storage locations to zero.
- 5. Calling of VARIABLES 1. (Refer to Section 6.2.2.2 for description.)
- 6. Checking of BACKUP. (Refer to Section 6.2.3.2 for description.)
- 7. Calculation of diffusion-node temperatures. (Refer to Section 6.2.5.1 for description.) Calculation differs from the other explicit routines, since diffusion nodes with CSGMIN less than DTIMEL receive steady state calculation (refer to Section 6.3.2.2.)

If DTIMEL < CSGMIN, the node temperature is calculated as,

$$T_{i,n+1} = T_{i,n} + \frac{\Delta t}{C_i} \begin{bmatrix} p \\ \Sigma \\ j=1 \end{bmatrix} G_{ij,n} (T_{j,n} - T_{i,n}) + q_{i,n}$$

If DTIMEL > CSGMIN, the node temperature is calculated using the steady state expression,

$$T_{i,n+1} = \begin{pmatrix} q_{i,n} + \sum_{j=1}^{p} G_{ij,n} & (T_{j,n} - T_{i,n}) \\ p_{j=1} & G_{ij,n} \end{pmatrix}$$

- 8. Calculation of arithmetic-node temperatures if the number of iterations equals NLØØP the temperatures are retained without user notification. (Refer to Section 6.2.5.1 for details.)
- 9. Calling of VARIABLES 2. (Refer to Section 6.2.2.3 for description.)
- 10. Advancing of time, checking of time to print, and the printing at the the output interval.
- 11. Calling of ØUTPUT CALLS.
- 12. Checking for problem end-time stored in user specified control constant TIMEND



Figure 6.3-4. Functional Flow Chart for CNFAST

6.3.3 Subroutine: CNEXPN

6.3.3.1 General Comments

Subroutine CNEXPN is an explicit routine based upon the exponential prediction method;¹, ¹⁷ the method being unconditionally stable permits any size time-steps and requires the short pseudo-compute sequence (SPCS). An infinite time-step reduces the transient equation to a steady state one. Although the method is unconditionally stable, stability should not be confused with accuracy. Comparison of several numerical methods, including the exponential approximation, is given in Reference 17.

If accuracy is an important consideration, time-steps should not be larger than those taken with the standard explicit method such as used in CNFRWD. If high accuracy is not an important consideration, considerable savings in computational time can be affected with the use of a large timestep. It should be noted that the same savings in computational time may be possible with the implicit routines. As another note of interest, CNEXPN solutions have a tendency to lag in time the true temperatures.

6.3.3.2 Finite Difference Approximation and Computational Algorithm

Diffusion Nodes

The expression for the numerical method used in subroutine CNEXPN for solving the diffusion-node temperatures may be derived from the heat balance equation (5.1-6).

$$\frac{d\mathbf{T}_{i}}{dt} = \frac{1}{C_{i}} \begin{bmatrix} q_{i} + \sum_{j=1}^{p} a_{ij} (\mathbf{T}_{j} - \mathbf{T}_{i}) + \sum_{j=1}^{p} \sigma b_{ij} (\mathbf{T}_{j}^{4} - \mathbf{T}_{i}^{4}) \end{bmatrix}$$
(equation
5.1-6 of
Section 5)
 $\mathbf{i} = 1, 2, \dots, N$
 $\mathbf{T}_{j} = \text{constant}, N < j \le p$

If $G_{ij} = a_{ij} + \sigma b_{ij} (T_j^2 + T_i^2)(T_j + T_i)$ equation (5.1-6) becomes,

$$\frac{\mathrm{d}\mathbf{T}_{\mathbf{i}}}{\mathrm{d}\mathbf{t}} = \frac{1}{C_{\mathbf{i}}} \left[\mathbf{q}_{\mathbf{i}} + \sum_{\mathbf{j=1}}^{\mathbf{p}} \mathbf{G}_{\mathbf{ij}} \left(\mathbf{T}_{\mathbf{j}} - \mathbf{T}_{\mathbf{i}} \right) \right]$$

$$\mathbf{i} = 1, 2, \dots, N$$
(6.3-6)

$$T_j = constant, N < j \leq p$$

If we further let G_{ij} , q_i and T_j be invariant with time and temperature, equation (6.3-6) may be integrated rather easily to yield,

$$T_{i,n+1} = T_{i,n} e^{-\alpha_n \Delta t} + \frac{q_{i,n} + \sum_{j=1}^{p} G_{ij,n} T_{j,n}}{p} \left(1 - e^{-\alpha_n \Delta t}\right) \quad (6.3-7)$$

$$\sum_{j=1}^{\sum} G_{ij,n}$$

where, n = nth time-step; i = 1, 2, ..., NND (number of diffusion nodes) $C_{i},q_{i},a_{ij},b_{ij} = may$ be optionally specified (refer to Tables 6.2-1 - 6.2-4) $T_{j,n} = constant, (NND + NNA) < j < p$ (NNA is the number of arithmetic nodes and p is the total number of nodes)

$$\alpha_n = \frac{\sum_{j=1}^{p} G_{ij,n}}{C_{i,n}}$$

 $\Delta t = time-step$ (refer to Section 6.2.4)

$$G_{ij,n} = a_{ij,n} + \sigma b_{ij,n} (T_{i,n}^2 + T_{j,n}^2) (T_{i,n} + T_{j,n})$$

Computationally equation (6.3-7) is applied to the diffusion nodes. It should be noted that the form of equation (6.3-7) represents a "block" change in temperatures since the evaluation of $T_{i,n+1}$ is based upon $T_{i,n}$.

Arithmetic Nodes (if any)

Arithmetic-node temperatures are calculated in the same manner as in CNFRWD (Section 6.3.1.2) or refer to Section 5.2.3 for finite difference algorithm.

6.3.3.3 Comments on the Computational Procedure

The important steps of the computational procedure used in subroutine CNEXPN are indicated in Table 6.3-3 and a functional flow chart is shown in Figure 6.3-5. A detailed computational procedure requires the examination of the CNEXPN computer listing which is presented in Appendix A but some general computational details are given in Section 6.2.5.1. The computational process of subroutine CNEXPN is essentially identical to CNFRWD with the difference being the finite difference expression used for the calculation of the diffusion nodes and the time-step which is calculated as CSGMIN*CSGFAC in lieu of CSGMIN/CSGFAC. The "look ahead" feature for

time-step calculation as well as a check with DTIMEH, DTIMEL and DTMPCA is identical to CNFRWD. Temperatures of arithmetic nodes are calculated after the diffusion nodes and utilize NL $\phi\phi$ P, ARLXCA, and DAMPA in exactly the same way as CNFRWD. The verbal flow description of CNFRWD (Section 6.3.1.3) applies here except for the differences indicated above.

6.3.3.4 Control Constants

Control constants ØUTPUT and TIMEND (> TIMEØ) must be specified as indicated in Table 6.2-5 and described in Section 6.2.3.2; otherwise the "run" will terminate with an error message. The function of optionally specified control constants ARLXCA, ATMPCA, BACKUP, CSGFAC, DAMPA, DTIMEH, DTIMEL, DTMPCA, NLØØP, and TIMEØ is described in Section 6.2.3.2. The user should take particular care in the selection of CSGFAC since too large of a time-step would lead to grossly inaccurate temperatures even though the solution is stable. Note also that TIMEØ may be set negative and that NLØØP is set to one if not specified.

6.3.3.5 Error and Other Messages

If control constants ØUTPUT and TIMEND are not specified, the following error message will be printed for each,

ØUTPUT	"NØ ØUTPUT		INTERVAL"	
TIMEND	"TIME	STEP	тøø	SMALL"

The reason for the TIMEND error message is that a direct check on TIMEND is not made; the resultant error message just happens to be a quirk in the coding.

If the short pseudo-compute sequence SPCS is not specified, the error message will be,

"CNEXPN REQUIRES SHORT PSEUDO-COMPUTE SEQUENCE"

If the dynamic storage allocation is not sufficient NDIM < (NND + NNA), the message will be,

" LOCATIONS AVAILABLE"

Note that the <u>number printed will be negative</u> indicating the additional storage locations required.

If the time-step used is less than the time-step allowed (DTIMEL) which may be optionally specified by the user, the message will be,

"TIME STEP TØØ SMALL"

If CSGMIN < 0, the message printed will be,

"CSGMIN ZERØ or NEGATIVE"

Checks on the control constants, the pseudo-compute sequence and the dynamic storage allocation are made in the following sequence with the run terminating if a single check is not satisfied,

ØUTPUT, pseudo-compute sequence, dynamic storage locations.

It should be particularly noted that <u>no message</u> is printed if ARLXCA is not satisfied with NLØØP iterations; ARLXCA and NLØØP are optionally specified control constants. Table 6.3-3. Basic Computational Steps for CNEXPN

- Specification of control constants. Control constants ØUTPUT and TIMEND must be specified. SPCS is required for CNEXPN. (Refer to Table 6.2-5 for values and Section 6.2.3.2 for description.)
- Sufficiency check on dynamic storage. Requirements = NND + NNA (NND = diffusion nodes and NNA = arithmetic nodes).
- Setting and/or calculation of time-step, Δt. (Refer to Section 6.2.4 for detailed procedure.) Calculated time-step = 0.95 * CSGMIN * CSGFAC.
- 4. Setting of source and diffusion node dynamic storage locations to zero.
- 5. Calling of VARIABLES 1. (Refer to Section 6.2.2.2 for description.)
- 6. Checking of BACKUP. (Refer to Section 6.2.3.2 for description.)
- 7. Calculation of diffusion-node temperatures. (Refer to Section 6.2.5.1 for description.)

Diffusion-node temperatures are calculated by using (refer to Section 6.3.3.2),

$$T_{i,n+1} = T_{i,n} e^{-\alpha_n \Delta t} + \frac{q_{i,n} + \sum_{j=1}^{p} G_{ij,n} T_{j,n}}{p} \begin{pmatrix} -\alpha_n \Delta t \\ 1 - e^{-\alpha_n} \end{pmatrix}$$

where,

$$\alpha_{n} = \frac{\sum_{j=1}^{\Sigma} G_{ij,n}}{C_{i,n}}$$

ъ

- 8. Erasure of all temperature calculations for latest time-step if allowable temperature change criterion DTMPCA is not satisfied and recalculation of temperatures with reduced time-step.
- 9. Calculation of arithmetic-node temperatures. If the number of iterations equal NL $\phi\phi$ P, the temperatures are retained without user notification
- Erasure of all temperature calculations for latest time-step if allowable temperature change criterion ATMPCA is not satisfied and recalculation of temperatures with reduced time-step.
- 11. Calling of VARIABLES 2 and checking of BACKUP. (Refer to Section 6.2.2.3 and 6.2.3.2 for description.)
- 12. Advancing of time, checking of time to print, and the printing at the output interval.
- 13. Calling of ØUTPUT CALLS.
- 14. Checking for problem end time stored in user specified control constant TIMEND.



Figure 6.3-5. Functional Flow Chart for CNEXPN

6.3.4 Subroutine: CNDUFR

6.3.4.1 General Comments

Subroutine CNDUFR is an explicit numerical solution routine that uses an unconditionally stable DuFort-Frankel method.⁹, ¹², ¹⁷ The DuFort-Frankel method replaces the present temperature of the node being operated on by the average of future and past temperatures in the forward differencing equation. In subroutine CNDUFR the present temperature of the node being operated on is replaced by a <u>time-weighted</u> average of future and past temperatures. CNDUFR requires the short pseudo-compute sequence (SPCS).

The intent of an unconditionally stable routine such as CNDUFR is the reduction of computational time by using time-steps greater than those allowed with the conditionally stable explicit methods as constrained by the stability criterion. However, less accuracy can be expected with a lengthened time-step. The time-step controlled with control constant CSGFAC represents a user decision that is difficult and must be aided by a trial and error procedure.

Examination of several CNDUFR solutions reveals a tendency to lead in time the true temperatures.

6.3.4.2 Finite Difference Approximation and Computational Algorithm

Diffusion Nodes

The DuFort-Frankel explicit finite difference expression⁹, ¹², ¹⁷ for calculating the diffusion-node temperatures may be readily determined as follows:

Using the standard explicit finite difference expression,

$$C_{i} \frac{(T_{i,n+1} - T_{i,n})}{\Delta t} = q_{i,n} + \sum_{j=1}^{p} G_{ij,n} (T_{j,n} - T_{i,n}) \qquad (6.3-8)$$
$$i = 1, 2, \dots, N$$
$$T_{j,n} = \text{constant}, N < j \le p$$

letting the present temperature, $T_{i,n}$, be replaced by the average of future temperature, $T_{i,n+1}$, and past temperature, $T_{i,n-1}$,

$$T_{i,n} = \frac{T_{i,n+1} + T_{i,n-1}}{2}$$
(6.3-9)

where, $\Delta t_i = \Delta t_{i+1}$, i = 1, 2, ..., M (equal time-steps) and defining,

 $\overline{C}_i = C_i / \Delta t$ (refer to Section 6.2.4 for discussion on Δt) (6.3-10) equation (6.3-8) can be expressed as,

$$T_{i,n+1} = \frac{\overline{C}_{i,n} T_{i,n-1} + 2 q_{i,n} + \sum_{j=1}^{p} G_{ij,n} (2 T_{j,n} - T_{i,n-1})}{\overline{C}_{i,n} + \sum_{j=1}^{p} G_{ij,n}}$$
(6.3-11)
 $i = 1, 2, ..., N$

n

In CNDUFR the present temperature, $T_{i,n}$, of equation (6.3-8) is replaced by a weighted average of future temperature, $T_{i,n+1}$, and past temperature, $T_{i,n-1}$. The weighting is based on unequal time-steps.

$$T_{i,n} = \frac{(\Delta t_{n-1} T_{i,n+1} + \Delta t_n T_{i,n-1})}{\Delta t_{n-1} + \Delta t_n}$$
(6.3-12)

where, $\Delta t_{n-1} = t_n - t_{n-1}$ (past time-step)

$$\Delta t_n = t_{n+1} - t_n \text{ (present time-step)}$$

Let

$$\tau_{n-1} = \frac{\Delta t_{n-1}}{\Delta t_{n-1} + \Delta t_n}$$
(6.3-13)

$$\tau_{n} = \frac{\Delta t_{n}}{\Delta t_{n-1} + \Delta t_{n}}$$
(6.3-14)

Equation (6.3-8) becomes,

$$T_{i,n+1} = \frac{\tau_{n} T_{i,n-1} \left(\overline{C}_{i,n} - \sum_{j=1}^{p} G_{ij,n} \right) + \sum_{j=1}^{p} G_{ij,n} T_{j,n} + q_{i,n}}{\overline{C}_{i,n} - \tau_{n-1} \left(\overline{C}_{i,n} - \sum_{j=1}^{p} G_{ij,n} \right)}$$
(6.3-15)

where, i = 1,2,...,NND (number of diffusion nodes)

 $T_{j,n} = constant$, (NND + NNA) < $j \le p$ (NNA is the number of arithmetic nodes and p is the total number of nodes)

$$G_{ij,n} = a_{ij,n} + \sigma b_{ij,n} (T_{j,n}^2 + T_{i,n}^2)(T_{j,n} + T_{i,n})$$

 $C_{i},q_{i},a_{ij},b_{ij} = may be optionally specified (refer to Tables 6.2-1 - 6.2-4)$

In CNDUFR, equation (6.3-15) is applied to the diffusion nodes with the computational procedure being a "block" change in temperature from one time-step to another.

Arithmetic Nodes (if any)

Arithmetic-node temperatures are calculated in the same manner as in CNFRWD (Section 6.3.1.2) or refer to Section 5.2.3 for the finite difference algorithm.

6.3.4.3 Comments on the Computational Procedure

The important steps of the computation procedure used in subroutine CNDUFR are indicated in Table 6.3-4 and a functional flow chart is shown in Figure 6.3-6. A computer listing of CNDUFR is found in Appendix A but some general computational details are given in Section 6.2.5.1. The computational procedure for CNDUFR follows the CNEXPN computational pattern, but with the temperatures of the diffusion nodes calculated by the DuFort-Frankel method of the exponential prediction method. Another significant difference is that CNDUFR must provide for two sets of past temperatures which are required for DuFort-Frankel algorithm; two timesteps for consecutive time-step calculations are also required. Otherwise, checks and control constant use are identical to CNEXPN. Thus, the verbal flow description of Section 6.3.1.3 applies directly except for the differences indicated above.

6.3.4.4 Control Constants

Control constants ØUTPUT and TIMEND (> TIMEØ) must be specified as indicated in Table 6.2-5 and described in Section 6.2.3.2; otherwise the "run" will terminate with an error message. The function of optionally specified control constants ARLXCA, ATMPCA, BACKUP, CSGFAC, DAMPA, DTIMEH, DTIMEL, DTMPCA, NLØØP, and TIMEØ is described in Section 6.2.3.2. The user should take particular care in the selection of CSGFAC since too large of a time-step would lead to grossly inaccurate temperatures even though the solution is stable. Note also that TIMEØ may be set negative and that NLØØP is set to one if not specified. 6.3.4.5 Error and Other Messages

If control constants ØUTPUT and TIMEND are not specified, the following error message will be printed for each,

ØUTPUT"NØ ØUTPUT INTERVAL"TIMEND"TIME STEP TØØ SMALL"

The reason for the TIMEND error message is that a direct check on TIMEND is not made; the resultant error message just happens to be a quirk in the coding.

If the short pseudo-compute sequence SPCS is not specified, the error message will be,

"CNDUFR REQUIRES SHORT PSEUDO-COMPUTE SEQUENCE"

If the dynamic storage allocation is not sufficient (NDIM < (2*NND + NNA)), the message will be,

" LOCATIONS AVAILABLE"

Note that the <u>number printed will be negative</u> indicating the additional storage locations required.

If the time-step used is less than the time-step allowed (DTIMEL), which may be optionally specified by the user, the message will be,

"TIME STEP TØØ SMALL"

If CSGMIN < 0, the message printed will be,

"CSGMIN ZERØ or NEGATIVE"

Checks on the control constants, the pseudo-compute sequence and the dynamic storage allocation are made in the following sequence with the run terminating if a single check is not satisfied,

ØUTPUT, pseudo-compute sequence, dynamic storage locations

It should be particularly noted that <u>no message</u> is printed if ARLXCA is not satisfied with NLØØP iterations; ARLXCA and NLØØP are optionally specified control constants.

Table 6.3-4. Basic Computational Steps for CNDUFR

- Setting of control constants to nominal values. Control constants
 ØUTPUT and TIMEND must be specified. SPCS is required for CNDUFR. (Refer to Table 6.2-5 for values and Section 6.2.3.2 for description.)
- 2. Sufficiency check on dynamic storage. Requirements = 2*NND + NNA (NND = diffusion nodes and NNA = arithmetic nodes).
- 3. Setting and/or calculation of time-step, Δt. (Refer to Section 6.2.4 for detailed procedure.) Calculated time-step = 0.95*CSGMIN*CSGFAC.
- 4. Setting of source and diffusion node dynamic storage locations to zero.
- 5. Calling of VARIABLES 1. (Refer to Section 6.2.2.2 for description.)
- 6. Checking of BACKUP. (Refer to Section 6.2.3.2 for description.)
- 7. Calculation of diffusion-node temperatures. (Refer to Section 6.2.5.1 for description.)

Diffusion-node temperatures are calculated by using (refer to Section 6.3.4.2),

$$T_{i,n+1} = \frac{\tau_{n} T_{i,n-1} (\overline{C}_{i,n} - \sum_{j=1}^{p} G_{ij,n}) + \sum_{j=1}^{p} G_{ij,n} T_{j,n} + q_{i,n}}{\overline{C}_{i,n} - \tau_{n-1} (\overline{C}_{i,n} - \sum_{j=1}^{p} G_{ij,n})}$$

where,

$$\tau_{n-1} = \frac{\Delta t_{n-1}}{\Delta t_{n-1} + \Delta t_{n}}$$

. .

$$\tau_{n} = \frac{\Delta t_{n}}{\Delta t_{n-1} + \Delta t_{n}}$$

- 8. Erasure of all temperature calculations for latest time-step if allowable temperature change criterion DTMPCA is not satisfied and temperature recalculation with reduced time-step.
- 9. Calculation of arithmetic-node temperatures; if the number of iterations equal NLØØP, the temperatures are retained without user notification (refer to Section 6.2.5.1 for details).
- 10. Erasure of arithmetic-node temperatures for latest time-step if allowable temperature change criterion ATMPCA is not satisfied and temperature recalculation with reduced time-step.
- 11. Calling of VARIABLES 2 and checking of BACKUP. (Refer to Section 6.2.2.3 and 6.2.3.2 for description.)
- 12. Advancing of time, checking of time to print, and the printing at the output interval.
- 13. Calling of ØUTPUT CALLS.
- 14. Checking for problem end time stored in user specified control constant TIMEND.



Figure 6.3-6. Functional Flow Chart for CNDUFR

6.3.5 Subroutine: CNQUIK

6.3.5.1 General Comments

Subroutine CNQUIK is a numerical solution routine that uses an algorithm composed of half DuFort-Frakel method⁹, ¹², ¹⁷ and half exponential prediction method.¹, ¹⁷ CNQUIK requires the short pseudocompute sequence (SPCS); characteristics of subroutines CNDUFR and CNEXPN, as described in Section 6.3.3 and 6.3.4, also apply to CNQUIK.

Why CNQUIK? Examination of CNDUFR and CNEXPN solutions reveals that CNDUFR has a tendency to yield temperatures which lead the true temperatures, whereas CNEXPN has a tendency to lag the true temperatures. Thus, it was theorized that a combination of CNDUFR and CNEXPN should yield a more accurate solution than either one. Preliminary results indicate that CNQUIK is more accurate than either CNDUFR or CNEXPN with approximately the same solution time. It can also be theorized that a more accurate combination of the DuFort-Frankel and exponential prediction is probably possible than the half and half used in CNQUIK. However, a detailed study will be required before a realistic evaluation of CNQUIK can be made.

6.3.5.2 Finite Difference Approximation and Computational Algorithm

Diffusion Nodes

Subroutine CNQUIK uses a numerical solution algorithm composed of half DuFort-Frankel and half exponential prediction. That is the temperature of the diffusion nodes is calculated by using,

$$T_{i,n+1} = \left(T_{CNDUFR} + T_{CNEXPN}\right)/2.0 \qquad (6.3-16)$$

$$T_{CNDUFR} = \frac{\tau_n T_{i,n-1} \left(\overline{C}_{i,n} - \sum_{j=1}^{p} G_{ij,n}\right) + \sum_{j=1}^{p} G_{ij,n} T_{j,n} + q_{i,n}}{\overline{C}} \qquad (6.3-16)$$

 $C_{i,n} (1 - \tau_{n-1}) + \sum_{j=1}^{n} G_{ij,n}$

(equation 6.3-15 of Section 6.3.4.2)

$$T_{\text{CNEXPN}} = T_{i,n} e^{-\alpha_n \Delta t} + \frac{q_{i,n} + \sum_{j=1}^{p} G_{ij,n} T_{j,n}}{p} \left(1 - e^{-\alpha_n \Delta t}\right)$$

(equation 6.3-7 of Section 6.3.3.2)

n = nth time-step

i = 1,2,...,NND (number of diffusion nodes)

 $T_{j,n} = \text{constant}, (NND + NNA) < j < p$ (NNA is the number of arithmetic nodes and p is the total number of nodes)

$$\alpha_{n} = \frac{\frac{\sum_{j=1}^{p} G_{ij,n}}{\sum_{i,n}}}{C_{i,n}}$$

$$G_{ij,n} = a_{ij,n} + \sigma b_{ij,n} (T_{i,n}^2 + T_{j,n}^2) (T_{i,n} + T_{j,n})$$

$$\tau_n = \frac{\Delta t_n}{\Delta t_{n-1} + \Delta t_n}; \quad \tau_{n-1} = \frac{\Delta t_{n-1}}{\Delta t_{n-1} + \Delta t_n}$$

C_i, q_i, a_{ij}, b_{ij} = optionally specified (refer to Tables 6.2-1 - 6.2-4)

 $\overline{C}_i = C_i / \Delta t$ (refer to Section 6.2.4 for discussion of Δt)

Arithmetic Nodes (if any)

Temperatures of arithmetic nodes are calculated in the same manner as in CNFRWD (Section 6.3.1.2) or refer to Section 5.2.3 for the finite difference algorithm.

6.3.5.3 Comments on the Computational Procedure

The important steps of the computational procedure used in subroutine CNQUIK are indicated in Table 6.3-5 and a functional flow chart is shown in Figure 6.3-7. A computer listing of CNQUIK is found in Appendix A. General computational details are given in Section 6.2. The computational procedure for CNQUIK follows CNEXPN or CNDUFR with the diffusion-node temperatures calculated with the half DuFort-Frankel and half exponential prediction algorithm being the only difference. Arithmetic-node temperatures are calculated in the same manner as the other SINDA explicit routines. Note that the time-step is calculated as CSGMIN*CSGFAC and checks are the same as CNEXPN or CNDUFR. Thus, the verbal flow description of Section 6.3.1.3 applies directly except for the differences indicated above.

6.3.5.4 Control Constants

Control constants ØUTPUT and TIMEND (> TIMEØ) must be specified as indicated in Table 6.2-5 and described in Section 6.2.3.2; otherwise the "run" will terminate with an error message. The function of optionally specified control constants ARLXCA, ATMPCA, BACKUP, CSGFAC, DAMPA, DTIMEH, DTIMEL, DTMPCA, NLØØP, and TIMEØ is described in Section 6.2.3.2. Again, caution must be exercised in the selection of CSGFAC since too large of a time-step would lead to grossly inaccurate temperatures even though the solution is stable. Note also that TIMEØ may be set negative and that NLØØP is set to one if not specified.

6.3.5.5 Error and Other Messages

If control constants ØUTPUT and TIMEND are not specified, the following error message will be printed for each,

ØUTPUT	"NØ ØUTPUT		INTERVAL"	
TIMEND	"TIME	STEP	тøø	SMALL"

The reason for the TIMEND error message is that a direct check on TIMEND is not made; the resultant error message just happens to be a quirk in the coding.

If the short pseudo-compute SPCS is not specified, the error message will be,

"CNQUIK REQUIRES SHORT PSEUDO-COMPUTE SEQUENCE"

If the dynamic storage allocation is not sufficient (NDIM < (2*NND + NNA), the message will be,

" LOCATIONS AVAILABLE"

Note that the <u>number printed will be negative</u> indicating the additional storage locations required.

If the time-step used is less than the time-step allowed (DTIMEL), which may be optionally specified by the user, the message will be,

"TIME STEP TØØ SMALL"

If CSGMIN ≤ 0 , the message printed will be,

"CSGMIN ZERØ or NEGATIVE"

Checks on the control constants, the pseudo-compute sequence and the dynamic storage allocation are made in the following sequence with the run terminating if a single check is not satisfied,

ØUTPUT, pseudo-compute sequence, dynamic storage locations.

It should be particularly noted that <u>no message</u> is printed if ARLXCA is not satisfied with NLØØP iterations; ARLXCA and NLØØP are optionally specified control constants.

Table 6.3-5. Basic Computational Steps for CNQUIK

- Specification of control constants. Control constants ØUTPUT and TIMEND must be specified. SPCS is required for CNEXPN. (Refer to Table 6.2-5 for values and Section 6.2.3.2 for description.)
- 2. Sufficiency check on dynamic storage. Requirements = 2(NND) + NNA (NND = diffusion nodes and NNA = arithmetic nodes).
- 3. Setting and/or calculation of time-step, Δt . (Refer to Section 6.2.4 for detailed procedure.) Calculated time-step = 0.95*CSGMIN*CSGFAC.
- 4. Setting of source and diffusion node dynamic storage locations to zero.
- 5. Calling of VARIABLES 1. (Refer to Section 6.2.2.2 for description.)
- 6. Checking of BACKUP. (Refer to Section 6.2.3.2 for description.)
- 7. Calculation of diffusion-node temperatures. (Refer to Section 6.2.5.1 for description.)

Diffusion-node temperatures are calculated by using (refer to Section 6.3.5.2),

 $T_{i,n+1} = (T_{CNDUFR} + T_{CNEXPN})/2.0$

(Refer to equation 6.3-17, Section 6.3.5.2.)

- 8. Erasure of all temperature calculations for latest time-step if allowable temperature change criterion DTMPCA is not satisfied and temperature recalculation with reduced time-step.
- 9. Calculation of arithmetic-node temperatures; if the number of iterations equal NLØØP, the temperatures are retained without user notification. (Refer to Section 6.2.5.1 for details.)
- Erasure of all temperature calculations for latest time-step if allowable temperature change criterion ATMPCA is not satisfied and temperature recalculation with reduced time-step.
- 11. Calling of VARIABLES 2 and checking of BACKUP. (Refer to Section 6.2.2.3 and 6.2.3.2 for description.)
- 12. Advancing of time, checking of time to print, and the printing at the output interval.
- 13. Calling of ØUTPUT CALLS.
- 14. Checking for problem end time stored in user specified control constant TIMEND.



Figure 6.3-7. Functional Flow Chart for CNQUIK

6.4 Transient Implicit Solution Routines

SINDA implicit solution routines number three; these routines are identified as follows:

CNBACK Implicit backward difference method. Requires long pseudo-compute sequence (LPCS). CNFWBK Implicit forward-backward differencing, using Crank-Nicolson method. Requires long pseudo-compute sequence (LPCS). CNVARB Combination of CNBACK and CNFWBK. Requires long pseudo-compute sequence (LPCS).

Implicit methods generally tend to be more accurate than explicit methods and are unconditionally stable as are some explicit methods. With implicit methods the time-step is specified in contrast to the calculated time-steps of explicit methods with their stability criterion. An important consideration in the use of implicit methods is that the time-step DTIMEI should be specified in conjunction with control constant NLØØP which represents the maximum number of computational iterations during each timestep. Since each iterative calculation is essentially equivalent to a time-step calculation for an explicit method, the combination of DTIMEI and NLØØP for a given time period should be set less than the total number of time-steps used by the explicit method for the same time period, if computational time is to be reduced; this of course assumes that during each time-step the maximum number of iterations is required. If the NLØØP iterations are required during a time-step, the temperature accuracy is affected but the magnitude would depend upon the value used for the maximum allowable relaxation temperature change criteria, ARLXCA and DRLXCA. It should be noted if NLØØP iterations are required during a time-step, the message "RELAXATION CRITERIA NOT MET" is printed.

A detailed description of each implicit routine, as presented on the pages to follow, relies on the general description of Section 6.2. A brief description of these routines is summarized first.

<u>CNBACK</u> uses the standard backward differencing algorithm and requires the long pseudo-compute sequence (LPCS). The time-step must be

specified via control constant DTIMEI and used in conjunction with the control constant NL $\phi\phi$ P. CNBACK uses the acceleration of convergence feature.

<u>CNFWBK</u> uses the Crank-Nicolson algorithm which is composed of half forward differencing and half backward differencing. CNFWBK solutions tend to be more accurate than CNBACK solutions with approximately 25% less iterations; however CNFWBK solutions have "blown" on occasions.

<u>CNVARB</u> uses a combination of forward differencing and backward differencing. Unlike CNFWBK which is half and half, CNVARB uses a variable beta factor which ranges from 0 to 1. Thus CNVARB uses a method that is somewhere between forward differencing and backward differencing.

6.4.1 Subroutine: CNEACK

6.4.1.1 General Comments

Subroutine CNBACK is an implicit routine that uses the standard backward difference expression and requires the long pseudo-compute (LPCS). Time-step must be specified via DTIMEI otherwise the "run" will terminate with an error message printout. The time-step value is arbitrary but the user should consider DTIMEI in conjunction with the control constant $NL\emptyset\emptysetP$ which represents the maximum number of computational iterations during each time-step (refer to Section 6.2.3.2 for description).

Implicit methods tend to be more accurate than explicit methods and are unconditionally stable, but implicit solutions often oscillate at start up or boundary step changes when heat transfer by radiation is present. CNBACK internally controls sudden radiation heat transfer changes by an averaging technique which is termed "radiation damping" (refer to Section 6.2.6 for details). This automatic damping has been very effective in many solutions that have been examined and lessens the need for the use of DAMPD and DAMPA.

6.4.1.2 Finite Difference Approximation and Computational Algorithm

The numerical solution algorithm used in subroutine CNBACK is the standard backward-difference expression¹²,¹³, ¹⁷ which may be expressed as:

$$C_{i} \frac{(T_{i,n+1} - T_{i,n})}{\Delta t} = q_{i,n} + \sum_{j=1}^{p} a_{ij} (T_{j,n+1} - T_{i,n+1})$$

$$+ \sum_{j=1}^{p} \sigma b_{ij} (T_{j,n+1}^{4} - T_{i,n+1}^{4})$$
(equation 5.2-5 of Section 5.2.2)
$$i = 1, 2, \dots, N$$

$$T_{j,n+1} = \text{constant}, N < j \le p$$

$$T_{i,n} \equiv T_{i} (n\Delta t)$$

The computational procedure for the backward difference formulation must necessarily be re-iterative because of the need to solve a set of simultaneous nonlinear equations.
Diffusion Nodes

Diffusion node temperatures are solved by "successive point" iteration but differs from the arithmetic-node temperature calculation because of the capacitance term and the use of "radiation damping" (refer to Section 6.2.5.2).

$$T_{i,k+1} = DD^{*} T_{i,k}$$

$$+ DN^{*} \frac{\overline{C}_{i,n} T_{i,n} + q_{i,n} + \frac{i}{j=1}C_{ij,n} T_{j,k+1} + \frac{p}{2}C_{ij,n} T_{j,k} - (q_{i})_{ave}}{\overline{C}_{i,n} + \frac{p}{j=1}a_{ij,n}} (6.4-1)$$
where, $i = 1, 2, \dots, NND$

$$n = nth time-step$$

$$k = kth iteration$$

$$DN = DAMPD (diffusion-node damping factor)$$

$$DD = 1.0 - DN$$

$$G_{ij,n} = a_{ij,n} + \sigma b_{ij,n} T_{j,k}^{3} (l = k \text{ if } j \ge 1 \text{ and } l = k+1 \text{ if } j < 1)$$

$$\overline{C}_{i,n}^{*} q_{i,n}^{*} a_{ij,n}^{*} b_{ij}^{*} = optionally specified (refer to Tables 6.2-1 - 6.2-4)$$

$$\overline{C}_{i,n} = C_{i,n}/\Delta t (\Delta t = time step, refer to Section 6.2.4)$$

$$(q_{i})_{ave} = \sum_{j=1}^{p} \sigma b_{ij,n} [(T_{i,k}^{4}) + (T_{i,k}^{4})_{2}]/2.0, \text{ average heat loss from the ith node (refer to Section 6.2.6 on radiation damping for details)$$

Details on the computational procedure for implicit routines are presented in Sections 5.2.2 and 6.2.5.2.

Arithmetic Nodes

Arithmetic-node temperatures are calculated identically the same in all the SINDA numerical solution routines. Thus, refer to either Section 6.3.1.2 or Section 6.2.5.2 for the finite difference algorithm.

6.4.1.3 Comments on the Computational Procedure

The important steps of the computational procedure used in subroutine CNBACK are indicated in Table 6.4-1. For a detailed step-by-step computational description, the user must examine the computer listing for

CNBACK in Appendix B, but some general computational details are given in Section 6.2.5.2. A functional flow chart of CNBACK is shown in Figure 6.4-1.

Both diffusion-node temperatures and arithmetic-node temperatures are calculated by "successive point" iteration. Each third iteration, diffusion-node temperatures which are decreasing over two time-steps are extrapolated in an attempt to accelerate convergence (refer to Section 6.2.7). Temperature convergence is examined during each time-step by checking DRLXCC and ARLXCC against the user control constants DRLXCA (for diffusion nodes) and ARLXCA (for arithmetic nodes), respectively. If temperatures have not converged with NLØØP iterations, the message "RELAXATION CRITERIA NOT MET" is printed. Control constant NLØØP is used to specify the maximum number of iterations allowed during each time-step.

VARIABLES 1 and VARIABLES 2 are performed only once for each timestep. Since this subroutine is implicit, the user must specify the timestep to be used through the control constant DTIMEI in addition to control constant TIMEND and ØUTPUT. The look ahead feature for the time-step calculation used in CNFRWD is also employed in CNBACK as are checks for maximum allowable time-step DTIMEH, maximum allowable temperature change between time-steps, DTMPCA (diffusion nodes) and ATMPCA (arithmetic nodes). The minimum time-step DTIMEL is not checked however. Damping of solutions can be achieved through the use of the control constants DAMPD and DAMPA but "radiation damping" (refer to Section 6.2.6) used by CNBACK lessens the need for the damping factors DAMPD and DAMPA.

6.4.1.4 Control Constants

Control constants ARLXCA, DRLXCA, DTIMEI, NLØØP, ØUTPUT, and TIMEND must be specified as indicated in Table 6.2-5 and as described in Section 6.2.3.2; otherwise "run" will terminate with an appropriate error message. The function of optionally specified control constants ATMPCA, BACKUP, DAMPA, DAMPD, DTIMEH, DTMPCA, and TIMEØ is described in Section 6.2.3.2.

Specification of time-step DTIMEI should be done in conjunction with control constant $NL \phi \phi P$ which represents the maximum number of computational iterations during each time-step. Since each iterative calculation is essentially equivalent to a time-step calculation for an explicit

method, the combination of DTIMEI and NLØØP for a given time period should be less than the total number of time-steps by the explicit method for the same period. Note also that TIMEØ may be set negative. Specification of ARLXCA and DRLXCA depends upon the problem but a typical value is 0.1.

6.4.1.5 Error and Other Messages

If control constants ARLXCA, DRLXCA, DTIMEI, NLØØP, ØUTPUT and TIMEND are not specified, the following error message will be printed for each,

ARLXCA	"NØ ARLXCA"
DRLXCA	"NØ DRLXCA"
DTIMEI	"NØ DTIMEI"
nløøp	"NØ NLØØP"
ØUTPUT	"NØ ØUTPUT INTERVAL"
TIMEND	"TRANSIENT TIME NØT SPECIFIED"

If the long pseudo-compute sequence LPCS is not specified, the error message will be,

"CNBACK REQUIRES LONG PSEUDO-COMPUTE SEQUENCE"

If the dynamic storage allocation is not sufficient (NDIM < (3*NND + NNA + NNB)), the message will be,

"_____ LOCATIONS AVAILABLE"

Note that the <u>number printed will be negative</u> indicating the additional storage locations required.

If CSGMIN < 0, the following message will be printed,

"CSGMIN ZERØ or NEGATIVE"

If either ARLXCA or DRLXCA is not satisfied with NLØØP iterations, the following message will be printed,

"RELAXATION CRITERIA NOT MET"

Checks on the control constants, the pseudo-compute sequence and the dynamic storage allocation are made in the following sequence with the "run" terminating if a single check is not satisfied,

NLØØP, TIMEND, ØUTPUT, ARLXCA, DTIMEI, DRLXCA, LPCS and dynamic storage allocation.

Table 6.4-1. Basic Computational Steps for CNBACK

- 1. Specification of control constants. Control constants ARLXCA (if NNA > 0), DRLXCA (if NND > 0), DTIMEI, NLØØP, ØUTPUT and TIMEND (TIMEND > TIMEØ) must be specified. LPCS is required. (Refer to Table 6.2-5 for values and Section 6.2.3.2 for description.)
- 2. Sufficiency check on dynamic storage. Requirements = 3*NND + NNA + NNB (NND = diffusion nodes, NNA = arithmetic nodes and NNB = boundary nodes).
- 3. Setting and/or calculation of time-step, ∆t. (Refer to Section 6.2.4 for detailed procedure.) Time-step = DTIMEI.
- 4. Setting of iterative DØ loop, 1 to NLØØP.
- 5. Setting of source locations to zero.
- 6. Calling of Variables 1. (Refer to Section 6.2.2.2 for description.)
- 7. Checking of BACKUP. (Refer to Section 6.2.3.2 for description.)
- 8. Diffusion-node temperature calculations, first iteration only.

Evaluation of q_i , C_i and G_k .

Damping of radiation heat transfer. (Refer to Section 6.2.5.2.)

Calculation of diffusion-node temperature.

The computational algorithm depends upon the presence of radiation heat transfer, but the method of solution is the standard implicit algorithm (refer to Section 6.2.5.2).

- 9. Conversion of T_{i.k+1} to degrees Rankine.
- 10. Diffusion-node temperature calculations, successive iterations after first. Repeating of step 8, except that q_i, C_i and G_k are not updated. Calculation of DRLXCC.
- 11. Acceleration of convergence every third iteration if linear extrapolation is met (refer to Section 6.2.7).
- 12. Conversion of T_{i.k+1} to degrees Fahrenheit.
- 13. Calculation of arithmetic-node temperatures, second and succeeding iterations; arithmetic-node temperatures are not calculated on the first iteration (refer to Section 6.2.5.2 for details).
- 14. Conversion of temperatures to degrees Rankine.
- 15. Checking of ARLXCA and DRLXCA for convergence and ØPEITR for output. If both ARLXCA and DRLXCA are satisfied, iterations during a time-step ceases, otherwise NLØØP iterations are performed.
- 16. Checking of ATMPCA and DTMPCA. If either one is not satisfied time-step is shortened, previous temperatures erased, and temperatures recalculated for shortened time-steps (refer to Section 6.2.5.2).
- 17. Conversion of temperatures back to degrees Fahrenheit.
- 18. Calling of VARIABLES 2 and checking of BACKUP (refer to Section 6.2.2.3 and 6.2.3.2).
- 19. Advancing of time, checking of time to print, and the printing of the output interval.
- 20. Calling of ØUTPUT CALLS.

۰.

21. Checking for problem end time stored in control constant TIMEND.



6.4.2 Subroutine: CNFWBK

6.4.2.1 General Comments

Subroutine CNFWBK is an implicit numerical solution routine that uses the Crank-Nocolson algorithm.^{7, 8, 12} The long pseudo-compute sequence (LPCS) is required and the nodal temperatures (both diffusion and arithmetic) are solved by "successive point" iterations. The iteration looping, convergence criteria and other control constant checks are identical to CNBACK. Time-step must be specified via control constant DTIMEI. Diffusion and arithmetic temperature calculations may be damped through the use of DAMPD and DAMPA, respectively. Thermal radiation heat transfer is uniquely "handled" via a so-called "radiation damping" (refer to Section 6.2.6), and acceleration of convergence (refer to Section 6.2.7) is also available in CNFWBK.

CNFWBK solutions which are based on a half forward differencing and a half backward differencing method tend to be more accurate than CNBACK solutions with approximately the same solution time.

6.4.2.2 Finite Difference Approximation and Computational Algorithm

The numerical solution algorithm used in subroutine CNFWBK is the Crank-Nicolson method, which is half forward differencing and half backward differencing, and may be expressed as:

$$T_{backward} = q_{j,n} + \sum_{j=1}^{r} a_{j,n} (T_{j,n+1} - T_{i,n+1}) + \sum_{j=1}^{r} \sigma b_{j,n} (T_{j,n+1}^{4} - T_{i,n+1}^{4}) (6.4-4)$$

$$n = nth time-step$$

$$i = 1, 2, \dots, N$$

$$p = total number of nodes$$

$$T_{j,n}; T_{j,n+1} = constant, N < j \le p$$

The computational procedure for the forward-backward difference formulation must be re-iterative because of the need to solve a set of simultaneous nonlinear equations. The pattern of computation is very similar to that used in CNBACK.

Diffusion-Nodes

Diffusion node temperatures are solved by "successive point" iteration but the algorithm differs from the algorithm used in CNBACK because of the additional terms arising from the forward difference portion of the expression.

$$T_{i,k+1} = DD* T_{i,k} + DN* [Q_{sum} - (q_i)_{ave}]/G_{sum}$$
 (6.4-5)

where,
$$Q_{sum} = Q_{i} + \sum_{j=1}^{i} a_{ij,n} T_{j,k+1} + \sum_{j=i+1}^{p} a_{ij,n} T_{j,k} + \sum_{j=i+1}^{i} \sigma b_{ij,n} T_{j,k+1}^{4} + \sum_{j=i+1}^{p} \sigma b_{ij,n} T_{j,k}^{4}$$
 (6.4-6)

$$Q_{i} = 2 q_{i,n} + 2 \overline{C}_{i,n} T_{i,n} + \sum_{j=1}^{p} a_{ij,n} (T_{j,n} - T_{i,n})$$
 (6.4-7)

$$G_{sum} = 2 \overline{C}_{i,n} + \sum_{j=1}^{p} a_{ij,n}$$
(6.4-8)

n = nth time-step; k = kth iteration $C_i, q_i, a_{ij}, b_{ij} = optionally specified (refer to Tables 6.2-1 - 6.2-4)$ DN = DAMPD (diffusion-node damping factor) DD = 1.0 - DN $\overline{C}_{i,n} = C_{i,n} / \Delta t$ (Δt = time-step) $(q_i)_{ave} = \sum \sigma b_{ij,n} [(T_{i,k}^4) + (T_{i,k}^4)_2]/2.0$, average heat loss from ith node (refer to Section 6.2.6 on radiation damping for details)

(Note that the known quantities at time-step, n, are indicated by Q_i , equation 6.4-7.)

Arithmetic Nodes

Arithmetic-node temperatures are calculated identically the same in all the SINDA numerical solution routines. Thus, refer to Section 6.3.1.2 or Section 6.2.5.2 for the finite difference algorithm.

6.4.2.3 <u>Comments on the Computational Procedure</u>

The important steps of the computational procedure used in subroutine CNFWBK are indicated in Table 6.4-2. For a detailed step-by-step computational description, the user must examine the computer listing for CNFWBK in Appendix B, but some general computational details are given in Section 6.2.5.2. A functional flow chart of CNFWBK is shown in Figure 6.4-2.

The computational flow pattern for CNFWBK is identical to CNBACK with the only difference between the routines being the diffusion-node temperature finite-difference algorithm. On the first iteration only the source locations zeroed out and the present temperatures stored, VARIABLES 1 is called and variable C_i , impressed source q_i and variable coefficients G_i (diffusion-diffusion and diffusion-arithmetic) evaluated. All quantities which are evaluated at time, t_n , are summed in accordance with equations (6.4-6) and (6.4-8). CSGMIN is evaluated and the diffusion-node temperatures calculated; note the arithmetic-node temperatures are not calculated on the first iteration.

On the second and succeeding iterations the quantities C_i , q_i and G_k (diffusion-diffusion and diffusion-arithmetic) are not updated. Diffusion-node temperatures are calculated and DRLXCC determined. Every third iteration, if a diffusion-node temperature is converging, a linear extrapolation to accelerate convergence is performed (refer to Section 6.2.7). If arithmetic nodes are encountered, the appropriate q_i and G_k (for arithmetic nodes) are evaluated once per time-step. Arithmetic-node temperatures are calculated and ARLXCC determined.

Control constants DRLXCC and ARLXCC are checked against DRLXCA and ARLXCA, respectively each time-step; if both criteria are satisfied the iterations cease, otherwise the iterations continue NLØØP times and the message "RELAXATION CRITERIA NOT MET" is printed.

Diffusion-node and arithmetic-node temperature changes between time-steps are calculated and stored in DTMPCC and ATMPCC, respectively.

If DTMPCC > DTMPCA or if ATMPCC > ATMPCA, the just completed calculations are erased and the time-step shortened (refer to Section 6.2.5.2).

6.4.2.4 Control Constants

The control constants for CNFWBK are used in exactly the same way as used in CNBACK. Control constants ARLXCA, DRLXCA, DTIMEI, NLØØP, ØUTPUT, and TIMEND must be specified as indicated in Table 6.2-5 and as described in Section 6.2.3.2; otherwise "run" will terminate with an appropriate error message. The function of optionally specified control constants ATMPCA, BACKUP, DAMPA, DAMPD, DTIMEH, DTMPCA, and TIMEØ is described in Section 6.2.3.2.

Specification of time-step DTIMEI should be done in conjunction with control constant NLØØP which represents the maximum number of computational iterations during each time-step. Since each iterative calculation is essentially equivalent to a time-step calculation for an explicit method, the combination of DTIMEI and NLØØP for a given time period should be less than the total number of time-steps by the explicit method for the same time period. Note also that TIMEØ may be set negative. Specification of ARLXCA and DRLXCA depends upon the problem but a typical value is 0.1.

6.4.2.5 Error and Other Messages

If control constants ARLXCA, DRLXCA, DTIMEI, NLØØP, ØUTPUT and TIMEND are not specified the following error message will be printed for each,

ARLXCA	"NØ ARLXCA"
DRLXCA	"NØ DRLXCA"
DTIMEI	"NØ DTIMEI"
nløøp	"nø nløøp"
ØUTPUT	"NØ ØUTPUT INTERVAL"
TIMEND	"TRANSIENT TIME NØT SPECIFIED"

If the long pseudo-compute sequence LPCS is not specified, the error message will be,

"CNFWBK REQUIRES LØNG PSEUDØ-CØMPUTE SEQUENCE"

If the dynamic storage allocation is not sufficient (NDIM < (3* NND + NNA + NNB)), the message will be,

LOCATIONS AVAILABLE"

Note that the <u>number presented will be negative</u> indicating the additional storage locations required.

If CSGMIN ≤ 0 , the following message will be printed,

"CSGMIN ZERØ OR NEGATIVE"

If either ARLXCA or DRLXCA is not satisfied with $NL \emptyset \emptyset P$ iterations, the following message will be printed,

"RELAXATION CRITERIA NOT MET"

Checks on the control constants, the pseudo-compute sequence and the dynamic storage allocation are made in the following sequence with the "run" terminating if a single check is not satisfied,

NLOOP, TIMEND, OUTPUT, ARLXCA, DTIMEI, DRLXCA, LPCS and dynamic storage allocation.

. Table 6.4-2. Basic Computational Steps for CNFWBK

- Specification of control constants. Control constants ARLXCA (if NNA > 0), DRLXCA (if NND > 0), DTIMEI, NLØØP, ØUTPUT and TIMEND (TIMEND > TIMEO) must be specified. LPCS is required. (Refer to Table 6.2-5 for values and Section 6.2.3.2 for description.
- 2. Sufficiency check on dynamic storage. Requirements = 3*NND + NNA + NNB (NND = diffusion nodes, NNA = arithmetic nodes and NNB = boundary nodes)
- Setting and/or calculation of time-step, Δt. (Refer to Section 6.2.4 for detailed procedure.) Time-step = DTIMEI.
- 4. Setting of iterative DØ loop, 1 to NLØØP.
- 5. Setting of source locations to zero.
- 6. Calling of Variables 1. (Refer to Section 6.2.2.2 for description.)
- 7. Checking of BACKUP. (Refer to Section 6.2.3.2 for description.)
- 8. Diffusion-node temperature calculations, first iteration only. Evaluation of q_i , C_i and G_k . Damping of radiation heat transfer. (Refer to Section 6.2.5.2.) Calculation of diffusion-node temperature. The computational algorithm depends upon the presence of radiation heat transfer, but the method of solution is the Crank-Nicolson algorithm (half forward and half backward, refer to Section 6.2.5.2).
- 9. Conversion of T_{i,k+1} to °R (Rankine).
- Diffusion-node temperature calculation, successive iterations after first. Repeating of step 8 except that q_i,C_i and G_k are not updated. Calculation of DRLXCC.
- 11. Acceleration of convergence every third iteration if linear extrapolation is met (refer to Section 6.2.7).
- 12. Conversion of $T_{i,k+1}$ to degrees Fahrenheit.
- 13. Calculation of arithmetic-node temperatures, second and succeeding iterations; arithmetic-node temperatures are not calculated on the first iteration (refer to Section 6.2.5.2 for details).
- 14. Conversion of temperatures to degrees Rankine.
- 15. Checking of ARLXCA and DRLXCA for convergence and ØPEITR for output. If both ARLXCA and DRLXCA are satisfied, iterations during a timestep cease, otherwise NLØØP iterations are performed.
- 16. Checking of ATMPCA and DTMPCA. If either one is not satisfied timestep is shortened, previous temperatures erased, and temperatures recalculated for shortened time-steps (refer to Section 6.2.5.2).
- 17. Conversion of temperatures back to degrees Fahrenheit.
- 18. Calling of VARIABLES 2 and checking of BACKUP (refer to Section 6.2.2.3 and 6.2.3.2).
- 19. Advancing of time, checking of time to print, and the printing of the output interval.
- 20. Calling of ØUTPUT CALLS.
- 21. Checking for problem end-time stored in user specified control constant TIMEND.



6.4-2. Functional Flow Chart for CNFWBK

6.4.3 Subroutine: CNVARB

6.4.3.1 General Comments

Subroutine CNVARB uses an implicit finite difference algorithm that is a composition of forward-differencing and backward-differencing. The proportion of forward to backward to be used is calculated internally by using a weighting factor, β , that is dependent upon the ratio of the explicit stability criterion as stored in the control constant CSGMIN divided by the computational time-step stored in DTIMEU. The weighting factor can vary each time-step but is constrained to range, $0 \le \beta \le 1/2$ (refer to Section 6.2.5.2 or Section 6.4.3.2). A β of one-half yields the Crank-Nicolson half-forward and half-backward expression, whereas a β of zero yields the standard backward-difference expression.

Except for the weighting factor, β , the computational procedure and the use of the various control constants in CNVARB is essentially identical to subroutine CNFWBK.

Solution characteristics should be very similar to CNFWBK solutions with expectation that CNVARB solutions would be more optimum in terms of accuracy and solution time. Solutions are not presently available to verify or refute the expected advantages of CNVARB solutions.

6.4.3.2 Finite Difference Approximation and Computational Algorithm

The numerical solution algorithm used in subroutine CNVARB is a combination of forward-differencing and backward-differencing with the weighting of each determined by the ratio of control constants CSGMIN/DTIMEU.

The combination forward-backward differencing with weighting can be expressed as:

$$\frac{c_{i}}{\Delta t} (T_{i,n+1} - T_{i,n}) = \beta \left(q_{i,n} + \sum_{j=1}^{p} a_{ij,n} (T_{j,n} - T_{i,n}) \right) + \sum_{j=1}^{p} \sigma b_{ij,n} (T_{j,n}^{4} - T_{i,n}^{4})$$

$$+ (1.0 - \beta) \left(q_{i,n} + \sum_{j=1}^{p} a_{ij,n} (T_{j,n+1} - T_{i,n+1}) \right) + \sum_{j=1}^{p} \sigma b_{ij,n} (T_{j,n+1}^{4} - T_{i,n+1}^{4}) \quad (6.4-9)$$

$$i = 1, 2, \dots, N$$

$$n = nth time-step$$

$$\beta = weighting factor (0 < \beta \le 1/2)$$

$$T_{j,n}; T_{j,n+1} = constant, N < j \le p$$

If equation (6.4-9) is multiplied by 2.0 and the known quantities (at time-step, n) and the unknown quantities (at time-step, n+1) separated, the algorithm used in CNVARB may be obtained readily.

Diffusion Nodes

where,

Diffusion-node temperatures are solved by "successive point" iteration. The finite difference iterative form as used in CNVARB can be found by multiplying equation (6.4-9) by 2.0 and by using appropriate time-step, n, and iteration, k subscripts.

$$T_{i,k+1} = DD^* T_{i,k} + DN^* [Q_{sum} - (q_i)_{ave}]/G_{sum}$$
(6.4-10)

$$Q_i = 2 q_{i,n} + 2 \overline{C}_{i,n} T_{i,n}$$

$$+ \beta' \left(\sum_{j=1}^{p} a_{ij,n} (T_{j,n} - T_{i,n}) + \sum_{j=1}^{p} \sigma b_{ij,n} (T_{j,n}^4 - T_{i,n}^4) \right)$$
(6.4-11)

$$Q_{sum} = Q_i + (2.0 - \beta') \left(\sum_{j=1}^{i} G_{ij,n} T_{j,k+1} + \sum_{j=i+1}^{p} G_{ij,n} T_{j,k} \right)$$
(6.4-12)

$$G_{sum} = 2 \overline{C}_{i,n} + (2.0 - \beta') \sum_{j=1}^{p} a_{ij,n}$$
 (6.4-13)

$$G_{ij,n} = a_{ij,n} + \sigma b_{ij,n} T_{j,\ell}^{3}$$
(6.4-14)

 $(l = k, \text{ if } j \ge i \text{ and } l = k+1, \text{ if } j < i)$

$$(q_i)_{ave} = \frac{(2.0-\beta')}{2} \sum_{j=1}^{p} \sigma b_{ij,n} [(T_{i,k}^4) + (T_{i,k}^4)_2]$$
 (6.4-15)

average heat loss from the ith node, called radiation
damping (refer to Section 6.2.6 for details)
= 0, if radiation is not present

 $\beta' = 2.0 \times CSGMIN/DTIMEU \text{ (range allowed, } 0 \le \beta' \le 1.0\text{, note } \beta' = 2\beta\text{)}$ n = nth time-step; k = kth iteration $C_{i}, q_{i}, a_{ij}, b_{ij} = \text{optionally specified (refer to Tables 6.2-1 - 6.2-4)}$ $\overline{C}_{i,n} = C_{i,n}/\Delta t$ $i = 1, 2, \dots, \text{NND}$ $T_{j,n}; T_{j,k} = \text{constant, (NND + NNA)} < j \le p \text{ (p is the total number of } j \le p \text{ (p is the total numb$

nodes and NNA is the number of arithmetic nodes)

Arithmetic Nodes

Arithmetic nodes are calculated in the same manner in all the SINDA numerical solution routines. For the finite difference algorithm refer to either Section 6.3.1.2 or Section 6.2.5.2.

6.4.3.3 Comments on the Computational Procedure

The important steps of the computational procedure used in subroutine CNVARB are indicated in Table 6.4-3. For a detailed step-by-step computational description, the user must examine the computer listing for CNVARB in Appendix B, but some general computational details are given in Section 6.2.5.2. A functional flow chart of CNVARB is shown in Figure 6.4-3.

The computational flow pattern for CNVARB is very similar to CNFWBK or CNBACK; the slight difference is shown in the flow chart of Figure 6.4-3. The basic difference between CNVARB and the other two implicit routines is the use of a variable beta, β ', which is calculated internally by the routine. Thus, the updating of the variable capacitance C_i , the impressed source q_i and the variable coefficients (a_{ij} for conduction and σb_{ij} for radiation) during the first iteration and the subsequent calculation of diffusion-node temperatures in subsequent iterations are identical to CNFWBK except for the finite difference algorithm. Use of the various control constants and checks are identical to CNFWBK.

6.4.3.4 Control Constants

Control constants for CNVARB are used in exactly the same way as used in CNFWBK. Control constant ARLXCA, DRLXCA, DTIMEI, NLØØP, ØUTPUT, and TIMEND must be specified as indicated in Table 6.2-5 and as described in Section 6.2.3.2; otherwise "run" will terminate with an appropriate error message. The function of optionally specified control constants ATMPCA, BACKUP DAMPA, DAMPD, DTIMEH, DTMPCA and TIMEØ is described in Section 6.2.3.2.

6.4.3.5 Error and Other Messages

If control constants ARLXCA, DRLXCA, DTIMEI, NLØØP, ØUTPUT and TIMEND are not specified, the following error message will be printed for each,

APLXCA	"NØ ARLXCA"
DRLXCA	"NØ DRLXCA"
DTIMEI	"NØ DTIMEI"
nløøp	"NØ NLØØP"
ØUTPUT	"NØ OUTPUT INTERVAL"
TIMEND	"TRANSIENT TIME NØT SPECIFIED"

If the long pseudo-compute sequence LPCS is not specified, the error message will be,

"CNVARB REQUIRES LØNG PSEUDØ-CØMPUTE SEQUENCE"

If the dynamic storage allocation is not sufficient (NDIM < (3*NND + NNA + NNB)), the error message will be,

" LØCATIONS AVAILABLE"

Note that the <u>number presented will be negative</u> indicating the additional storage locations required.

If CSGMIN \leq 0, the following message will be printed,

"CSGMIN ZERØ or NEGATIVE"

If either ARLXCA or DRLXCA is not satisifed with NLØØP iterations, the following message will be printed,

"RELAXATION CRITERIA NØT MET"

Checks on the control constants, the pseudo-compute sequence and the dynamic storage allocation are made in the following sequence with the "run" terminating if a single check is not satisfied,

NLØØP, TIMEND, ØUTPUT, ARLXCA, LPCS and dynamic storage allocation.

Table 6.4-3. Basic Computational Steps for CNVARB

- 1. Specification of control constants. Control constants ARLXCA (if NNA > 0), DRLXCA (if NND > 0), DTIMEI, NLØØP, ØUTPUT and TIMEND (TIMEND > TIMEØ) must be specified. LPCS is required. (Refer to Table 6.2-5 for values and Section 6.2.3.2 for description.)
- Sufficiency check on dynamic storage. Requirements = 3*NND + NNA + NNB (NND = diffusion nodes, NNA = arithmetic nodes and NNB = boundary nodes).
- 3. Setting and/or calculation of time-step, ∆t. (Refer to Section 6.2.4 for detailed procedure.) Time-step = DTIMEI.
- 4. Setting of iterative DØ loop, 1 to NLØØP.
- 5. Setting of source locations to zero.
- 6. Calling of Variables 1 (refer to Section 6.2.2.2 for description).
- 7. Checking of BACKUP (refer to Section 6.2.3.2 for description).
- 8. Diffusion-node temperature calculations, first iteration only.

Checking of stable stability criteria. Calculation of weighting factor $\beta'= 2.0$ *CSGMIN/DTIMEU. $(0 \le \beta' \le 1.0)$ Conversion of temperatures to degrees Rankine. Damping of radiation heat transfer (refer to Section 6.2.5.2). Calculation of diffusion-node temperatures using forward-backward algorithm with variable beta (β'). Calculation of DRLXCC.

- 9. Diffusion-node temperature calculations, successive iterations after first. Repeating of step 8 except that q_i , C_i and G_k are not updated. Calculation of DRLXCC.
- 10. Acceleration of convergence every third iteration if linear extrapolation criterion is met (refer to Section 6.2.7).
- 11. Conversion of $T_{i,k+1}$ to degrees Fahrenheit.
- 12. Calculation of arithmetic-node temperatures every iteration (refer to Section 6.2.5.2 for details).
- 13. Conversion of temperatures to degrees Rankine.
- 14. Checking of ARLXCA and DRLXCA for convergence and ØPEITER for output. If both ARLXCA and DRLXCA are satisfied, iterations during a time-step cease, otherwise NLØØP iterations are performed.
- 15. Checking of ATMPCA and DTMPCA. If either one is not satisfied timestep is shortened, previous temperatures erased, and temperatures recalculated for shortened time-steps (refer to Section 6.2.5.2).
- 16. Conversion of temperatures back to degrees Fahrenheit.
- 17. Calling of VARIABLES 2 and checking of BACKUP (refer to Section 6.2.2.3 and 6.2.3.2).
- 18. Advancing of time, checking of time to print, and the printing of the output interval.
- 19. Calling of ØUTPUT CALLS.
- 20. Checking for problem end time stored in user specified control constant TIMEND.



6.5 Steady State Numerical Solution Routines

SINDA steady state numerical solution routines number three. These steady state routines are identified as follows:

> CINDSS Block iterative method Requires short pseudo-compute sequence (SPCS)

CINDSL Successive point iterative method Requires long pseudo-compute sequence (LPCS)

CINDSM Modified CINDSL for radiation-dominated problems Requires long pseudo-compute sequence (LPCS)

A detailed description of steady state routines is presented in the pages to follow with liberal reference to materials presented in Section 6.2. A brief description of these routines follows.

<u>CINDSS</u> which uses the short pseudo-compute sequence (SPCS) was the first steady state routine developed for SINDA (via CINDA and CINDA-3G); as a result, some of the features contained in subsequent steady state routines are not used in CINDSS. If a transient analysis is to be performed following a steady state analysis, CINDSS must be used with a transient routine that also requires SPCS. The "block" iterative method (refer to Section 5.2.3) used by CINDSS should lend itself to some types of problems which are highly nonlinear with terms such as G_{ij} $(T_j^4 - T_i^4)$. With "block" iteration, both T_j and T_i are changed simultaneously. Solution convergence is based upon a <u>temperature</u> relaxation criterion stored in DRLXCA for diffusion nodes and ARLXCA for arithmetic nodes.

<u>CINDSL</u> requires the long pseudo-compute sequence (LPCS) and uses the "successive point" iteration method (refer to Section 5.2.3). Any transient analysis routine coupled with CINDSL must require LPCS. Solution time for CINDSL is less than CINDSS; as a result, it is used more often than CINDSS. A major problem with CINDSL is that a highly nonlinear problem can present convergence difficulties unless considerable amount of damping is used. For example, a radiation-dominated problem contains many σb_{ij} ($T_j^4 - T_i^4$). With "successive point" iteration, T_j may be updated and T_i not for a given conductor; as a result, the resultant heat flow calculation could present difficulties because of large change in values. CINDSL has the acceleration of convergence feature, whereas CINDSS does not.

Solution convergence is based upon <u>temperature</u> relaxation criterion stored in DRLXCA for diffusion nodes and ARLXCA for arithmetic nodes.

- V 44

.

<u>CINDSM</u> is the latest addition to the SINDA library of steady state routines. CINDSM requires the long pseudo-compute sequence and uses "successive point" iteration. The routine was specifically developed to solve radiation-dominated problems. Solution convergence is based upon system energy criterion stored in BALENG.

.

6.5.1 Subroutine: CINDSS

6.5.1.1 General Comments

Subroutine CINDSS is a steady state routine that requires the short pseudo-compute sequence (SPCS) and ignores the capacitance values of diffusion nodes to calculate steady state temperatures. Diffusion nodes are solved by a "block" iterative method as discussed in Section 6.5.2.3, whereas arithmetic nodes are solved by a "successive point" iterative method also discussed in Section 6.5.2.3. For steady state solutions diffusion nodes are not necessary; as a matter of fact, solutions will be achieved more quickly if all diffusion nodes are specified as arithmetic. The use of diffusion nodes in a steady state solution allows for the direct use of the transient model.

A series of steady state solutions at various points in a time period can be accomplished by specifying control constants TIMEN and ØUTPUT. ØUTPUT is used both as the output interval and the computational interval. The instructions with the appropriate call are made in VARIABLES 1 to modify boundary conditions with time.

The CINDSS call can be followed by a call to one of the transient solution subroutines which has the same short pseudo-compute sequence requirements such as CNFRWD. In this manner the steady state solution becomes the initial conditions for the transient analysis. It is important to remember that control constants specified for the steady state routine will be used by the transient routine unless initialized to the desired values. Since CINDSS utilizes control constants TIMEND and ØUTPUT for the steady state-transient problem, the user must specify their values in the execution block after the steady state call and prior to the transient analysis call. CINDSS does not utilize the acceleration of convergence feature as discussed in Section 6.2.7.

Solution convergence is based upon a temperature relaxation criterion stored in control constants DRLXCA for diffusion nodes and ARLXCA for arithmetic nodes. Normally, identical values are specified for both DRLXCA and ARLXCA. Sufficient information is not presently available to indicate different values for DRLXCA and ARLXCA. A method to indicate the accuracy of the "converged" temperatures is not presently available. It

should also be noted that "converged" temperatures could have large system energy unbalance.

6.5.1.2 Finite Difference Approximation and Computational Algorithm

The steady state heat balance equation at the ith node may be readily expressed as,

$$q_{i} + \sum_{j=1}^{p} a_{ij} (T_{j} - T_{i}) + \sum_{j=1}^{p} \sigma b_{ij} (T_{j}^{4} - T_{i}^{4}) = 0$$
(6.5-1)
$$i = 1, 2, ..., N$$
$$T_{j} = \text{constant}, N < j \le p$$

Equation (6.5-1) represents a set of nonlinear algebraic equations to be solved simultaneously. Since CINDSS solves temperature of nodes specified as diffusion (nodes with capacitance even though a steady state solution is desired) by the "block" iteration method and temperatures of nodes specified as arithmetic (no capacitance) by the "successive point" iteration method, two successive approximation algorithms are used.

Diffusion Nodes (if any)

$$T_{i,k+1} = DD* T_{i,k} + \frac{DN*(q_{i,k} + \sum_{j=1}^{p} G_{ij,k} T_{j,k})}{\sum_{j=1}^{p} G_{ij,k}}$$
(6.5-2)

Arithmetic Nodes (if any)

$$T_{i,k+1} = AD* T_{i,k} + \frac{AN* (q_{i,k} + \sum_{j=1}^{i} G_{ij,k} T_{j,k+1} + \sum_{j=i+1}^{p} G_{ij,k} T_{j,k})}{\sum_{j=1}^{p} G_{ij,k}}$$

6 - 102

where,
$$k = kth$$
 iteration; $i = (NND + 1), (NND + 2), ..., (NND + NNA)$
 $q_{j}, a_{jj}, b_{jj} = optionally specified (refer to Tables 6.2-1 - 6.2-4)$
 $T_{j,k} = constant, (NND + NNA) < j \le p$ (NNA is the number of
arithmetic nodes and p is the total number of nodes
 $G_{ij,k} = a_{ij,k} + \sigma b_{ij,k} (T_{j,\ell}^2 + T_{i,j}^2) (T_{j,\ell} + T_{i,k})$
 $(\ell = k, if j \ge i and \ell = k+1, if j < i)$
AN = DAMPA (arithmetic node damping factor)
 $AD = 1, 0 - AN$

6.5.1.3 Comments on the Computational Procedure

The important steps of the computational procedure used in the steady state subroutine CINDSS are indicated in Table 6.5-1. For a detailed procedural description, the user must examine the computer listing for CINDSS in Appendix C, but some general computational details are given in Section 6.2.5.3. A functional flow chart of CINDSS is shown in Figure 6.5-1. The user is required to specify the maximum number of iterations to be performed via control constant NLØØP and the diffusion-node temperature change relaxation criteria DRLXCA and the arithmetic-node temperature change criteria ARLXCA. The iterations continue until either NLØØP is satisfied or both DRLXCA and ARLXCA are satisfied. If DRLXCA and ARLXCA are not satisfied with NLØØP iterations, an appropriate message is printed. VARIABLES 1 and ØUTPUT CALLS are performed at the start and VARIABLES 2 and ØUTPUT CALLS are performed upon completion. Control constants DAMPD for diffusion nodes and DAMPA for arithmetic nodes are so-called damping factors which are multipliers of the "new" temperatures; the factor 1.0 - DAMPD (or 1.0 - DAMPA) is a multiplier for the "old" temperatures. This weighting of "old" and "new" temperatures is useful for damping oscillations due to nonlinearities. For nonlinear systems, the damping factors are specified to be less than one. If not specified, the damping factor is set to 1.0. As a point of interest, it appears that if a linear system is to be solved, the convergence could be accelerated by using the damping factor greater than one. The diffusion nodes receive a "block" iteration, whereas the arithmetic nodes receive a "successive point" iteration; acceleration features are not utilized.

6.5.1.4 Control Constants

Control constant NLØØP must be specified and control constants ARLXCA and DRLXCA must be specified if NNA > 0 and NND > 0, respectively; otherwise "run" will terminate with an appropriate error message. Control constants DAMPA and DAMPD may be optionally specified among others. Control constant characteristics are tabulated in Table 6.2-5 and description of these control constant is presented in Section 6.2.3.2. Specification of NLØØP is dependent upon the values of ARLXCA and DRLXCA and thus the accuracy of solution. Since the type of problem will influence accuracy, it appears that a trial and error procedure is the only practical way of determining realistic control constant values.

6.5.1.5 Error and Other Messages

If control constants ARLXCA, DRLXCA and NLØØP are not specified, the following error message will be printed for each,

ARLXCA	"NØ	ARLXCA"
DRLXCA	"NØ	DRLXCA"
nløøp	"NØ	nløøp"

If the short pseudo-compute sequence SPCS is not specified, the error message will be,

"CINDSS REQUIRES SHØRT PSEUDO-CØMPUTE SEQUENCE"

If the dynamic storage allocation is not sufficent (NDIM < NND) will be,

" LØCATIONS AVAILABLE"

Note that the number printed will be negative indicating the additional storage locations required.

If both temperature change relaxation criteria ARLXCA and DRLXCA are not met with NLØØP iterations, the message will be,

"ITERATION COUNT EXCEEDED, LOOPCT = "

Checks on the control constants, the pseudo-compute sequence, and the dynamic storage allocation are made in the following order with the "run" terminating if a single check is not satisfied.

NLØØP, ARLXCA, DRLXCA, SPCS, and dynamic storage allocation.

Table 6.5.1. Basic Computational Steps for CINDSS

- Specification of control constants. Control constants ARLXCA (if NNA > 0), DRLXCA (if NND > 0) and NLØØP must be specified. SPCS is required. (Refer to Table 6.2-5 for values and Section 6.2.3.2 for description.)
- Sufficiency check on dynamic storage. Requirements = NND (NND = diffusion nodes).
- 3. Setting of TIMEN for first iteration and succeeding iterations.

TIMEN = $TIME\phi$, first iteration

TIMEN = TIME ϕ + ϕ UTPUT, succeeding iterations

- 4. Setting of iterative loop for all nodes, k1 = 1, $NL \phi \phi P$
- 5. Setting of source locations to zero.
- 6. Calling of VARIABLES 1 (refer to Section 6.2.2.2 for description).
- Calculation of diffusion-node temperatures by "block" iteration if NND > 0 (refer to sections 6.2.5.3 and 6.5.1.2).

$$T_{i,k+1} = DD^* T_{i,k} + \frac{DN^* (q_{i,k} + \sum_{j=1}^{p} G_{ij,k} T_{j,k})}{\sum_{j=1}^{p} G_{ij,k}}$$

$$DN = DAMPD \text{ and } DD = 1.0 - DN$$

- 8. Calculation of DRLXCC.
- 9. Calculation of arithmetic-node temperatures by "successive point" iteration if NNA > 0 (refer to Sections 6.2.5.3 and 6.5.1.2).

$$T_{i,k+1} = AD* T_{i,k} + \frac{AN* (q_{i,k} + \sum_{j=1}^{i} G_{ij,k} T_{j,k+1} + \sum_{j=i+1}^{p} G_{ij,k} T_{j,k})}{\sum_{j=1}^{p} G_{ij,k}}$$

$$AN = DAMPA$$

$$AD = 1.0 - DAMPA$$

- 10. Calculation of ARLXCC.
- 11. Checking of DRLXCC and ARLXCC against the relaxation criteria DRLXCA and ARLXCA, respectively, for convergence. If both ARLXCA and DRLXCA are satisfied, iterations cease, otherwise NLØØP iterations are performed.
- 12. Calculation of system energy balance which is stored in ENGBAL.
- 13. Call VARIABLES 2 and ØUTCAL, print ENGBAL and LØØPCT.
- 14. Check if TIMEND = TIMEN.

6 - 105

1



Figure 6.5-1. Functional Flow Chart for CINDSS

6.5.2 Subroutine: CINDSL

6.5.2.1 General Comments

Subroutine CINDSL is a steady state routine that requires the long pseudo-compute sequence (LPCS). Both diffusion- and arithmetic-node temperatures are calculated by a "successive point" iteration computational technique. Every third iteration a linear extrapolation is performed to accelerate convergence. CINDSL generally yields significantly faster solutions than CINDSS, but nonlinear problems such as those with radiation heat transfer can pose considerable convergence difficulties unless a large amount of damping (low values of DAMPA and DAMPD) is imposed.

A series of steady state solutions at various points in time can be generated by specifying control constants TIMEND and \emptyset UTPUT. \emptyset UTPUT is used both as the output interval and the computation interval; this requires appropriate <u>calls</u> in VARIABLES 1 to modify boundary conditions with time.

CINDSL can be followed by a call to one of the transient numerical solution routines which have the same LPCS requirements. Used in this manner the steady state solutions become the initial conditions for the transient analysis. Note that since CINDSL utilizes control constants TIMEND and ØUTPUT for the coupled steady state-transient problem, the user must specify the values of TIMEND and ØUTPUT in the execution block after the steady state call and prior to the transient analysis call.

Solution convergence is based upon a temperature relaxation criterion stored in control constants DRLXCA for diffusion nodes and ARLXCA for arithmetic nodes. Normally, identical values are specified for both DRLXCA and ARLXCA for lack of anything better. The damping factors DAMPD for diffusion nodes and DAMPA for arithmetic nodes are merely multipliers of "new" temperatures and the factor 1.0 - DAMPD (or 1.0 - DAMPA) is a multiplier of the "old" temperatures. Normally, these damping factors are specified to be less than 1.0, but for a linear system the convergence probably could be accelerated by using a damping factor greater than one.

6.5.2.2 Finite Difference Approximation and Computational Algorithm

The set of steady state heat balance equations,

$$q_{i} + \sum_{j=1}^{p} a_{ij} (T_{j} - T_{i}) + \sum_{j=1}^{p} ob_{ij} (T_{j}^{4} - T_{i}^{4}) = 0$$

$$i = 1, 2, \dots, N$$

$$T_{j} = \text{constant } N < j \le p$$

is solved by a re-iterative scheme called a "successive point" iterative method here. Both diffusion-node and arithmetic-node temperatures are solved in this manner. The only difference between the two algorithms is that control constant DAMPD is used with diffusion nodes and control constant DAMPA is used with arithmetic nodes.

Diffusion Nodes (if any)

Y

$$T_{i,k+1} = DD* T_{i,k} + DN* \frac{(q_{i,k} + \sum_{j=1}^{j} G_{ij,k} T_{j,k+1} + \sum_{j=i+1}^{p} G_{ij,k} T_{j,k})}{\sum_{j=1}^{p} G_{ij,k}} (6.5-4)$$

Arithmetic Nodes (if any)

$$T_{i,k+1} = AD* T_{i,k} + AN* \frac{(q_{i,k} + \sum_{j=1}^{p} G_{ij,k} T_{j,k+1} + \sum_{j=i+1}^{p} G_{ij,k} T_{j,k})}{\sum_{j=1}^{p} G_{ij,k}}$$
(6.5-5)

where,
$$i = (NND + 1), (NND + 2), ..., (NND + NNA)$$

 $q_{j}, a_{jj}, b_{jj} = may be optionally specified (refer to Tables 6.2-1 - 6.2-4)$
 $T_{j,k} = constant (NND + NNA) < j \le p$ (NNA is the number of arithmetic nodes and p is the total number of nodes)
 $AN = DAMPA$ (arithmetic-node damping factor)
 $AD = 1.0 - AN$
 $G_{ij,k} = a_{ij,k} + \sigma b_{ij,k} (T_{j,l}^2 + T_{i,k}^2) (T_{j,l} + T_{i,k})$
 $(l = k \text{ if } j \ge i \text{ and } l = k+1 \text{ if } j < i)$

6.5.2.3 Comments on the Computational Procedure

The important steps of the computational procedure used in the steady state subroutine CINDSL are indicated in Table 6.5-2. For a detailed procedural description, the user must examine the computer listing for CINDSL in Appendix C, but some general computational details are given in Section 6.2.5.3. A functional flow chart of CINDSL is shown in Figure 6.5-2.

The computational pattern of CINDSL is very similar to CINDSS with the differences being that CINDSL uses the long pseudo-compute sequence, whereas CINDSS uses the short pseudo-compute sequence, and that CINDSL contains the acceleration convergence feature, whereas CINDSS does not. The user is required to specify the maximum number of iterations to be performed via control constant NLØØP and the diffusion-node temperature change relaxation criteria DRLXCA and the arithmetic-node temperature change relaxation criteria ARLXCA. The iterations continue until either NLØØP is satisfied or both DRLXCA and ARLXCA are satisfied. If DRLXCA and ARLXCA are not satisfied with NLØØP, an appropriate message is printed. Acceleration of convergence is performed every third iteration if a temperature is converging over two time-steps.

6.5.2.4 Control Constants

Control constant NLØØP must be specified and control constants ARLXCA and DRLXCA must be specified if NNA > 0 and NND > 0, respectively; otherwise "run" will terminate with an appropriate error message. Control constants DAMPA and DAMPD may be optionally specified among others. Control constant characteristics are tabulated in Table 6.2-5 and description of these control constants is presented in Section 6.2.3.2. Specification of NLØØP is dependent upon the values of ARLXCA and DRLXCA and thus the accuracy of the solution. Since the type of problem will influence accuracy, it appears that a trial and error procedure is the only practical way of determining realistic control constant values.

6.5.2.5 Error and Other Messages

If control constants ARLXCA, DRLXCA and NLØØP are not specified, the following error message will be printed for each,

ARLXCA	"NØ	ARLXCA"
DRLXCA	"NØ	DRLXCA"
nløøp	"NØ	nløøp"

If the long pseudo-compute sequence LPCS is not specified, the error message will be,

"CINDSL REQUIRES LØNG PSEUDØ-COMPUTE SEQUENCE"

If the dynamic storage allocation is not sufficient, (NDIM < 2* (NNA + NND)), the message will be,

" LØCATIONS AVAILABLE"

Note that the number printed will be negative indicating the additional storage locations required.

"LØØPCT = and ENGBAL = "

If both temperature change relaxation criteria, ARLXCA and DRLXCA, are not met with NLØØP iterations, the message will be,

"ITERATIØN CØUNT EXCEEDED, LØØPCT = ____"

Checks on the control constants, the pseudo-compute sequence, and the dynamic storage allocation are made in the following order with the "run" terminating if a single check is not satisfied.

NLØØP, ARLXCA, DRLXCA, LPCS, and dynamic storage allocation.

Table 6.5.2 Basic Computational Steps for CINDSL

- 1. Specification of control constants. Control constants ARLXCA (if NNA > 0), DRLXCA (if NND > 0) and NLØØP must be specified. LPCS is required. (Refer to Table 6.2-5 for values and Section 6.2.3.2 for dexcription.)
- 2. Sufficiency check on dynamic storage. Requirements = 2* (NND + NNA) (NND = diffusion nodes and NNA = arithmetic nodes).
- 3. Setting of TIMEN for first and succeeding iterations.

TIMEN = TIMEØ, first iteration TIMEN = TIMEØ + ØUTPUT, succeeding iterations

- 4. Setting of iterative loop for all nodes, kl = 1, NLØØP.
- 5. Setting of source locations to zero.
- 6. Calling of VARIABLES 1 (refer to Section 6.2.2.2 for description).
- 7. Calculation of diffusion-node temperatures by "block" iteration if NND > 0 (refer to Section 6.2.5.2 and 6.5.1.2).

$$T_{i,k+1} = DD* T_{i,k} + \frac{\frac{DN* (q_{i,k} + \sum_{j=1}^{i} G_{ij,k} T_{j,k+1} + \sum_{j=i+1}^{p} G_{ij,k} T_{j,k})}{\sum_{j=1}^{p} G_{ij,k}}$$

$$DN = DAMPD \text{ and } DD = 1.0 - DN$$

- 8. Calculation of DRLXCC.
- 9. Calculation of arithmetic-node temperatures by "successive point" iteration if NNA > 0 (refer to Sections 6.2.5.3 and 6.5.1.2).

$$T_{i,k+1} = AD* T_{i,k} + \frac{AN* (q_{i,k} + \sum_{j=1}^{i} G_{ij,k} T_{j,k+1} + \sum_{j=i+1}^{p} G_{ij,k} T_{j,k})}{\sum_{j=1}^{p} G_{ij,k}}$$

- 10. Calculation of ARLXCC.
- 11. Checking of DRLXCC and ARLXCC against the relaxation criteria DRLXCA and ARLXCA, respectively, for convergence. If both ARLXCA and DRLXCA are satisfied, iterations cease, otherwise NLOOP iterations are performed.
- 12. Acceleration of convergence each third iteration, if linear extrapolation criterion is met (refer to Section 6.2.7).
- 13. Calculation of system energy balance which is stored in ENGBAL.
- 14. Call VARIABLES 2 and ØUTCAL, print ENGBAL and LØØPCT.
- 15. Check if TIMEND = TIMEN.



6.5.3 <u>Subroutine: CINDSM</u>

6.5.3.1 General Comments

Subroutine CINDSM is a steady state routine specifically generated for radiation dominated problems. CINDSM requires the long pseudo-compute sequence (LPCS) and is considerably different from CINDSL. CINDSM is based on the use of pseudo linear equations which are the result of linearizing the radiation conductors. These equations are solved by using the "successive point" method with LAXFAC iterations. Updating of the properties as well as the linearized conductors occur outside of the iterative loops. Temperature convergence is based on a criterion that is continually tightened until either the NLØØP iterations or the system energy balance criterion stored in BALENG has been satisfied.

The acceleration of convergence by linear extrapolation as used in CINDSM is essentially the same as used in the other SINDA numerical solution routines, but in lieu of limiting the extrapolation by an allowable slope value (refer to Section 6.2.7) the maximum temperature change of the network on the last iteration is used as the allowable value.

Information available at this time indicates that each problem appears to have an optimum combination of NLØØP, DAMPD, and LAXFAC values. An NLØØP of 100, a DAMPD of 0.5 and a LAXFAC of 10 has been successfully applied to spacecraft problems with radiation domination, but the solution time is rather long.

6.5.3.2 Finite Difference Approximation and Computational Algorithm

The set of steady state heat balance equations,

$$q_{i} + \sum_{j=1}^{p} a_{ij} (T_{j} - T_{i}) + \sum_{j=1}^{p} \sigma b_{ij} (T_{j}^{4} - T_{i}^{4}) = 0$$

$$i = 1, 2, \dots, N$$

$$T_{i} = \text{constant}, N < j \le p$$

is solved by a re-iterative "successive point" method after linearization. Linearization is achieved by letting $\sigma b_{ij} (T_j^4 - T_i^4) = G_r (T_j - T_i)$ with $G_r = \sigma b_{ij} (T_j^2 + T_i^2)(T_j + T_i)$. This yields

$$q_{i} + \sum_{j=1}^{p} a_{ij} (T_{j} - T_{i}) + \sum_{j=1}^{p} G_{r} (T_{j} - T_{i}) = 0$$
 (6.5-6)

Diffusion and Arithmetic Nodes

No distinction is made between diffusion and arithmetic nodes. As a result, the following algorithm applies to both types of nodes,

$$T_{i,k} = DD* T_{i,k} + DN* \frac{(q_{i,L} + \sum_{j=1}^{i} G_{ij,L} T_{j,k+1} + \sum_{j=i+1}^{p} G_{ij,L} T_{j,k})}{\sum_{j=1}^{p} G_{ij,L}} (6.5-7)$$

where, i = 1, 2, ..., (NND + NNA); p = total number of nodes<math>k = kth iteration L = before each LAXFAC iterative loop $T_{j,k} = constant, (NND + NNA) < j \le p$ DN = DAMPD (diffusion-node damping factor; DAMPA is not used) DD = 1.0 - DAMPD $G_{ij,L} = a_{ij,L} + \sigma b_{ij,L} (T_{j,L}^2 + T_{i,L}^2) (T_{j,L} + T_{i,L})$

 $(G_{ij,L}$ is updated once before each LAXFAC iterative loop NNA = number of arithmetic nodes

NND = number of diffusion nodes

q_i,a_{ij},b_{ij} = may be optionally specified (refer to Tables 6.2-1 - 6.2-4) 6.5.3.3 <u>Comments on the Computational Procedure</u>

A detailed step-by-step computational procedure as used in the steady state routine CINDSM is presented in Table 6.5-3. For a more detailed procedural description, the user must examine the computer listing in Appendix C. A functional flow chart that is compatible with the step-by-step description of Table 6.5-3 is shown in Figure 6.5-3.

CINDSM is considerably different from either CINDSS or CINDSL because of the use of a variable convergence criterion which is internally updated. Overall, from a total system basis, control constants $NL \phi \phi P$ and BALENG are the ultimate criteria.

It should be particularly noted here that unlike CINDSS or CINDSL, which use both DAMPA and DAMPD, CINDSM uses only DAMPD. The reason for this is that CINDSM does not treat the nodal types as diffusion or arithmetic.

6.5.3.4 Control Constant

Control constants BALENG, LAXFAC AND NLØØP must be specified; otherwise the "run" will terminate with an appropriate error message. Control constant DAMPD may be optionally specified among others. Control constant characteristics are tabulated in Table 6.2.5 and description of these control constants is presented in Section 6.2.3.2. Specification of BALENG, LAXFAC and NLOOP appears to be a trial and error procedure.

6.5.3.5 Error and Other Messages

If control constants BALENG, LAXFAC, and NL $\phi\phi$ P are not specified, the following error message will be printed for each,

BALENG	"NØ	BALENG"
LAXFAC	"NØ	LAXFAC"
nløøp	"NØ	NLØØP"

If the long pseudo-compute sequence LPCS is not specified, the error message will be,

"CINDSM REQUIRES LØNG PSEUDØ-CØMPUTE SEQUENCE"

If the dynamic storage allocation is not sufficient, (NDIM < (3* NNA + 3* NND + NGT)), the message will be,

" LOCATIONS AVAILABLE"

Note that the <u>number printed will be negative</u> indicating the additional storage locations required.

If either NL $\phi\phi$ P iterations has been made or if ENGBAL \leq BALENG, the following message is printed,

"LØØPCT = and ENGBAL "

Checks on the control constants, the pseudo-compute sequence and the dynamic storage allocation are made in the following order with the "run" terminating if a single check is not satisfied,

NLØØP, LPCS, BALENG, LAXFAC and dynamic storage allocation.

Table 6.5.3. Basic Computational Steps for CINDSM

- Specification of control constants. Control constants BALENG, LAXFAC and NLØØP must be specified. The long pseudo-compute sequence (LPCS) is required. (Refer to Table 6.2-5 for values and Section 6.2.3.2 for description.)
- 2. Sufficiency check on dynamic storage. Requirement = 3* (NND + NNA) + NGT (NND = diffusion nodes, NNA = arithmetic nodes and NGT = total number of conductors).
- 3. Setting of TIMEN for the first and succeeding iterations.

TIMEN = TIMEØ, first iteration TIMEN = TIMEØ + ØUTPUT, succeeding iterations

4. Constants used in CINDSM

 \cdot

- NLAX = NLØØP/LAXFAC (both NLØØP and LAXFAC are specified by the user)
- RELAX = .05 (initial value used in CINDSM as the allowable temperature change)
- DELXXX = .05/NLAX (a number used in reducing RELAX for a tighter criterion)
- XXXDUM = .001 (a value of RELAX used in CINDSM for a tighter criterion)
 - = .001/5 (a subsequent value of RELAX for a tighter criterion)
- DAMP = DAMPD (damping factor for all nodes; DAMPA is not used)
- 5. Updating of variables and linearization of radiation .

Variable q; and Gk are evaluated by calling subroutine NØNLIN.

Linearization means that the radiation exchange expressed as $\sigma b_{ij} (T_j^4 - T_i^4)$. Normally, G_{ij} would be updated each iteration as done in CINDSS or CINDSL, but in CINDSM G_{ij} is not updated within the DØ-LØØP (kl = 1,LAXFAC) but is updated outside of the loop.

6. Iterative $D\emptyset - L\emptyset\emptyset P$ (k1 = 1, LAXFAC) is established.

Temperatures of all nodes are calculated by "successive point" iteration with no damping.

$$T_{i,k+1} = \frac{q_{i} + \sum_{j=1}^{i} G_{ij} T_{j,k+1} + \sum_{j=i+1}^{p} G_{ij} T_{j,k}}{\sum_{j=1}^{p} G_{ij}}$$
(6.5-5)

where, $G_{ij} = a_{ij} + \sigma_{b_{ij}} (T_j^2 + T_i^2) (T_j + T_i) (q_i and G_{ij} are not updated during the LAXFAC iterations)$

Check on temperature convergence. Temperatures have converged if,

$$|\mathbf{T}_{i,k+1} - \mathbf{T}_{i,k}|_{\max} \leq \text{RELAX} (= .05)$$

If temperatures have converged, the computation goes out of the iteration loop to step (7).

6 - 116

e.
Every third iteration, acceleration of convergence is attempted if linear extrapolation criterion is met (refer to Section 6.2.7).

Iteration ceases if LAXFAC iterations have been performed or if the temperatures have converged.

7. Check on NLAX iterations.

If in step (6) the number of iterations, $LOOPCT \ge NLAX$, the computational procedures go to step (9). However, in step (6) if the number of iterations LOOPCT < NLAX, then a set of temperature calculations is made using "successive point" method with a damping factor and no iterations.

$$T_{i,k+1} = DD* T_{i,k} + DN* \frac{(q_i + \sum_{j=1}^{r} G_{ij} T_{j,k+1} + \sum_{j=i+1}^{p} G_{ij} T_{j,k})}{\sum_{j=1}^{p} G_{ij}}$$

where, DN = DAMPD (diffusion node damping factor; note DAMPA is not used)
 G_{ij} = constant

Allowable temperature change criterion RELAX is reduced to,

RELAX = .05 - (.05/NLAX)

and computational procedure goes to step (5).

8. Repetition of steps (5) through (7) except for temperature convergence criterion.

Temperatures have converged if,

$$|T_{i,k+1} - T_{i,k}|_{max} \le \text{RELAX} (= .05 - .05/\text{NLAX})$$

- 9. Assuming step (7) has been satisfied, $L\phi\phi$ PCT is checked against NL $\phi\phi$ P. If LOOPCT > NLOOP, the computation proceeds to step (12). If LOOPCT < NLOOP computation proceeds to step (10).
- 10. Reduce RELAX to .001.
- 11. Check on temperature convergence. If $|T_{i,k+1} - T_{i,k}| \leq \text{RELAX} (= .001)$ go to step (12). $|T_{i,k+1} - T_{i,k}| > \text{RELAX} (= .001)$, LAXFAC is reduced to LAXFAC = NLØØP - LØØPCT, and steps (5) through (11) are repeated.
- 12. Compute system energy balance and store in control constant ENGBAL.
- 13. If LØØPCT > LAXFAC (original user input value), go to step (15)
- 14. If LOOPCT < LAXFAC (original user input value), ENGBAL is checked against BALENG. If ENGBAL < BALENG, go to step (14) If ENGBAL > BALENG, RELAX is set to, RELAX = .001/5, and steps (5) through (14) are repeated with the new RELAX values.
- 15. Print ENGBAL; call VARIABLES 2; call ØUTCAL; check if TIMEND = TIMEØ.



6.5-3. Functional Flow Chart for CINDSM

6 - 118

REFERENCES

- J. D. Gaski and D. R. Lewis, "Chrysler Improved Numerical Differencing Analyzer," TN-AP-66-15, April 30, 1966, Chrysler Corporation Space Division, New Orleans, Louisiana.
- D. R. Lewis, J. D. Gaski and L. R. Thompson, "Chrysler Improved Numerical Differencing Analyzer for 3rd Generation Computers," TN-AP-67-287, October 20, 1967, Chrysler Corporation Space Division, New Orleans, Louisiana.
- J. D. Gaski, L. C. Fink and T. Ishimoto, "Systems Improved Numerical Differencing Analyzer, Users Manual," 11027-6003-RO-00, September 1970, TRW Systems Group.
- 4. J. P. Smith, "SINDA User's Manual," 14690-H001-RO-00, June 1971, TRW Systems Group.
- 5. TRW Thermal Analyzer Program, Users Manual, Part II, Program Description, CDRC Heat Transfer Group, Program No. 3123.23-31, May 22, 1967.
- T. Ishimoto, J. D. Gaski and L. C. Fink, "Final Report, Development of Digital Computer Program for Thermal Network Correction; Phase II -Program Development, Phase III - Demonstration/Application," September 1970, 11027-6602-RO-00, TRW Systems Group.
- 7. W. J. Karplus, <u>Analog Simulation</u>, <u>Solution of Field Problems</u>, McGraw-Hill Book Company, 1958.
- J. Crank and P. Nicolson, "A Practical Method for Numerical Evaluation of Solutions of Partial Differential Equations of the Heat Conduction Type," Proc. Camb. Phil Soc., vol. 43, 1947, pp. 50-54.
- E. C. Dufort and S. P. Frankel, "Stability Conditions in the Numerical Treatment of Parabolic Differential Equations," <u>Math. Tables and</u> Other Aids to Computation, vol. 7, pp. 135, 1953.
- 10. J. Douglas, Jr., "On the Numerical Integration of $\frac{\partial^2 u}{\partial x^2} + \frac{\partial^2 u}{\partial y^2} = \frac{\partial u}{\partial t}$ by Implicit Methods," <u>Journal of the Society of Industrial and</u> Applied Mathematics, 3:42-65, March 1955.
- J. Douglas, Jr., and D. W. Peaceman, "Numerical Solution of Two-Dimensional Heat Flow Problems," <u>A.I.Ch.E. Journal</u>, vol. 1, No. 4, 1955, pp. 505-512.
- 12. R. D. Richtmyer, Difference Methods for Initial-Value Problems, Interscience Publishers, Inc., 1957.

- J. B. Diaz, R. F. Clippinger, B. Friedman, E. Isaacson and R. Richtmyer, "6. Partial Differential Equations," ch. 14, Numerical Analysis, <u>Handbook of Automation, Computation and</u> <u>Control</u>, vol. 1, Control Fundamentals, 1958.
- 14. G. M. Dusinberre, <u>Heat Transfer Calculations by Finite Differences</u>, International Textbook Company, 1961.
- 15. G. R. Gaumer, "The Stability of Three Finite Difference Methods of Solving for Transient Temperatures," Fifth U. S. Navy Symposium on Aeroballistics, McDonnell Aircraft Corporation, 18 October 1961.
- 16. R. S. Varga, "On Higher Order Stable Implicit Methods for Solving Parabolic Partial Differential Equations," <u>Journal of Mathematics</u> <u>and Physics</u>, vol. 40, no. 3, October 1961.
- B. K. Larkin, "Some Finite Difference Methods for Problems in Transient Heat Flow," <u>A.I.Ch.E Preprint 16</u>, Seventh National Heat Transfer Conference, August 9 to 12, 1964, Cleveland, Ohio.
- 18. S. R. Allada and D. Quon, "A Stable Explicit Numerical Solution of the Conduction Equation for Multi-Dimensional Non-Homogeneous Media," <u>A.I.Ch.E. Preprint 27</u>, Eighth National Heat Transfer Conference, August 8 to 11, 1965, Los Angeles, California.
- R. S. Barker and H. M. Stephens, "Investigation of Numerical Methods for Determining Transient Temperatures in Flowing Fluids," A.I.A.A. Fourth Aerospace Sciences Meeting, 27 June 1965, Los Angeles, California, Douglas Paper No. 3823.
- 20. H. Z. Barakat and J. A. Clark, "On the Solution of the Diffusion Equations by Numerical Methods," <u>Journal of Heat Transfer</u>, November 1966.
- 21. S. Gross, "An Improved Finite-Difference Method for Heat Transfer Calculation," J. of Spacecraft, vol. 4, no. 4, pp. 538, 1967.
- 22. B. P. Jones, "A Study of Several Numerical Methods for Solving a Particular System of Ordinary Differential Equations," TMX-53121, NASA/MSFC.
- 23. G. L. Kusic and A. Lavi, "Stability of Difference Methods for Initial-Value Type Partial Differential Equations," J. of Computational Physics 3, pp. 358-378, 1969.
- 24. J. H. Smith, "Survey of Three-Dimensional Finite Difference Forms of Heat Equation," SC-M-70-83, Sandia Laboratories, March 1970.
- O. C. Zienkiewicz, <u>The Finite Element Method in Structural and</u> <u>Continuum Mechanics</u>, McGraw-Hill Publishing Company, pp. 148-169, 1967.

- 26. W. Visser, "A Finite Element Method for the Determination of Nonstationary Temperature Distribution and Thermal Deformations," <u>Matrix Methods in Structural Mechanics</u>, Proceedings of the Conference held at Wright-Patterson AFB, Ohio, October 26-28, 1965, AFFDL-TR-66-80, November 1966, pp. 925-943.
- 27. A. F. Emery and W. W. Carson, "Evaluation of Use of the Finite Element Method in Computation of Temperature," A.S.M.E. 69-WA/HT-38, A.S.M.E. Winter Annual Meeting, November 16-20, 1969, Los Angeles, California.
- 28. E. C. Lemmon and H. S. Heaton, "Accuracy, Stability and Oscillation Characteristics of Finite Element Method for Solving Heat Conduction Equation," 69-WA/HT-35, A.S.M.E. Winter Annual Meeting, November 16-20, 1969, Los Angeles, California.
- P. D. Richardson and Y. M. Shum, "Use of Finite Element Methods in Solution of Transient Heat Conduction Problems," 69-WA/HT-36, A.S.M.E. Winter Annual Meeting, November 16-20, 1969, Los Angeles, California.
- 30. H. S. Carslaw and J. C. Jaeger, "Conduction of Heat in Solids," Oxford University Press, 1959.
- 31. "Temperatures in Structures," NAVORD Report 5558,
 - Part 1. General discussion and one-dimension, one layer problem; analytic solutions, C. J. Thorne, 10 July 1957.
 - Part 2. One dimension, two- and three-layer plate problems; analytic solutions, C. J. Thorne, 28 March 1958.
 - Part 3. Two- and three-layer cylindrical shields; one-space variable, linear, 0. H. Strand and C. J. Thorne, 20 August 1958.
 - Part 4. Two- and three-layer spherical shell segments, one-space variable, linear,
 R. E. Smithson and C. J. Thorne, 4 September 1958.
 - Part 5. One layer, two-space variables, linear problems, L. C. Barrett, H. J. Fletcher and C. J. Thorne, 27 May 1958.
- S. S. Abarbanel, "On Some Problems in Radiation Heat Transfer," O.S.R. Technical Report No. 59-531, Massachusetts Institute of of Technology, April 1959.
- 33. P. L. Chambre', "Nonlinear Heat Transfer Problem," J. of Applied Physics, vol. 20, no. 11, pp. 1683-1688, November 1959.

ж.,

- 34. D. C. Stickler and L. S. Han, "Three Linear Approximations to the Stefan-Boltzmann Radiation Problem for the Plane Slab," RADC-TDR-62-557, Supl. 1, Ohio State University, July 1962.
- 35. R. H. MacNeal, "An Asymmetrical Finite Difference Network," Quarterly Applied Mathematics, vol. 11, pp. 295-310, 1953.
- 36. T. Ishimoto and J. T. Bevans, "Temperature Variance in Spacecraft Thermal Analysis," TRW Systems Group, A.I.A.A. Paper No. 68-62, <u>J. of Spacecraft</u>, vol. 5, no. 11, pp. 1372-1376, 1968.
- 37. B. Bussell, "Properties of a Variable Structure Computer System in the Solution of Parabolic Partial Differential Equations," Report No. 62-46, September 1962, University of California, Los Angeles, California.
- 38. P. F. Strong and A. G. Emslie, "The Method of Zones for the Calculation of Temperature Distribution," A.S.M.E. 65-WA/HT-47, A.S.M.E. Winter Annual Meeting, November 7-11, 1965, Chicago, Illinois.
- 39. A. L. Edwards, "TRUMP: A Computer Program for Transient and Steady State Temperature Distributions in Multidimensional Systems," TID-4500, UC-32, Lawrence Radiation Laboratory, University of California, Livermore, May 1, 1968.
- 40. A. K. Oppenheim, "Radiation Analysis by the Network Method," Trans. A.S.M.E., 78, pp. 725-735, 1956.
- 41. G. L. Polyak, "Radiative Transfer Between Surfaces of Arbitrary Spatial Distribution of Reflection," Translation TT-9, School of Aeronautical and Engineering Sciences, Purdue Univeristy, Lafayette, Indiana; translated from Konvek tionyi i Luchisty i Teploobmen, pp. 118-132, Akad. Nauk. SSSR, Moskva (1960).
- 42. E. M. Sparrow, E. R. G. Eckert and V. K. Jonsson, "An Enclosure Theory for Radiative Exchange Between Specularly and Diffusively Reflecting Surface," J. of Heat Transfer, Trans. A.S.M.E., 84, Series C, pp. 924-300, 1962.
- 43. R. P. Bobco, "Gray Surface Radiation Between a Differential Area and Infinite Plane: Nonuniform Local Heat-Flux," <u>ARS Journal</u>, August 1962.
- 44. E. M. Sparrow, "Heat Radiation Between Simply-Arranged Surfaces Having Different Temperatures and Emissivities," <u>A.I.Ch.E. Journal</u>, March 1962.
- 45. J. T. Bevans and D. K. Edwards, "Radiation Exchange in an Enclosure with Directional Wall Properties," Trans. A.S.M.E., <u>J. Heat Transfer</u> <u>87C</u>, pp. 388-396, 1965.

7 - 4

46. J. T. Bevans, T. Ishimoto, B. R. Loya and E. E. Luedke, "Prediction of Space Vehicle Thermal Characteristics," AFFDL-TR-65-139, TRW Systems, October 1965.

- 47. E. M. Sparrow and R. D. Cess, <u>Radiation Heat Transfer</u>, Brooks/Cole Publishing Company, 1966.
- 48. R. Viskanta, J. R. Schornhorst and J. S. Toor, "Analysis and Experiment of Radiant Heat Exchange Between Simply Arranged Surfaces," AFFDL-TR-67-94, Purdue University, June 1967.
- 49. O. W. Clausen and T. Ishimoto, "Thermal Test of a Model Space Vehicle, Part I. Description and Thermal Analysis Model, AFFDL-TR-67-42, Part I, TRW Systems Group, June 1967.
- 50. R. G. Hering and R. P. Bobco, "Local Radiant Flux and Temperature in Semigray Nondiffuse Enclosures," A.I.A.A. No. 68-60, A.I.A.A. 6th Aerospace Sciences Meeting, New York, New York, January 22-24, 1968.
- 51. T. Ishimoto and J. T. Bevans, "Method of Evaluating F Variance for Thermal Variance Analysis," <u>A.I.A.A. Journal</u>, vol. 6, no. 6, p. 1178-1180, June 1968.
- 52. J. R. Howell and R. Siegel, <u>Thermal Radiation Heat Transfer</u>, <u>Volume II, Radiation Exchange Between Surfaces and in Enclosures</u>, NASA SP-164, 169.
- R. P. Bobco, "Radiation Heat Transfer in Semigray Enclosures with Specularly and Diffusely Reflecting Surfaces, <u>J. Heat Transfer</u>, Trans. A.S.H.E., Ser. C, vol. 86, no. 1, pp. 123-130, February 1964.
- 54. A. F. Sarofim and H. C. Hottel, "Radiative Exchange Among Non-Lambert Surfaces," J. Heat Transfer, Trans, A.S.M.E., Ser. C, vol. 88, no. 1, February 1966.
- 55. T. Ishimoto and J. T. Bevans, "Method of Evaluating Script F for Radiant Exchange Within an Enclosure," <u>A.I.A.A. Journal</u>, vol. 1, no. 6, pp. 1428-1429, June 1963.
- 56. K. A. Toups, "A General Computer Program for the Determination of Radiant Interchange Configurations and Form Factors - CONFAC II," North American Aviation, Inc., Space and Information Systems Division, S.I.D. Report 65-1043-2, October 1965
- 57. R. P. Bobco, "Analytical Determination of Radiation Interchange Factors," SSD 90190R, Hughes Aircraft Company, June 1969.
- 58. R. J. McGrath and F. L. Egendorf, "A Computer Program for Determination of Radiation Interchange Factors - RADFAC, User's Manual," SSD 90191R, Hughes Aircraft Company, June 1969.
- "Measurement of Radiation Interchange Factors, Final Report and Facility Design Description," Thermophysics Department Staff, Project Manager, R. P. Bobco, SSD 00577R, NASA Contract NAS 9-10271, Nov. 1970.

.

		Page
CNFRWD		A-2
CNDRDL	• • • • • • • • • • • • • • • • • • • •	A-13
CNFAST	•••••	A-24
CNEXPN	•••••••	A-33
CNDUFR	•••••	A-44
CNQUIK	• • • • • • • • • • • • • • • • • • • •	A55

CNFRWD . CNF PWD

.

))

.

GIN FOR,* CNFRWD,CNFRWD Univac 1108 Fortran V Athena Version 131K-10D Created on 20 Aug 70 This compilation was done on 09 Jun 70 at 14:00:37

SUBROUTINE CNFRWD ENTRY POINT 003522

STORAGE USED (BLOCK, NAME, LENGTH)

	003535	TEMP 000072	VAR 000050	000000	000000	00000	000001	000001	100001	000001	000001	000001	100001	00001	100000	00003	010000	
	+CODE	+C0115T+	*SINPLE	*ARRAYS	*BLANK	TITLE	TEMP	CAP	SOURCE	COND	PC1	PC2	KONST	ARKAY	FIXCON	XSPACE	DIMENS	
	1000	0000	0005	000	0005	0006	0007	0100	0011	0012	0013	0014	0015	0016	0017	0.020	0021	28 - 2 - 2
ъ																		

EXTERNAL REFERENCES (BLOCK, NAME)

and a second and

	VARBLI	MATOTO	PLYAWM	DZD1WM	VARULZ	OUTCAL	EXIT	NERK25	NWDCI &	N1025	NER105
1	0022	0023	0024	0025	0026	0027	0030	0031	0032	0033	0034

STORAGE ASSIGNMENT FOR VARIABLES (BLOCK, TYPE, RELATIVE LOCATION, NAME)

	0.002 R 000021 C1	0012 R 000000 6	0002 I 000004 I	0002 I 000013 J2	0302 I 000042 LAX	0021 I 000006 L501	0021 1 000004 NCT	0002 I 000001 NNC	0020 I 000001 NTH	0.02 R 000035 000T	0007 R 000000 T	0002 R 000010 TSUM	0001 000125 101
	0017 R 000000 CON	0002 R 000040 DN	0006 R 000000 H	0002 I 000012 JI	0002 I 000017 LA	0002 I 000020 LK	0021 I 000005 NAT	0021 I 000001 NNA	0014 I 000000 NSQ2	0011 R 000000 G	0002 R 000047 SUMCV	0002 R 000006 TSTEP	0015 A 000000 XK
	0002 R 000015 CKM	0002 R 000037 DELTA	0002 R 000031 62	0002 I 000044 JJ2	0002 I 000045 L	0002 I 000025 LG	0002 I 000005 L1	0002 I 000003 NLA	102 I 000000 I 200	0002 R 000000 PASS	0002 R 000046 SUMC	0002 R 000007 TPRINT	0020 R 000002 X
ATTACHED TOPOCAL TILLE N	0010 R 000000 C	0002 R 000041 DD	0002 R 000030 G1	0002 I 000043 JJI	0017 1 000000 KON	0002 I 000036 LEA	0002 I 000026 LTA	0021 1 000003 NGT	0U21 1 000002 NNT	0020 00002 NX	0002 R 000024 02	0002 R 000027 TM	0002 H 000033 T2
NOTAL PUBLICIUM	0016 R 000000 A	0002 R 000022 C2	0002 R 000034 GV	0002 I 000002 IE	0015 I 000000 K	0002 I 000011 LE	0021 1 000007 LSQ2	0020 I 000000 NUIM	0021 I 000000 NND	0002 I 000016 NIYPF	0002 R 000023 G1	0002 R 000014 TCGM	0002 R 000032 T1

. Parios de la como de la

21 FEB 71

)

A - 2

CNFRWD, CNFRWD

10121

1032L	1151	1401	180L	1 6661	20106	20301	2045L	2126	30	3015L	3032L	3050L	3999L	4015L	4030L	4998L	5005L	5020L	5037L	5736	6026	8975	995L		
000534	002123	003246	003360	000670	001262	001405	001520	000225	000161	002616	002743	003057	003206	000765	001042	001137	002272	002335	002446	002177	002220	000016	003442		
1000	0001	0001	0001	0001	1000	0001	0001	0001	0001	1000	1000	0001	0001	1000	0001	0001	0001	0001	0001	0001	0001	0000	1000		•
13 1030L	71 105L	00 135L	27 175L	65 1998L	42 2007L	45 2025L	61 2042L	23 2065L	55 2999L	13 3010L	36 3030L	51 3045L	60 3998L	46 4012L	34 4025L	37 45L	46 5000L	20 5017L	32 5035L	15 55L	00 6000L	11 886F	44 95L	73 999L	
0002	0020	0025	0033	0000	0012	0013	0014	0016	0016	0026	0027	0030	0031	2000	0010	0005	0011	0.023	0024	0017	0025	0000	0020	0034	
0001	0001	1000	0001	0001	0001	0001	0001	1000	0001	0001	1000	0001	0001	0001	0001	0001	0001	0001	000	1000	0001	0000	0001	0001	
1 1025L	5 1045L	1 130L	5 165L	5 195L	5 2005L	1 2020L	0 2040L	2060L	1899AL	5 3007L	5 3025L	2042L	1 3065L	5 4010L	14022L	1 4040L	50L	5015L	1 5030L	5356) 60L) 885F	200	1 998L	
00046	000643	002150	003316	003423	001235	001337	001440	001617	001627	002573	002676	003012	003154	000745	00101	001131	001146	002317	002374	002075	001720	000000	002025	194600	
0001	0001	0001	1000	0001	0001	0001	0001	1000	0001	0001	0001	0001	0001	0001	0001	0001	1000	0001	0001	1000	0001	0000	0001	0001	
000427 10201	000600 1040L	002142 125L	000134 156	003420 190L	000673 2000L	001303 2017L	001435 2035L	001571 2055L	000146 25L	002566 3005L	002670 3020L	002771 3040L	003150 3060L	000740 4005L	001003 4020L	001114 4037L	000113 5L	002300 5012L	002366 5025L	002463 5040L	002475 5999L	001776 85L	000030 889F	003456 997L	
0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	0000	0001	•
000421 1015	000572 1035L	002134 120L	003251 145L	003371 185L	000206 2006	001265 2015	001412 2032L	001526 2050L	000256 2326	001660 3000L	002634 3017L	002766 3035	003122 3055L	003211 4000L	000766 4017L	001100 4035L	001143 4999L	002277 5010L	002351 5022L	002004 5046	002471 5998L	001767 65L	000023 8685	003450 996L	
0001	1000	1000	1000	1000	0001	0001	0001	1000	1000	0001	0001	1000	0001	1000	1000	0001	1000	1000	0001	1000	1000	1000	0000	1000	
	0001 000421 10151 0001 000427 10204 0001 000467 1025L 0001 000513 1030L 0301 000534 1032L	0001 000421 1015L 0001 000427 1020L 0001 000467 1025L 0001 000513 1030L 0301 000534 1032L 0001 000572 1035L 0001 000600 1040L 0001 000643 1045L 0001 002071 105L 0001 002123 115L	0001 000421 1015L 0001 000427 1020L 0001 000467 1025L 0001 000513 1030L 0301 000534 1032L 0001 000572 1035L 0001 000600 1040L 0001 000643 1045L 0001 002071 105L 0001 002123 115L 0001 002134 120L 0001 002142 125L 0001 002150 130L 0001 002500 135L 0001 003246 140L	0001 000421 1015L 0001 000467 1025L 0001 000134 1032L 0001 0002134 1035L 0001 000450 1045L 0001 002134 1032L 0001 0002134 125L 0001 002134 02242 105L 0001 002234 103246 140L 0001 002234 145L 0001 002347 105L 0001 002244 140L 0001 002234 155L 0001 002346 140L 0001 003246 140L 0001 0022341 145L 0001 0033271 175L 0001 003560 180L	0001 000421 1015L 0001 000467 1025L 0001 000534 1032L 0001 000572 1035L 0001 0001467 1025L 0001 002071 105L 0011 002123 115L 0001 002134 125L 0001 002142 125L 0001 002501 135L 0001 003243 140L 0001 002134 125L 0001 002160 135L 0001 003246 140L 0001 0023251 145L 0001 0033271 155L 0001 003346 140L 0001 003420 199L 0001 0033271 175L 0001 000540 199L	0001 000421 1015L 0001 000467 1025L 0001 000534 1032L 0001 000572 1035L 0001 000643 1045L 0001 0022123 115L 0001 0002734 1035L 0001 002643 1045L 0001 0022123 115L 0001 002134 125L 0001 002501 135L 0001 003244 140L 0001 0023251 155L 0001 0033271 155L 0001 003264 140L 0001 0023151 155L 0001 003327 175L 0001 00570 1994L 0001 003371 155L 0001 003371 155L 0001 00556 180L 0001 003371 155L 0001 003377 175L 0001 000556 189L 0001 002371 155L 0001 0012423 199L 0011 0012423 199L 00	0001 000421 1015L 0001 000467 1025L 0001 000534 1035L 0001 000534 1032L 1030L 000534 1032L 1030L 0001 002123 115L 0001 002572 1035L 0001 002572 1035L 0001 002572 1035L 0001 002572 1035L 0001 003254 140L 0001 002154 125L 0001 002150 130L 0001 003254 140L 0001 002551 145L 0001 002154 155L 0001 003327 175L 0001 003546 140L 0001 003551 145L 0001 003134 155L 0001 003556 199L 0001 003551 165L 0001 001242 195L 0001 000570 1999L 0001 001242 2010L 0011 012355 2027L 0001 001242 2031L 0001 001242	0001 000421 1015L 0001 000467 1025L 0001 000534 1035L 0011 000534 1035L 0001 000572 1035L 0001 000504 1035L 0001 002234 1035L 0001 002234 1035L 0001 002234 1032L 1011 0022123 115L 0001 002134 120L 0001 002144 120L 0001 0022451 105L 0001 0022451 105L 0001 0022451 107L 0001 0022451 107L 0001 0022451 107L 0001 003246 140L 0001 002144 15C 0001 0023423 155L 0001 003246 180L 0001 003246 140L 0001 0023423 155L 0001 0032423 195L 0001 0031 003665 199L 0001 003246 190L 0001 001245 0001 00124 0001 00140 0014	0001 000421 1015L 0001 000467 1025L 0001 000534 1032L 0001 000517 1015L 0001 000504 1032L 0001 002234 1032L 0001 0002134 120L 0001 00214 15L 0001 002241 105L 0001 002244 105L 0001 002234 1035L 0001 002346 15L 0001 002244 105L 0001 003244 107L 0001 003244 107L 0001 003246 1001 003246 107L 0001 003246 1001 003246 1001 003246 1001 0001 003246 1001 0001 003246 1001 0001 003246 1001 0001 001265 2001 0001 001265 2001 0001 0001 0001	0001 000421 10151 0001 000467 10251 0001 000534 10351 0001 000534 10351 0001 000534 10351 0001 000234 10351 0001 0022071 10551 0001 003245 1401 0001 002143 1551 0001 002142 1551 0001 003260 1361 0001 003245 1401 0001 002142 1551 0001 002142 1551 0001 003267 1364 1401 0001 002344 151 0001 0023423 1551 0001 003567 1391 0001 003420 1901 0013423 1951 0001 003567 1991 0001 001347 1091 001337 10551 0001 001242 1011 0001 00145 2001 001242 0001 001455 20311 0001 001412 20351 0001 001242	0001 000421 10151 0001 000467 10251 0001 000534 10351 0001 000534 10351 0001 000534 10351 0001 000534 10351 0001 000534 10324 10401 0001 002504 10310 0001 0022423 1151 0001 003244 1401 0001 003244 1401 0001 003244 1401 0001 003244 1401 0001 003244 1401 0001 003244 1401 0001 003247 1751 0001 003246 1401 0001 003246 1401 0001 003247 1751 0001 003246 1401 0001 003247 1751 0001 003246 1401 0001 003246 1401 0001 003246 1401 0001 003246 1401 00101 003246 1401 00101 003246 1401 00101 003246 1401 00101 003446 1401 00101	0001 000421 10151 0001 000467 10251 0001 000534 10351 0001 000534 10351 0001 000534 10351 0001 000534 10321 10311 0001 002534 10321 1151 0001 0023514 1551 0001 002501 1351 0001 003244 1401 0001 002351 1451 0001 002542 1351 0001 003244 1401 0001 002344 1551 0001 003251 1551 0001 003264 1401 0001 002342 1551 0001 003251 1551 0001 003264 1401 0001 003371 1551 0001 0032423 1551 0001 003264 1401 0001 002371 1551 0001 001242 1571 0011 001242 1561 0101 001262 1561 1016 0011 001245	0001 000421 10151 0001 000467 10254 10351 0001 002213 1151 0001 0002134 1201 0001 002145 1301 0001324 103246 0001 0022134 1201 0001 002145 1351 0001 002244 1401 0001 0022134 151 0001 002346 154 0001 003244 1401 0001 002134 151 0001 002371 1551 0001 003244 1401 0001 002144 154 0001 001347 1051 0001 003247 1751 0001 003246 1401 0001 001347 1901 001342 1951 0001 001242 1971 0001 001462 2011 0001 001247 2051 0001 001347 2061 0001 001462 2011 001462 2011 001462 2011 001462 2011 0	0001 000421 10151 0001 000467 10251 0001 000534 10351 0001 000534 10351 0001 000534 10351 0001 0002337 10351 0001 002234 103246 1001 0002334 103246 1001 0002345 103246 1001 0002337 103246 1001 00023367 103246 1001 0003234 103246 1001 0003244 1051 0001 0033767 103246 1001 0033267 103246 1001 0033267 103246 1001 0033267 103246 1001 0033267 103246 1001 0033267 103246 1001 0033267 10367 0001 0033267 10367 0001 0033267 10367 0001 0033267 10367 0001 003246 1001 0033267 10367 0001 003266 1001 003266 1001 003266 1001 0011605 0001 0011605 0001 00011 0011262	0001 000421 1015 0001 000467 10251 1031 000534 10324 10324 10324 10324 10324 10324 10324 10324 10324 10324 10324 10324 10324 103244 1040 0001 003247 1055 0001 003247 1055 0001 003247 1046 0001 003247 1046 0001 003247 1046 0001 003247 1046 0001 003247 1046 0001 003247 1046 0001 003247 1046 0001 003247 1046 0001 003247 1046 0001 003247 1046 10670 10994 0001 103246 1401 0001 001347 1001 001347 1001 001347 1001 00147 1001 00147 1001 00147 1040 1040 1040 1040 1040 1040 1040 1040 1040 1040 1040 1040 1040	0001 000421 1015 0001 000467 10251 1015 0001 0005234 10324 1015 0001 000273 10351 0001 000257 10351 0001 002247 10551 0001 002244 1401 0001 002351 1451 0001 002351 1551 0001 003246 1401 0001 002351 1451 0001 0023716 1551 0001 003254 1401 0001 002342 1551 0001 003571 1551 0001 003257 1751 0001 003767 2041 0001 001412 20501 0011 001242 2071 0001 001245 2031 0001 001412 20511 0001 001447 2051 0001 001452 2061 0001 001452 2061 0001 001452 2061 0001 001452 2051 0001 001452 2051 0011 <t< th=""><th>0001 000421 1010 000427 1020 000457 10251 1031 10324</th><th>0001 000421 1015 0001 000572 10351 10351 10351 10351 10351 10351 10351 10351 10351 10351 10351 10351 10351 10351 10351 10551 0001 000157 1051 0001 002153 1151 0001 002153 1151 0001 002525 1151 0001 002525 1151 0001 002525 1151 0001 002525 1151 0001 002525 1151 0001 001262 103546 1401 003546 1401 00356 10367 13994 00356 1301 000157 10357 10351</th></t<> <th>0001 000421 1015 0001 000534 10351 0001 000534 10351 0001 000534 10351 1031 0001 000534 1031 0013 <th013< th=""> <th013< th=""> 0013</th013<></th013<></th> <th>0001 000421 10151 0001 000500 10151 0011 000530 10311 0001 0001372 10351 0001 000137 10351 0001 000134 1151 0001 00134 1251 0001 00134 1251 0001 001242 1041 0011 001245 1011 001274 14101 0001 0013451 1551 0001 001347 1551 0001 001242 2011 001247 10356 1491 0001 001245 2051 0001 001242 2071 0001 001242 2011 001267 20451 0001 001267 20451 0001 001267 20451 0001 001267 20451 0001 001267 20451 0001 001267 20451 0001 001267 20451 0001 001267 20451 0001 001267 20451 0001 001267 20451 0001 001267 20451 0</th> <th>0001 0000421 10151 0001 000539 10351 0001 000539 10351 0001 000539 10351 0001 000536 1401 0001 001253 11351 0001 0012531 1151 0001 0012535 11351 0001 001256 11351 0001 001256 11351 0001 001356 11491 0001 001356 11491 0001 001356 11491 0001 001356 11491 0001 001265 21401 0001 001245 2011 0011 001265 21401 0001 001245 2011 0011 001265 21401 0001 001245 2011 0011 001265 21401 0001 001245 2011 0011 001265 21251 0001 001245 2011 0011 001265 2126 0001 001435 20261 0001 001465 20251 0001 001465 20251 2011 0001 0011615 2126 0001<</th> <th>0001 000421 10201 0001 000571 1051 0001 000571 1051 0001 000571 1051 0001 000571 1051 0001 000571 1051 0001 000571 1051 0001 000571 1051 0001 000571 1051 0001 000571 1051 0001 000574 1471 0001 000574 1471 0001 000574 1471 0001 000574 1471 0001 000574 1471 0001 000574 1471 0001 001245 2017 0001 001245 2017 0001 001245 2017 0001 001245 2017 0001 001245 2017 0001 001245 2017 0001 001245 2017 0001 001245 2017 0001 001245 2017 0001 001245 2017 0001 001245 2017 0001 001245 2017 0001 001245 2017 0001 001245 2017</th> <th>0001 000421 1025L 0001 000513 1035L 0001 000513 1035L 0001 000513 1035L 0001 000513 115L 0001 0011 001245 213C 0001 0011405 203C 0001 0011405 203C 0001 0011405 203C 0001 00115C 203C 0001 0011613 201C 00011</th> <th>0001 000421 1055 0001 000457 1055 0001 000553 1055 0001 000553 1055 0001 0001534 1055 0001 0001534 1055 0001 0001534 1155 0001 0001534 1155 0001 0001534 1155 0001 0001534 1155 0001 0001534 1155 0001 0001534 1155 0001 0001534 1156 0001 001534 1156 0001 001534 1156 0001 001534 1156 0001 001534 1156 0001 0011405 2014 0001 0011405 20131 1156 20111 1156 20111 1156 20111 1156 20111 1156 20111 1156 20111 1156 20111 1156 20111 1156 20111 1156 20111 1156 20111 1156 20111 1156 20111 1156 20111 1156 201111 1156 20111 <t< th=""><th>0001 000421 1055 0001 000513 1035 0011 000513 1035 1035 0011 000513 1135 0011 000513 1135 0011 001135 1155 0011 001233 1155 0011 001233 1155 0011 001234 1261 0011 001242 1251 0011 001235 1155 0011 001242 1261 0011 001242 1261 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 101242 10011 001242 1961 0011 101242 10011 001242 10101 101242 10101 101242 10101 101242 10101 <</th></t<></th>	0001 000421 1010 000427 1020 000457 10251 1031 10324	0001 000421 1015 0001 000572 10351 10351 10351 10351 10351 10351 10351 10351 10351 10351 10351 10351 10351 10351 10351 10551 0001 000157 1051 0001 002153 1151 0001 002153 1151 0001 002525 1151 0001 002525 1151 0001 002525 1151 0001 002525 1151 0001 002525 1151 0001 001262 103546 1401 003546 1401 00356 10367 13994 00356 1301 000157 10357 10351	0001 000421 1015 0001 000534 10351 0001 000534 10351 0001 000534 10351 1031 0001 000534 1031 0013 <th013< th=""> <th013< th=""> 0013</th013<></th013<>	0001 000421 10151 0001 000500 10151 0011 000530 10311 0001 0001372 10351 0001 000137 10351 0001 000134 1151 0001 00134 1251 0001 00134 1251 0001 001242 1041 0011 001245 1011 001274 14101 0001 0013451 1551 0001 001347 1551 0001 001242 2011 001247 10356 1491 0001 001245 2051 0001 001242 2071 0001 001242 2011 001267 20451 0001 001267 20451 0001 001267 20451 0001 001267 20451 0001 001267 20451 0001 001267 20451 0001 001267 20451 0001 001267 20451 0001 001267 20451 0001 001267 20451 0001 001267 20451 0	0001 0000421 10151 0001 000539 10351 0001 000539 10351 0001 000539 10351 0001 000536 1401 0001 001253 11351 0001 0012531 1151 0001 0012535 11351 0001 001256 11351 0001 001256 11351 0001 001356 11491 0001 001356 11491 0001 001356 11491 0001 001356 11491 0001 001265 21401 0001 001245 2011 0011 001265 21401 0001 001245 2011 0011 001265 21401 0001 001245 2011 0011 001265 21401 0001 001245 2011 0011 001265 21251 0001 001245 2011 0011 001265 2126 0001 001435 20261 0001 001465 20251 0001 001465 20251 2011 0001 0011615 2126 0001<	0001 000421 10201 0001 000571 1051 0001 000571 1051 0001 000571 1051 0001 000571 1051 0001 000571 1051 0001 000571 1051 0001 000571 1051 0001 000571 1051 0001 000571 1051 0001 000574 1471 0001 000574 1471 0001 000574 1471 0001 000574 1471 0001 000574 1471 0001 000574 1471 0001 001245 2017 0001 001245 2017 0001 001245 2017 0001 001245 2017 0001 001245 2017 0001 001245 2017 0001 001245 2017 0001 001245 2017 0001 001245 2017 0001 001245 2017 0001 001245 2017 0001 001245 2017 0001 001245 2017 0001 001245 2017	0001 000421 1025L 0001 000513 1035L 0001 000513 1035L 0001 000513 1035L 0001 000513 115L 0001 0011 001245 213C 0001 0011405 203C 0001 0011405 203C 0001 0011405 203C 0001 00115C 203C 0001 0011613 201C 00011	0001 000421 1055 0001 000457 1055 0001 000553 1055 0001 000553 1055 0001 0001534 1055 0001 0001534 1055 0001 0001534 1155 0001 0001534 1155 0001 0001534 1155 0001 0001534 1155 0001 0001534 1155 0001 0001534 1155 0001 0001534 1156 0001 001534 1156 0001 001534 1156 0001 001534 1156 0001 001534 1156 0001 0011405 2014 0001 0011405 20131 1156 20111 1156 20111 1156 20111 1156 20111 1156 20111 1156 20111 1156 20111 1156 20111 1156 20111 1156 20111 1156 20111 1156 20111 1156 20111 1156 20111 1156 201111 1156 20111 <t< th=""><th>0001 000421 1055 0001 000513 1035 0011 000513 1035 1035 0011 000513 1135 0011 000513 1135 0011 001135 1155 0011 001233 1155 0011 001233 1155 0011 001234 1261 0011 001242 1251 0011 001235 1155 0011 001242 1261 0011 001242 1261 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 101242 10011 001242 1961 0011 101242 10011 001242 10101 101242 10101 101242 10101 101242 10101 <</th></t<>	0001 000421 1055 0001 000513 1035 0011 000513 1035 1035 0011 000513 1135 0011 000513 1135 0011 001135 1155 0011 001233 1155 0011 001233 1155 0011 001234 1261 0011 001242 1251 0011 001235 1155 0011 001242 1261 0011 001242 1261 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 001242 1961 0011 101242 10011 001242 1961 0011 101242 10011 001242 10101 101242 10101 101242 10101 101242 10101 <

CNFRWD	
INE	
-	9
∍	
0	
Č.	
UB.	1
_vn	
	1
· · ·	

10100

.

EXPLICIT FORWARD DIFFERENCING EXECUTION SUBROUTINE FOR SINDA F-V THE SHORT PSEUDO-COMPUTE SEQUENCE IS REQUIRED 00 *

00103 00104 00105 00105

00107 01100

INCLUDE COMM.LIST COMMON /TITLE/H(1) /TEMP/T(1) /CAP/C(1) /SOURCE/Q(1) /COND/G(1) COMMON /PC1/NS01(1) /PC2/NS02(1) /KONST/K(1) /ARRAY/A(1) COMMON /FIXCON/KON(1) /XSPACE/NDIM.NTH.X(1) #

ž

#

#

•

COMMON /DIMENS/ NND.NNA.NNT.NGT.NCT.NAT.LSQ1.LSQ2 DIMENSION CON(1),XK(1),NX(1) EQUIVALENCE (KON(1),CON(1)),(K(1),XK(1)),(X(1)),X(1))

EKO

.............................

ł

CNFRWD , CNFRWD

CONTROL CONSTANT 18 CONTAINS THE OUTPUT INTERVAL CC19 CONTAINS THE ARITHMETIC RELAXATION CRITERIA ALLOWED (ARLXCA) CC20 CONTAINS THE NUMBER OF RELAXATION LOOPS USED.INTEGER (LOOPCT) CC22 CONTAINS THE NUMBER OF RELAXATION LOOPS USED.INTEGER (LOOPCT) CC22 CONTAINS THE C/SG MAXIMUM CC23 CONTAINS THE C/SG RANGE ALLOWED CC26 CONTAINS THE C/SG RANGE CALCULATED CC25 CONTAINS THE C/SG RANGE CALCULATED CC26 CONTAINS THE C/SG RANGE CALCULATED CC27 CONTAINS THE C/SG RANGE CALCULATED CC27 CONTAINS THE DIFFUSION RELAXATION CRITERIA ALLOWED CC26 CONTAINS THE DIFFUSION RELAXATION CRITERIA ALLOWED CC26 CONTAINS THE DIFFUSION RELAXATION CRITERIA ALLOWED CC27 CONTAINS THE DIFFUSION RELAXATION CRITERIA ALLOWED CC26 CONTAINS THE DIFFUSION RELAXATION CRITERIA (LIFICA) CC26 CONTAINS THE DIFFUSION RELAXATION CRITERIA (LIFICA) CC26 CONTAINS THE PAGE COUNTER, INTEGER CC20 CONTAINS THE PAGE COUNTER, INTEGER CC23 CONTAINS THE PAGE COUNTER, INTEGER CC23 CONTAINS THE DIFFUSION RELAXITON CLIANEE (LIFICA) CC23 CONTAINS THE PAGE COUNTER, INTEGER CC23 CONTAINS THE DIFFUSION RELAXITON CLIANEE (CCCCC) CONTAINS THE DIFFUSION RELAXITON CLIANEE (ARLXCA) CC23 CONTAINS THE DIFFUSION RELAXITON CLIANEE (ARLXCA) CC23 CONTAINS THE PAGE COUNTER, INTEGER CONTAINS THE DIFFUSION RELAXITON CLIANEE (ARLXCA) CC23 CONTAINS THE DIFFUSION RELAXITON CLIANEE (ARLXCA) CC23 CONTAINS RELATIVE NODE NUMBER OF TAMPCC CC33 CONTAINS RELATIVE NODE NUMBER OF CONSTANTS (I-O-VTEST) CC49 IS THE OUNSILATION NUMBER OF ARLXCC CC39 CONTAINS RELATIVE NODE NUMBER OF CSGMIN CC39 CONTAINS RELATIVE NODE NUMBER OF CONSTANTS (I-O-VTEST) CC39 CON DOES THE TIME SUM PLUS THE TIME STEP EXCEED OUTPUT INTERVAL IF(TSUW+TSTEP-CON(10)) 25,30,20 DOES OLD TIME PLUS THE OUTPUT INTERVAL EXCEED THE STOP TIME IF(CON(13)+CON(18).LE.CON(3)) 60 TO 10 DONT EXCEED IT CHECK FOR EXTRA LOCATIONS FOR CALCULATED NODES 1 = NLA-NNC TTEP = CON(18) PRINT = CON(13) INITALIZE TIME SUM BETWEEN OUTPUT INTERVALS IS THE TIME STEP LARGER THAN ALLOWED IF (TSTEP.LE.CON(8)) 60 TO 15 $\begin{array}{l} \mbox{IF}(CON(4) . LT . 1 . 0) CON(4) = 1.0 \\ \mbox{IF}(KOW(5) . LE . 0) KON(5) = 1 \\ \mbox{IF}(CON(6) . LE . 0.) CON(6) = 1 . E + 8 \\ \mbox{IF}(CON(9) . LE . 0.) CON(6) = 1 . E + 8 \\ \mbox{IF}(CON(9) . LE . 0.) CON(9) = 1 . E + 8 \\ \mbox{IF}(CON(19) . LE . 0.) CON(11) = 1 . E + 8 \\ \mbox{IF}(CON(19) . LE . 0.) CON(11) = 1 . E + 8 \\ \mbox{IF}(CON(19) . LE . 0.) CON(19) = 1 . E + 8 \\ \mbox{IF}(CON(19) . E + 18 . E + 18 \\ \mbox{IF}(CON(19) . E + 18 . E + 18 \\ \mbox{IF}(CON(19) . E + 18 . E + 18 \\ \mbox{IF}(CON(19) . E + 18 . E + 18 \\ \mbox{IF}(CON(19) . E + 18 . E + 18 \\ \mbox{IF}(CON(19) . E + 18 . E + 18 \\ \mbox{IF}(CON(19) . E + 18 \\ \mbox{IF}(CON(19)$ CON(1B) = CON(3) - CON(13)LF(1.LT.0) GO TO 998 L1 = NND+1 TSTEP = CON(IB)-TSUM NDIM = NDIM-NNC DONT LACEED IT TSTEP = CON(8) NLA = NDIM NTH = NTH+NNC 1SUM = 0.0 IE = NTHEND 2 50 ŝ -1-P O U) Q U Q ့ပ U 3 * 8 5 26# **** 909 000 000 00144 00145 00145 00151 00152 00152 00155 00155 00155 00156 00157 00157 00157 00161 00162 00162 00165 0.0T44

A - 4

		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	UUJ 22* ***	* * * NN * NN *
	sno	TYPE		
N.ED	0CAT I	2) • N		
ALLO	TRA L	0.404		
STEP LOCAT	HE EX	+01 ·5		
TIME TIONS XTRA	T 0 T	2000 0.103		
N THE ITERA AND E	RES	0 T0 5,103 C(I))	C	
D 997 D 997 D 200 D 200 D 200 D 100 D 100	ERATU	•0) 6 0,102 (LK) •		
D LES 60 T ME + TSTE E HET N (13) LOCAT	5 TEMP To 10	()).E0 ()).E0 ()2))) ()2)) ()2))) ()2)) ()2)) ()2)))()())())())())())())())())())())()	2)) 22)) 4).XK	21)
P USE (21)) EW TI +TSUM N TIM URCE URCE	METIC C TO 4 60	ST ST NSS2 SS2 SS2 SS2 SS2 SS2 SS2 SS2 SS2 SS	7.41L S02(J NS02(),A(L LA)	502 (J N502 (LA)
DILLCONAL DI	ARITH 001111.000 1000000	11. NO	.17.1 2.14. M(T(I))*XK()*XK(•17•N 2•14•] *XK (
0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	+0 +1 +1 +1 +1 +1 +1 +1 +1 +1 +1 +1 +1 +1 +1 +1		VILLO I	JZ+1 FLD (5 FLD (2 XK (LK XK (LK
THE CONCLUSE THE C	C C C C C C C C C C C C C C C C C C C		COLLENE COLLEN	
5	ດ ດທີ ທີ່ ຊີຊີ	005	012 015 015 020	
υυυυ		,,	નું અનું તેનું છે.	
4 4 4 4 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	500 500 500 500 500 500 500 500 500 500	10000000000000000000000000000000000000	766***	76+ 76+
1173 1173 1173 1173 202 203 203	20700007700000000000000000000000000000	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	00000000000000000000000000000000000000	261 261 262
	46* 30 CONICY = TOTER 47* C IS THE TIME STEP USED LESS THAN THE TIME STEP ALLOWED 48* F(TSTEP-LT-CON(21)) 60 TO 997 49* C CALCULATE THE NEW TIME 50* CONI(1) = TPRINT+TSUM+TSTEP 51* C CONI(1) = TPRINT+TSUM+TSTEP 51* C CONI(1) = TONI(1) f(2) TO 997 52* CONI(1) = TONI(1) f(2) TO 5.0 STER ITERATIONS 53* C ZER0 OUT ALL SOURCE LOCATIONS AND EXTRA LOCATIONS 53* C ZER0 OUT ALL SOURCE LOCATIONS AND EXTRA LOCATIONS 54* D0 35 I = 1.NND 55* L = 1. SCURCE LOCATIONS AND EXTRA LOCATIONS 54* D0 35 I = 1.NND 55* C 2.00 56* C 0.00 57* 0.10 2.0	30 CUNIC: TIRE STEP USED LESS THAN THE TIME STEP ALLOWED 48 15 THE TIME STEP USED LESS THAN THE TIME STEP ALLOWED 49 C CALCULATE THE NEW TIME 51 C CALCULATE THE NEW TIME 51 C CONIL1) = TRINIFISED 52 CONIL1) = TRINIFISED 50 52 CONIL1) = CON(1)+CON(13))/2.0 EXTRA LOCATIONS 53 CONIL1) = CON(1)+CON(13))/2.0 EXTRA LOCATIONS 53 CONIL1) = CON(1)+CON(13))/2.0 EXTRA LOCATIONS 54 DO TALL SOURCE LOCATIONS AND EXTRA LOCATIONS 55 CONIL1) = CON(1)+CON(13))/2.0 55 CONIL1) = CON(1)+CON(13))/2.0 55 CONITALL SOURCE LOCATIONS AND EXTRA LOCATIONS 55 CONIL1) = 0.0 56 S1 = 1.0NUD 57 S1 = 1.0NUC 58 CONTINUE 59 CIII = 0.0 51 CONTINUE 51 S1.0NO 51 S1.0NO 51 CONTINUE 61 CONTINUE 61 CONTINUE 61 CONTINUE	<pre> 0 10 10 10 10 10 10 10 10 10 10 10 1</pre>	 30 CONTRACT = 1315FE USED LEES THAN THE TIME STEP ALLOWED 31 CONTRISTENTION STEP USED LEES THAN THE RETWER ITERATIONS 42 COUNT = TRAINTSTANTISTENTIE 43 COUNT = TRAINTSTANTISTENTIE 44 COUNT = TRAINTSTANTISTEN 55 CONTINUE 56 CONTINUE 57 CONTINUE 58 CONTINUE 58 CONTINUE 59 STETTIONS 50 CONTINUE 50 CONTINUE 50 CONTINUE 50 CONTINUE 51 E 1100 51 E 1000

-'

 $\bigcirc .$

 $\langle \rangle$

 $\langle \rangle$

.

•

.•

A - 5

	+ 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	N NUI VII * *	3330 WWW1 222+ ***	BAO BAO UUI UUI BUUI 22.8 8 8 8 8 8 8	王 王 王 王 王 王 王 王 王 王 王 王 王 王 王 王 王 王 王
CNFRWD.CNFRWD	<pre>76. 1030 CALL PLYAWM(A(LA),T(I),A(LA+1),XK(LK),C1) 76. 1032 J2 = J2+1 76. LA = FLD(5.17,NSQ2(J2)) 76. LA = FLD(22,14,NSQ2(J2)) 76. CALL PLYAWM(A(LA),T(I),A(LA+1),XK(LK),C2) 76. 1035 C1 = XK(LK)+XK(LA) 76. 1035 C1 = XK(LK)+XK(LA) 76. 1040 CALL PLYAWM(A(LA),T(I),A(LA+1),XK(LK),C1)</pre>	76* J2 = J2+1 76* LA = FLD(5,17,NSQ2(J2)) 76* LK = FLD(5,14,NSQ2(J2)) 76* C2 = XK(LK)*XK(LA) 76* 1045 CALL D2D1WM(T(I),CON(14),A(LA),XK(LK),C(I)) 76* 1099 C(I) = C1+C2 76* 1999 J2 = J2+1 76* END 76* END	<pre>77* INCLUDE VARGALIST 77* INCLUDE VARGALIST 77* IF(FLD(4/1/NSQ1(J)41)).E0.0) 60 T0 5000 77* NTYPE E = ELD(0.5.NSQ2(J2)) 77* LA = FLD(22/14/NSQ2(J2)) 60 T0 (4005/4015/4025/4020,4025,4030,4035), NTYPE 77* 4005 G(1) = XK(LK).4015/4025,4030,4035,4040,4030). NTYPE 77* 4010 G1 = 0.0 77* 4010 G1 = 0.0 77* 4015 G1 T0 4998 77* 4015 G1 T0 4998</pre>	77* LA = FLD(5,17,NSG2(J2)) 77* LK = FLD(22,14,NSG2(J2)) 60 TO 4017 77* 4025 01 = XK(LK) *XK(LA) 77* 4030 CALL D1D1WM(CON(14),A(LA),XK(LK),01) 77* 4030 CALL D1D1WM(CON(14),A(LA),XK(LK),01) 77* 4030 CALL D1D1WM(CON(14),A(LA),XK(LK),01) 77* 4035 CALL D1D1WM(CON(14),A(LA),XK(LK),01) 77* 4035 CALL D1D1WM(CON(14),A(LA),XK(LK),01)	77: 4040 DI = FLD(5,17,NSG2(J2)) 77: 4040 DI = XK(LA) 77: 4040 DI = XK(LA) 77: 4099 DI = $2 \times (LK) + XK(LA)$ 77: 4999 J2 = 2×1 77: 4000 J2 = 1141 77: 4000 J2 = 1141 78: 4000 J2 = 1141 79: 4000 J2 = 1000
	00265 00267 00271 00271 00272 00273 00273 00273	00277 00300 00302 00304 00304 00305 00305 00305 00305 00310	0003313	10000000000000000000000000000000000000	000355 003555 0003555 0003555 0003555 0003555 0003555 0003555 0003555 000555 000555 000000

A - 6

.)

.

• .

•

CNFRWD . CNFRWD

3330 222 ***	801 801 8 8 8 8 8 8	9 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	* * * * * * * * * * * * * * * * * * *	N N N N U N U N U N N N N N N N N N N N	
					., €
145,2050,2051]	.,
00 2035,2040,2(•6(L6)) K) •61)	((LK),62) K),61)	+6(LG)) 61+62	
R DUCTOR DUCTOR 0) 60 T0 30 0,2025,2030, (LK),6(L6)) KK(LK),61)))•XK(LK)•62) XK(LK)•61)) La+1) • XK (LK) A (La+1) • XK (L) • A (LA+1) • XK A (LA+1) • XK (L) a(La)•xk(Lk) .) 6(L6) =	
ST CONDUCTOF 60 T0 85 //LIST 01/LIST 01/LION 0	7, NSG2 (J2)) 14, NSG2 (J2) 7 (LTA), A (LA) XK (LA) 1 (1), A (LA),	7.HISO2(J2)) 14.NSO2(J2) XK(LA) (LTA))/2.0 (LTA))/2.0 A(LA).TM.A(7,NS02(J2)) 14,NS02(J2) A(LA),T(LTA XK(LA) A(LA),T(I),	7,NSOZ(J2)) 14,NSO2(J2) XK(LA) (LTA))/2,0 TM,CON(14). TM,CON(14). 501(J1)).E0	
HECK FOR LA FF(LG.EG.0.) A = FLD(22 NCLUDE VARG NCLUDE VARG NCLUDE VARG NCLUDE VARG A = FLD(22, 010(22, 00)(10)(10)(10)(10)(10)(10)(10)(10)(10)(2 = J2+1 A = FLD(5,1 K = FLD(22, ALL D101WM(0 T0 2017 0 T0 2017 ALL D101WM(ALL D101WM(2 = J2+1 A = FLO(5,1 2 = XK(LK)+ 0 TO 2998 ALL PLYAMM(0 TO 2999 0 TO 2999 0 TO 2092 0 TO 2032 ALL PLYAMM(0 TO 2032	Z = JZ+1 A = FLD(5,1 K = FLD(5,1 ALL PLYAWM(0 TO 2998 1 = XK(LK)* ALL PLYAWM	Z = 2.51 Z = 2.51 Z = 2.51 Z = 2.51 Z = 2.52 Z =	
004004 0012 4041 90 1 6 70 8	2017 J 2020 6 6 C L L L J 2020 6 6 C L L L J 2020 6 6 C L L L J	00010000000000000000000000000000000000	00000000000000000000000000000000000000	00 7 1 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	•.
8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		x u u u u u u u u u x * * * * * * * * *	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	•
00362 00365 00365 00366 00366 00372 00377 00377 00377 00377 00376 00377 00401 00402	00400 00400 004404 004407 004407 004407 004407 004407 004407 004407 004407 004407 004407 004407 004407 004407 00400 004407 00400 00400 00400 00400 00400 00400 00400 0004000 000400 0004000 0004000 0004000 0004000 000000	00415 00416 00416 00415 00421 00421 00421 00424 00424 00424	004431 004431 004431 004431 004433 0004433 00000000	004.55 004437 004441 004442 004442 004445 004451 004551 004551 004553 004553	

;

1

ļ

•

CNFRWD , CNFRWD

IS THE TIME STEP USED LESS THAN THE TIME STEP CALCULATED IF (ISTEP.LE.UELTA) 60 TO 130 COMPUTE THE TIME STEP KON(35) = 1 COMPUTE NEW TEMPERATURES USING CALCULATED SOURCE TERMS T1 = TSTEP+Q(1)/C(1) CALCULATE THE ABSOLUTE VALUE TEMPERATURE CHANGE FireLot3.1.NSo1(J1)).E0.0) 60 T0 55
T1 = T(1)+460.0
T2 = T(LTA)+460.0
6V = 6(L6).(T1+T1+T2+T2).(T1+T2)
60 T0 60
60 T0 60
55 6V = 6(L6)
55 6V = 6(L6)
60 B0T = 6(V,(T(LTA)-T(1))
60 B0T = 6(V,(T(LTA)-T(1))
70(1) = 0(1)+800T
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15+17
70(1) = 15 OUTAIN NEW DIFFUSION TEMPERATURES, DTMPCC AND CSGMIN DO 100 I ± 1,4410 Le = IE+I CON(17) = CKM DELTA = CKM/CON(4) IF(CKM.LE.0.0) GO TO 996 C CHECK FOR FIRST PASS IF(PASS.GT.0.0) GO TO 115 C UNDO THE TEMPERATURE CALCULATIONS LE = IE+I LE = IE+I T2 = ABS(T1) SAVE THE LARGEST TEMPERATURE CHANGE IF(TCGM.GE.T2) GO TO 95 X(LEA) = X(LEA)+6V 0(LTA) = 0(LTA)-0D0T CHECK FOR LAST CONDUCTOR IF(11501(J1).6T.0) 60 T0 50 TCGM = T2 TCGM = T2 KON(36) = I STORE THE TEMPERATURES 95 X(LE) = T(1)+T1 100 CONTINUE IF (PASS.GT.0.0) 60 TO 15 CALCULATE C/SK MINIMUM T1 = C(1)/X(LE) IF (T1.GE.CKM) 60 TO 90 CKM = T1 TSTEP = UELTA+0.95 60 TO 195 PASS = 1.0 CON(1) = TPRINT CON(2) = 0.0110 CONTINUE 85 CONTINUE ENO 20 65 60 115 ų ų J v Ų U U J Ų U Ų J U U. 126* 128* 128* 130* +001 132+ 141+ 142+ 142+ 131+ 137+ 138* 139+ 140+1 00545 00545 00545 00546 00550 00551 00551 00552 00552 00457

i. Ar

																																				•.	*NEW						HEN#	A JN+	**-5								
CNFRWD, CNFRWD	TSTEP = DELTA+0.95	GO TO 105	20 TSTEP = 0.95*TSTEP+CON(6)/TCGM	SGU 10 103 FLICTED-CONVILLIATER		TO I U LO U LO UNITATION DE LO LO UNITATION DE LO LO UNITATION DE LO LO UNITATION DE LO UNITAT	SCH IT IC ICRICATION CANNON AND TO PARAGE		STURE THE MAXIMUM UTFTUSION TEMPERATORE CHANGE	CONTRACT AT FIRE ADD ANY ADDAUDTO MADE	CHELK 10 SEE IT THERE AKE ANT ANTITUMETEL NOUES	IF (NATIENT) OF 10 10 100	CONFUL AN INMETIC FEMERALORED DE SOCHEDURE FOIN CEENSA										INCLUDE VROZALIST	IF(FLD(4,1,NSQ1(JJ1+1)),EQ.0) 60 TO 6000	NTYPE = FLD(0,5,NSQ2(JJ2))	LA = FLD(5,17,NS02(JJ2))	LK = FLD(22) 14,NS02(JJ2)	241 N 40000000000000000000000000000000000			I2 CALL DIDIWM(T(L),A(LA),XK(LK),Q2)	GO TC 5998	15 01 = 0.0	IT CALL UIDIWM(CON(14),A(LA),XK(LK),QZ)	20 Control 2000 (14) - 4 (14) - 4 (14) - 4 (14) - 50 Control 2000 (14) - 4 (14) - 50 Control 2000 (14) - 50 Contro	22 U2 = U2+1	LA = FLD(5,17,NSQ2(JJ2))	LK = FLD(22+14+NSQ2(JU2))		25 GI I XYICK)*XXICA) 22 T R003	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		LA = FLD(5,17,NS02(JJ2))	LK = FLD(22,14,NS02(JJ2))	02 = XK(LK)+XK(LA)	G0 T0 5998	35 CALL DIDIMM(CON(14) A(LA) XK(LK) Q1)				40 (1) = XK(LK)+XK(LA)	G0 T0 5037	96 Q(L) = Q1+Q2+G(L)
			*	* *			د ج	•	ບ		•	•	د *		* :		• •	•	1		 			*		*		* 1			-un -	*	*	÷.	بر	م ا ±	*	•		n 2 1	2	ר י <u>י</u>		*	*	•	ទា	ດ * 1			<u>ب</u>	±	ີ <u>ສ</u>
	144	145	140	+	Ť	* 1			Å.						1	10					104	165	169	16	16	10	9	91		16.0	169	169	169	169		165	165	169	169	507 1	041	169	169	169	169	165	165	165	071	169	105	169	165
	1554	1555	1556	222		100	100		202	1 0 0 0	100	202	00	200						003	1000	6090	0100	1190	0613	1014	0615	116	1000	0621	1622	623	1624	1625	500	0630	1631	1632	1633		10.00	6.37	0.79	641	642	643	944	010		020	1051	1652	653

• 1

•

•

		RWD . CNFRWD
		CNFR

8 y 16 •		'л ц		
	3 4 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		N N N N N N N N N N N N N N N N N N N	1 X 4 1 X 4 1 X 4
JJ2 = JJ2+1 CONTINUE END JJ1 = JJ1+1 LG = FLD(5,16,NSO1(JJ1)) LT = FLD(22,14,NSO1(JJ1)) LT = FLD(22,14,NSO1(JJ1)) TF(1.6T.1) 60 T0 4000 TF(FLD(2.1,NS01(JJ1)).EQ.0) 60 T0 4000 TF(FLD(2.1,NS01(JJ1)).EQ.0) 60 T0 4000 TF(FLD(2.1,NS02(JJ2)) LA = FLD(5,17,NS02(JJ2)) LA = FLD(5,17,NS02(JJ2)) COTO(3005,3010,3015,3020,3025,3030,3035,3040,3045,3050,3055) T = T(LL)+T(LTA))/2.0 T = T(LL)+T(LTA))/2.0 CALL DIDIWM(TM,A(LA),XK(LK),6(LG))	TM = 1(L) 60 TO 3007 Call DDIUW(T(L),A(LA),XK(LK),61) JJ2 = JJ2+1 LA = FLD(5,17,NSQ2(JJ2)) LK = FLD(22,14,NSQ2(JJ2)) Call D101WM(T(LTA),A(LA),XK(LK),62) 61 = XK(LK) *XK(LA) 61 = XK(LK) *XK(LA) 61 = ZK(LK) *XK(LA)	JJZ = JJZ+1 LK = FLD(5,17,NSQ2(JJ2)) LK = FLD(5,17,NSQ2(JJ2)) G2 = XK(LK)+XK(LA) G3 T0 3998 TM = T(L)+T(LA))/2.0 TM = T(L) G4L PLYAWM(A(LA),TM,A(LA+1),XK(LK).6(L6)) G4 T0 3032 TM = T(L) G4L PLYAWM(A(LA),T(L),A(LA+1),XK(LK).6(L6)) LA = FLD(5,17,NSQ2(JJ2))	LK = FLD(22,14,NSQ2(JJ2)) Call Plyam(A(LA),T(LTA),A(LA+1),XK(LK),G2) GO TO 3998 61 = XK(LK)+XK(LA) GO TO 3042 1 Call Plyam(A(LA),T(L),A(LA+1),XK(LK),G1) 1 = JJ2+1 1 a = FLD(5,17,NSO2(JJ2))	<pre>LK = FLD(22:14,NSQ2(JJ2)) 62 = XK(LK)*XK(LA) 60 T0 3998 1 M = (T(L)+T(LTA))/2.0 60 T0 3999 60 T0 3999 1 M = T(LTA) 60 T0 3007 1 M = T(LTA) 60 T0 3032</pre>
5999 6000 1.35 2005 30075	3015 3025 3025	3030 3032 3032 3040	3045 3050	3055 3060 3060
11 12 12 12 12 12 12 12 12 12 12	177555 177555 177555 177555 177555 177555 177555 177555 177555 177555 177555 177555 177555 177555 177555 177555 177555 1	17755 177555 177555 177555 177555 177555 177555 177555 177555 177555 177555 17	175* 175* 175* 175* 175*	1755
10655 10655 10655 10655 10665 10665 10665 10665 10665 10667 10667 10672 10672 10675 10675 10675 10675	0675 0677 0677 0702 0703 0705 0710	07112 07115 07715 07720 07722 07724 07724 07723 07724 07723 07724	00726 00731 00731 00733 00733 00733	00735 00735 007441 0074441 0074441 0074441 0074441 0074443 0074443 0074443 0074443 0074443 0074443 0074443 0074443 00774 00773 00774 00000000

۰.

CNFRWD . CNFRWD

3996 6(L6) = 1,/(1,/61+1,/62)	IF(FLD(3,1,NSQ1(JUL)),EQ.1) 6(L6) = 61=62	0444 00K - 00641 4000 fontinie		IF (FLD(3,1,NSQ1(JJ1)).EQ.0) 60 TO 140	T1 = T(L)+460.0	T2 = T(LTA)+460.0	6V # 6(L6)*(1]*1]*12*12]*(1]+12] 20 to 145		145 SUMC # SUMC+6V	SUMCV = SUMCV+6V+T(LTA)	CHECK FOR LAST CONDUCTOR	IF (NSQ1(JJ1).6T.0) GO TO 135	T2 = DD+T(L)+DN+(SUMCV+Q(L))/SUMC	OBTAIN THE CALCULATED TEMPERATURE DIFFERENCE	I = AUS(I(L)+IZ) CAANT TUE KEW EDATIDE	LIVE THE YEAR LEATERATORE	SAVE THE MAXIMUM ARITHMETIC RELAXATION CHANGE	IF(TC6M.6E.T1) 60 TO 165	TC6M = T1	KON(37) = L	165 CONTINUE	SEE IF RELAXATION CRITERIA WAS WET	IF (ICOMPLETCON(191) 60 10 1/3	CTORE THE MAYINUM ARITHMETIC RELAXATION CHANGE	175 CON(30) = TCGM	COMPUTE THE ARITHMETIC TEMPERATURE CHANGE	TCGM = 0.0	DO 180 I = L1,NNC	T ARC(T(T)_V(F))	IF(T1.LT.TCGM) 60 T0 180	TC6M = T1	KON(38) = I	160 CONTINUE	SEE IF ATMPCA WAS SATISFIED	TTITCOM.61.CUNILLY 60 10 165 Tonith I TCGM	185 KON(12) = 0	CALL VARBL?	CHECK THE BACKUP SWITCH	IF(KON(12), NE.0) 60 TO 105	ADVANCE TIME	CONTAT = LUNLI Telm = Telm+Teteb	TSTEP # DELTA+0.95	CHECK FOR TIME TO PRINT	IF(TSUM.GE.CON(18)) 60 TO 190	CHECK FOR PRINT EVERY ITERATION	IF (KON(7).EQ.03 60 10 10	CALL UUICAL	THY TO FUEL THE DUTPUT INTERVALS	190 TPRINT = TPRINT+TSUM	195 CALL OUTCAL	IS TIME GREATER THAN END COMPUTE TIME
						. .							:	-					•								_		فد م					- -												.	×.				
175	175		175	176	177	178	1001		281	1834	1844	1854	1861	187			101	192+	1934	1941	195	1961	100	000	200	2014	2024	203	205	2064	207	2084	2091	210	112	213	2144	215	216	217		2204	2214	2224	10 10 10 10 10 10 10 10 10 10 10 10 10 1	225		200	228	229	2301
00750	00751	12200	00755	00756	00760	00761	00762	00100	00765	00.766	00766	00767	00771	00771	21100	200 - 10 201 - 10	22200	00774	00776	00777	01000	01000	20010		01006	01006	01007	01010		01015	01017	01020	01021	01021	52010	01026	01027	01027	01030	01030	01032	01034	01034	01035	01035	01037	14010	201010	21010	01044	01044

4

• .

CNFPWD , CNFRWD

۰.

۰,

·.

•

 IF(CON(1)+1.000001.LT.CON(3)) 60 TO 5	NTH F 1E	NDIM = NLA	RETURN	995 WRITE(6,885)	60 TO 1000	996 WRITE(6,886)	G0 T0 1000	997 WRITE(6,887)	GO TO 1000	998 WRITE(6,884) I	GO TO 1000	999 WRITE(6,889)	1000 CALL OUTCAL	CALL EXIT	885 FORMAT (46H CNFAWD REQUIRES SHORT PSEUDO-COMBLIF SEGUENCE)	886 FORMAT (24H CS6MIN ZERO OR NEGATIVE)	887 FORMAT(201 TIME STEP TOD SMALL)	888 FORMAT(18,20H LOCATIONS AVAILABLE)	889 FORMAT(194 NO OUTPUT INTERVAL)	END	NIVAC 1108 FORTRAN V COMPILATION. 0 *DIAGNOSTIC* MESSAGE(S) Symbolic De relocatable	
+16	32+	*??	**	35*	36*	37+	*8 £	39.	#0 #	41+	42.4	-#Ct	* 7 7	454	46*	+1+	48+	*6†	50+	51+	D OF L	
2	2	0	1	0 0	ค ะ	به 2	2	۵ 0	2	2	0	2	2	2	5	N t	20	ŝ,	2	2		
0104	0104	0105	5010.	0105	0105	0105	0105	0106	0106	0106	0106	0106	0107	0107	0107.	0102	0107	01070	10107	0110	55	

.• ~;~~;

•

21 FE8 71				CUDZ R 000021 C1 0012 R 000000 G 0002 I 000040 1 0002 I 000040 JX 00021 I 000040 JX 0021 I 000040 JX 0021 I 000010 NYPE 0002 R 000020 NYPE 0002 R 000024 G7 00012 R 000024 G7 00012 R 000024 J7 00012 R 000027 TM
				0017 R 00000 CON 0002 R 000036 DN 0002 I 000012 JI 0002 I 000017 LA 0021 I 000017 LA 0021 I 000017 LA 0022 I 00001 NH 0002 R 00001 NH 0002 R 000014 TCGM 0002 R 000014 TCGM
EATED ON 20 AUG 70			RELATIVE LOCATION, NAME)	0002 R 00015 CKM 0002 R 000035 DELTA 0002 I 000042 JJ2 0002 I 000042 JJ2 0002 I 000042 LK 0002 I 0000020 LK 0021 I 000005 NAT 0021 I 000000 NNA 0011 R 000000 NS02 0011 R 000000 0 0001 R 000000 1 00012 R 000000 1 00012 R 000000 1 00012 R 000000 1
CNFRDL,CNFRDL L/CNFRDL Hena Version 131K-100 CRI E on 09 Jun 70 AT 14:00:31 Entry Point 003463	KAR 0000172 VAR 0000172 000001 000001 000001 000001 000001 000001 000001 000001 000001 000001 000001 000001 000001		R VARIARLES (BLOCK, TYPE, 1	0010 R 000000 C 0002 R 000037 DD 0002 R 000030 G1 0007 I 0000091 JJ1 0002 I 000005 LG 0002 I 000005 LG 0002 I 000005 L1 0002 R 000005 L1 0002 R 000000 PASS 0002 R 000000 PASS 0002 R 000000 FASS 0002 R 000000 FASS
BIW FOR * UNIVAC 1108 FORTRAW V ATT UNIVAC 1108 FORTRAW V ATT THIS COMPILATION WAS DON SUBROUTINE CNFROL E STORAGE USED (BLOCK,)	0001 *CODE *CODE *SIMPT 0000 *SIMPT *ARRAYS 0005 *BLANK 0005 *BLANK 0001 CAP 0010 CAP 0010 CAP 0010 CAP 0011 CONU 0011 PC1 0011 PC1 0011 PC1 0011 PC1 0011 PC1 0012 CONU 0011 PC1 0012 CONU 0012 PC1 0012 CONU 0012 CONU 00012 CONU 0012 CONU 0012 CONU 0012 CONU 0012 CONU 0012 CONU 0012 CONU 0012 CONU 0012 CONU 0012 CONU 00012 CONU 00012 CONU 0000 0000 CONU 0000 0000 0000 0000 0000 0000 0000 0	0022 VARULI 0023 VARULI 0024 PLYAWM 0025 D201WM 0025 VARULZ 0025 VARULZ 0031 NERK25 0033 NIO25 0034 NERV25	STORAGE ASSIGNMENT FO	0016 R 000000 A 0002 R 000022 C2 0002 I 000024 6V 0015 I 000000 K 0002 I 000000 K 0021 I 000002 LTA 0021 I 000002 NNT 0021 I 000002 NNT 0022 R 000002 NNT

•

A - 13

...

ē
2
٠.
z
Q.
الس
ö
ar i
٤.
z
U.

	•	1040L	1251	1601	190L	2000L	2017	2035L	2055L	25L	3005L	3020L	30401	30601	4005L	4020L	4037L	16601	5012L	5125L	5256	6000L	BA6F	956	76.6	
		000430	002103	003212	003361	000674	001301	001433	001567	000147	002527	002631	002732	003111	000741	001004	001115	001144	002241	002327	002036	002441	110000	002005	10100	
		0011	0001	0001	1000	0001	0001	1000	1000	0001	1000	1000	0001	1000	0001	1000	0001	0001	1000	1000	0001	0001	0000	1000	1000	
		000422 1015L	002075 120L	003207 155L	003332 1856	000207 2006	001263 2015L	001410 2032L	001524 2050L	000257 2326	001656 3000L	002575 3017L	002727 3035L	003063 3055L	003152 4000L	000767 4017L	001101 4035L	001140 4998L	002240 5010L	002312 5022L	002424 5040L	002436 5999L	000000 885F	001766 90L	003425 998L	
		000	1000	0001	1000	0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	1000	0001	0001	0001	0001	0001	0000	0001	1000	
)		000367 1012L		000135 156	003321 18AL	000671 1999L	001260 2010L	001403 2030L	001516 2045L	000226 2126	000162 30L	002557 3015L	002704 3032L	003020 3050L	003147 3999L	000766 4015L	001043 4030L	001745 4746	002233 5005L	002276 5020L	002407 5037L	002432 5998L	001716 B0L	000030 889F	003417 997L	
		1000		1000	1000	000	0001	0001	0001	0001	0001	0001	0001	0001	0001	1000	1000	0001	0001	0001	0001	1000	0001	0000	1000	
	-DC-	000351 1010L	000344 1030C	002441 1456	003270 175L	000666 1998L	001240 2007L	001343 2025L	001457 20421	001621 2065L	001653 2999L	002554 30101	002677 3030L	003012 3045L	003121 399AL	000747 4012L	001035 4025L	000240 451	001147 5000L	002261 5017L	002373 5035L	002161 5726	001713 75L	000023 888F	003411 996L	
	NFROLICNE	1000	1000	1000	0001	000	0001	1000	0001	0001	0001	0001	1000	0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	0000	0001	:
	υ	000330 1005	10201 044000		003257 1650	003364 1950	001233 2005	001335 20201	001436 2040L	001615 20601	001625 2998	12002 12500	002637 3025L	002753 3042L	003115 30654	000746 40100	001020 4022L	001132 40401	000114 55	002260 50151	002335 5030L	002140 5636	001147 70L	000015 8475	003403 995L	
		1000		1000	1000	1000	1000	1000	1000	0001	1000	1000	1000	0001	0001	1000	1000	1000	1000	1000	0001	1000	1000		1000	

										(N)	5	<u>ç</u>		(d)	A)	29	(4)	0	(A)	(d	0	(S	50		E	
	INDA F-V		(1)9/0N						******	SWIT)	MI LO	LIMEN	T (CSGF1	UNLOC	DAWLD	COPE I 1	(DAMF	(DAMF	E (ATMPO	BACK	LTME T	HIL)		L N S S S S S S S S S S S S S S S S S S	(OUTP)	
	N SUBROUTINE FOR S	BUIRED) /SOURCE/0(1) /CO	T/K(1) /ARRAY/A(1)	T.LS01,LS02		((T)YN*(T)X)*((AND NAMES +++++	OBLEN TIME	TEP USED	M STOP TIME	TEP FACTOR EXPLICE	O LOOPS, INTEGER	CHANGE ALLOWED	SWITCH	DAMPING FACTOR	DAMPING FACTOR	HMETIC TEMP. CHANG	D AFTER VARIABLES	LEM START TIME	I TERATION	L CHANGL CALCULATE	E CASE MINIMUM	T INTERVAL	
	ENCING EXECUTION	SEQUENCE IS RE	VT(1) /CAP/C(1	PARE ANTWANTH	NNT.NGT.NCT.NA	4X(1)	(T))))))))))))))))))))))))))))))))))))		ANT DEFINITIONS	AINS THE NEW PR	AINS THE TIME S	VINS THE PROBLE	VINS THE TIME S	OF ITERATION D	ION TENPERATURE	EACH ITERATION	THMETIC TEMP.	FFUSION TEMP.	M ALLOWED ARIT	SWITCH CHECKE	AT TIME OR PROB	LIME BETWEEN AN	SION TEMPERATUR	DEMPERATORE C	LAINS THE DUTPU	
56 T++0000	CNFI-DL WARD DIFFERE	CUDO-COMPUTE	E/H(1) /TEM	VNSQ1(1) /PC2	NN NND NNA	14(1) XX (1) 10	(KON(T) CON	-,LIST	DNTROL CONST	STANT 1 CONT	STANT 2 CONT	STANT 3 CONT	STANT 4 CONTI	INPUT NUMBER	S THE DIFFUS	S THE OUTPUT	THE NEW AR	IS THE NEW DI	US THE MAXIM	IS THE BACKUR	VS THE PRESEN	IS THE MEAN	IS THE UIFFUS	VS ANT 17 TS F	STANT 18 COUL	
	SUBROUTINE (THE LONG PSE	COMMON /TITL	COMMON /PC1/	COMMON /DIME	DIMENSION CO	EULIVALENCE END	INCLUDE DEFF		CONTROL CONS	CONTROL CONS	CONTROL CONS	CONTROL CONS	CC5 IS THE	CC6 CONTAINS	CC7 CONTAINS	CC9 CONTAINS	CC10 CONTAIN	CC11 CONTAIN	CC12 CONTAIN	CC13 CONTAIN	CC14 CONTAIN	CC15 CONTAIN	CONTROL CONTROL	CONTROL CON	
66 £0+C00	U 	ບ •					* *		+ C+++		U •	U •	U	U H	U •		<u>ں</u> د		ں •	.u	ບ •	ں •			بر	ر •
1000	00101 2.	00101	10100	00105 4	100100	00110 4	00112 4:	00113 5	00113 5	00113 5	00113 5	00113 5	00113 5	00113 5	00113 5	00113 5		00113 5-	00113 5	00113 5	00113 5	00113 5	00113 5	C C1100		

۳.,

CNFRDL, CNFRDL

00113	÷.	U	CC19 CONTAINS THE ARITHMETIC RELAXATION CHITEMIA ALLUMEU	5
2113	in i	<u>.</u>	CC20 CONTAINS THE NUMBER OF RELAXATION LOOPS USED INTEGER (LOOPC	
21100	6 u	U L	CC21 CUNIAINA INC FINIMUM ALLUMED TIME STEP VIEW (DTIME COS) IS EDD THE TAME STEP TAPI TOTIC	
		ې د	CCC 13 TON THE INTO THE STEP AND THE STEP AND TO	XX
	- 4 0 14	ر ب	CC20 CONTAINS THE CASE BANKE BILOWED (CSGR	R
21130	1 1	, 0	CC25 CONTAINS THE C/SG RANGE CALCULATED	2
0113	÷.	U	CC26 CONTAINS THE DIFFUSION RELAXATION CRITERIA ALLOWED (DRLXC	S
00113	.	υ	CC27 CONTAINS THE DIFFUSION RELAXATION CHANGE CALCULATED (DRLXC	ទូរ
00113	ۍ ۴	U	CC28 CONTAINS THE LINE COUNTER. INTEGER	53
00113	ເດີ ເ	U	CC29 CONTAINS THE PAGE COUNTER. INTEGER	55
51100	# 1 0 u	U L	CCJU CUNIAINS ANINYETIC RELAXATION CRANGE CALCULATED TARGAE Craits is truteated, ofthermal SPCS,12thFremail PCS,22GENFRAL (1SPC	20
CT100	n u	<u>ل</u>	TOLE IN INVESTIGATION CONTRIBUTION OF OUR THE SYSTEM. IN - DUT (ENGRA	AL
00113	н á) (J	CC33 CONTAINS THE DESIRED ENERGY BALANCE, USER INPUT	ဗ်
0113	*	.u	CC34 CONTAINS THE NUCOPY SWITCH FOR MATRIX USERS (NOCOP	Lλd
0113	ۍ ۴	U	CC35 CONTAINS RELATIVE NODE NUMBER OF CSGMIN	
0113		U	CC36 CONTAINS RELATIVE NODE NUMBER OF DIMPCC	
0113		U.	CC37 CONTAINS RELATIVE NODE NUMBER OF ARLACC	
00113	ព័ រ	.	CC38 CONTAINS PELATIVE NODE NUMBER OF AIMPCC	
51100	# - 4 D ¥	U C	CC39+4U-41-442-43 COMIAIN UNMMI INICGER CONSTANIS (1-0-8-6-462) Preud-45-46-44-48 Contrain frimmy Fi Dating Constants (8-5-1-40-VTES	i f
CT100	n y	י נ	CLATTACTORIZATION CONTRACTOR INTERVAL FOR CINCLE (LAYE	Ş
	2 <u>10</u>	20	CC50 IS NOT USED AT PRESENT	
0114	-	, . •		
00115	•9		IF(CON(4).LT.1.0) CON(4) = 1.0	
0117	**		IF (KON(5). LE.0) KON(5) = 1	
00121	6 0 (IF(CON(6).LE.0.) CON(6) = 1.E+8	
00123	* ·			
122100	# 1 D		JETCON(9). LE. 0. 1 CON(3) + 1.0	
10131	12.		IF (CON(18) .LE.0.) 60 TO 999	
0133	*0		IF (CON(19), LE.0.) CON(19) = 1,E+8	
0135	14		IF(KOH(31),NE,1) 60 TO 995	
00137	15*		PASS = -1.0	
00140	16*		NNC = NND+PINA	
14100	17*		IE = NTH	
00142	18			
0143	#61		NTH = NTH+INC	
1111	50	:	NOIM # NDIM-NNC	
	214	U.	CHECK FOR EXIRA LOCATIONS FOR CALCULATED NUMES	
00145	50		IF(I.LT.0) 60 TO 998	
00150	24*		L1 = NND+1	
00151	25*		TSTEP = CON(18)	
00152	56	(TPRINT = CON(13)	
20152	28.	ب	TITTALLE TIME SUM BETWEEN UUIPUT INTERVALS R Term # 0.0	
00153	562 562	Ų	DOES OLD TIME PLUS THE OUTPUT INTERVAL EXCEED THE STOP TIME	
00154	#0E		IF (CON(13)+CON(18) .LE.CON(3)) GO TO 10	
00154	*16	υ	DONT EXCEED IT	
00100		ç	CONTRATE CONTOPECONTLOP 14 The store step i argers Than allowed	
00157	1 1 1 1 1 1 1	د	LO [F(TSTEP-LE-CON(8)) 60 TO 15	
00161	10°+		TSTEP = CONI(8)	
00101	0	J	DOES THE TIME SUM PLUS THE TIME STEP EXCEED OUTPUT INTERVAL	
00162		,u	LU TELESONTESCETEORIENSE ZURUNE DAUT EXCEFA IT	
00165	*6£,		20 TSTEP = CON(16) -TSUM	
00166	+0+		60 TO 30	

•

~
~~
. 🖳
z
U U
- -
~
<u>.</u>
<u> </u>
۰.
-Z
0
-

																													#JN#	*NEW					NUN+						*NEW	0 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1			
CNFROL, CNFRDL	DOES TIME SUM PLUS TWO TIME STEPS EXCEED GUTPUT INTERVAL	APPROACH THE OUTPUT INTERVAL GRADUALLY	TSTEP # (CON(18)+TSUM)/2+0 Store deita time step in the Constants	0 CON(2) = TSTEP	IS THE TIME STEP USED LESS THAN THE TIME STEP ALLOWED	IF(TSTEP'LT.CONIZI)) 60 10 997 Calculate THE MEW TIME	CON(1) = TPRINT+TSUM+TSTEP	COMPUTE THE MEAN TIME HETWEEN ITERATIONS	COM(14) = (COM(1)+COM(13))/2+0 2ERO OUT ALL SOURCE LOCATIONS AND EXTRA LOCATIONS	D0 35 I = 1.0ND	LE = 1E+1	X(LE) = 0.0	5 CONTINUE	SHIFT THE ARITHMETIC TEMPERATURES INTO THE EXTRA LOCATIONS	IF (NNA.LE.0) 60 TO 45 Do 40 T = 11.NNC		LE = IE+I	X(LE) = T(1)	0 CONTINUE		IF (KON(12).NE.D) 60 TO 10			1.06M = 0.0	CALCULATE O SUM AND 6 SUM	DO 85 I = 1. NND	LE = IE+I THATION VARCIITCT	IF(FLD(1,1,NSQ1(J)+1)),FQ.D) 60 TO 2000	NTYPE = FLD(0.5.NSQ2(J2))	LA = FLD(5,17,NS02(J2))	GO TO (1005,1010,1015,1020,1025,1030,1035,1040,1045), NTYPE	5 CALL DIDIWM(T(!),A(LA),XK(LK),C(!))	GOTOL CONTRACTORIA STATISTICS		LA = FLD(5,17,NS02(J2))	LK # FLD(22+14+NSQ2(J2))	CALL VIUTUTITITITICATION CONTRACTOR 00 TO 1998	5 C1 = XK(LK)•Xk(LA) 50 T0 1013	0 CALL DIDIWM(T(I),A(LA),XK(LK),CI)	J2 = J2+1	LA = FLD(5,17,NSQ2(J2))	C2 = XX (LK) *XX (LA)	GO TO 1998	5 CALL PLYAWW'(A(LA),T(I),A(LA+1),XK(LK),C(I)) Goi to 1999	0 CALL PLYAWM(A(LA),T(1),A(LA+1),XK(LK),C1)
	ة بن	ہ در	c	n.	J	L	а Э. с.	U	્ય	,			-m	J				1	a 2	F.					J							100	5	101				101	102					102	103
			- -					-						-				.	۰. ه		с [.] да																				* •				
	1	10	3 ú 3 3	ų ų tr	1	80 0 47 51	ŝ	15	លំហំ លំហំ	ŝ	52	9 F 9 S	ຮັດ	65	9	0.0	59	÷ 19	9		89	69	21	22	i ĉ	74	5.4	ġ,ġ	16	9 	9.9	92	0 F	92	16	9 4 7 7	2.92	01	92	16	9 ç 7 ç	20	92	16	20,
	00166	00167	00171	00172	00172	00173	00175	00175	00176	00177	00202	00203	00205	00205	00202	00214	00215	00216	00217	12200	00223	00225	00226	00227	00230	00231	00234	00236	00540	00241	00243	00244	C # 2 0 0	00247	00250	00251	00253	00254	00256	00257	00200	00262	00263	00264	00266

•

•

...

A - 16

•

.

NFROL
Ū,
ROL
CNF

*NEW	AUN						*NEW	*NEW										*NEW	*NEW	*NEW	n=++									-NEW)		•	M-SN-	*NEW	2-##			#NEW	*NEN	8=++							
1032 J2 = J2+1 LA = FLD(5,17,N502(J2))	LK = FLD(22,14,NS02(J2))	CALL PLYAWM(A(LA),T(I),A(LA+1),XK(LK),C2)	60 TO 1998	1035 C1 = XK(LK)+XK(LA)	60 TO 1032	1040 CALL PLYANWIA(LA), I'L) ALLATI PANILNICI'					TO A CONTRACT ON THE AND A CONTRACT A CONTRA		1998 C(1) = C1+C2		2000 CONTINUE	Etio	INCLUVE VARGALIST			LA = FLO(21.1.1.1.1.1.2.4.1)	60 T0 (4005,4010,4015,4020,4025,4030,4035,4030,000,000) NTYPE	4005 0(1) = XK(LK)+0(1)	60 T0 4999	4010 01 = 0.0	4012 CALL DIDIWM(T(I).4(LA).XK(LK).02)	60 T0 4998	4015 01 = 010 	401 CAL DALLWILLOUT 114	4020 CALL DIDIWM(COM(14),A(LA),XK(LK),01)	4022 J2 = J2+1	LA = FLD(5,17,1502(J2))	LK = FLD(22,14,NSQ2(JZ))			4030 CALL DIDIWM(CON(14),A(LA),XK(LK),Q1)	J2 = J2+1	LA = FLD(5):17:022(42))		G0 T0 4998	4035 CALL DIDIWM(CON(14) A(LA) XK(LK) POID			G0 T0 4012	4040 01 H XK(LK)+XK(LA)		1101 110 110 110 110 110 110 110 110 11	5000: CONTINUE			LTA = FLD(22+1++NSG1(J1))
76*	76*	76.	76+	76*	16+	16*		0	# 0 /	#0.L		101	764	704	76+	76+	+22	*11	***	+		17+	+11	+22	17+	+	**	*/-	-22	77*	17*	11	* * *	*24	+17+	-22	+11	*	77+	17+			- 2.2	-17		8/L	+11+	-11-	10.	80
00267	00271	00272	00273	00274	00275	00276	11200	00200	10000	200202		00200	00200	20200	00310	0(311	00312	00313	00315	6100	002200	00321	00322	00323	00324	00325	00326	002200	00331	0.0332	00333	00334	20000	00337	00240	00341	00342	00344	00345	00346	00347	00000	00152	1.153	00354	00356	00357	00360	19000	00363

CNFRDL, CNFRDL Include VARG, LIST

Ś

8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	* * * * * * * * *	0 * NEW * *	* * * * * * * *	8 3 QI JULI I Z Z 8 8 4 6
<pre>C INCLUDE VARG.LIST C CHECK FOR RADIATION CONDUCTOR IF(FLD(2-1.NSG1(J1)).5G.0) GO TO 3000 NTYPE = FLD(0.5.NSQ2(J2)) LA = FLD(2.17.NSQ2(J2)) LA = FLD(2.17.NSQ2(J2)) LA = FLD(22.14.NSQ2(J2)) GOTO(2005.2010.2015.2020.2025.2030.2035.2040.2045.2050.2055. GOTO(2005.2010.2015.2020.2025.2030.2035.2040.2045.2050.2055. 2005 TM = (1(1)+T(LTA))/2.0 2007 CALL DIDUMM(TM.A(LA).XK(LK).6(LG)) GO TO 2099 GO TO 2007 GO TO 2007 GO TO 2007</pre>	2015 CALL DIDIWM(T(I),A(LA),XK(LK),G1) 2017 J2 = J2+1 LA = FLO[5,17,NS02(J2)) LK = FLO[22,14,NS02(J2)) CALL DIDIWM(T(LTA),A(LA),XK(LK),G2) 60 T0 2998 2020 G1 = XK(LK)+XK(LA),XK(LK),G1) 2025 G1 = D10WM(T(1)),A(LA),XK(LK),G1)	LA = JC-1 LA = FU15.17.1502(J2)) LK = FLD(22.14.NS02(J2)) 62 = XK(LK) *XK(LA) 60 T0 2998 2032 CAL PLYAWM(A(LA).7K.A(LA+1).XK(LK).6(L6)) 2035 TM = T(1) 2040 CAL PLYAWM(A(LA).TK.A(LA+1).XK(LK).6(L6)) 2040 CAL PLYAWM(A(LA).T(1).A(LA+1).XK(LK).6(L6)) 2040 CAL PLYAWM(A(LA).T(1).A(LA+1).XK(LK).A(LA))2000 CAL PLA) 2040 CAL PLYAWM(A(LA)	LA = FLD(5,17,NSQ2(J2)) LK = FLD(22,14,NSQ2(J2)) CALL PLYAWM(A(LA),T(LTA),A(LA+1),XK(LK),62) CALL PLYAWM(A(LA)) 2045 G1 = XK(LK)+XK(LA) 60 T0 2042 2050 CALL PLYAWM(A(LA),T(1),A(LA+1),XK(LK),61) J2 = J2+1	LA = FLU(5,17,NSG2(J2)) LA = FLU(5,14,NSG2(J2)) 62 = XK(LK) + XK(LA) 63 TO 2996 Call D2DuMM(TM,CON(14).A(LA).XK(LK).6(L6)) 64 TO 2999 64 TO 2999 2065 TM = T(LTA) 65 TO 2007 2055 TM = T(LTA) 66 TO 2007 2055 TM = T(LTA) 66 TO 2007 2998 6(L6) = 1.(1./(1./(51+1./62)) 16 F(LD(3.1.NSG1(J1)).EG.1) 6(L6) = 61•62 2999 J2 = J2+1 3000 CONTTINUE 1 = T(1)+460.0 1 = T(1)+460.0
66666666666 6666666666666666666666666				* * * * * * * * * * * * * * * * * * *
000347400003474000034740000034740000034740000034740000034740000034740000034740000034740000003474000000347400000034740000003474000000347400000034740000003474000000347400000034740000000347400000000			00000000000000000000000000000000000000	00000000000000000000000000000000000000

•

A - 18

CNFRDL , CNFPDL

÷

CNFRDL , CNFRDL

.)

).

													AUN+	*NEW	NEM	ウーチョ																			*NEM												
CNFRDL . CNFRDL	CHECK TO SEE IF THERE ARE ANY ARITHMETIC NODES	COMPUTE ARTITHKETIC TEMPERATURES BY SUCCESSIVE POINT OVER-RELAX	DD = 1.0-DN	LAX = KON(5)		JU2 = J2	TC6M = 0.0 KON(20) = 1	DO 165 L = L1.NNC	SUMC = 0.0	SU4CV = 0.0	IFILIE VROFITST	IF(FLD(4,1,NSQ1(JU1+1)).EQ.0) 60 70 6000	NTYPE = FLD(0,5,NSO2(JJ2))	LA = FLD(5,17,NSQ2(JU2))	LK = FLD(22.14.NSQ2(JJ2))	60 T0 (5005,5010,5015,5020,5025,5030,5035,5040,5030) NTYPE	05 Q(L) # XK(LK)+0(L)		12 CALL DIDIWM(T(L), A(LA), XK(LK), 02)		15 01 = 0,0	17 CALL DIDIWM(CON(14)•A(LA)•XK(LK)•02) Go to food	20 CALL DIDIWM(CON(14),A(LA),XK(LK),Q1)	22 JJ2 = JJ2+1	LA = FLD(5,17,NS02(JJ2))	LK I FLD(22*14*N502(JJ2)) GO TO 5017	55 60 10 111 X (I K) # XK (I A)		30 CALL DIDINM(CON(14) A(LA) XK(LK) Q1)		LA = FLU(D)1/1/1/2042(UUC)] 1 k - FLU(D)1/1/1/2042(UUC)]	C2 = XK(LK) * XK(LA)	60 TO 5998	32 LALE ULUISPIICUULTITYTAATAANIENTAULUITUUTU	LA = FLD(5.17.NSQ2(JJ2))	LK = FLD(22,14,NSQ2(JJ2))		40 41 - ANICAIFANCEAN Go To 5037	98 G(L) = 01+02+0(L)	99 JU2 = 2J2+1	UD CONTINUE	END	16 m FLD(5,16,4501(1,41))	LIA = FLD(22, [4, NSQ1(JJ])	IF(I.6T.1) GO TO 4000	INCLUDE VRG2.LIST	CHECK FOR RAULATION COMPUCTOR If (FLD(2+1+HSO1(JJ1)).EQ.0) GO TO 4000
	U	U														1	50	5 US	200	2	50	50	50	50			50	2	50				ŭ				2	0	59	65	60	-	4			ç	د
	145+		1494	150+	152+	153*	154+	156*	157+	158*	* 0 0	160+	1.00+	160+	100+	160+	1604	150.	160+	160+	160*	160*	160*	160+	160+	160*	1604	160+	160+	160*	160	160*	160+	1001	160+	160*	160*	1604	160+	160*	160+	160+	162#	103+	164+	1654	1664
	48930		00560	00561	00565	00566	00567	00571	4200	00575	00400	0000	0000	00-04	00-05	00,000	00607	00611	00012	00613	00614	00615	00617	00620	00621	00622	00624	00025	00626	00027	0000	00632	00633	00635	00636	00637	00000	44000	00643	00044	00645	0.0646	0000	00651	00652	90054	00655

.

a B B B B B B B B B B B B B B B B B B B	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	330 WU 229 * * *		2 2 V U U U 2 V 2	
NTYPE = FLD(0,5,NS02(JJ2)) La = FLD(5,17,NS02(JJ2)) Lk = FLD(2,14,NS02(JJ2)) Lk = FLD(22,14,NS02(JJ2)) Goto(3005,3010,3015,3020,3025,3030,3035,3040,3045,3050,3055) S = 3060,3065), NTYPE 3005 TM = (T(L)+T(LTA))/2,0 3007 CALL DID1WM(TM,A(LA),XK(LK),6(L6)) Go TO 3099 S010 TM = T(L) GO TO 3007 S015 CALL DID1WM(T(L),A(LA),XK(LK),61) S015 CALL DID1WM(T(L),A(LA),XK(LK),61)	3017 JU2 = JU2+1 LA = FLD(5,17,NSQ2(JJ2)) LK = FLD(22,14,NSQ2(JJ2)) Call DD1MM(T(LTA),A(LA),XK(LK),62) 60 TO 3998 3020 G1 = XK(LK)*XK(LA) 60 TO 3017 3025 CALL DD1UMM(T(L),A(LA),XK(LK),61)	JJZ = JJZ+1 La = FLD(5,17,NSQ2(JJZ)) LK = FLD(22,14,NSQ2(JJZ)) G2 = XK(LK)*XK(LA) G0 TO 3999 S030 TM = (T(L)+T(LTA)/2,0 3032 CALL PLYAWM(A(LA),TM,A(LA+1),XK(LK),G(LG)) G0 TO 3999 G0 TO 3999 5040 CALL PLYAWM(A(LA),T(L),A(LA+1),XK(LK),G1)	3042 JJ2 = JJ241 LA = FLD(5,17,NS02(JJ2)) LK = FLD(22,14,NS02(JJ2)) CALL PLYAWM(A(LA),T(LTA),A(LA+1),XK(LK),G2) GO TO 3998 3045 G1 = XK(LK)+XK(LA) GO TO 3042 GO TO 3042 SOLL PLYAWM(A(LA),T(L),A(LA+1),XK(LK),G1)	JJ2 = JJ2+1 LA = FLD(5:17.NSQ2(JJ2)) LK = FLD(22.14.NSQ2(JJ2)) G2 = XK(LK) *XK(LA) G0 T0 3996 CALL D2D1WM(TM.CON(14),A(LA),XK(LK).G(LG)) G0 T0 3999 3060 TM = T(L7A) G0 T0 3007 3065 TM = T(L7A) G0 T0 3007 3065 TM = T(L7A) G0 T0 3007 3066 TM = T(L7A) G0 T0 3007 3065 TM = T(L7A) G0 T0 3007 1 L = 11/1,V(1,VG1+1./G2) 1 F(FLD(3.1,NSQ1(JJ1)).EG.0) G0 T0 155 1 F(FLD(3.1,NSQ1(JJ1)).EG.0) G0 T0 155 1 T = T(L)+460.0	T2 = T(LTA)+4460.0 GV = G(LG)*(T1+T1+T2+T2)*(T1+T2) G0 T0 160
44444444444444444444444444444444444444	10000000000000000000000000000000000000	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	16666666	16666555555555555555555555555555555555	170+171+
006557 006661 006662 006663 006663 006663 006663 006663 006663 006663 006663 006663 006663 006663 006663 006663 006663 006663 006653 006662 006665 006665 006665 006665 006665 006665 006665 006665 006665 006665 006665 006665 006665 006665 0006653 000665 000665 000665 000665 000665 000665 000665 000665 000665 000665 000665 000665 000665 000665 0006662 0006663 000665 00000000	00671 00672 00673 00674 00674 00676 00677 00677 00677	00202 00703 00704 00705 00705 00705 00711 00712 00712 00712 00712 00712	00723 007215 00717 00717 00721 00723	007224 007255 007231 007331 0073331 007335 007335 007335 007335 007441 007441 007445 007445 007445 007445 007445 007455 007465 007465 007465 007465 007765 007765 007765 007755 0007555 000555 000555 0005555 000555 000555 00055555 00055555 00055555 00055555 00055555 00055555 00055555 000555555	00752 00752 00753

CNFRDL, CNFRDL

ក
ē
ц,
ΰ.
ឥ
ã.
ц.
ΰ.

.)

1)

.

<pre>155 6V = 6(LG) 160 SUMC = SUMC+6V SUMC = SUMC+6V*T(LTA) C CHECK FOR LAST CONUUCTOR IF(ISO1(JJ1).6T.0) 60 T0 135 T2 = DD*T(L)+DN*(SUMCV+Q(L))/SUMC C 0BTAIN THE CALCULATED TEMPERATURE DIFFERENCE C 0BTAIN THE CALCULATED TEMPERATURE C STORE THE NEW TEMPERATURE C T1(1) = T2 C T1(1) = T2</pre>	C SAVE THE MAXIMUM ARTHHMETIC RELAXATION CHANGE 1 FITCOM.E.T1) GO TO 165 TCGM = T1 TCGM = T1 TCGM = T1 TCGM = C 105 CONTINUE C SEE FF RELAXATION CRITERIA WAS MET 165 CONTINUE C SEE FF RELAXATION CRITERIA WAS MET 170 CONTINUE C STORE THE MAXIMUM ARITHMETIC RELAXATION CHANGE 170 CONTINUE C STORE THE ARITHMETIC TEMPERATURE CHANGE TCGM = 0.0 DO 180 I = L1.00 C E = 16.1 T = ARCITIL-VIEI)	IF(T1.LT.TCGM) GO T0 180 TCGM = T1 KON(30) = I KON(30) = I C SEE F ATMPCA WAS SATISFIED IF(TCGM.GT.CON(11)) GO TO 125 CON(16) = TCGM 185 KON(12) = 0 CALL VARE 2 CON(12) = 0 CALL VARE 2 CON(12) = 0 CALL VARE 2 CON(12) = 0 CALL VARE 2 CON(12) = 0 TF(XON(12) NE.0) GO TO 105 CON(12) = CON(1) TSTEP = DELTA+0.95	C CHECK FOR TIME TO PRINT IF(TSUM.GE.CON(18)) 60 TO 190 IF(TSUM.GE.CON(18)) 60 TO 190 IF(NON(7)) 60 TO 10 IF(NON(7)) 60 TO 10 CALL OUTCAL GG TO 10 CALL OUTCAL 190 TPRINT = TRINIT+TSUM 195 CALL OUTCAL IFFINITESUM 195 CALL OUTCAL IFFINIT+TSUM 195 CALL OUTCAL NTH = IE NTH = IE NTH = IE NTH = NC 995 WHITE(6,885) 60 TO 1000 994 WHITE(6,887) 60 TO 1000 997 WHITE(6,887) 60 TO 1000 997 WHITE(6,887) 60 TO 1000 997 WHITE(6,887) 60 TO 1000 997 WHITE(6,887)
175+++ 175++ 1775+ 1775+ 1775+ 181++ 181++		1974 1974 2005 2005 2005 2005 2005 2005 2005 200	2128 21138 21138 21138 21138 21138 21138 22138 22138 22138 22138 22138 22138 22138 22138 22138 22138 22138 2214 22138 2222 2222
00754 00755 00755 00755 00755 00755 00762 00762 00762			010025 010025 010022 010022 010010022 010010022 010010022 0100100022 0100100022 0100100022 0100100022 0100100022 0100100022 0100100022 010022 00022 00022 00022 00022 00022 00022 00022 00022 00022 00022 00022 00022 00022 000022 00000 000000

A - 22

Ž	
CNFF	
gr.	
CNF	

•.

•

....

•

SEQUENCE)	MESSAGE (S)
NG PSEUDO-COMPUTE ATIVE) ABLE)	0 *DIAGNOSTIC*
WRITE(6.888) I 50 TO 1000 WRITE(6.889) ALL EVIT ALL EVIT ALL EVIT COMMI(45H CNFRDL REQUIRES LO Commit(45H CNFRDL REQUIRES LO Commit(24H CSGMIN ZERO OR NEG CORMAT(24H CSGMIN ZERO OR NEG CORMAT(24H CSGMIN ZERO OR NEG CORMAT(24H CSGMIN ZERO OR NEG CORMAT(24H CSGMIN ZERO OR NEG CORMAT(19H NO OUTPUT INTERVAL CORMAT(19H NO OUTPUT INTERVAL	LIOB FORTRAN V COMPILATION. Mbolic Elocatable
998 1000 00 00 00 00 00 00 00 00 00 00 00 00	UNIVAC S
5445 5445 5445 5445 557 567 567 567 567 567 567 567 567 56	RDL OF
01062 01062 01062 01062 01065 01066 01066 01066	CNF

•

.

•

•

CNF AST, CNFAST

)

BIN FOR.* CHFAST.CHFAST UNIVAC 1108 FORTRAN V ATHEWA VERSION 131K-10D CREATED ON 20 AUG 70 This compilation was done on 09 JUN 70 at 14:00:25

SUBROUTINE CHFAST ENTRY POINT 003360

STORAGE USED (BLOCK, NAME, LENGTH)

003373 000070 000047 000004	000000		
TEMP	000000000000000000000000000000000000000		
*CODE *COUST+ *SIMPLE *ARKAYS	•ULATIX TITLE CAP SOURCE COURCE	PC2 PC2 KONST ARRAY FIXCON FIXCON DIMENS	
0001 0000 0002 0002	00000 0010 00110 00110	0015 0015 0015 0015 0017 0021 0021	
•.		and the second	

EXTERNAL REFERENCES (BLOCK . NAME)

:•

•

VARIAL1	OUTCAL	MMICIO	PLYAWM	DZU1WM	VARULZ	EXIT	NERK25	NWDU\$	N1025	NEH105
0.022	0023	0.0.24	0025	0026	0027	0030	0031	0.0.32	0033	0034

STORAGE ASSIGNMENT FOR VARIABLES (BLOCK, TYPE, RELATIVE LOCATION, NAME)

N G Z J
TTOPYTOPY
0000011 0000000 0000000000000000000000
<u></u>
LIN MICON
000000000000000000000000000000000000000
· · · · · · · · · · · · · · · · · · ·
0 >0
10000110000000000000000000000000000000
00000000000000000000000000000000000000
2 2
APRICE C C C C C C C C C C C C C C C C C C
00000000000000000000000000000000000000
STORATION STORAGE
00000000000000000000000000000000000000
000000000000000000000000000000000000000
аахиминий хах
55555555555555555555555555555555555555

21 FEB 71

A - 24

.

\$
_
u.
z
υ
÷.
+
ŝ
1
z
U.

-

1015L 175L	1706	20171	2035L	20551	18642	JOIOC	3030L	30451	186bC	47101	40221	10401	291	5012L	5025L	5376	60L	BAGE	956	7666	
000565 000565 003153	021000	001274	001426	001562	001620	002520	002643	002756	003065	000736	001010	001122	001137	002205	002273	002104	001705	110000	002032	003331	•
1000	1000	1000	1000	1000	1000	1000	1000	1000	1000	1000	1000	0001	0001	1000	1000	0001	0001	0000	0001	0001	
1012L	150L	2015L	1 2032L	2050L	25L	1 3007L	1 3025L	3042L	3065L	4005	4020L	1037L	51	5010L	5022L	1 5040L	5999L	1 885F	90L	1998L	
000361 000527 002405	003240	001256	001100	001517	000120	002500	002603	002717	003061	000731	000774	001105	000072	002200	002256	002370	002402	000000	002014	003323	
0001000101	1000	1000	1000	0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	1000	0001	0001	1000	0000	1000	0001	
000343 1010L 000506 1030L 002055 105L	003236 155L	000133 2010L	001376 20301	001511 2045L	000251 2256	002473 3005L	002575 3020L	002676 3040L	003055 3060L	003116 4000L	000757 40176	001071 4035L	001134 49996	002177 5005L	002242 5020L	002353 5037L	002376 5998L	001762 85L	000026 889F	003314 997L	••••
0001 1000	1000	1000	0001	1000	0001	0001	0001	1000	0001	0001	0001	1000	1000	1000	1000	1000	0001	1000	0000	1000	
000322 1005L 000462 1025L 000636 1045L	003223 140L	000663 1999L	001336 2025L	001452 2042L	001614 2065L	001651 3000L	002541 3017L	002673 3035L	003027 3055L	000212 40L	000756 4015L	001033 4030L	001130 4998L	001137 5000L	002225 5017L	002337 5035L	001702 55L	001753 65L	000023 888F	003306 996L	
00010001	0001	1000	0001	0001	0001	0001	0001	0001	0001	0001	0001	1000	0001	1000	0001	1000	0001	0001	0000	0001	
1 003336 1000. 1 000422 10200. 1 000523 10401	1 003161 120L	1 000660 1998L	1 001330 2020L	1 001431 2040	1 001610 2060L	1 001646 2999L	1 002523 J015L	1 002650 3032L	1 002764 3050L	1 003113 3999L	1 000737 4012L	1 001025 4025L	1 000237 456	1 001773 5006	1 002224 5015L	1 002301 5030L	1 002125 5466	1 002405 6000L	0 00016 887F	1 003300 995L	** <u>*</u>
000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	

SUBROUTINE CNF.ST AN EXPLICIT EXECUTION SUBROUTINE FOR SINDA FORTRAN V THE SHORT PSEUDO COMPUTE SEQUENCE IS REGUIRED NODES WITH CSG RELOW DITMEI RECEIVE STEADY STATE SOLUTION NO RACKING UP IS DONE OR ALLOWED.	INCLUDE COMMALIST COMMON /TITLE/H(1) /TEMP/T(1) /CAP/C(1) /SOURCE/Q(1) /COND/G(1)	COMMON /PCL/NSOI(1) /PC2/NSO2(1) /KONST/K(1) /ARRAY/A(1) COMMON /FIYCON/KON/1) /YCPACF/ND1M.*XTN.X(1)	COMMON /DIMENS/ MND/NNA, WIT/NGT/NCT/NAT/LS02	DIMENSION CON(1) (XK(1) (X(1)))	EQUIVALENCE INOVILITICOVILITICANTICATIONICATIONICATIC	INCLUCE DEFF.LIST	######################################	CONTROL CONSTANT 1 CONTAINS THE NEW PRORLEM TIME (TIMEN)	CONTROL CONSTANT 2 CONTAINS THE TIME STEP USED (DTIMEU)	CONTRUL CONSTANT 3 CONTAINS THE PROBLEM STOP TIME (TIMEND)	CONTROL CONSTANT 4 CONTAINS THE TIME STEP FACTOR EXPLICIT (CSGFAC)	CCS IS THE INPUT NUMBER OF ITERATION DO LOOPS. INTEGER (NLOOP)	CC6 CONTAINS THE DIFFUSION TEMPERATURE CHANGE ALLOWED (DTMPCA)	CC7 CONTAINS THE OUTPUT EACH ITERATION SWITCH (OPEITR)	CCB CONTAINS THE MAXIMUN ALLOWED TIME STEP (DTIMEH)	CC9 CONTAINS THE NEW ARITHMETIC TEMP. DAMPING FACTOR (DAMPA)	CC10 CONFAINS THE NEW DIFFUSION TEMP. DAMPING FACTOR (DAMPD)	CC11 CONTAINS THE MAXIMUM ALLOWED ARITHMETIC TEMP. CHANGE (ATMPCA)	CC12 CONTAINS THE HACKUP SWITCH CHECKED AFTER VARIABLES (BACKUP)	CC13 CONTAINS THE PRESENT TIME OR PROBLEM START TIME (TIMEO)	CC14 CONTAINS THE NEAN TIME BETWEEN AN ITEPATION	CCIS CONTAINS THE DIFFUSION TEMPERATURE CHANGE CALCULATED (DTMPCC)	CCI6 CONTAINS ARITH/SETIC TEMPERATURE CHANGE CALCULATED (ATMPCC)	CONTROL CONSTANT 17 IS RESERVED FOR THE CASE MINIMUM (CSGMIN)
0000	•						Cetter	Ū	U	Ű	Ű	۔ ں	U	U	۔ ي	Ū	Ū	U	U	U	J	U	U	U U
*****	* * • • • •			* 1 \Q \	+ • •	1*	7*	*2	**		7+	*~	**	7*	-2	+2	*4	**	*	- 7.	4.4	1*	*	*
00101 00101 00101 00101	00103	00105	20100	01100	00112	00113	00113	00113	00113	00113	00113	00113	00113	E1100	00113	00113	00113	00113	00113	E1100	00113	00113	00113	00113

	(OUTPUT) (ARLXCA)	(DTIMEL)	(CSGMAX)	(CSGRCL)	(DRLXCA)	(LINECT)	(PAGECT)	(ARLACC)	(ENGBAL)	(BALENG) (NOCOPY)				-L-MTEST) -U-VTEST)	(LAXFAC)																								
	L A ALLOWED ED. INTEGED				ALLOWED	ALCULATED		ULATED S.2=GENERAL	1N - 0UT	INPUT				7-C-1-C-X-1-X-1-X-1-X-1-X-1-X-1-X-1-X-1-X	MSON														1817100	(OT MOA			181-75(141)	14001-101					
	UT INTERVA ON CRITERI	E STEP		ED	IN CRITERIA	KN CHARIGE C	ER	HANGE CALC	HE SYSTEP.	NACE, USER	CSGMIN	ARLACC	ATMPCC	IR CONSTANT	VAL FOR CI														- (-),100 -		100000.1		0.54 CONC						
	IS THE OUTP	ALLOWED TIM		SE ALLOWED	A RELAXATIO	TER INTEG	VTER INTEG	ELAXATION C	ALANCE OF	ENERGY BALA	E NUMBER OF	E NUMBER OF	E NUWRER OF	JMMY INTEGE	ATION INTER			[.E+8	.	= 1.E+B									CONVERSION	= CON(B)	= CON(21)4		all Tetep =			1101			
-	LA CONTAI	E MINIMUM	E C/SG MAX	E C/SG RAN	E DIFFUSIO	E LINE COU	E PAGE COU	ITHMETIC K	E FNERGY R	E DESIRED E NUCOPY S	LATIVE NOD	LATIVE NOD	LATIVE NOD	CONTAIN D	I-LINEARIZ	AI PHESEN	KON(5) = 1	CON(8) =	CON(9) =	CON(19)	0 60 10 99 60 10 995					TO 997			01. ET CONT	B)) TSTEP	21)) TSTEP	TSUM	P. GT. CONCI		TSUM+TSTEP	INDOL TIMO			TO 40
IFAST, CNFAS	ONTAINS TH	ONTAINS TH	ONTAINS TH	ONTAINS TH	TH SULAINS TH	UNTAINS TH	ONTAINS TH	CONTAINS AR	ONTAINS TH	ONTAINS TH	ONTAINS RE	ONTAINS RECONTAINS RE	UNTAINS RE	10-41-42-43	IS THE OUAS	IS NOT USED	((2) • FE • 0)	((8) LE.C.)	1(18) LE.O.)	1(19) LE.0.	4(21).LE.0. 1(31).NE.0)	-1.0	NNA+NND	WION	NTH+NND	[M.LT.0) 60	I+ONI	= CON(21)		TEP.6T.CON	TEP.LT.CON	= CON(18)-	25 M+2_0+TCTF	I = TSTEP.	= TPRINT+		= 0.0 r + 1	0.0	A.LE.0) 60 I = NN.NFIC
Û	CC19 CC19		0000	CC25 C	CC26 C	CC28	CC29		CC32	CC33	CC35	CC36 CC37	CC38 0	1-6000	CC49		IF (KO	IF (CO)	IF (COT	IF (CO)	IF (CO	PASS		NLA	NTH =			TSTEP	5 TSUM	10 IF (TST	IF (TS)	15 TSTEP	60 T0	25 CON(2)	CONCI	00 30	10	30 X(LE)	IF (NN/ D0 35
	000	00	J U I	υu	U	U U	U	U,U		υU	0	ÚU	U	υų	0	υ																							
	***		2,21	* *	-1	**	FI	* *	: # 1	**	*	**	- 7	**	F	**		#6 6	+ +	-	+ + 	15+	10+ 12+	18#	194	21+	22*	5 # 1 5 # 1	22 +	274	28*	1 4 1 0 1 1 1	419 405	***	# 1 50 10	* * 9 E	17+	+ 6 P	* * 0 1 *
	21100	1100	00113	00113	00113	CT100	00113	£1100	21100	00113	21100	00113	00113	00113	51100	00113	00115	21100	00123	00125	00121	00133	00134	00136	00137	14100	00143	24100	00146	00151	00153	00160	00161	00164	00165	00167	00172	00174	00176 00200

J

e 1.	•																NEW	#NEW	HUN+	n=##				#UEA	NUN+	2=++			4		NUN N					•								*NEW	NSN+							
CNFAST, CNFAST										CON(1) = TPRINT+TSTEP	CON(2) = TSTEP					INCLUCE THE PARTING FOR THE POOL		LA = FLO(5,17,4502(22))	LK = FLD(22,14,NSQ2(J2))	60 T0 (1005,1010,1015,1020,1020,1035,1040,1045), NTYPE	1005 CALL DIDIWRTT(1),ACLA),XXKLK1,C(1))			LA = FLD(5,17,NS02(J2))	LK = FLD(22+14,NSQ2(J2))	CALL DIDIWM(T(T), A(LA), XK(LK), C2)			1020 CALL DIDIMM(T(1),A(LA),XK(LK),C1)	J2 = J2+1	<pre>LA = FLD(5,17,1)5Q2(J2))</pre>	LK = FLD[22+14-NSQ2(J2))	CZ Z XYLK + XK (LA)	60 10 1995 10 1995 10 1915 10 10 10 10 10 10 10 10 10 10 10 10 10		IO30 CALL PLYAWM(A(LA),T(I),A(LA+I),XK(LK),CI)	1032 U2 = U2+1				1035 C1 = XK(LK) * XK(LA)	60 TO 1032	ID40 CALL PLYAWM(A(LA),T(I),A(LA+I),XK(LK),CI)			• C2 = XK(LK) + XK(LA)	60 T0 1998	1045 CALL USDIWM(I(I))(COM(I4))A(LA))AK(LK)(CI))			• 2000 CONTINUE	END
	424	すりま	オリオリ			4.9		105		55	53	135	ц ц				200	28.	584	581	105		285	28	581	583	200	200	200	284	581	585	283		5.00	584	28	58	200		2.84	58.	585		285	58.	5.8	80.4		99	581	284
	00203	00204	00200	10200	01200	00214	1200	0.0216	01212	00220	00221	00222	10000	19900	12200	11000	12200	00234	00235	00236	0.237	0.0200	00242	00243	00244	00245	01200	0.025.0	00251	00252	00253	00254	00255	002570	00250	00261	00262	00263	90200	00200	00267	00270	00271	2/200	00274	00275	00276	00277	10200	00302	00303	00204

CNFAST, CNFAST

2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2		2 2 0 2 2 4 7 4 9	3 3 0 3 3 0 3 4 4 4 3 4 4 4	• • • • • • • • • • • • • • • • • • •
Include Varoilist If(FLDUE Varoilist If(FLDUE Varoilist If(FLDUE Varoilist NTYPE = FLD(05:17,HS02(J2)) NTYPE = FLD(22:14,MS02(J2)) La = FLD(22:14,MS02(J2)) La = FLD(22:14,MS02(J2)) La = FLD(22:14,MS02(J2)) La = FLD(22:14,MS02(J2)) 60 T0 (4005,4010,4015,4020,4025,4030,4035,4040,4030), NTYPE 4005 G(1) = XX(LK)+0(1) 4010 G1 = 0.0 4012 G1 0 4999 4015 G1 0 4999 4015 G1 0 4999 4015 G1 0 4998 4015 G1 0 4998 4017 Call D11MM(CON(14))A(LA))XK(LK),021 4020 Call D11MM(CON(14))A(LA))XK(LK),021 4020 Call D11MM(CON(14))A(LA))XK(LK),021 4020 Call D11MM(CON(14))A(LA))XK(LK),021	LA = FLD(5.17,11502(J2)) LK = FLD(22,14,NS02(J2)) 60 T0 4017 4025 G1 = XK(LK)*XK(LA) 61 T0 4022 4030 Call D11WM(CON(14).A(LA).XK(LK).01) J2 = J2+1 LA = FLD(5.17,11502(J2)) LX = FLD(22,14,NS02(J2)) 1 LX = FLD(22,14,NS02(J2)) 2 = XK(LK)*XK(LA).XK(LK).01) 4035 Call D101WM(CON(14).A(LA).XK(LK).01)	<pre>4037 J2 = J2+1 LA = FLD(5,17,NS02(J2)) LA = FLD(22,114,NS02(J2)) LA = FLD(22,114,NS02(J2)) LA = XK(LK) *XK(LA) 4040 01 = XK(LK) *XK(LA) 60 T0 4037 4999 01 = 01+02+0(1) 4999 J2 = J2+1 5000 CONTINUE 50 U = 01 = 01+02+0(1) 50 U = J1+1 50 U = J1</pre>	<pre>F(FL0[2:1.NS01(J1)).E0.0) 60 T0 3000 NTYPE = FL0[0:5.NS02(J2)) LA = FL0[5:14.NS02(J2)) LA = FL0[2:14.NS02(J2)) Ex = FL0[2:14.NS02(J2)) G0T0(2005.2010.2015.2020.2025.2030.2035.2040.2045.2050.2055. 2005 TM = 1(1).HT(LTA))./2.0 2005 TM = 1(1).HT(LTA))./2.0 2007 CALL D101WM(TM.A(LA).XK(LK).6(LG)) 2010 TM = T(1) 2010 299 2010 CALL D101WM(T(1).A(LA).XK(LK).61) 2015 CALL D101WM(T(1).A(LA).XK(LK).61) 2017 22 = J2+1</pre>	LA = FLD(5,17,4:502(J2)) Lk = FLD(22,14,NSU2(J2)) Call D101WM(TT(LTA),A(LA),XY(LK),G2) GU TU 2998
0,00,00,00,00,00,00,00,00,00,00,00,00,0		5 5 6 6 6 7 7 7 9 9 9 9 9 9 9 9 9 9 9 9 9 9	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.000
003022 003114 003114 0003114 0003114 0003222 000322 00032 00032 00030 000000	00327000333100033310003331000033310000333130000333130000333130000333130000333130000333130000033313000003313000003313000003313000000	00342 00344 00344 00344 00345 00355 00355 00355 00355 00355 00355 00355 00355 00355 00355	00362 00364 00365 00367 00367 00377 00377 00377 00375 00375	00377 00400 00401 00401

÷

	* * * * * *	
		10 65
·((LG))	((LK),62) K),61)	.6(LG)) .6(LG)) .61*62
(17) *XK ([K) (17) *(17)	A(LA+1),XK LA+1),XK(L	LA) , XK(LK) LA) , XK(LK)) 6(L6) = ([11+72) ([11+72) , NSal (J1))
-A) - A(LA) - XK SG2(J2) -A) -A) - X -A) - X - X - X - X - X - X - X - X - X - X	502(J2)) VS02(J2)) N).T(LTA). A).T(LTA).	<pre>502(J2) 502(J2)) 40)/2.0 4)/2.0 5)/2.0 5)/1.1.62) 5)/1.1.62) 5)/2.0 1.1.1.62) 5)/2.0 1.1.1.62) 5)/2.0 1.1.1.62) 5)/2.0 5) 60 10 5 5) 50 10 5</pre>
K(LK) *XK(L 2017 1014 2017 2017 2017 2014 2014 2017 2014 2014 2014 2014 2014 2014 2014 2014	241 ED (5, 17, MS ED (22, 14, M ED (22, 14, M (A (L/ 2998 EV AWM (A (L/ 2042 EV AWM (A (L/	The second secon
	CALCER II CONCERNENCE	Image: 100 million Image: 100 million Image: 100 mil
2020 2025 2025 2032 2032 2035 2035	2045	2065 2065 2065 2065 2066 2999 655 665 655
00000000 3333333334444444 668864844444444	00000000 4444444 444444	6644 6644 6644 6644 6644 6644 6644 664
00000000000000000000000000000000000000	004400044000 00445000445000 0044500004450000 0044500004500000000	00000000000000000000000000000000000000

CNFAST, CNFAST

j. A - 29

•.•
		333 10 10 10 10 10 10 10 10 10 10 10 10 10	3 3 5 7 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	Q BRQ BI 4 UU 4 U 4 V 4 V 4 V 4 V 4 V 4 V	2 U 1 Z 2 H 4 H
CNFAST.CNFAST	ON(35) = 1 = T(1)+TSTEP=(G(1)-X(LE)=T(1))/C(1) 1 = G(1)/X(LE) 1 = G(1)/X(LE) 2 = AUST1-T(1)) 1 = AIST1-T(1)) 1 = AIST1-T(1) = AIST1-T(1)) 1 = AIST1-T(1	LX = 0.0 UL = J1 J2 = J2 O 145 L = NN.PNC UMC = 0.0 UMC = 0.0 C(1.61.1) 60 T0 6000 F(1.61.1) 50 T0 6000 F(1.61.1) 50 T0 6000 TYPE = FLD(0.5.NS02(JJ2)) A = FLD(5.177(S02(JJ2)) TYPE = FLD(2.14.NS02(JJ2)) TYPE = FLD(2	0 T0 5999 1 = 0.0 ALL D101WMA(T(L),A(LA),XK(LK),02) ALL D101WM(CON(14),A(LA),XK(LK),02) ALL D101WM(CON(14),A(LA),XK(LK),01) 0 T0 599U 2 = JJ2+1 2 = FL0152,14,NS02(JJ2)) A = FL0152,14,NS02(JJ2)) K = FL0122,14,NS02(JJ2))	0 TO 5017 1 = XK(LK) + XK(LA) 0 TO 5022 ALL DIDIMM(CON(14) • A(LA) • XK(LK) • 01) J2 = JJ2+1 A = FLD(5,17,15G2(JJ2)) A = FLD(5,17,15G2(JJ2)) C = XK(LK) + XK(LA) 0 TO 5998 ALL DID14M(CON(14) • A(LA) • XK(LK) • 01) J2 = JJ2+1 A = FLD(5,47,155Q2(JJ2)) A = FLD(5,47,155Q2(JJ2))	H = FLD(27.14.1)SQ2(JJ2)) U TO 5012 1 = XK(LK)+XK(LA)
•		20000000000000000000000000000000000000	200 200 200 200 200 200 200 200 200 200	5025 5020 5025 5030 5025 5033 5033 5033	2040
	88888888888888888888888888888888888888	**************************************			1134
	00000000000000000000000000000000000000	00000000000000000000000000000000000000	00000000000000000000000000000000000000	000000000000000000000000000000000000000	00613 00614 00615

.).

CNFAST . CNFAST

2 2 2 0 2 2 2 2 0 2 2 0 0 2 0 0 0 2 0 0 0 0	A 9 = 8 8 =		8 8 9 8 9 8 9 8 9 8 9 8 9 8 9 8 9 8 9 8	104 *****
0 TO 5037 (L) = 01402+0(L) U2 = JJ2+1 0NTINUE 01 = JJ1+1 6 = FLO(5.16.NS01(JJ1)) AD TA = FLO(22.14.NS01(JJ1)) TA = FLO(22.14.NS01(JJ1)) F(FLOT2.1.NS01(JJ1)).E0.0) F(FLOT2.1.NS01(JJ1)).E0.0) F(FLOT2.1.NS02(JJ2)) A = FLO(22.14.NS02(JJ2)) A = FLO(22	3060.3063), NITPE M = (T(L)+T(LTA))/2.0 ALL DIOLWM(TM,A(LA),XK(LK),G(LG)) 0 TO 3999 M = T(L) ALL DIOLWM(T(L),A(LA),XK(LK),G1) ALL DIOLWM(T(L),A(LA),XK(LK),G1) AL = JJ2+1 M = FLO(22,17,NSO2(JJ2)) A = FLO(22,17,NSO2(JJ2))	ALL DIDIWM(T(LTA),A(LA),XK(LK),62) 0 T0 3998 0 T0 3017 1 = XK(LK)+XK(LA) 0 T0 3017 ALL DIDIWM(T(L),A(LA),XK(LK),61) 2 = JJ2+1 2 = JJ2+1 2 = JJ2+1 3 = FLD(22,14,NS02(JJ2)) 5 = XK(LK)+XK(LA) 6 T0 3998 M = (T(L)+T(LTA))/2.0 M = (T(L)+T(LTA))/2.0 M = T(L) 7 = T(L)	ALL PLYAWM(A(LA),T(L),A(LA+1),XK(LK),G1) J2 = JJ2+1 J4 = FLD(5,17,NSO2(JJ2)) K = FLD(2,14,NSO2(JJ2)) ALL PLYAWM(A(LA),T(LTA),A(LA+1),XK(LK),G2) O T0 3998 O T0 3092 ALL PLYAWM(A(LA),T(L),A(LA+1),XK(LK),G1) ALL PLYAWM(A(LA),T(L),A(LA+1),XK(LK),G1)	U2 = JJ2+1 A = FLD(5.17, MS02(JJ2)) K = FLD(22.14, NS02(JJ2)) 2 = XK(LK) *XK(LA) 0 TO 3998 M = (T(L)+T(LTA))/2.0 M = (T(L)A) ALL D?D1MA(TMACON(14), A(LA), XK(LK), G(LG)) ALL D?D1AA(TMACON(14), A(LA), XK(LK), G(LG)) 0 TO 3009 M = T(LTA) 0 TO 3007
0070W7JJHHUHZJJ0 8600 6600 1 6600	5015 01 02 1 2015 0 00 0 2017 0 0 0 2017 0 0 0 2017 0 0 0 0 2017 0 0 0 0 0 2017 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 4 6 C 4 6 6 F F C C 6 6 6 C 0 3 3 3 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	2040 2040 2040 2040 2040 2040 2040 2040	2022 2022 2022 2022 2022
00000000000000000000000000000000000000	00000000000000000000000000000000000000	000 000 000 000 000 000 000 000 000 00	00667 00671 00671 00671 00673 00673 00675 006775 006775	00700 00705 00705 00705 00705 00705 00705 00705 00705 00705 00711

.

					5																										×	•				
	CNFAST.CNFAST	3065 TM = T(LTA) 60 TO 3032	3996 GiLG) = 1./(1./G1+1./G2) If(FLD(3.1.NS01(JJ1)).E0.1) G(LG) = 61+62	3999 JJZ = JJZ+1 4000 continue	END IF(FLD(3,1,NSq1(JJ1)).EQ.0) 60 TO 115	11 = 1(L_1)+460.0 12 = 1(L_1)+460.0	6V = 6(10) + 11 + 12 + 12 + 12 + 12 + 12 + 12 + 1		LCU SUMC = SUMC+64	C CHECK FOR LAST CONDUCTOR TO THIS NUDE IF(NSO1(JJ1).61.0) 60 TO 110	TI = DAMPN+(SUMCV+G(L))/SUMC+DAMPO+T(L) T2 = AHS(T(1)-T1)	IF (RLX+6E+T2) 60 TO 140	RLX = 12 KON(37) = L	140 T(L) = T1	IF(RLX.LE.CON(19)) GO TO 155	150 CONTINUE 155 CONTINUE	155 CONTSUL # KLX 160 CALL VARBL2	CON(13) = CON(1)	TSUM = TSUM+TSTEP	IF(TSUM.LT.CON(18)) 60 70 10 YUDINT - TDDINT+TCHM	CALL OUTCAL	IF(CON(1)*1+000001.LT.CON(3)) 60 TO 5 NTH = IE	NDIM # NLA	WEIUKN 995 WRITE(6,865)	60 TO 1000 606 Lette (4.886)	G0 T0 1000	997 WRITE(6+887) NDIM	998 WRITE (6,888)	G0 T0 1000 G0D M01TF16.8800	1006 CALL OUTCAL	CALL EXIT Abs formatiunal cufast requires short pseudo-couputs sequence.	086 FORMAT(22H C/SK ZERO OR NEGATIVE)	867 FORMATTISHZUM LUCAILONS AVAILABLET 888 Furmattion No Diimel)	889 FORMAT(19H NO OUTPUT INTERVAL) End	UNIVAC 1108 FURTRAN V COMPILATION, 0 *DIAGNOSTIC* MESSAGE(S) Symbolic DDF Hei ocatanie	
		119*	119*	119+	120+	121+	124	126+	128*	130+	132+	103+	135+	136+	138+	+621	***	142+	****	145+	+44	149*	150+	152*	153+	155+	156*	158+	159*	161*	162+	104	100	167* 168*	AST CC	
\supset		00712	00714	00717	00721	00725	00727	00731	00733	00733	00736	00740	00743	00744	00/43	00751	00754	00755	00757	00760	00763	00764	00767	00771	00773	00776	00177	20010	01005	01010	1010	21010	01015	01010	CNF	

· ****

•

 \mathbf{i}

A - 32

z
٥.
×
لب
z
U
_
÷
<u>e</u> .
<u>.</u>
Ψ,
£.
-

.

GIW FOR.* CHEXPH.CHEXPN Univac 1108 Fortran V Athena Version 131K-10D Created on 20 Aug 70 This compilation was done on 09 Jun 70 at 14:00:19

SUBROUTINE CNEXPN ENTRY POINT 003536

STORAGE USED (BLOCK, NAME, LENGTH)

003551 000072 000047	000000											
EMP VAR		100000	100000	000000	100000	00000	100000	100000	100000	100000	£00000	000010
*CODE *COLST+T *SIMPLE	* ARKAYS *BLANK	TITLE	CAP	SOURCE	COND	PC1	PC2	KONST	ARRAY	FIXCON	XSPACE	DIMENS
0000 0000	0004	9000	0010	1100	0012	0013	0014	0015	0016	0017	0020	0021

EXTERNAL REFERENCES (BLOCK . NAME)

VARBLI	MATOTO	PLYAWM	D2U1WM	VARULZ	OUTCAL	EXII	NERK25	ЕХР	SUDWN	N1025	NEKI05
0022	0023	0024	0025	0026	0027	0030	0031	0032	0033	0034	0035

STORAGE ASSIGNMENT FOR VARIABLES (BLOCK, TYPE, RELATIVE LOCATION, NAME)

0002 R 000021 C1 0012 R 000000 G 0002 I 000004 I 0002 I 000015 J2 0002 I 000004 LA2 0021 I 000004 LA2 0021 I 000004 LA2 0022 I 000004 NCT 0002 R 000010 NTH 0002 R 000014 TCGM
0017 R 000000 CON 0002 R 000037 DN 0002 I 000012 J1 0002 I 000012 J1 0021 I 000017 LA 0021 I 000005 NAT 0021 I 000000 NNA 0011 R 000000 0 SQ2 0011 R 000000 0 SQ2 0017 R 000000 0 SQ2 0017 R 000000 0
0002 R 000015 CKM 0002 R 000036 DELTA 0002 I 000044 L 0002 I 000044 L 0002 I 000044 L 0002 I 000004 L 0002 I 000005 L1 0002 R 000000 PASS 0002 R 000000 PASS
0010 R 000000 C 0002 R 000040 C 0002 R 000040 C 0002 I 000042 JJI 0002 I 000005 LEA 0002 I 000035 LEA 0021 I 000035 LEA 0021 I 000026 LTA 0022 R 000022 NNT 0002 R 000042 SUMC
0016 R 000000 A 0002 R 000022 C2 0002 R 000024 GV 0002 I 0000024 EV 0002 I 0000010 K 0021 I 000001 LE 0021 I 000007 L502 1021 I 000007 NUM 0021 I 000007 NUM 0022 I 000007 NUM

21 FEB 71

Z
a.
~×
- U
z
ü
Z
ā
x
64
₹
-75

. .

•

#15500

5010L 5032L 5938L

0022555 0022565 002565 001765 001765 001765

003136 003225

002650

001765 65L 000023 886F 003464 996L

SUBROUTINE CNEAPN EXPLICIT EXPONENTIAL DIFFERENCING ROUTINE FOR SINDA FORTRAN V THE SHORT PSEUDD-COMPUTE SEQUENCE IS REQUIRED INCLUDE COMMALLIST COMMON /TITLE/H(1) /TEMP/T(1) /CAP/C(1) /SOURCE/G(1) /COND/G(1) COMMON /TITLE/H(1) /YENSQ2(1) /KONST/K(1) /ARRAY/A(1) COMMON /DIMENS/ NND.NNA.HNT.NGT.NAT.LSOI.LSO2 DIMENSION CON(1) /XK(1).NX(1)).(X(1).NX(1)) EQUIVALENCE (KON(1)).CON(1)).(K(1).XK(1)).(X(1).NX(1)) υu

INCLUDE DEFFILIST END <u>................</u>

CNEXPN. CNEXPN

(CSGMIN) (OUTPUT) (ARLXCA) (LOOPCT) (DTIMEL) (DTIMEL) (DTIMEL) (CSGRAL) (CS	(ARLXCC) L (LSPCS) (BALENG) (BALENG) (NOCOPY) (NOCOPY) -U-VTEST) (LAXFAC)		AL K
CONTROL CONSTANT 17 IS RESERVED FOR THE C/SG MINIMUM CONTROL CONSTANT 18 CONTAINS THE OUTPUT INTERVAL CC19 CONTAINS THE ARITHMETIC RELAXATION CRITERIA ALLOWED CC20 CONTAINS THE NUMBER OF RELAXATION LOOPS USED.INTEGER CC22 IS FOR THE INFUT TIME STEP CC22 IS FOR THE INFUL TIME STEP CC22 CONTAINS THE C/SG MAXIMUM CC22 CONTAINS THE C/SG MAXIMUM CC22 CONTAINS THE C/SG MAKIE ALLOWED CC23 CONTAINS THE C/SG MAKIE ALLOWED CC23 CONTAINS THE C/SG MAKIE ALLOWED CC22 CONTAINS THE C/SG MAKIE ALLOWED CC22 CONTAINS THE C/SG MAKIE CALCULATED CC23 CONTAINS THE C/SG MAKIE CALCULATED CC23 CONTAINS THE C/SG MAKIE CALCULATED CC23 CONTAINS THE DIFFUSION RELAXATION CRITERIA ALLOWED CC23 CONTAINS THE DIFFUSION RELAXATION CRITERIA ALLOWED CC29 CONTAINS THE DIFFUSION RELAXATION CRITERIA CALCULATED CC29 CONTAINS THE DIFFUSION RELAXATION CRITERIA CALCULATED CC29 CONTAINS THE DIFFUSION RELAXATION CRITERIA CALCULATED	CC30 CONTAINS ARTHMETIC RELAXATION CHANGE CALCULATED CC31 IS INDICATOR. 02THERMAL SPCSVILATHERMAL LPCS.2=GENERAL CC32 CONTAINS THE ENERGY BALANCE OF THE SYSTEM. IN - OUT CC33 CONTAINS THE NESTRED FNERGY BALANCE. USER INPUT CC34 CONTAINS RELATIVE NODE NUMMER OF CSGMIN CC35 CONTAINS RELATIVE NODE NUMMER OF CSGMIN CC35 CONTAINS RELATIVE NODE NUMMER OF TALXUSERS CC37 CONTAINS RELATIVE NODE NUMMER OF ATLXCC CC39 CONTAINS RELATIVE NODE NUMMER OF ATLXCC CC39 CONTAINS RELATIVE NODE NUMMER OF ATMOCC CC39 CONTAINS RELATIVE NODE NUMMER OF ATMOCC	FID F(COH(4).LE.0.0) CON(4) = 1.0 F(CON(6).LE.0.) KON(5) = 1.10 F(CON(6).LE.0.) CON(5) = 1.E+8 F(CON(6).LE.0.) CON(6) = 1.E+8 F(CON(11).LE.0.) CON(1) = 1.E+8 F(CON(11).LE.0.) CON(1) = 1.E+8 F(CON(11).LE.0.) CON(1) = 1.E+8 F(CON(19).LE.0.) CON(19) = 1.E+8 F(CON(19).CON(19).E+8 F(CON(19).LE.0.) C	IF(I.LT.0) G0 T0 998 TF(I.LT.0) G0 T0 998 TSTEP = CON(18) TPRINT = CON(13) TRITALIZE TIME SUM RETWEEN OUTPUT INTERVALS TSUM = 0.0 DOES OLD TIME PLUS THE OUTPUT INTERVAL EXCEED THE STOP TI IF(CON(13)+CON(13)) E0 T0 10 DONT EXCEED IT CON(13)+CON(13)+CON(3)) G0 T0 10 DONT EXCEED IT TSTEP = CON(13)) G0 T0 15 IF(TSTEP_LECCON(13)) G0 T0 15 TSTEP = CON(10)) 25.30.20 DOFS THE TIME SUM PLUS THE TIME STEP EXCEED OUTPUT INTERV
•			1 0 1 1 0 0
000000000000000000000000000000000000000	000000000000000	Ų	νυνυνο
ស៊ីស៊ីស៊ីស៊ីស៊ីស៊ីស៊ីស៊ីស៊ីស៊ីស៊ីស៊ីស៊ីស	ស៊ីស៊ីស៊ីស៊ីស៊ីស៊ីស៊ីស៊ីស៊ីស៊ីស៊ីស៊ីស៊ី - ស៊ីស៊ីស៊ីស៊ីស៊ីស៊ីស៊ីស៊ីស៊ីស៊ីស៊ីស៊ីស៊ីស	**************************************	**************************************
511100 511100 511100 511100 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 5111000 511000 511000 511000 511000 511000 511000 511000 511000 511000 511000 511000 511000 511000 511000 511000 511000 511000 511000 5110000 5110000 5110000 51100000 51100000000000000000000000000000000000		00115 00115 00115 00115 001125 001125 001125 001125 001125 001125 001125 001125 001125 001125 001125 001125 001125 001125 001125 001125 001125 00115 00115 00115 00115 00115 00115 00115 00115 00115 00115 00115 00115 00115 00115 00115 00115 00115 00015 0015 00005 00015 00005 00015 00005 00015 00005 00015 00005 00015 00005 00015 000000	00000000000000000000000000000000000000

:	CNEXPN

																													#NEI												1974 *	1		
		D OUTPUT INTERVAL				E STEP ALLOWED		Ŭ	n	LOCATIONS					THE EXINA LOCATIONS																35,1040,1045), NTYPE										, manatar and a			
CNEXPN.CNEXPN	TSTEP = CON(18)+TSUM 60 to 30	DOES TIME SUM PLUS TWO TIME STEPS EXCEED	APPROACH THE OUTPUT INTERVAL GRADUALLY	TSTEP = (CON(18)-TSUM)/2.0 STORE DELTA TIME STEP IN THE CONSTANTS	CON(2) = TSTEP	IS THE TIME STEP USED LESS THAN THE TIME TETTSTEP IT CONTOINT GO TO 497	CALCULATE THE NEW TIME	COV(1) = TPRINT+TSUM+TSTEP complife the mean time between itebation	COMPUTE THE MEAN TIME DEFINERY LIEAN ANY CONTINE TO ANY ANY CONTINE TO ANY ANY CONTINE TO ANY	ZERO OUT ALL SOURCE LOCATIONS AND EXTRA	00 35 I = 1. NND	LE = IE+I V(: F) = 0.0		CONTINUE	SHIFT THE ARITHMETIC TEMPERATURES INTO TF/NNA.LE.D) GO TO 45	DO 40 I = L1+NNC			CONTINUE	KON(12) = 0	CALL VARBLI		U2 = 1	TCGM = 0.0	CAM - 1.578 Calculate o Sum and 6 Sum	DO 85 I = 1.NND		INCLUGE VARC.LIST TF(FLD(1.1.NSq1(J1+1)).FQ.N) GO TO 2000	NTYPE = FLD(0,5+NS02(J2))	LA = FLD(5,17,14502(J2)) 1	GO TO (1005-1010-1015-1020-1025-1030-10	CALL DID1WM(T(I),A(LA),XK(LK),C(I))	CALL DIDIWM(T(I),A(LA),XK(LK),C1)	U2 = J2+1	LA = FLD(5;17;1502(J2)) 18 = FLD(2;17;1502(J2))	CALL DIDIWM(T(1),A(LA),XK(LK),C2)	60 TO 1998	CI = Xk(LK)*Xk(LA) 60 T0 1012	CALL DIDIWH(T(I),A(LA),XK(LK),C1)	J2 = J2+1	LA = FLD(5,17,NSG2(J2)) 1 v - FI D(20,14,NSG2(J2))	C2 = XK(LK)+XK(LA)	60 T0 1998 Call PI Vauntary Al. III) . A/I A+1) . YY /I Y) . C	CALL TLIANPIALLA/ III LIZIEATETTIAN
	20	Ļ	ŝ		30									35					40	5 1												002	010	012				015	020				0.21	220
		U.	U	ų	•	υ	U	. C	: د	Ų				. (ų										. U							-	-	-				-	-				-	4
	*60 *07		1 1 1 1 1 1 1	* * # 4 # 4	*9	474	+6.+	• • •	52+	53+	***	204 4 4	57+	185	59#	61+	429	+ 20	65+	+9 9	67*	• 6 9	70+	11+	13+	24*	15*	76*	76*	76*	76*	16+	76*	16.	76*	76*	16*	76*	76+	16*	76*	16.	76+	0
÷	00165	00166	00167	00171	00172	00172	00173	00175	00176	00176	22100	00202	00204	00202	00205	00211	00214	00215	00217	00221	00222	00225	00226	00227	002200	00231	00234	00235	00540	00241	06243	900244	00246	00247	00250	00252	00253	00254	00256	00257	00260	00202	00263	10201

-

-
۵.
×
u.
z
U.
٠
z
ã
×
١Ĵ.
z
D.
_

•

2 2 0 2 2 1 2 2 4 4 4 4 4	820 UUI 224 455	2320 2000 2000 2000 2000 2000 2000 2000	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	3330 223 44
<pre>60 T0 1999 103u CALL PLYAWM(A(LA).T(I).A(LA+1).XK(LK).C1) 1032 J2 = J2+1 LA = FLD(5.17.NS02(J2)) LK = FLD(22.14.NS02(J2)) CALL PLYAWM(A(LA).T(I).A(LA+1).XK(LK).C2) CALL PLYAWM(A(LA).T(I).A(LA+1).XK(LK).C2) 1035 C1 = XK(LK)*XK(LA) 1040 CALL PLYAWM(A(LA).T(I).A(LA+1).XK(LK).C1) 1040 CALL PLYAWM(A(LA).T(I).A(LA+1).XK(LK).C1)</pre>	UZ = UZ + 1 LX = FLD(5.14.NSQ2(J2)) LX = FLD(22.14.NSQ2(J2)) C2 = XK(LK)*XK(LA) G0 T0 1998 G0 T0 1999 G0 T0 1999 1999 C(T) = C1+C2 1999 UZ = U2+1 2000 CONTINUE END	INCLUDE YARGALIST INCLUDE YARGALIST ITYPE = FLD(0,5)NS02(J2)) LA = FLD(5,17,1502(J2)) LA = FLD(2,14,NS02(J2)) LA = FLD(22,14,NS02(J2)) GG TO (4005,4010,4015,4020,4025,4030,4035,404030), NTYPE 4005 GT 0 4999 4010 G1 = 0.0 4015 CALL DID1WM(T(1),A(LA),XK(LK),02) 4015 G1 - 0.0 4015 G1 - 0.0 4025 G1 - 0.0 4025 G1 - 0.0 4026 G	4 = FLD15.17.MS02(J2)) LK = FLD15.17.MS02(J2)) 60 T0 4017 4024 01 4030 CALL 12 - 22+1 12 - 22+1 12 - 22+1 14 MS02(J2)) 12 - 22+1 14 MS02(J2)) 12 - 22+1 13 = FLD(25+17+MS02(J2)) 14 MS02(J2)) 15 = X(LK)-XK(LA)	60 T0 4996 4035 CaLL DIDINM(CON(14).A(LA).XK(LK).01) LA = FLD(5.17.N502(J2)) LA = FLD(22.14.N502(J2)) LX = FLD(22.14.N502(J2)) 60 T0 4012 4040 01 = XK(LK)*XK(LA) 4099 J2 = J2+1 4099 J2 = J2+1 4099 J2 = J2+1 50 D1 = J1+1 50 J1 = J1+1
7664 7664 7664 7664 7664	2000 2000 2000 2000 2000 2000 2000 200	*****		777 777 777 777 777 777 777 777 777 77
00265 00265 00270 00271 00277 00277 00275 00275 00275			0001255 0001255 0001255 0001255 000140 00140 001412 001412 001412 001413 001413	000345 0003445 0003445 000355445 00033554 00033555 00033555 0003555 0003555 0003555 0003555 00000000

N
×
ž
Ü
÷
a
X
ž
Ü

•

8 8 9 8 9 8 9 8 9 8 9 8 9 8 9 8 9 8 9 8	N I I I I I I I I I I I I I I I I I I I		NEN NGN NGN NGN NGN NGN NGN NGN NGN NGN	21 * *
LG = FLD(5.16.4501(J1)) LG = FLD(5.20.0) G0 70 85 LTA = FLD(22.14.4.NS01(J1)) INCLUE VARG.LTST C CHECK FOR RADIATION CONDUCTOR IF(FLD(2.11.NS01(J1)).E0.0) G0 T0 5000 NTTPE = FLD(5.17.4.S02(J2)) LA = FLD(5.17.4.S02(J2)) LA = FLD(5.17.4.S02(J2)) LA = FLD(22.14.9.NS02(J2)) LA = FLD(22.14.9.NS02(J2)) LA = FLD(22.14.9.NS02(J2)) LA = FLD(22.14.9.NS02(J2)) LA = FLD(22.14.9.NS02(J2)) S = FLD(22.14.9.NS02(J2)) LA = FLD(22.14.9.NS02(J2)) LA = FLD(22.14.9.NS02(J2)) LA = FLD(22.14.9.NS02(J2)) LA = FLD(22.14.9.NS02(J2)) S = FLD(22.14.9.NS02(J2)) LA = FLD(22.14.9.NS02(J2)) S = FLD(22.14.9.NS02(J2)) LA = FLD(22.14.9.NS02(J2)) S = FLD(22.14.9.NS02(J2)) S = FLD(22.14.9.NS02(J2)) S = FLD(22.14.9.NS02(J2)) S = FLD(22.14.9.NS02(J2)) S = FLD(22.14.9.NS02(J2)) LA = FLD(22.14.9.NS02(J2)) S = FLD(22.14.9.NS02(J2))	2015 CALL DIDIAMATITINATION (AND AND AND AND AND AND AND AND AND AND	LA = FLD(5,17,NSd2(J2)) LA = FLD(22,14,NSd2(J2)) G2 = XK(LK)+XK(LA) G2 = XK(LK)+X(LA))/2.0 2030 TM = (1(1)+T(LTA))/2.0 2035 TM = 1(1) 2035 TM = 1(1) 2040 CALL PLYAWM(A(LA),T(1),A(LA+1),XK(LK).6(L6)) 2040 CALL PLYAWM(A(LA),T(1),A(LA+1),XK(LK).61) 2040 CALL PLYAWM(A(LA),T(1),A(LA+1),XK(LK).61)	Z045 UL = FLD(5,17,NSO2(J2)) LK = FLD(22,14,NSO2(J2)) LK = FLD(22,14,NSO2(J2)) Call Plyawm(a(La),T(LTA),A(LA+1),XK(LK),G2) GO TO 2998 2045 G1 = XK(LK) *XK(LA) 2050 CALL PLYAWM(A(LA),T(1),A(LA+1),XK(LK),G1) J2 = J2+1 LX = FLD(22,14,NSO2(J2)) LX = FLD(22,14,NSO2(J2))	62 XK(LK) *XK(LA) 60 TO 2998 60 TO 2998 60 TO 2999 60 TO 2007 60 TO 2007 60 TO 2032 2956 TM = T(LTA) 705 TM = T(LTA) 60 TO 2032 295 G(G) Z0107 205 TM = T(LTA) 60 TO 2032 2996 G(LG) T/(1,/G1+1./G2) 2999 JZ<= JZ+1 3000 CONTINUE
 Contraction Contract		2 8 8 8 8 8 8 8 8 8 9 8 8 8 8 8 8 8 8 8		ទេ ១ ២ ២ ២ ២ ២ ២ ២ ២ ២ ២ ២ ២ ២ ២ ២ ២ ២ ២
003774 00365 00365 00365 00374 00377 003774 003774 003774 003774 003774 003774 003774 003774 00037774 0003774	00400 00400 00400 00400 00411 00411 00411 00411 00411	00000000000000000000000000000000000000		00000000000000000000000000000000000000

n an an Araba an Arab Araba an Arab

·. .

IS THE TIME STEP USED LESS THAN THE TIME STEP CALCULATED IF (TSTEP_LE_UELTA) 60 TO.130 COMPUTE THE TIME STEP KON(35) = I KON(35) = I COMPUTE NEW TEMPERATURES USING CALCULATED SOURCE TERMS T2 = 1.0/EXP(TSTEP*X(LE)/C(1)) T1 = (1.0-T2)+0(1)/X(LE)+T2#T(1) T1 = (1.0-T2)+0(1)/X(LE)+T2#T(1) CALCULATE THE ABSOLUTE VALUE TEMPERATURE CHANGE ABS(11-T(1)) SAVE THE LARGEST TEMPERATURE CHANGE IF(TCGM.GE.T2) G0 T0 95 OUTAIN NEW DIFFUSION TEMPERATURES, DTMPCC AND CSGMIN DO 100 I = 1.NND LE = IE+I Calculate C/SK MINIMUM T1 = C(I)/X(LE) X(LE) = X(LE)+GV CHECK FOR ADJOINING DIFFUSION NODE FFLTAGE.NNU.OR.FLD(21,1)NSQ1(J1)).EG.1) GO TO 65 SAVE SUMMATION OF CONDUCTORS FOR ADJOINING NODE LEA = IE+LTA T1 = T(1)+460.0 T2 = T(LTA)+460.0 6V = 6(LG)+(T1+T2+T2)*(T1+T2) 60 T0 60 50 T0 60 50 T0 60 51 THE 0 RATE THRU THE CONDUCTOR 0 0(1) = 0(1)+6V+T(LTA) 5 AVE SUMMTION OF CONDUCTORS $\begin{array}{l} \text{CONTIT} = \text{CKM} \\ \text{DELTA} = \text{CKM*CONT4} \\ \text{DELTA} = \text{CKM*CON14} \\ \text{IFICKM*LE.0.0} & \text{GO TO 996} \\ \text{CHECK FOR FIRST PASS} \\ \text{IFICKASCT.0.0} & \text{GO TO 115} \\ \text{IFICMSCSCT.0.0} & \text{GO TO 115} \\ \text{UNDO THE TEMPERATURE CALCULATIONS} \\ \text{IS DO 110 I = 1.NNC} \\ \text{LE = IE+I \\ LE = IE+I \end{array}$ IF(FLD(3,1,NSQ1(J1)).EQ.0) 60 70 55 X(LEA) = X(LEA)+6V 0(LTA) = 0(LTA)+6V*T(I) CHECK FOR LAST CONDUCTOR IF(NSO1(J1).6T.0) 60 TO 50 IF(PASS.61.0.0) 60 T0 15 KON(36) = I STORE THE TEMPERATURES X(LE) = T(1) IF(T1.GE.CKM) GO TO 90 CKM = T1 TSTEP = DELTA+0.95 60 T0 195 **CNEXPN**, CNEXPN CON(1) = TPRINTCON(2) = 0.0T(T) = X(LE)PASS = 1.0 TCGM = T2 $T(1) = T_1$ CONTINUE CONTINUE CONTINUE ENO 35 100 105 55 60 <u> 9</u> 110 115 85 8 U U U U U ų Ú U U U ų υ Ų Ū, U 141+ 87 88 88 91 4 91 +96 1004 122+ 26* 86* 92+ 94+ 95+ *66 108+ *601 110* 111* 112* 113* 114* 115* 115* 119* 120+ 121* 128* 35* 440 140* * * 8 85# +96 \$7\$ 24.* 25* 31+ 32* +nn *** 36* 38* 39* ໍ້ຄື 00475 00475 00475 00475 00475 00500 00500 00502 00502 00506 00551 00552 00552 00465 00466 00470 00511 00512 00512 00513 00514 00514 00515 00516 00515 00520 00520 00543 00546 00550 00466 00524 00534 0.0545 00457 09400 00462 00463 00464 00467 00467 00471 0.0521 00522 00527 00530 00532 00537 0.0540 0.0541 00551 00526 0.0530

		월 월 20 이 월 년 년 년 국 군 군 부 부 북 북 북	N	30 101 22 5 5 5 5 5	11 12 12 12 12 12 12 12 12 12 12 12 12 1
CWEXPN.CNEXPN	TSTEP = DELTA+0.95 60 T0 105 515FE = 0.95*T5F+C:.(11)/TCGM 60 T0 105 55E IF THE TEMPERATURE CHANGE WAS TOO LARGE 550 E IF THE TEMPERATURE CHANGE WAS TOO LARGE 510 IF (TCGM, 617, CON(6)) GO TO 120 510 IF (TCGM, 617, CON(6)) GO TO 120 510 IF (TCGM, 617, CON(6)) GO TO 120 510 IF (TAMA, LE.0) GO TO 185 510 CON(12) = TGGM 60 T0 105 76 CM 105 76 CM 105 76 CM 106 76 CM 11 77 CM 11 70 LAX = K04(5) 70 LAX = K04(5	<pre>F(1.6T.1) GO TO 6000 INCLUDE VR02*LIST INCLUDE VR02*LIST INCLUDE VR02*LIST INTYPE FLD(4,1NS01(JJ)+1)).E0.0) GO TO 6000 NTTYPE FLD(4,1NS02(JJ2)) LA = FLD(22,14,NS02(JJ2)) LA = FLD(22,14,NS02(JJ2)) LA = FLD(22,14,NS02(JJ2)) CO TO (5005.5010.5015,5020.5025,5030.5035,5040.5030). NTYPE 60 TO (5999 010 01 = 0.0 012 CALL DIDIWM(T(L),A(LA),XK(LK),02) 012 CALL DIDIWM(T(L),A(LA),XK(LK),02) 015 01 = 0.0 015 01 0.0 015 0.0 00 0.0</pre>	022 JJZ = JJZ+1 LK = FLD(5-17+NSQ2(JJZ)) LK = FLD(22-14+NSQ2(JJZ)) 60 T0 5017 125 01 = XK(LK)+XK(LA) 120 21 5022 131 JJZ = JJZ+1 132 JJZ = JJZ+1	LA = FLD(5,17,17502(JJ2)) LK = FLD(22,14,11502(JJ2)) 02 = XK(LK)+XK(LA) 03 TO 5990 335 CALL DID1+M(CON(14),A(LA),XK(LK),01) 337 JJ2 = JJ2+1	LA = FLD(5,17,4802(JJ2)) LA = FLD(22,14,4802(JJ2)) LA = FLD(22,14,4802(JJ2)) D44 01 = xK(LA) 60 TO 5037 998 Q(L) = Q1+02+6(L)
	υυυυ	ស ស៊ស ស៊ស ស៊	ស ស ស	ស័ណៈ	<u> </u>
	1 1 1 1 1 1 1 1 1 1 1 1 1 1	11111111111111111111111111111111111111	1 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	168* 168* 168*	1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
	00000000000000000000000000000000000000	24654371046666666716 226554571046667167 255557104667167106 2565555710467167 2555557104670 2565555710 2565555710 2565555710 25655555710 2565555710 2565555710 2565555710 2565555710 2565555710 2565555710 2565555710 2565555710 25655555710 2565555710 25655555710 25655555710 25655555710 25655555555555555555555555555555555555	000531 000531 000533 00654 00654 00655 00635	000044	00646 006551 006551 006551 006551

**. • *

)

and the second second

-
a.
×
Lu
z
Û
ंक
z
٥.
×
ш
Z
U

.

B	- 				2 QI U I 2 8 9
JJZ = JJ2+1 CONTINUE END JJJ = JJ1+1 LT = FLD(22:14.NSO1(JJ1)) FF(1.6T.1) 60 T0 4000 IF(1.6T.1) 60 T0 4000 INCLUE VR62.11ST INCLUE VR62.11ST INCLUE VR62.11ST INCLUE VR62.11SO1(JJ1)) FF(FLD(2:1,NSO2(JJ2)) LA = FLD(0.5.NSO2(JJ2)) LA = FLD(5:17.NSO2(JJ2)) LA = FLD(5:17.NSO2(JJ2))	TM = (T(L)+T(LTA))/2.0 TM = (T(L)+T(LTA))/2.0 Call DIDLWM(TM.A(LA).XK(LK).G(LG)) Go TO 3999 GO TO 3907 Go TO 3007 Call DIDLWM(T(L).A(LA).XK(LK).G1) JJ2 = JJ2+1	LA = FLD(5)1/(1502(5027)) LK = FLD(52(14,NS02(5027)) CALL DIDWM(T(LTA),A(LA),XK(LK),62) 60 T0 3998 61 = Xk(LK)*XK(LA) 60 T0 3017 CALL DIDIMM(T(L),A(LA),XK(LK),61) JU2 = JU2+1	LA = FLD(5,17,11502(JJ2)) LK = FLD(22,14,NS02(JJ2)) 62 = XK(LK)*XK(LA) 7M = (T(L)+T(LTA))/2.0 7M = (T(L)+T(LA),TM.A(LA+1),XK(LK).6(LG)) 60 TO 3999 60 TO 3999 60 TO 3032 CALL PLYARW(A(LA),T(L).A(LA+1),XK(LK).61) JJ2 = JJ2+1	LA = FLD(5,17,NSQ2(JJ2)) LK = FLD(22,14,NSQ2(JJ2)) CAL PLYAWM(A(LA),T(LTA),A(LA+1),XX(LK),62) GO TO 3998 G1 = XX(LK)+XX(LA) G1 = XX(LX)+XX(LA),T(L),A(LA+1),XX(LK),G1) CAL PLYAWM(A(LA),T(L),A(LA+1),XX(LK),G1) LA = -JJ2+1 LA = FLD(5,17,NSQ2(JJ2))	LK = FLD(22,14,NSQ2(JJ2)) 62 = XK(LK)*XK(LA) 60 TO 3998 Call D2D1WM(TM,CCN(14),A(LA),XK(LK).6(LG)) 60 TO 3999 TM = T(LTA) 60 TO 3007 TM = T(LTA) 60 TO 3032
C 2000	3005 3007 3010 3015 3015	3020	3030 3032 3032 3040 3040	3045	3055 3060 3060
1688 1688 1688 1708 1718 1718 1718 1748 1748 1748 1748	********* 1777744 17777444 17777444 17777444 17777444 17777444 1777744	**************************************	************ 17775555 1777555555 17775555555555	174* 174* 174* 174* 174*	1744 17744 17744 17744 17744 17744 17744
	006773 006773 006773 006673 006675 006675 006675 006675	00702 00703 00705 00705 00705 00705 00710	00712 00713 00713 00715 00716 00722 00722 00723 00723	00725 00727 00730 00731 00731 00733 00733	00735 00737 00741 00741 00741 00742 007443 007443 007445 007445

		16 ^{- 1}		ta ana gina ang ang ang ang ang ang ang ang ang a	and the California and States in
		() 成一。 ● (15) () ● (15) () ● (15) () ● (15) ()			
		1) = 61+62 0 140	IC IE DIFFERENCE Xation Charge	IET AXATION CHANGE Re change	
	EXPN	./61+1./62) 31(JJ1)).E0.1) 6(L(31(JJ1)).E0.1) 6(L(11(JJ1)).E0.0) 60 7 50.0 1.11+T2*T2)*(71+T2)	/ +6V+T(LTA) T CONDUCTOR T CON	TION CRITERIA WAS 1 V(19)) GO TO 175 Imum Arithmetic Rem Aithmetic Temperatu Ainc -X(LE)) CO TO 180	MAS SATISFIED ((1)) 60 T0 125 (UP SWITCH 0) 60 T0 105 (1) (1) (1) (1) (1) (1) (1) (1)
•.	CNEXPN.CN	98 6(LG) = 1./(1 17(FLD(3,1,MS) 99 JJZ = JJ2+1 00 CONTINUE 10 CONTINUE 11 = T(L1,4460 11 = T(L1,4460 12 = T(LTA)+44 60 T0 145 60 T0 145	45 SUMC = 5(LG) 45 SUMC = SUMC+6 5 SUMC = SUMC+6 5 SUMC = SUMC+6 17 (L)+1 12 = DD+1(L)+1 12 = DD+1(L)+1 12 = ABS(T(L)+1 11 = ABS(T(L)+1 11 = ABS(T(L)+1 11 = ABS(T(L)+1) 11 = ABS(T(L	65 CONTINUE 15 CONTINUE 16 CONTINUE 16 CONTINUE 16 CONTINUE 16 CONTINUE 17 COMPUTE 14 MAX 16 COMPUTE 14 MAX 16 COMPUTE 14 MAX 16 M	80 CONTINUE 55E TTREA 55E TTREA 65 CONTINUE 65 CONTINUE 65 CONTINUE 65 CONTINUE 65 CONTISUE 65 CONTISUE 65 CONTISUE 75 CONTISUE 70 CONTISUE 75 CONTISUE 75 CONTISUE 76 CONTISUE 77 CONTISUE 75 CONTISUE 75 CONTISUE 75 CONTISUE 75 CONTISUE 75 CONTISUE
ŧ,					
		1700		20000000000000000000000000000000000000	
		0762 0755 0755 0755 0755 0762 0762 0762 0762 0762 0762 0762 0762	07664		

A - 42

٠

· · ·

EXPN	00001.LT.CON(3)) 60 TO 5									EXPN REQUIRES SHORT PSEUDO-COMPUTE SEQUENCE)	GWIN ZERO OR NEGATIVE)	ME STEP TOO SMALL)	LOCATIONS AVAILABLE)	OUTPUT INTERVAL)		V COMPILATION. D +DIAGNOSTIC+ MESSAGE(S)
CNEXPN. CNEXPN	IF(CON(1)+1.000001.L' NTH = IE	NDIM = NLA RETURN	WRITE (6+885) Go To 1000	WKITE(6,886) Go to 1000	WRITE(6,887) 60 TO 1000	WRITE (6+888) I	GU TO 1000	WRITE (6,889)	CALL EXIT	FORMAT (46H CNEXPN RE	FORMAT (24H CSGWIN 2E	FORMATIZOH TIME STEP	FORMATIIB. 20H LOCATI	FORMAT(19H NO OUTPUT	END	1108 FORTRAN V COMPI Symbolic Relocatable
• :			* 995	966 *	+ 997	466 +		666 *		÷ 885		* * 887	. 888	. 889		OF UNIVAC CODE
	01045 230 01047 231	01050 232 01051 233	01052 234	01055 236	01060 238	01063 240	01066 241	01067 242	01072 244	01073 245	01074 246	01075 247	01076 248	01077 249	01100 250	CNEXPN

'n

A - 43

•

•

•

2

CNDUFR

BI FOR: CNDUFR UNIVAC 1108 FORTRAN V ATHENA VERSION 131K-100 CREATED ON 20 AUG 70 THIS COMPILATION WAS DONE ON 09 JUN 70 AT 23115:00

SUBROUTINE CNDUFR ENTRY POINT 003620

.

STORAGE USED (BLOCK, NAME, LENGTH)

·

003635	00100	000054	000000	000000	_	_						_					
	LEND	VAR			00000	00000	00000	00000	00000	00000	00000	00000	00000	00000	00000	00001	
+CODE	+CONST+	*SIMPLE	*ARHAYS	+BLANK	TITLE	TEMP	CAP	SOURCE	COND	PCI	PC2	KONST	ARRAY	FIXCON	XSPACE	DIMENS	•
0001	0000	0002	1000	0002	0000	0001	0100	1100	0012	0013	100	0015	0016	1100	00200	0021	•
					•											1	

•

EXTERNAL REFERENCES (BLOCK, NAME)

÷

VARBLI	PLYAWM D2D1WM VARBL2	OUTCAL EXIT NERR25	NWDUS NIO25 NER105
0022	0024 0025 0025	0020	0032

. L

STORAGE ASSIGNMENT FOR VARIABLES (BLOCK, TYPE, RELATIVE LOCATION, NAME)

+

10

R

So

10

	U	0	Ø	Ż	¥	2	هتر	Z	2	z	0	-	-
	000023	010000	000033	000047	000000	000037	000001	000000	000000	000050	000026	1000031	000034
*	0002 R	0002 8	C002 R	6002 1	0017 1	0002 1	0021 I	00200	0021 I	0002 1	0002 R	0002 R	0.002 R
	z	•		I			01	- د	U	I		GM	M
	000 000	NG 440	032 61	003 IE	X 000	013 LE	006 LS	004 NC	001 NN	001 NT	025 01	016 TC	012 15
	R 000	R 000	R 000	1 000	1 000	000 1	1 000	1 000	1 000	000 1	R 000	R 000	R 000
	0017	0005	0005	0005	0015	0005	0021	0021	0005	0200	0002	0005	0005
	W	LTA		1.4	•••	X		1	A b	502			TEPO
	017 CH	043 DE	036 61	002 16	015 J2	046 L	022 LK	005 NJ	001 NN	000 NS	000	000 T	010 TS
	000 1	000 2	000 2	000	000	000	000	000	000 1	000	000 2	000 2	000 1
3	002 9	002 6	002 9	002	002]	002 1	002	021]	021]	014	011 5	007 F	002 F
	C	0		0	0	Ģ	0		0	0	0	Ð	0
		ģ			11		.0		Ľ.	1501	ASS	V DWUS	STEP
	000000	000045 0	000000	000005	000014	000021 L	000027 L	000000	100000	000000	000000	000053 5	1 200000
	0010 8	0002 R	0012 8	0002 1	0002 1	0002	0002	0002	0002	1 6100	0002 8	0002 R	0002 R
201												U	INT
		20	0.12	ÍI	200	اني ا اني ا	EH S	LTA	S NGT	NNT O	ž	SUM	THR
	000000	000024	00004	000000	00000	000051	00004	00003	000000	00000	00000	00000	10000
	1 × R	200	. a	: œ	1 4 0	10	10	100	101	15	- 02	02 8	02 R
	00	00			00		00		20	50	00	00	8

lan an an an tara an a

NTYPE WI ON

03

i.r

,

n ***** Marine ar train Tar

A - 44

	10001	10201	10356	1151	1400	140	1999L	20156	20306	20456	2146	30L	3015L	3032L	3050L	3999L	40151	4030L	1866#	5005L	5020L	5037L	5566	6753	887F	J 566			2 2	
	903574	1111000	000607	002174	003317	124200	000705	001300	001420	001533	000232	000166	002667	003014	003130	003257	001000	001055	001152	002343	002406	002517	02162	002250	000016	003536			•	.,
	1000	1000	1000	1000	1000	0001	1000	1000	1000	0001	0001	0001	0001	0001	0001	0001	0001	1000	0001	0001	1000	1000	1000	0001	0000	0001				
	LOL	ICISL	1032L	10506	135 L	175L	1998L	2010L	2025 L	2042L	2065L	1 666i	5010L	3030L	5045L	5998L	1012L	1025L	151	5000L	5017L	in35L	55L	2000L	386F)5L .	166C			
	000132	000436	000551	003462	002551 1	003400	000702 1	001275 2	001360	001474 2	001636	001670 2	002664 3	003007 3	003122 3	003231 3	000761 4	001047 4	000244	001161	002371 5	002503 5	001724 5	002551 6	000011 6	002101	003567 9			
	1000	1000	1000	0001	0001	1000	0001	0001	1000	0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	0000	0001	0001			
	×	012L	030L	05 L	30L	65L	95L	007L	020L	Otol	060L	9986	007L	025L	0426	065L	0101	022L	040L	0L	015L	030L	416	0	85F	٥ ر	98L			
	000000 X	000403 1	000530 1	002125 1	002221 1	003367 1	003517 1	001255 2	001352 2	001453 2	001632 2	001642 2	002644 3	002747 3	003063 3	0032255 3	000.760 4	001032 4	001144 4	001161 5	002370 5	002445 5	002131 5	001727 6	000000 8	002044 9	003560 9			
	0015 R	1000	0001	0001	1000	1000	0001	0001	1000	0001	1000	1000	1000	0001	1000	0001	0001	1000	1000	1000	1000	0001	1000	0001	0000	1000	0001	:		
		010L	025L	045L	25L	51	90 L	005L	026	035L	055L	51	005L	020L	040	060L	005L	020L	037L		012L	025L	076	1666	51	89F	97L			
	000002 X	000365 1	000504 1	000660 1	002213 1	C00141 1	003514 1	001250 2	000213 2	001450 2	001604 2	000153 2	002637 3	002741 3	003042 3	003221 3	4 ECT000	001016 4	001127 4	000120 5	002351 5	002437 5	002020 5	002546 5	001775 6	000030 A	003552 9			
DUFR	0020 R	0001	0001	0001	0001	0001	1000	0001	0001	0001	0001	0001	0001	0001	0001	0001	1000	0001	0001	0001	0001	0001	0001	0001	0001	0000	0001			
U		05L	236	401	٥ ר	56	5	00	171	32L	SOL	46	DOL	171	351	55L	DOL	176	35L	-99L	101	221	HOL	98L	0 7	8F	و ر			
	000035 72	000344 10	003407 10	000615 10	002205 12	003322 14	003442 16	00710 20	01316 20	01425 20	01541 20	00273 23	001673 3 0	J02705 30	103037 30	03173 30	003262 40	01001 40	001113 40	001156 49	02350 50	02422 50	02534 50	02542 59	02271 61	00023 86	203544 95	 		
	0002 R (0001	1000	1000	1000	1000	0001	1000	0001	0001	1000	1000	0001 0	0001	0001	1000	0001 0	0001	0001	1000	1000	0001 0	0001 0	1000	1000	0000	0001 0			

10100	*	SUBRI	DUTINE CNDUFR	
00101	\$*	C EXPL	ICIT DUFORT-FRANKEL EXECUTION SUBROUTINE FOR SINDA F_V	
00101	ň	C THE	SHORT PSEUDO-COMPUTE SEQUENCE IS REGULAED	
20103	**	INCL	JDE COMM.LIST	
00104	*#	COMMI	DN /TITLE/H(1) /TEMP/T(1) /CAP/C(1) /SOURCE/Q(1) /COND/G(1)	
00105	# 17	COMMI	DN /PCI/NSOI(1) /PC2/NSO2(1) /KONST/K(1) /ARRAY/A(1)	
00100	# :7	COMMI	DN /FIXCON/KON(1) /XSPACE/NDIM+NTH+X(1)	
00102	# 7	COMMI	DN /DIMENS/ NND+NNA+NNT+NGT+NAT+LSA1+LSQ2	
00110	* 5	DIME	V2ION CON(1)*XK(1)*NX(1)	
11100	**	EGUI	/ALENCE (KON(1)*CON(1))*(K(1)*XK(1))*(X(1)*NX(1))	
00112	*	END		
00113	ໍ້ຄ	INCLU	JDE DEFF'LIST	
00113	ສູ	C+++++++	**** CONTROL CONSTANT DEFINITIONS AND NAMES ####################################	
00113	.*	C CONTI	ROL CONSTANT 1 CONTAINS THE NEW PROBLEM TIME	
00113	ۍ پ	C CONTI	COL CONSTANT & CONTAINS THE TIME STEP USED (DTIMEU)	
00113	* 0	C CONTI	COL CONSTANT 3 CONTAINS THE PROBLEM STOP TIME (TIMEND)	
00113	ب	C CONTI	ROL CONSTANT & CONTAINS THE TIME STEP FACTOR. EXPLICIT (CSGFAC)	
00113	،	C CC5	IS THE INPUT NUMBER OF ITERATION DO LOOPS, INTEGER (NLOOP)	
00113	ب م	c c c c c	CONTAINS THE DIFFUSION TEMPERATURE CHANGE ALLOWED (DTMPCA)	
00113	ŝ	C	CONTAINS THE OUTPUT EACH ITERATION SWITCH	
00113	ٿ	CCB CCB	CONTAINS THE MAXIMUM ALLOWED TIME STEP (DTIMEH)	
00113	5 *	600 0	CONTAINS THE NEW ARITHMETIC TEMP. DAMPING FACTOR (DAMPA)	
00113	ۍ ۴	C CC10	CONTAINS THE NEW DIFFUSION TEMP. DAMPING FACTOR (DAMPD)	
00113	ئ	C CC11	CONTAINS THE MAXIMUM ALLOWED ARITHMETIC TEMP. CHANGE (ATMPCA)	
00113	ເ	C CC12	CONTAINS THE RACKUP SWITCH CHECKED AFTER VARIABLES (BACKUP)	
00113	ٿ	C CC13	CONTAINS THE PRESENT TIME OR PROBLEM START TIME (TIMEO)	
00113	*	C CC14	CONTAINS THE REAN TIME BETWEEN AN ITERATION (TIMEM)	
00113		C CC15	CONTAINS THE DIFFUSION TEMPERATURE CHANGE CALCULATED (DIMPCC)	
00113	÷.	c cc16	CONTAINS ARITHMETIC TEMPERATURE CHANGE CALCULATED (ATMPCC)	

A - 45

CNDUFR

CONTROL CONSTAIT 17 IS RESERVED FOR THE C/SG MINIMUNI CONTROL CONSTANT 18 CONTAINS THE OUTPUT INTERVAL CC19 CONTAINS THE ARITHMETIC RELAXATION CRITERIA ALLOWED CC22 CONTAINS THE MUNMUM ALLOWED TIME STEP CC22 CONTAINS THE NUMMER OF RELAXATION LOOPS USED.INTEGER (LOOPCT) CC22 CONTAINS THE NUMMER OF RELAXATION LOOPS USED.INTEGER (LOOPCT) CC22 CONTAINS THE C/SG MANGE ALLOWED TIME STEP CC22 CONTAINS THE C/SG MANGE ALLOWED CC22 CONTAINS THE C/SG MANGE ALLOWED CC22 CONTAINS THE C/SG RANGE CALCULATED CC22 CONTAINS THE C/SG RANGE CALCULATED CC22 CONTAINS THE C/SG RANGE CALCULATED CC22 CONTAINS THE LINE COUNTER. INTEGER CC22 CONTAINS THE LINE COUNTER. INTEGER CC23 CONTAINS THE LINE COUNTER. INTEGER CC33 CONTAINS THE LINE COUNTER. INTEGER CC33 CONTAINS THE RELAXATION CHANGE CALCULATED CC23 CONTAINS THE RELAXATION CHANGE CALCULATED CC23 CONTAINS THE LINE COUNTER. INTEGER CC33 CONTAINS THE RELAXATION CHANGE CALCULATED CC33 CONTAINS THE RELAXION CHANGE OF THE SYSTEM. IN - OUT (RAGERAL CC33 CONTAINS THE RELAXION CHANGER OF CSGMIN CC33 CONTAINS THE RELAXION CHANGER OF CSGMIN CC33 CONTAINS THE NOCOPY SWITCH FOR MATRIX USERS (NOCOPY SWITCH FOR ANTRIX USERS (C339-40-41-49-43-43 CONTAINS RELATIVE HODE ONUMER OF ARLXCC CC339-40-41-49-44-43 CONTAINS RELAXION INTERVAL FOR CONSTANTS (C449 IS NOT USED AT PRESENT CC34 IS NOT USED AT PRESENT CC35 IS NOT USED AT PRESENT TSTEPO = 0.0 TPRINT = CON(13) INTALIZE TIME SUM BETWEEN OUTPUT INTERVALS TSUM = 0.0 DOES OLD TIME PLUS THE OUTPUT INTERVAL EXCEED THE STOP TIME DOES OLD TIME PLUS THE OUTPUT INTERVAL EXCEED THE STOP TIME DOES OLD TIME PLUS THE OUTPUT INTERVAL EXCEED THE STOP TIME DUES THE TIME SUM PLUS THE TIME STEP EXCEED OUTPUT INTERVAL CHECK FOR EXTRA LOCATIONS FOR CALCULATED NODES CON(18) = CON(3)-CON(13) IS THE TIME STEP LARGER THAN ALLOWED IF(TSTEP.LE.CON(8)) GO TO 15 $\begin{array}{l} \label{eq:constraints} & \mbox{IF}(\mbox{Con}(4), \mbox{LT}, 1, 0) & \mbox{Con}(4) = 1, 0 \\ \mbox{IF}(\mbox{Con}(5), \mbox{LE}, 0, 0) & \mbox{Con}(6) = 1, \mbox{E} + 8 \\ \mbox{IF}(\mbox{Con}(9), \mbox{LE}, 0, 0) & \mbox{Con}(9) = 1, \mbox{E} + 8 \\ \mbox{IF}(\mbox{Con}(1), \mbox{LE}, 0, 0) & \mbox{Con}(1) = 1, \mbox{E} + 8 \\ \mbox{IF}(\mbox{Con}(1), \mbox{LE}, 0, 0) & \mbox{Con}(1) = 1, \mbox{E} + 8 \\ \mbox{IF}(\mbox{Con}(1), \mbox{LE}, 0, 0) & \mbox{Con}(1) = 1, \mbox{E} + 8 \\ \mbox{IF}(\mbox{Con}(1), \mbox{LE}, 0, 0) & \mbox{Con}(1) = 1, \mbox{E} + 8 \\ \mbox{IF}(\mbox{Con}(1), \mbox{LE}, 0, 0) & \mbox{Con}(1) = 1, \mbox{E} + 8 \\ \mbox{IF}(\mbox{Con}(1), \mbox{LE}, 0, 0) & \mbox{Con}(1) = 1, \mbox{E} + 8 \\ \mbox{IF}(\mbox{Con}(1), \mbox{LE}, 0, 0) & \mbox{Con}(1) = 1, \mbox{E} + 8 \\ \mbox{IF}(\mbox{Con}(1), \mbox{LE}, 0, 0) & \mbox{Con}(1) = 1, \mbox{E} + 8 \\ \mbox{IF}(\mbox{Con}(1), \mbox{LE}, 0, 0) & \mbox{Con}(1) = 1, \mbox{E} + 8 \\ \mbox{IF}(\mbox{Con}(1), \mbox{LE}, 0, 0) & \mbox{Con}(1) = 1, \mbox{E} + 8 \\ \mbox{IF}(\mbox{Con}(1), \mbox{LE}, 0, 0) & \mbox{Con}(1) = 1, \mbox{E} + 8 \\ \mbox{IF}(\mbox{Con}(1), \mbox{LE}, 0, 0) & \mbox{Con}(1) = 1, \mbox{E} + 8 \\ \mbox{IF}(\mbox{Con}(1), \mbox{LE}, 0, 0) & \mbox{Con}(1) = 1, \mbox{E} + 8 \\ \mbox{IF}(\mbox{Con}(1), \mbox{LE}, 0, 0) & \mbox{Con}(1) = 1, \mbox{E} + 8 \\ \mbox{IF}(\mbox{Con}(1), \mbox{LE}, 0, 0) & \mbox{Con}(1) = 1, \mbox{E} + 8 \\ \mbox{IF}(\mbox{Con}(1), \mbox{LE}, 0, 0) & \mbox{Con}(1) = 1, \mbox{E} + 8 \\ \mbox{IF}(\mbox{Con}(1), \mbox{LE}, 0, 0) & \mbox{Con}(1) = 1, \mbox{E} + 8 \\ \mbox{IF}(\mbox{Con}(1), \mbox{IE}, 0, 0) & \mbox{Con}(1) = 1, \mbox{E} + 8 \\ \mbox{IE}(\mbox{Con}(1), \mbox{IE}, 0, 0) & \mbox{Con}(1) = 1, \mbox{E} + 8 \\ \mbox{IE}(\mbox{Con}(1), \mbox{IE}, 0) & \mbox{Con}(1) = 1, \mbox{E} + 8 \\ \mbox{IE}(\mbox{Con}(1), \mbox{IE}, 0) & \mbox{Con}(1) = 1, \mbox{E} + 8 \\ \mbox{IE}(\mbox{Con}(1), \mbox{IE}, 0) & \mbox{IE}(\mbox{Con}(1), \mbox{IE}, 0) & \mbox{IE}(\mbox{Con}(1), \mbox{IE}, 0) & \mbox{IE}(\mbox{Con}(1), \mbox{IE},$ IEH = NTH+NNC NLA = NDIM NTH = NTH+NNC+14ND NDIM = NDIM-NNC-NND I = NLA-NNC-NND IF(I.LT.0) G0 T0 998 L1 = NND+1 STEP = CON(18)ISTEP = CON(B) ŝ 10 U U U, U c 254 26* *** * 15 1000 00152 00154 00154 00155 00155 00155 00155 00155 00160 00161 00163 00163

CNDUFR

00164	400 0		15 JF(TSUM+157EP-CON(18)) 25+30+20	
00164		» د	DUNI EACEEU II 20 TETED II CONCIRVETSING	
00170		•		.•
00110	+03	Ű	DOES TIME SUM PLUS TWO TIME STEPS EXCEED OUTPUT INTERVAL	
12100	444		25 [F(TSUM+2.0+TSTEP.LE.CON(18)) 60 T0 30	
00171	404 474	ų	APPROACH THE OUTPUT INTERVAL GRADUALLY TSTEP = (CONVIA)-TSUM)/2.D	
00173	+2+	<u>,</u> Ú	STORE DELTA TIME STEP IN THE CONSTANTS	
00174	484	73	30 CON(2) = TSTEP	
00174	*6*	U	IS THE TIME STEP USED LESS THAN THE TIME STEP ALLOWED	
00175	+	Ļ	IF(TSTEP.LT.CON(ZI)) GO TO 997 Calculate THE NEW TIME	
00177	52#	,	CON(1) = TPRINT+TSUM+TSTEP	
00177	53+	U	COMPUTE THE MEAN TIME RETWEEN ITERATIONS	
00200	* † 5	ł	CON(14) = (CON(1)+CON(13))/2.0	
00200	100 100 100	U	ZERO OUT ALL SOURCE LOCATIONS AND EXTRA LOCATIONS	
10200	* * 0 /-			
00205	185		X(LE) = 0.0	
00206	29*			
00207	÷09	ر ی د	35 CONTINUE	
00211	+ + 29	َ د	IF (NNA-LE-D) 60 TO 45	
00213	63+		DO 40 I # L1.NNC	
00216	64#		0(1) = 0.0	
00217	65# 66		LE # 1E+1 VIE1 # 1111	
00221	- 1 - F	ų.		
00223	68*	- 4	45 KON(12) # 0	
00224	*69		CALL VARBLI	
00225	+04		IF(KON(12).NE.0) 60 TO 10	
02200	12*			
00231	13.		TCGM = 0.0	
00232	* + ~		CKM H 1.6+10	
00232	76.	U	CALCULATE Q SUM AND G SUM	
00236	77*			
00237	78*		INCLUDE VARCALIST	
00240	78+		IF(FLD(1,1,NSQ1(J1+1)),E0.0) 60 T0 2000	
00243	78*		NITE - FLUIDSTASKE (22) LA = FLU(5,17,NS02(J2))	*NEW
00244	78+		LK = FLD(22,14,N502(J2))	M3N+
00245	78*		60 TO (1005.1010.1015.1020.1025.1030.1035.1040.1045); NYYE	
00240	101	201	LUUD CALL ULUIMMIILITTAILAITANILNITLILITT Gñ Tñ 1999	
00250	78*	101	1010 CALL DIDIWM(T(I),A(LA),XK(LK),CI)	
00251	18+	101	1012 J2 = J2+1	
00252	18+		LA # FLD(5,17,1/502(J2))	
00254	181		LK = FLU(22/14/NSU2/OC)) fail Didikmititi.a(La).kk(k).C9)	
00255	78+		60 TO 1998	ł
00256	78*	101	1015 C1 = XY (LK) + XK (LA)	
00257	18+			
00200	- 91	105	LOZU CALL DIDINMII(IJ/#AILA/#AK/LA/#CL// 10 t = 2041	
00262	78*		LA = FLD(5,17,NS02(J21)	*NEW
00263	78+		LK = FLU(22+14+NS02(J2))	NUN -
00264	78+		C2 = XK (LK) + XK (LA)	244

				a segment of the	ng star e e	n Na tradición Na	a shi aadala - a bada B	le source and and source of the source of th
					•			
)								
	•.				·.			
		2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	2 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	A B B B B B B B B B B B B B B B B B B B	2 7	* NEK * NEK	8 8 0 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	330 111 211 211 211 211 211 211 211 211 21
	•							
					NTYPE			
					. 4030) .			
	(II) (I	5	1	, 6 ,	35.4040			
	X(LK),C	אנראס יכ	K (LLK) S C	K (LK) , C	4030.40) ,02)	(10)	(10	
	X•(1+4- X•(1+4	X • (T + Y	X+(T+V]	A(LA).X	0,4025, (LK),02 ,XK(LK)		י אנורא	
	1)¥*(1).	J2)) (J2)) (1),A(I	((1)•A(I) (12))	11(14).1	15,402(15,402(LA),XK	- A(LA)	(12)) (12))	
	A(LA) • 7 A(LA) • 7	7. NS02 (14. NS02 (A (LA) . 1	XK (LA) A (LA) - 1 7 - NSQ2 (14 - NSQ2 XK (LA)	1(1),CC ,LIST 501(J1 7,NS02(14.NS02 4013.46 1+0(1) 7(1),A(7(1),A(CON(14) 7,NSQ2 14,NSQ2 XK(LA) XK(LA) CON(14)	7,11502 (14,11502 (XK (LA) CON(14)	7,NS02(14,NS02 XK(LA) +G(I)
DUFR	1998 1998 1999 1999	J2+1 LD(5+1 LD(22+ LYAWM(1998	К (LK) 1032 2 Г ҮАЖН (1241 1241 1241 1265 1265 160 (224 1998	2201WM(1999 - C1+C2 12+1 	-LD(22. (4005, 4999 4999 1.0 101WM(4998 1.0 1.0 4998 4998	01014M(12+1 -12+1 -10(5,1 -10(22) 4017 4017 4022 4022 01014M(J2+1 FLD(5;1 FLD(22; (K(LK); 4998 JID1WM(12+1 + 10(22+ + 10(22+ + 10(22+ + 10(22+ + 10(22+ + 10(22+ 10(2))))))))))))))))))))))))))))))))))))
5	GO TO CALL F GO TO CALL F			CALL GOLL CC(1): J2 = J2 L2 =	LK = 7 60 T0 60 T0 61 = 0 60 T0 60 LL 1 60 LL 1 60 LL 1 60 LL 1 60 LC 10	CALL CALL CALL CALL CALL CALL CALL CALL	U2 12 12 12 12 12 12 12 12 12 12 12 12 12	CONTINUE 601 10
	1025	101	1035	1045 1998 2000 2000	4005 4010 4012 4015	4020 4022 4022	4035	4 4 40 4 40 4 40 4 40 4 60 4 6 4 4 6 4 6 4 6 4 6 4 6 4 6 7 1 6 7 1 7 7 1 7 7 7 7 7 7 7 7 7 7 7
	78+ 78+ 78+	78+ 78+ 78+	**************************************	798 798 798 798 798 798	799************************************	* 662 * 662 * 662 * 662 * 662	79# 79# 79#	*667 *667 *667 *667

A - 48

-	CNDUFR	

ė

00362	79+ 80+	50	END	
00364	81+		LG = FLD(5,16,NSQ1(J1))	
00365	82.		IF(LG.EQ.0) GO TO 65	
00367	83		LTA = FLD(22,14+NS01(J])	
00370	+ + 10 0 - + +	•	INCLUDE VARG/LIST	
01270		ر	IF (FLD(2+1+NS01(J1)).E0.0) 60 70 3000	·
50373	85		NTYPE = FLD(0,5,NS02(J2))	ISN.
00374	85+		LA = FLD(5,17,11502(J2))	
00375	85.		LK = FLD(22,14,NS02(J2))	
00376	10 10 10		6010(2005,2010;2015;2020;2025;2030;2035;2040;2045;2050;2050;2050; 2020 2025 2 2 217755	
01000	* 4 10 10	3000	ZUDUS200211 NTTE	
00400	85 8	2002	CALL DIDIWH(TM:A(LA),XK(LK),G(LG))	
10400	85#		60 10 2999	
00402	854	2010	TM = T(1)	
	+ u 0	2100	60 10 2007 Call Didiwiritit, Ali Al. YK (1 K). 61)	
00402	85*	2017		ļ
00406	85+		LA = FLD(5,17,1502(J2))	
10407	85+		LK = FLD(22,14,NSQ2(J2))	
01+00	4 4 0 4 1 4		CALL UIUIWM(((L+A)+A(LA)+AK(LA)+944) Go to 2004	
00412	82*	2020	GI = XK(LK) *XK(LA)	
51400	85*		60 T0 2017	
#T #00	854	2025	CALL DIDIWM(T(I),A(LA),XK(LK),GI)	
21+00	+ 00 0 0			*NEI
			LX = FLD(22,14,NSG2(JZ))	+NEI
00420	85*		62 = XK(LK) + XK(LA)	
00421	85*		GO TO 2998	
0422	854	2030	TM = (T(I)+T(LTA))/2.0	
04400		2002	CALE FEIRWINNERTTING, 2000 10 2000 10 2000 10 10 10 10 10 10 10 10 10 10 10 10	
00425	85*	2035	TM = T(I)	
00426	85*		60 TO 2032	
00427	85*	2040	CALL PLYAWM(A(LA),T(1),A(LA+1),XK(LK),61)	
001100	80.4 0.0	2042	J2 # J2+1 A = Eln(5.17.HSA2(491)	-NEI
10431			LK = FLD(22+14,NSQ2(J2))	I INE
20433	85*		CALL PLYAMM (A(LA) +T(LTA) +A(LA+1) +XK(LK) +62)	
10434	85*		60 T0 2998	
00435	80.8 1	2045	61 = XK(LK)*XK(LA)	
00100	- 	2050	60 10 6046 Cail Plyamm(a(LA),7(1),4(LA+1),XK(LK),61)	
0440	854		J2 = J2+1	
1,000	85*		LA = FLD(5, 17, NS02(J2))	
00442	85*		LK = FLD(22,14,NSQ2(JZ))	
00443	85#		62 = XK(LK) *XK(LA) co to 3000	
00445	808 808	2055	60 10 2330 TM = (T(I)+T(LTA))/2.0	
00446	85+		CALL D2D1+M(TM.COU(14).A(LA).XK(LK).G(LG))	
64400	85\$		60 T0 2999	
00450	* •	2060	1M = 1(LIA) co to 2007	
00452	100 100 100	2065	50 10 5007 TN II (LTA)	
0.453	858		GO TO 2032	
00454	•58 •58	2998	6(L0) = 1./(1./61+1./62)	
00422	854		14 (FLD()211,M201(01)) (CO.1) 6160 E 01402	

.

CNDUFR

.

 \sim

J2 = J2+1 CONTINUE END T1 = T(1)+460.0 72 = T(LTA)+460.0 6V = 6(LG)*(T1*T1+T2) 60 T0 60 61 = 0 PLUS SUMMATION 6*TA TERM 05TAIN THE 0 PLUS SUMMATION 6*TA TERM 05TAIN THE 0 PLUS SUMMATION 6*TA TERM 05TAIN THE 0 PLUS SUMMATION 6*TA TERM	X(LE) = X(LE)+GV CYCLE) = X(LE)+GV F(LTA.6T.NND.OR.FLD(Z11.NS01(J1)).E0.11 60 TO 65 SAVE SUMMATION OF CONDUCTORS FOR ADJOINING NODE E(A = IE+LTA X(LEA) = X(LEA)+6V (ITA) = 0(LTA)+6V (ITA) = 0(LTA)+7V (ITA) = 0(LTA)	TI=[DT1+x(LEH)+(C(I)/TSTEP-X(LE))+0(I))/(C(I)+L)-DT2)/TSTEP+DT2+ \$x(LE)) CALCULATE THE ARSOLUTE VALUE TEMPERATURE CHANGE T2 = ADS(T(I)-T1) SAVE THE LARGEST TEMPERATURE CHANGE IF(TCGM.GE.T2) 60 79 95 TF(TCGM.GE.T2) 60 70 95 XON(36) = 12 YON(36) =
29995 2005 2005		
	****************	***
60 50 60 60 60 60 60 60 60 60 60 60 60 60 60	00000000000000000000000000000000000000	
00451 00465 00465 00465 00465 00465 00465 00470 00470 00471 00471		00520 00521 005220 005220 005221 0052242 005224242 005224444 0052254442 00522524 005224244 005252444 005252444 0052524 0052524 0052524 0052524 0052524 0052524 0052524 0052524 0052524 0052524 0052524 0052524 0052524 0052524 0052524 0055555 00555555

•

n fransk filmer fan ster ster ster fan ster filmer fan ster filmer fan ster ster filmer fan ster filmer. En filmer

.● 19

A - 50

•

																				•	+NEW	ANEW -								*NEW	+NEN			:	ABN #	#11CH	
CNDUFR	TSTEP = CKM+0.9 Do 112 I = 1+N ^{HD} Leh = IeH+I	112 X(LEH) = T(I) 60 TO 195 7	L IS THE TIME STEP USED LESS THAN THE TIME STEP PARCOLATION 115 IF (TSTEP-LE-DELTA) 60 TO 130	C CUMPOLE THE TIME STEP TSTEP = DELTA+0.95	GO TO 105 120 TSTEP = 0.95*TSTEP*CON(6)/TCGM	GO TO 105 125 TSTEP = 0.95±TSTEP±CON(11)/TCGM		C SEE IF THE TEMPERATURE CHANGE WAS TOO LARGE	C STORE THE MAXIMUM DIFFUSION TEMPERATURE CHANGE	C CHECK TO SEE IF THERE ARE ANY ARITHMETIC NODES	IF(NNA.LE.0) GO TO 185 C compute anithmetic temperatures by successive point over-relax	DN = CON(9)		D0 170 I = 1, LAX	JUI = UI JU2 = J2	TCGM = 0.0	D0 165 L = L1.NNC	SUMC = 0.0	SUMCV = 0.0 [F(1.6T.1) 60 TO 6000	INCLUDE VRQ2.LIST	IF(FLD(4,1,NSQ[(JJ]+1)).E0.0) 60 TO 6000 HIVPF = FLD(A.5.NGO2(JJ2))	LA = FLD(5,17,NSQ2(JU2))	LK = FLD(22+14+NSQ2(JJ2)) Gn to (5005.5010.5015.5020.5025.5030.5035.5040.5030). NTYPE	2005 0(L) = XK(LK)+0(L)	60 10 5999 5010 01 = 0.0	5012 CALL DIDIWM(T(L),A(LA),XK(LK),02)	5015 01 = 0.0	5017 CALL DIDIWM(CON(14),A(LA),XK(LK),Q2) Go to 5998	5020 CALL DIDIWM(CON(14) +A(LA) +XK(LK) +01)	BUGK JJK = JJK+1 LA = FLD(5,17,43622(JJ2))	LK = FLD(22,14,NSQ2(JJ2))	60 T0 5017 5025 01 = XK(LK)+XK(LA)	G0 T0 5022	5030 CALL DIDIMM(COH(14)+A(LA)+XK(LK)+Q1)	LA = FLO(5, 17, NS02(JU2))	LK = FLO(22+14+NSQ2(JU2)) 02 = Xk(1x1+xx(1A)	60 TO 5998
	1444	146#		151*	153+	154*	156+	157*	159+	161*	162*	164+	165*	167.	168*	170+	172*	173+	175*	176*	176+	176+	176+	176*	176*	176*	1764	176*	176+	176*	176*	176*	176*	176*	176+	176*	176.
	00554 00555 00560	00563	00564	00566	00567	00571	00573	00573	00574	00576	00577	00601	00602	0000	000010	00611	00613	00616	00620	0.0622	00623	00626	00627	00631	00633	00634	00636	00637	00641	00043	00644	00045	00647	00650	00652	00653	00655

4

•

	10 10 10 10 10 10 10 10 10 10 10 10 10 1	3330 300 300 300 300 300 300 300 300 30	3 3 0 3 3 0 1 2 4 4 4 4 4	3 3 0 U U I 2 2 4 4 4 4	8 8 0 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	10
\sim	×K (LK) • 01)	1 570R 3) 60 TO 4000 6025,3030,3035,3040,3045,3050,3055,	(),6(LG)) Kk (LK),62)	(LK), G1)	-4+1),XK(LK),G1) A(LA+1),XK(LK),G2) A+1),XK(LK),G1)	
CNDUFR	Call DiDiwm(Con(14).A(LA). JJ2 = JJ2+1 LA = FLD(5.17.NSQ2(JJ2)) LK = FLD(22.14.NSQ2(JJ2)) 60 T0 5012 01 = XK(LK)+XK(LA) 61 = XK(LK)+XK(LA) 61 0 5012 61 0 501	JUZ = JJ2+1 CONTINUE END L1 = JJ1+1 L6 = FLD(5.16,MS01(JJ1)) F(1.6T.1) 60 T0 4000 IF(1.6T.1) 60 T0 4000 INCLUDE VRC2.14,NS01(JJ1)) F(FLD(2.1.0501(JJ1)).E000 NTYPE = FLD(0.17,NS02(JJ2)) LA = FLD(5.17,NS02(JJ2)) LA = FLD(2.14,MS02(JJ2)) LA = FLD(2.14,MS02(JJ2)) C0T0(3005.3010,3015,3020)	TH = (1(L)+1(LA))/2:0 CALL DIDNW(TM.A(LA),XK(LM 60 T0 3999 60 T0 3099 60 T0 3007 CALL DIDNW(T(L),A(LA),XK(LA = FLD(5.17,NSQ2(JJ2)) LA = FLD(5.17,NSQ2(JJ2)) LK = FLD(22,14,NSQ2(JJ2)) CALL DIDNW(T(LTA),A(LA),)	61 = XK(LX)*XK(LA) 60 TO 3017 1JJ2 = JJ2+1 LA = FLD(5,17,4502(JJ2)) LK = FLD(22,14,N502(JJ2)) 62 = XK(LX)*XK(LA) 62 = XK(LX)*XK(LA) 7M = (T(L)+T(LA))/2.0 7M = (T(L)+T(A))/2.0 60 TO 3999	TM = T(L) 60 T0 3034 JJ2 = JJ2+1 1J2 = JJ2+1 LA = FLD(5,17*1502(JJ2)) LK = FLD(5,17*1502(JJ2)) CALL PLYARM(A(LA),T(LTA), 60 T0 3998 61 = XK(LK)+XX(LA) 61 = XK(LK)+XX(LA)	UZ = JJ2+1 LA = FLG(5,17+":502(JJ2)) LK = FLD(22+14,NSQ2(JJ2)) 62 = Xk(LK)+XK(LA)
	5035 5035 5040 5040	C 6994	3015 3015 3017 3017	3020 3033 3033 3033 3032 3032 3032 3032	3035 3040 3042 3042 3045	2))
	176+ 176+ 176+ 176+ 176+	11111111111111111111111111111111111111		**************************************	**************************************	182+
)	0006557 0006557 0006657 0006661 0006661 0006621	000000000000000000000000000000000000000		00721 00724 00724 00724 00724 00725 00724 00725 00725 00725 00725 00725	000735 00735 00735 00735 00735 00745 00744 000744 00744 00744 00744 00744 00744 00744 00744 00744 00744 00744 00744 00744 00744 00744 00744 00746 00746 00735 00775 00000000	0074500745000751

٠

retarije na bila se kara na serije do na darena karejerana doverte kara ta kara ta kara ta kara ta serije se s Na serije se se serije se serije

`.**.**

 \mathcal{I}

A - 52

CNDUFR

4

60 TO 3998	0000 TM T (1 (1)+(1)(1)/2000 TM (1 (1) (1) (1) (1) (1) (1) (1) (1) (1)	60 TO 3999	3060 TM = T(LTA)	60 TO 3007	3065 TM = T(LTA)	60 TO 3032	3440 61_61 ~ 1./11./42111/461 15/51 / 1./211/11)1.50.11 6/16) 2 61462		4000 CONTINUE	END	IF(FLD(3,1,NSQ1(JJ1)).EQ.0) 60 TO 140	T1 = T(L)+460.0	T2 = T(1.TA)+460.0	GV = G(LG) + (T1 + T2 + T2) + (T1 + T2)	60 T0 145	3 + 0 = 6 + 16	145 SUMC = SUMC+6V	SUMCV = SUMCV+6V+T(LTA)	C CHECK FOR LAST CONDUCTOR	IF (NSQ1 (JU1), 61.0) 60 TO 135	T2 = 0D+T(L)+DN+(SUMCV+Q(L))/SUMC	C DETAIN THE CALCULATED TEMPERATURE ULTERENCE	T # AHS(T(L)-12) CTADE THE VEW TENDEDATINE	C DIORE THE REFERENCE	PARTIC THE MAYTHIM ARTHMETTE DELAYATION PUINGE	IFITCAM. GETTI GOTO 165	TCGM = T1	KON(37) = L	165 CONTINUE	C SEE IF RELAXATION CRITERIA WAS MET .	IF(TCGM.LE.CON(19)) 60 TO 175	170 CONTINUE C CTODE THE WAYTHIM ADITHMETIC DELAVATION CHAMES	175 CON(30) = TCGM	C COMPUTE THE ARITHMETIC TEMPERATURE CHANGE	TCGM = 0.0	D0 180 I # L1,NNC		T T AUSTILITAKLET	1		140 CONTINUE	C CEFTE ATMPCA WAS SATISFIED	IF (TCGW.6T.CON(11)) 60 TO 125	CON(16) = TCGM	185 KON(12) = 0	CALL VARBL2	C CHECK THE PACKUP SWITCH	IF (KON(12) .NE.0) 60 TO 105	C ADVANCE TIME		TSUX H TSUX+TSTEP METERS H TETERS		00 200 I = I NHD	- L. 1
182+	*29T	182#	182*	182*	182#	182	1004	1.82#	182*	182+	183+	1.84*	185+	186*	187*	188*	189*	190*	191+	192#	#06T		+96T				200+	201+	202+	2034	204+	8002 8002	202	208+	209*	210+	2114		101 v	110	2164	217+	218	219*	220+	221+	222+	223#	224#	225	226	\$122	228+	1.10
752		755	1756	757	1760	1910	201		766	767	770	772	773	174	175	176	777	000	000	1001	000	200				000	010	011	012	012	014	016	020	050	021	0.25	025		220		10.11		0.35	1037	0401	1041	041	1042	1042	1044	5.0	1046	1047	

3

. .

...

 CNOUFR
 CNOUFR

 01053
 230
 LEH = IEH+I

 01054
 231+
 200
 LEH = IEH+I

 01055
 233+
 C KICEH) = x(LE)
 = x(LE)

 01055
 233+
 C KICEH) = x(LE)
 = x(LE)

 01057
 234+
 C FICK FOR TIRE TO PRINT
 FIXON:055

 01057
 234+
 C FICK FOR TIRE TO PRINT
 FIXON:055

 01051
 235+
 C FIECK FOR PRINT
 FIXON:055

 01051
 239+
 C FICK FOR PRINT
 FIXON:04

 01055
 244+
 00170.1
 LEN
 FIXON:05

 01055
 244+
 00170.1
 LEN
 FIXON:05

 01055
 244+
 00170.1
 LEN
 FIXON:05

 01057
 244+
 PRINT
 FIXON:05
 PRINT

 01057
 244+

-

END OF UNIVAC 1108 FORTRAN V COMPILATION. 0 +DIAGNOSTIC+ MESSAGE(S) CNDUFR CODE RELOCATABLE

• , .

۰.

.

-

×.
Э.
ō
z
U

DI FOR** CNOUIK UNIVAC 1108 FORTRAM V ATHENA VERSION 131K-10D CREATED ON 20 AUG 70 This compilation was done on 09 JUN 70 AT 23:15:06

SUBROUTINE CNOUIK ENTRY POINT 003643

STORAGE USED (BLOCK . NAME . LENGTH)

027200	001000	000024	000000	000000													
	TENP	VAR			00000	00000	000000	00000	00000	000000	000001	100000	000000	00000	000000	000010	
10001	+CONST+	*SIMPLE	*ARRAYS	*BLANK	TITLE	TEMP	CAP	SOURCE	COND	PCI	PC2	KONST	ARRAY	FIXCON	XSPACE	DIMLNS	
.000	10000	0002	000	0005	0000	0001	0010	0011	0012	0013	0014	0015	0016	0017	0020	0021	

EXTERNAL REFERENCES (BLOCK . NAME)

VAR6L1 D1D1WM	PLYAWM D2D1WM VARHL2 OUTCAL EXIT	NERR25 EXP NWDU5 NIO25 NER105
0022	0024 0025 0026 0027	0031 0032 0033 0034 0034

.

STORAGE ASSIGNMENT FOR VARIABLES (BLOCK, TYPE, RELATIVE LOCATION, NAME)

000000	000032	000003	000000	000013	000000	000000	100000	100000	000025	000016
~ 0	< ac	₩.	H	H	H		m	-	œ	œ
100	0005	0005	0015	0005	0021	0021	0002	00200	0005	0005
CKM DEI TA	GV 6V	IE	20	LAX	Ę	NAT	NIJA	NS/92	0	F
000017	000036	000002	000015	000046	000022	000005	000001	000000	0000000	006000
œ o	c œ	н	H	-	-	-	-	P 4	Υ	œ
0005	0005	0005	0005	0005	0005	0021	0021	0014	0011	0001
0010 R 000000 C	0012 K 000000 G	0002 I 000005 I	0002 I 000014 JI	0002 1 000021 LA	0002 I 000027 LG	0002 I 000006 L1	0002 I 000004 11LA	0013 I 000000 4501	DUUZ R DUUDUU PASS	0002 H 000053 SUMCV
00000 A	000024 C2 000041 DT2	000000 H	000050 JJ2	000051 L	000042 LEH	C00030 LTA	000003 NGT	00002 001	000002 142	000052 SUMC
OC I	<u> </u>	. œ				1 HH 1 A				_≆
0016	2000	0000	0005	000	000	000	0021	0.021	0020	0005

0002 R 000A23 C1 0002 R 000040 CT1 0002 R 000047 JJ1 0017 I 000007 JJ1 0017 I 000007 K0N 0021 I 000007 L592 0021 I 000000 NDIM 0002 R 000000 NDIM 0002 R 000000 NDIM

18 FEB 71

.

CNQUIK

•

																					_				۰.				
T1 1000	10201	10351	1151	1401	1.80L	19991	2015L	20301	20451	2146	30	30151	30321	30501	39991	40151	40306	18664	5005L	5020L	5037L	5606	6076	897F	3 995L				
003617	000404	000000	002217	003342	003454	000705	001300	001420	001533	000232	000166	002712	003037	003153	003302	001000	001055	001152	002366	002431	002542	002205	002273	000016	003561			•.	
0002 H	1:00	0001	0001	1000	1000	0001	0001	1000	0001	1000	0001	0001	1000	0001	0001	0001	0001	0001	0001	1000	0001	0001	0001	0.00	0001		•		
TSUM 10L	10151	1032L	10526	1351	1756	1998L	2010L	20251	20421	2065L	16662 1	3010L	3030L	3045L	3998L	4012L	40256	45L	5000L	5017L	5035L	551	6000L	886F	195L	7666			
000122	000436	000551	003505	002574	003423	000103	001275	001360	001474	001636	001670	002707	003032	941500	003254	000761	001047	000244	001161	002414	002526	001724	002574	000011	002124	003612	ے۔ د		
0002 R	0001	1000	0001	0001	0001	1000	1000	0001	0001	0001	0001	0001	0001	0001	0001	1000	0001	0001	0001	0001	1000	0001	0001	0000	0001	0001			
TSTEP0 XX	10126	1030L	1056	130L	165L	195L	2007L	2020L	2040L	2060L	299AL	3007L	3025L	3042L	3065L	4010L	4022L	40401	50L	5015L	5030L	5436	60L	835F	90L	998L	:		
000010	000403	000530	002150	002244	003412	0.03542	001255	001352	001453	001632	001642	0.02667	002772	003106	003250	000760	001032	001144	001161	002413	002470	002154	001727	000000	002044	003603			1
0002 R	0001	1000	000	0001	0001	1000	0001	0001	0001	0001	0001	0001	0001	0001	0001	1000	1000	1000	0001	0001	1000	0001	0001	0000	1000	0001			
7 TSTEP	5 1010L	4 1025L	0 1045	6 125L	1 15	7 190L	0 2005L	3 2026	0 2035L	4 2055L	3 25L	2 3005L	4 3020L	5 3040L	4 3060L	3 4005L	6 4020L	7 4037L	0 51	4 5012L	2 5025L	0 5076	1 59991	5 65L	0 889F	5 997L			
00000	00036	000201	000661	002231	00014	00353	00125	00021	00145	00160	00015	00266	00276	00306	00324	00075	00101	00112	00012	00237	00246	0.02.02	00257	00177	00003	00357			•
0002	1000	1000	0001	0001	0001	1000	0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	0001	1000	0001	0001	0001	0001	0001	0001	0000	1000			
TPRINT	10056	10256	10401	1201	1451	1451	20001	2017L	2032L	2050L	2346	3000	30171	3035L	3055L	4000	40171	40351	19991	SOLOL	5022L	50401	59981	6166	BBEF	996L			
110000	000344	003432	000615	002230	003345	003465	000110	001316	001425	144100	000273	001673	002730	003062	003216	003305	00100	001113	001156	002373	002445	02557	002505	002314	000023	003567			
0002 8	0001	1000	1000	1000	000	0001	1000	1000	0001	1000	0001	1000	000	1000	1000	0001	1000	1000	1000	1000	1000	0001	0001	1000	0000	1000			

<pre>103 44 INCLUDE COMM.LIST 104 44 INCLUDE COMM.LIST 105 44 COMMON /TITLE/HIL) /TEMP/T(1) /CAP/C(1) /SOURCE/0(1) /COND/G(1) 105 44 COMMON /FIL/NOI(1) /XENS/K(1) /AFRAY/A(1) 106 44 COMMON /FIL/NOI(1) /XENS/K(1) /AFRAY/A(1) 107 44 COMMON /FIL/NOI(1) /XENS/MITH/X(1) 108 44 COMMON /FIL/NOI / NOI / NOI / SOURCE/0(1) /COND/G(1) 109 44 COMMON /FIL/NOI / NOI / NOI / SOURCE/0(1) /COND/G(1) 100 44 END 100 44 END 100 44 COMMON / NOI / NOI / NOI / NOI / SOURCE/0(1) / NOI / NOI 100 44 END 100 44 END 100 44 COMMON / NOI /</pre>	101	*** •••	်ပုပ	SUBROUTINE CNOLLY COMBINED EXPONENTIAL PREDICTION AND DUFORT-FRANKEL SINDA ROUTINE The short pseudo-compute sequence is required
 4. COMMON /PCI/NSO1(1) /PC2/NSO2(1) /KONST/K(1) /AFRAY/A(1) 4. COMMON /FIXCON/KON(1) /XSPACE/NDIM.NTH.X(1) 4. COMMON /FIXCON/KON(1) /XSPACE/NDIM.NTH.X(1) 4. COMMON /FIXCON/KON(1) /XSPACE/NDIM.NTH.X(1) 4. COMMON /FIXCON/KON(1) /XSPACE/NDIM.NTH.SO1.LSO2 5. COMMON /FIXCON/LOCON(1)) /(K(1) , XK(1)) , (X(1)) , NX(1)) 5. CONTROL CONSTANT 1 CONTANT DEFINITIONS AND NAMES ************************************	50	**		INCLUDE COMM.LIST COMMON /TITLE/H(1) /TEMP/T(1) /CAP/C(1) /SOURCE/0(1) /COND/G(1)
06 4* COMMON /FIXCON/KON(1) /XSPACE/NDIM.NTH.X(1) 07 4* COMMON /FIXCON/KON(1) /XSPACE/NDIM.NT.LS01.LS02 01 4* COMMON /FIXCON/KON(1) /XK(1) /XK(1) /X(1) /X(1) /X(1) 11 4* EQUIVALENCE (KON(1) /CON(1) / (K(1) /X(1) / (X(1) / NX(1)) 12 5* C CONTROL CONSTANT LONTAINS THE NAME ROBLEM TIME 13 5* C CONTROL CONSTANT LONTAINS THE TIME STEP USED 13 5* C CONTROL CONSTANT LONTAINS THE TIME STEP USED 13 5* C CONTROL CONSTANT LONTAINS THE TIME STEP PAGIOR TIME 13 5* C CONTROL CONSTANT LONTAINS THE TIME STEP PAGIOR TIME 14 END CONTROL CONSTANT LONTAINS THE TIME STEP PAGIOR CONTAINS TIMEND 15 C CONTROL CONSTANT LONTAINS THE TIME STEP PAGIOR (DTIMEND CONTROL CONSTANT LONTAINS THE TIME STEP PAGIOR (DTIMEND) 15 C CONTROL CONSTANT LONTAINS THE TIME STEP PAGIOR (DTIMEND) CONTROL CONSTANT LONTAINS THE TIME STEP PAGIOR (DTIMEND) 15 C CONTROL CONSTANT LOUND NUMBER OF TIME STEP PAGIOR (DTIMEND) CONTROL CONSTANT LOUND NUMBER OF TIME STEP PAGIOR (DTIMEND) 15 C CONTROL CONSTANT THE DISTON DO LOOPS. INTEGER (DTIMEND)	02	*		COMMON /PCI/NSO1(1) /PC2/NSO2(1) /KONST/K(1) /AFRAY/A(1)
07 44 COMMOU / DIMENS/ NND.NNA.NNT.NGT.NCT.NAT.LS01.LS02 11 44 DIMENSION CON(1).XK(1).NX(1) 12 44 EQUIVALENCE (KON(1).XK(1)).(K(1).XK(1)).NX(1)) 13 55 CONTROL CONSTANT DEFINITIONS AND NAMES ************************************	06	**		COMMON /FIXCON/KON(1) /XSPACE/NDIM+NTH+X(1)
10 4+ DIMENSION CON(1),XK(1),MX(1) 11 4+ EQUIVALENCE (KON(1),CON(1)),(K(1),XK(1)),(X(1),NX(1)) 12 5+ TINCLUDE DEFF,LIST 13 5+ CONTROL CONSTANT DEFINITIONS AND NAMES ************************************	07	*		COMMON /DIMENS/ HUND+NNA+MNT+NGT+NCT+NAT+LSO1+LSO2
11 4+ EQUIVALENCE (KON(1)).CON(1)).(K(1).XK(1)).(X(1).NX(1)) 12 4+ EQUIVALENCE (KON(1)).CON(1)).(K(1).XK(1)).(X(1)).NX(1)) 13 5+ C CONTROL DEFF/LIST 13 5+ C CONTROL CONSTANT 1 CONTAINS THE NEW PROBLEM TIME (TIMEN) 13 5+ C CONTROL CONSTANT 2 CONTAINS THE NEW PROBLEM TIME (TIMEN) 13 5+ C CONTROL CONSTANT 3 CONTAINS THE TIME STEP USED (DTIMEN) 13 5+ C CONTROL CONSTANT 3 CONTAINS THE TIME STEP USED (DTIMEN) 13 5+ C CONTROL CONSTANT 3 CONTAINS THE TIME STEP USED (DTIMEN) 13 5+ C CONTROL CONSTANT 3 CONTAINS THE TIME STEP PACION*EXPLICIT (CSGFAC 13 5+ C CONTROL CONSTANT 4 CONTAINS THE TIME STEP PACION*EXPLICIT (DTIMEN) 13 5+ C CCCT CONTAINS THE MAXIMUM ALLOWED TIME STEP PACION* (DTIMEN) (DTIMEN) 13 5+ C CCT CONTAINS THE MAXIMUM ALLOWED ANTHMETIC TEMP. DAMPING FACTOR (DTIMEN) 13 5+ C CCT CONTAINS THE NEW DIFFORED ANTHMETIC TEMP. CONTAINS THE NEW DIFFORED ANTHMETIC TEMP. CONTAINS THE NEW DIFFORED ANTHMETIC TEMP. CONTANON THE NEW DIFFORED ANTH	10	*		DIMENSION CON(1) * XK(1) * NX(1)
 4. END 5. CONTROL CONSTANT DEFINITIONS AND NAMES *********************************** 5. CONTROL CONSTANT 1 CONTAINS THE NEW PROBLEM TIME 5. CONTROL CONSTANT 2 CONTAINS THE TIME STEP USED 6. CONTROL CONSTANT 3 CONTAINS THE TIME STEP USED 7. CONTROL CONSTANT 3 CONTAINS THE TIME STEP USED 7. CONTROL CONSTANT 3 CONTAINS THE TIME STEP USED 7. CONTROL CONSTANT 3 CONTAINS THE TIME STEP USED 7. CONTROL CONSTANT 4 CONTAINS THE TIME STEP FACTOR*EXLICIT 7. COS IS THE INPUT NUMBER OF TERRATION DO LOOPS. INTEGER 7. CCCS IS THE INPUT NUMBER OF TERRATION SWITCH 7. CCCS CONTAINS THE DUTPUT EACH TERRATION SWITCH 7. CCC3 CONTAINS THE NEW ALLOWED TIME STEP FACTOR 7. CCC3 CONTAINS THE NEW DIFFICIT TERPTION SWITCH 7. CCC3 CONTAINS THE NEW DIFFICIAN ALLOWED ARTHMETIC TEMP. CHANGE (ATMPCA 7. CCC3 CONTAINS THE NEW DIFFICIAN ALLOWED ARTHMETIC TEMP. CHANGE (ATMPCA 7. CCC3 CONTAINS THE NEW DIFFICIAN ALLOWED ARTHMETIC TEMP. CHANGE (ATMPCA 7. CCC3 CONTAINS THE NEW DIFFICIAN ALLOWED ARTHMETIC TEMP. CHANGE (ATMPCA 7. CCC3 CONTAINS THE NEW DIFFICIAN ALLOWED ARTHMETIC TEMP. CHANGE (ATMPCA 7. CCC3 CONTAINS THE NEW DIFFICIAN ANTHMETIC TEMP. CHANGE (ATMPCA 7. CCC3 CONTAINS THE NEW DIFFICIAN AND THE NEW DI	11	* #		EQUIVALENCE (KON(1), CON(1)), (K(1), XK(1)), (X(1), NX(1))
13 5. INCLUDE DEFF,LIST 13 5. CONTROL CONSTANT DEFINITIONS AND NAMES ************************************	12	+ +		END
13 5: CONTROL CONSTANT DEFINITIONS AND NAMES ************************************	13	ي. 4		INCLUDE DEFF,LIST
 5. C CONTROL CONSTANT 1 CONTAINS THE NEW PROBLEM TIME (TIMEN) 5. C CONTROL CONSTANT 2 CONTAINS THE TIME STEP USED (TITMEU) 5. C CONTROL CONSTANT 3 CONTAINS THE TIME STEP USED (TITMEU) 5. C CONTROL CONSTANT 3 CONTAINS THE PROBLEM STOP TIME (TIMEN) 5. C CONTROL CONSTANT 4 CONTAINS THE PROBLEM STOP TIME (NLOOP) 5. C CONTROL CONSTANT 4 CONTAINS THE PROBLEM STOP TIME (NLOOP) 5. C CONTROL CONSTANT 4 CONTAINS THE PROBLEM STOP TIME (NLOOP) 5. C CC5 IS THE INPUT NUMBER OF TERRATION DO LOOPS, INTEGER (NLOOP) 5. C CC5 CONTAINS THE DIFFUSION TEMPERATURE CHANGE ALLOWED (DTMPCA CC5 CONTAINS THE MATINHMETIC TEMP. DAMPING FACTOR (DAMPD 5) 5. C CC10 CONTAINS THE NATINHMETIC TEMP. DAMPING FACTOR (DAMPD 5) 5. C CC10 CONTAINS THE NATINHMETIC TEMP. DAMPING FACTOR (DAMPD 1) 5. C CC10 CONTAINS THE NATINHMETIC TEMP. DAMPING FACTOR (DAMPD 1) 5. C CC10 CONTAINS THE NATINHMETIC TEMP. DAMPING FACTOR (DAMPD 1) 5. C CC10 CONTAINS THE NATINHMETIC TEMP. DAMPING FACTOR (DAMPD 1) 5. C CC10 CONTAINS THE NATINHMETIC TEMP. DAMPING FACTOR (DAMPD 1) 5. C CC10 CONTAINS THE NEW DIFFUSION TEMP. SATING FACTOR (DAMPD 1) 5. C CC10 CONTAINS THE NEW DIFFUSION TEMP. SATING FACTOR (DAMPD 1) 5. C CC10 CONTAINS THE NATINHMETIC TEMP. DAMPING FACTOR (DAMPD 1) 5. C CC10 CONTAINS THE NEW DIFFUSION TEMP. SATING FACTOR (DAMPD 1) 5. C CC10 CONTAINS THE NEASUP SATICH CHECKED AFTER VARIABLES (BACKUP 1) 5. C CC10 CONTAINS THE PRESENI TIME OFFERATURE CIANGE (ATUMPCA 1) 5. C CC10 CONTAINS THE PRESENI TIME OFFERATURE OFFERATURE (INCOMPCA 1) 5. C CC10 CONTAINS THE PRESENI TIME OFFERATURE (INCOMPCA 1) 5. C CC10 CONTAINS THE PRESENI TIME OFFERATURE (INCOMPCA 1) 5. C CC10 CONTAINS THE PRESENI TIME OFFERATURE (INCOMPCA 1) 5. C CC10 CONTAINS THE PRESENI TIME OFFERATURE (INCOMPCA 1) 	513	* <u>\$</u>	C++++	######################################
 5* C CONTROL CONSTANT 2 CONTAINS THE TIME STEP USED 5* C CONTROL CONSTANT 3 CONTAINS THE PROBLEM STOP TIME 5* C CONTROL CONSTANT 3 CONTAINS THE PROBLEM STOP TIME 5* C CONTROL CONSTANT 3 CONTAINS THE PROBLEM STOP TIME 5* C CONTROL CONSTANT 4 CONTAINS THE PROBLEM STOP TIME 5* C CONTROL CONSTANT 4 CONTAINS THE PROBLEM STOP TIME 5* C CCS IS THE INPUT NUMBER OF TIRERATION DO LOOPS. INTEGER 5* C CCT CONTAINS THE DIFFUSION TEMPERATURE CHAMGE ALLOWED 60 CCT CONTAINS THE DUTUT EACH TTERATION SWITCH 60 CCT CONTAINS THE NATINUM ALLOWED TIME STEP 60 CCT CONTAINS THE NATINUM ALLOWED ANTTHMETIC TEMP. DAMPTUG FOR THE STEP 60 CCT CONTAINS THE NATINUM ALLOWED ANTTHMETIC TEMP. CHANGE (DAMPA 55. C CC10 CONTAINS THE NATINUM ALLOWED ANTTHMETIC TEMP. CHANGE (DAMPA 55. C CC10 CONTAINS THE NATION ALLOWED ANTTHMETIC TEMP. CHANGE (DAMPA 13 55. C CC10 CONTAINS THE NATION ALLOWED ANTTHMETIC TEMP. CHANGE (DAMPA 13 55. C CC10 CONTAINS THE NATION ALLOWED ANTTHMETIC TEMP. CHANGE (DAMPA 13 55. C CC10 CONTAINS THE NEW DIFFUSION TEMP. ANTTHMETIC TEMP. CHANGE (DAMPA 13 55. C CC10 CONTAINS THE NEEDED ANTTHMETIC TEMP. CHANGE (DAMPA 13 55. C CC10 CONTAINS THE NEEDED ANTTHMETIC TEMP. CHANGE (TIMECA 15 55. C CC10 CONTAINS THE NEEDED ANTTHMETIC TEMP. CHANGE (TIMECA 15 55. C CC10 CONTAINS THE NEEDED ANTTHMETIC TEMP. CHANGE (TIMECA 15 55. C CC10 CONTAINS THE NEEDED ANTTHMETIC TEMP. CHANGE (TIMECA 15 55. C CC10 CONTAINS THE NEEDED ANTTHMETIC TEMP. CHANGE (TIMECA 15 55. C CC10 CONTAINS THE NEEDED ANTTHMETIC TEMP. CHANGE (TIMECA 15 55. C CC10 CONTAINS THE PRESENT TIME (TIME 07 PANTANE 15 75. C CC10 CONTAINS THE PRESENT TIME ON PHONEMERA 100 CONTAINS THE PR	13	ۍ ۵	U	CONTROL CONSTANT 1 CONTAINS THE NEW PROBLEM TIME (TIMEN)
 5* C CONTROL CONSTANT 3 CONTAINS THE PROBLEM STOP TIME 5* C CONTROL CONSTANT 4 CONTAINS THE TIME STEP FACTOR'EXPLICIT (CSGFAC) 5* C COLDIALNS THE INPUT NUMBER OF ITERATION DO LOOPS, INTEGER (NLOOP) 5* C CC5 SI THE OUTPUT FAMPERATURE CLANDE ALLOWED (DTMPCA 5* C CC7 CONTAINS THE DUTPUT FACH ITERATION SWITCH (DTMPCA 5* C CC7 CONTAINS THE OUTPUT FACH ITERATION SWITCH (DTMPCA 5* C CC7 CONTAINS THE OUTPUT FACH ITERATION SWITCH (DTMPCA 5* C CC7 CONTAINS THE OUTPUT FACH ITERATION SWITCH (DTMPCA 5* C CC7 CONTAINS THE NAXIMUM ALLOWED TIME STEP (DTMPCA 5* C CC10 CONTAINS THE NEW DIFFUSION TEMP. DAMPTING FACTOR (DAMPA 5* C CC10 CONTAINS THE NEW DIFFUSION TEMP. DAMPTING FACTOR (DAMPA 5* C CC11 CONTAINS THE NEW DIFFUSION TEMP. DAMPTING FACTOR (DAMPA 5* C CC12 CONTAINS THE NEW DIFFUSION TEMP. DAMPTING FACTOR (DAMPA 5* C CC12 CONTAINS THE NEW DIFFUSION TEMP. DAMPTING FACTOR (DAMPA 5* C CC11 CONTAINS THE NEW DIFFUSION TEMP. DAMPTING FACTOR (DAMPA 5* C CC12 CONTAINS THE NEW DIFFUSION TEMP. DAMPTING FACTOR (DAMPA 5* C CC12 CONTAINS THE NEW DIFFUSION TEMP. DAMPTING FACTOR (DAMPA 5* C CC12 CONTAINS THE NEW DIFFUSION TEMP. DAMPTING (TIMPCA 5* C CC12 CONTAINS THE NEW DIFFUSION TEMP. DAMPTING (TIMPCA 5* C CC12 CONTAINS THE NEW DIFFUSION TEMP. DAMPTING (TIMPCA 5* C CC12 CONTAINS THE NEW DIFFUSION TEMP. DAMPTING (TIMPCA 5* C CC12 CONTAINS THE NEW DIFFUSION TEMP. DAMPTING (TIMPCA 5* C CC12 CONTAINS THE NEW DIFFUSION TEMP. DAMPTING (TIMPCA 5* C CC12 CONTAINS THE NEW DIFFUSION TEMP. DAMPTING (TIMPCA 5* C CC14 CONTAINS THE NEW DIFFUSION TEMP. DAMPTING (TIMPCA 5* C CC14 CONTAINS THE NEW DIFFUSION TEMP. DAMPTING (TIMPCA 5* C CC14 CONTAINS THE NEW DIFFUSION DAMPENDED (TIMPCA 	13	ب م	U	CONTROL CONSTANT 2 CONTAINS THE TIME STEP USED (DTIMEU)
 5* C CONTROL CONSTANT 4 CONTAINS THE TIME STEP FACTOR.EXPLICIT (CSGFAC) 5* C COLS IS THE INDUT NUMBER OF ITERATION DO LOOPS. INTEGER (NLOOP) 5* C COLS CONTAINS THE DIFFUSION TEMPERATURE CHANGE ALLOWED (DTMPCA) 5* C CCC CONTAINS THE DIFFUSION TEMPERATURE CHANGE ALLOWED (DTMPCA) 5* C CCC CONTAINS THE MAXIMUM ALLOWED TIME STEP 5* C CCO CONTAINS THE NAXIMUM ALLOWED TIME STEP 5* C CCO CONTAINS THE NAXIMUM ALLOWED TIME STEP 5* C CCO CONTAINS THE NAXIMUM ALLOWED TIME STEP 5* C CCO CONTAINS THE NAXIMUM ALLOWED ANTHHMETIC TEMP. CHANGE ACTOR (DAMPA) 5* C CCII CONTAINS THE NAXIMUM ALLOWED ARTITHMETIC TEMP. CHANGE (DAMPA) 5* C CCII CONTAINS THE PACKUP SWITCH CHECKED AFTER VARIABLES (BACKUP) 5* C CCII CONTAINS THE PACKUP SWITCH CHECKED AFTER VARIABLES (BACKUP) 5* C CCII CONTAINS THE PACKUP SWITCH CHECKED AFTER VARIABLES (BACKUP) 5* C CCII CONTAINS THE PRESENT TIME OR PHORLEM START TIME (TIMPCA) 5* C CCII CONTAINS THE PRESENT TIME OR PHORLEM START TIME (TIMPCA) 5* C CCII CONTAINS THE PRESENT TIME OR PHORLEM START TIME (TIMPCA) 5* C CCII CONTAINS THE PRESENT TIME OR PHORLEM START TIME (TIMPCA) 	13	÷.	ų	CONTROL CONSTANT 3. CONTAINS THE PROBLEM STOP TIME (TIMEND)
 5* C CC5 IS THE INPUT NUMBER OF ITERATION DO LOOPS, INTEGER (NLOOP) 5* C CC6 CONTAINS THE DIFFUSION TEMPERATURE CHANGE ALLOWED (DTMPCA) 5* C CC7 CONTAINS THE NUTH ALLOWED TIME STEP 5* C CC9 CONTAINS THE NAXIMUM ALLOWED TIME STEP 5* C CC9 CONTAINS THE NEW ARTINHETIC TEMP. DAMPING FACTOR (DAMPA) 5* C CC10 CONTAINS THE NEW DIFFUSION TEMP. DAMPING FACTOR (DAMPA) 5* C CC10 CONTAINS THE NEW DIFFUSION TEMP. DAMPING FACTOR (DAMPA) 5* C CC11 CONTAINS THE NEW DIFFUSION TEMP. DAMPING FACTOR (DAMPA) 5* C CC11 CONTAINS THE NEW DIFFUSION TEMP. DAMPING FACTOR (DAMPA) 5* C CC11 CONTAINS THE NEW DIFFUSION TEMP. THENTIC TEMP. CHANGE (ATMPCA) 5* C CC12 CONTAINS THE PRESENT TIME OR PHORLEM START TIME (TIMEC) 5* C CC13 CONTAINS THE PRESENT TIME OR PHORLEM START TIME (TIMEC) 5* C CC14 CONTAINS THE PRESENT TIME OR PHORLEM START TIME (TIMEC) 	ŝ	م	U	CONTROL CONSTANT 4 CONTAINS THE TIME STEP FACTOR. EXPLICIT (CSGFAC)
135+CCC66CONTAINSTHEDIFUSIONTEMPERATURECHANGEALLOWED(DIMPCA)135+CCC7CONTAINSTHEOUTPUTEACHITERATIONSWITCH(OPTIME)135+CCC6CONTAINSTHEMANALLOWED(DTIME)135+CCC9CONTAINSTHENEWDIAPTONSMITCH(DTIME)135+CCC10CONTAINSTHENEWDIAPTONFEMP.DAMPOR135+CCC11CONTAINSTHENAXICHCHECKEDATTMPCA135+CCC12CONTAINSTHENAXICHCHECKEDAFTERVARIABLES(BACKUP135+CCC13CONTAINSTHEPRESENTTIMEORPRESENTTIME135+CCC13CONTAINSTHEPRESENTTIMEORPRESENTTIME135+CCC13CONTAINSTHEPRESENTTIMEORPRESENTTIME145+CCC14CONTAINSTHEPRESENTTIMEORPRESENTTIME155+CCC14CONTAINSTHEPRESENTTIMEORPRESENTTIME155+CCC14CONTAINSTHEPRESENTTIMEORPRESENTTIME155+CCC14CONTAINSTHEPRESENTT	13	÷.	U	CC5 IS THE INPUT NUMBER OF ITERATION DO LOOPS. INTEGER (NLOOP)
13 5+ C CC7 CONTAINS THE OUTPUT EACH ITERATION SWITCH (OPEITR) 13 5+ C CC8 CONTAINS THE MAXIMUM ALLOWED TIME STEP (DTIMEH) 13 5+ C CC9 CONTAINS THE NEW ARITHMETIC TEMP. DAMPTUG FACTOR (DTIMEH) 13 5+ C CC10 CONTAINS THE NEW DIFFICIENT TEMP. DAMPTUG FACTOR (DAMPD) 13 5+ C CC11 CONTAINS THE NEXTCH ALLOWED ARITHMETIC TEMP. CHANGE (ATMPCA) 13 5+ C CC12 CONTAINS THE NEXTCH ALLOWED ARITHMETIC TEMP. CHANGE (ATMPCA) 13 5+ C CC13 CONTAINS THE PRESENT THE OR PHOLEME START TIME (TIMEO) 13 5+ C CC13 CONTAINS THE PRESENT TIME OR PHOLEME START TIME (TIMEO) 13 5+ C CC13 CONTAINS THE PRESENT TIME OR PHOLEME START TIME (TIMEO) 14 5+ C CC13 CONTAINS THE PRESENT TIME OR PHOREMATORE (TIMEO) 15 5+ C CC13 CONTAINS THE PRESENT TIME OR PHOREMATORE (TIMEO) 15 5+ C CC14 CONTAINS THE PRESENT TIME OR PHO	51	ۍ ۴	Ų	CC6 CONTAINS THE DIFFUSION TEMPERATURE CHANGE ALLOWED (DTMPCA)
 5. C CCB CONTAINS THE MAXIMUM ALLOWED TIME STEP 5. C CCB CONTAINS THE NAWITH TIC TEMP. DAMPTHG FACTOR (DAMPA 5. C CC10 CONTAINS THE NEW DIFFUSION TEMP. DAMPTHG FACTOR (DAMPA 5. C CC11 CONTAINS THE NEW DIFFUSION TEMP. DAMPTHG FACTOR (DAMPA 13 5. C CC11 CONTAINS THE NAKINUA ALLOWED ARITHMETIC TEMP. CHANGE (ATMPCA 13 5. C CC11 CONTAINS THE BACKUP SWITCH CHECKED AFTER VARIABLES (BACKUP 13 5. C CC12 CONTAINS THE BACKUP SWITCH CHECKED AFTER VARIABLES (BACKUP 13 5. C CC13 CONTAINS THE BACKUP SWITCH CHECKED AFTER VARIABLES (BACKUP 13 5. C CC13 CONTAINS THE BACKUP TIME OR PHONEEM START TIME (TIME 13 5. C CC14 CONTAINS THE PRESENT TIME OR PHONEEM START TIME (TIME 13 5. C CC14 CONTAINS THE PRESENT TIME OR PHONE AN ITENTION (TIME 13 5. C CC14 CONTAINS THE PRESENT TIME OR PHONE (ATTMPCA 14 5. C CC14 CONTAINS THE DEFENT ON TEMPERATURE CLANGE (ATTMPCA 15 5. C CC14 CONTAINS THE DEFENT ON TEMPERATURE CLANGE (ATTMPCA 15 5. C CC14 CONTAINS THE DEFENT ON TEMPERATURE CLANGE (ATTMPCA 16 7. C CC15 CONTAINS THE DEFENT ON TEMPERATURE CLANGE (ATTMPCA 17 5. C CC14 CONTAINS THE DEFENT ON TEMPERATURE CLANGE (ATTMPCA 17 5. C CC15 CONTAINS THE DEFENT ON TEMPERATURE CLANGE (ATTMPCA 17 5. C CC15 CONTAINS THE DEFENT ON TEMPERATURE CLANGE (ATTMPCA 18 5. C CC14 CONTAINS THE DEFENT ON TEMPERATURE CLANGE (ATTMPCA 18 5. C CC15 CONTAINS THE DEFENT ON TEMPERATURE CLANGE (ATTMPCA 18 5. C CC15 CONTAINS THE DEFENT ON TEMPERATURE CLANGE (ATTMPCA 18 5. C CC15 CONTAINS THE DEFENT ON TEMPERATURE CLANGE (ATTMPCA 18 5. C CC15 CONTAINS THE DEFENT ON TEMPERATURE CLANGE (ATTMPCA 18 5. C CC15 CONTAINS THE DEFENT ON TEMPERATURE CLANGE (ATTMPCA 18 5. C CC15 CONTAINS THE DEFENT ON TEMPERATURE CLANGE (ATTMPCA 18 5. C CC15 CONTAINS THE DEFENT ON TEMPERATURE CLANGE (ATTMPCA 18 5. C CC15 CONTAINS THE DEFENT ON TEMPERATURE CLANGE (ATTMPCA 18 5. C CC15 CONTAINS THE DEFENT ON TEMPERATURE CLANGE (ATTMPCA 18 5. C CC15 CONTAINS THE DEFENT ON TEMPERATURE CLANGE (ATTMPCA 18 5. C CC15 CONTAINS THE DEFENT ON TEMPERATURE CLANGE (ATTMPCA 18 5. C CC15 CONTAINS THE D	5		U	CC7 CONTAINS THE OUTPUT EACH ITERATION SWITCH (OPEITR)
13 5* C CC9 CONTAINS THE NEW ARITHWETIC TEMP. DAMPING FACTOR (DAMPA 13 5* C CC10 CONTAINS THE NEW DIFFUSION TEMP. DAMPING FACTOR (DAMPD 13 5* C CC11 CONTAINS THE NEW DIFFUSION TEMP. DAMPING FACTOR (DAMPCA 13 5* C CC11 CONTAINS THE MAXIMUM ALLOWED ARITHWETIC TEMP. CHANGE (ATMPCA 13 5* C CC12 CONTAINS THE BACKUP SWITCH CHECKED AFTER VARIABLES (BACKUP 13 5* C CC13 CONTAINS THE PRESENT TIME OR PHOBLEM START TIME 13 5* C CC13 CONTAINS THE PRESENT TIME OR PHOBLEM START TIME 13 5* C CC14 CONTAINS THE MEAN TIME BETWEEN AN ITERATION 13 5* C CC14 CONTAINS THE DIFFUSION TEMPERATURE CLANCE (ATMPCA 14 5* C CC14 CONTAINS THE DIFFUSION TEMPERATURE CLANCED ANTOCCONTAINS THE DIFFUSION TEMPERATURE CLANCED AN	13	÷.	U	CCB CONTAINS THE MAXIMUM ALLOWED TIME STEP
13 5* C CC10 CONTAINS THE NEW DIFFUSION TEMP. DAMPING FACTOR (DAMPD) 13 5* C CC11 CONTAINS THE NAXIAUM ALLOWED ARITHMETIC TEMP. CHANGE (ATMPCA 13 5* C CC11 CONTAINS THE BACKUP SWITCH CHECKED AFTER VARIABLES (BACKUP 13 5* C CC12 CONTAINS THE PRESENT TIME OR POBLEM START TIME (TIMEO 13 5* C CC13 CONTAINS THE PRESENT TIME OR POBLEM START TIME (TIMEO 13 5* C CC14 CONTAINS THE MEAN TIME BETWERU AN ITERATION (TIMEO 13 5* C CC14 CONTAINS THE MEAN TIME OR POBLEM START TIME (TIMEO 13 5* C CC14 CONTAINS THE DEFUSION TEMPERATURE CLARCULATED (DIMPCC 13 5* C CC14 CONTAINS THE DEFUSION TEMPERATURE CLARCULATED (DIMPCC 14 5* C CC14 CONTAINS THE DIFFUSION TEMPERATURE CLARCULATED (DIMPCC 14 5* C CC14 CONTAINS THE DIFFUSION TEMPERATURE CLARCULATED (DIMPCC 14 5* C CC14 CONTAINS THE DIFFUSION TEMPERATURE CLARCULATED (DIMPCC 14 5* C CC14 CONTAINS THE DIFFUSION TEMPERATURE CLARCULATED (DIMPCC 14 5* C CC14 CONTAINS THE DIFFUSION TEMPERATURE CLARCULATED (DIMPCC 14 5* C CC14 CONTAINS THE DIFFUSION TEMPERATURE CLARCULATED (DIMPCCC 14 5* C CC14 CONTAINS THE DIFFUSION TEMPERATURE CLARCULATED (DIMPCCC 14 5* C CC14 CONTAINS THE DIFFUSION TEMPERATURE CLARCULATED (DIMPCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCC	51	ئ	U	CC9 CONTAINS THE NEW ARITHMETIC TEMP. DAMPING FACTOR (DAMPA)
13 5.4 C CCII CONTAINS THE MAXIMUM ALLOWED ARITHMETIC TEMP. CHANGE (ATMPCA) 13 5.4 C CCI2 CONTAINS THE BACKUP SWITCH CHECKED AFTER VARIABLES (BACKUP 13 5.4 C CCI3 CONTAINS THE PRESENT TIME OR PHOBLEM START TIME (TIMEO) 13 5.4 C CC14 CONTAINS THE MEAN TIME BETWEEN AN ITEMATION (TIMEO) 13 5.4 C CC14 CONTAINS THE MEAN TIME BETWEEN AN INFRATION (TIMEO) 13 5.4 C CC15 CONTAINS, THE DEFUSION TEMPERATURE CLANCULATED (DIMPCC) 13 5.4 C CC15 CONTAINS, THE DEFUSION TEMPERATURE CLANCULATED (DIMPCC)	13	*5	U	CC10 CONTAINS THE NEW DIFFUSION TEMP. DAMPING FACTOR (DAMPD)
13 5* C CC12 CONTAINS THE BACKUP SWITCH CHECKED AFTER VARIABLES (BACKUP) 13 5* C CC13 CONTAINS THE PRESENT TIME OR PHOHLEM START TIME (TIMEO 13 5* C CC14 CONTAINS THE MEAN TIME BETWELU AN ITERATION (TIMEM) 13 5* C CC14 CONTAINS THE DIFFUSION TEMPERATURE CLANCE ATED (DIMPCC)	13	÷.	U	CCII CONTAINS THE MAXIMUM ALLOWED ARITHMETIC TEMP, CHANGE (ATMPCA)
13 5* C CC13 CONTAINS THE PRESENT TIME OR PROBLEM START TIME (TIMEO) 13 5* C CC14 CONTAINS THE MEAN TIME BETWEEN AN ITERATION (TIMEM 13 5* C CC15 CONTAINS THE DIFFUSION TEMPERATURE CHANGE CALCULATED (DTMPCC	13	ۍ ۵	U	CC12 CONTAINS THE BACKUP SWITCH CHECKED AFTER VARIABLES (BACKUP)
13 5* C CC14 CONTAINS THE REAN TIME BETWEEN AN ITERATION (TIMEM) 13 5* C CC15 CONTAINS THE DIFFUSION TEMPERATURE CHANGE CALCULATED (DIMPCC	13	24	υ υ	CC13 CONTAINS THE PRESENT TIME OR PROBLEM START TIME (TIMEO)
13 5. C CONTAINS THE DIFFUSION TEMPERATURE CHANGE CALCULATED (DTMPCC)	2	<u>ي</u>	Ų	CC14 CONTAINS THE REAN TIME BETWEEN AN ITERATION (TIMEM)
	F.		ų	COIS CONTAINS THE DIFFUSION TEMPERATURE CHANGE CALCULATED (DTMPCC)

•

....

in en en Sectores

e. General and the control of the second sec

٠.,

CNOUTK

،

•

• •

•

. . .

•

• •

						333 100 100 100 100 100 100 100 100 100	₩₩ ₩ ₩ ₩ ₩ ₩	AUN*
: . ,)	IE TIME STEP EXCEED OUTPUT INTERVAL #30:20 We stede evteen output interval	ME SIEPS CACEU VOLPUL TRIERVAL 18) 60 TO 30 VAL GRADVALLY O THE CONSTANTS	IS THAN THE TIME STEP ALLOWED 0 997 P Ween iterations)/2.0	TONS AND EXTRA LOCATIONS Peratures into the fitra locations		.0) 60 TO 2000 0.1025,1030,1035,1040,1045), HTYPE	((TK),C(1)) (LK),C2) (LK),C2)	
ÇNGUTK	DOES THE TIME SUM PLUS TH IF(TSUMATSTEP-CON(18)) 25 DONT EXCEED IT TSTEP = CON(18)-TSUM 60 TO 30	UDES 140 500 FLOS 140 1 IF(ISUM+2.0*ISTEP.LE.CON APPROACH THE UUTPUT INTER ISTEP = (CON(18)-TSUM)/2. STORE DELTA TIME STEP IN CON(2) = TSTEP IN	IS THE TIME STEP USED LES IF (TSTEP.LT.CON(21)) GO 1 CALCULATE THE NEW TIME CON(1) = TPRINT+TSUM+TSTE COMPUTE THE MEAN TIME HEI COM(1) = (CON(1)+CON(13)	ZERO OUT ALL SOURCE LOCAT DO 35 I = 1.NND LE = IE+I X(LE) = 0.0 Q(I) = 0.0 CONTINUE CONTINUE	Ffinate of the second s	J1 = 0 J2 = 1 TC6M = 1.E+10 CALCULATE Q SUM AND G SUN D0 05 1 = 1.NND D0 05 1 = 1.NND TNCLUE VARC.LIST INCLUE VAR	Call DIDIWM(T(I).4(LA).X ⁺ 60 T0 1999 Call DIDIWM(T(I).4(LA).X ⁺ 2 = J2+1 La = FLD(5,17,N502(J2)) LA = FLD(22,14,N502(J2)) LK = FLD(22,14,N502(J2)) Call DIDIWM(T(I).4(LA).X ⁺ 60 T0 1999 61 = XK(LK).4XK(LA) 61 = XK(LK).4XK(LA)	J2 = J2+1 LA = FLU(5,17,4502(J2)) LK = FLU(22,14,NSQ2(J2))
	2 2 1 2 2 2	ີ ເຊິ່ງ ເຊິ່ງ		ີ ຕິ ບໍ່ບ	00) # # 1	⁻ υ	1010 1010 1012 1012 1015	
				10000000 ********	00000000000000000000000000000000000000	77777777777777777777777777777777777777	778 778 778 778 778 778 778 778 778 778	78+ 78+
0	00164 00164 00164 00164	00173	00175 00175 00175 000175	002001000000000000000000000000000000000	0005210 005210 0052216 0052216 0052520 0052520 0052520 0055520 00055520 00055520 00055520 00055520 00000000	20000000000000000000000000000000000000	00000000000000000000000000000000000000	00261 00262 00263

n a star har she and a star a star

· ,

-

•

•

A - 58

€4 ● ●	A NEA * S NEA * *	≥ ≥ 0 2 = = 2 = =	N N N N N N N N N N N N N N N N N N N	N 3 0 N 3 0 N 4 0 N 	A PIEW A PIEW
CNOUIK C2 = XK(LK)+XK(LA) G0 T0 1998 1025 Call Plyawm(a(La),T(I),A(La+1),XK(LK),C(I)) 1030 Call Plyamm(a(La),T(I),A(La+1),XK(LK),C1)	1032 J2 = J2+1 LA = FLD(5.17.NS02(J2)) LK = FLD(22.14.NS02(J2)) Call Plyawm(A(LA).7(1).A(LA+1).XK(LK).C2) Coll Plyawm(A(LA).7(1).A(LA+1).XK(LK).C2) 1035 C1 = XK(LK)*XK(LA) 1040 Call Plyawm(A(LA).7(1).A(LA+1).XK(LK).C1) 1040 Call Plyawm(A(LA).XK(LA).XK(LK).C1) 1040 Call Plyawm(A(LA).XK(LA).XK(LK).C1) 1040 CALL Plyawm(A(LA).XK(LA).XK(LK).C1) 1040 CALL Plyawm(A(LA).XK(LA).XK(LA).XK(LK).C1) 1040 CALL Plyawm(A(LA).XK(L	LX = FLD(5,17,11502(J2)) LX = FLD(5,17,11502(J2)) LX = FLD(2,14,41502(J2)) C2 = XK(LK)*XK(LA) G0 T0 1994 G0 T0 1999 1998 C(T) = C1+C2 1998	<pre>include Varg.List if(FLD(4.1.NSGI(J1+1)).EQ.0) GO TO 5000 NTYPE = FLD(5.17.NSQ2(J2)) LA = FLD(5.17.NSQ2(J2)) Lk = FLD(22.14.NSQ2(J2)) GO TO (4005.4010.4015.4020,4025.4030.4035.4040.4030). NTYPE 4005 G(1) = XK(LK)+G(1) 4005 G(1) = XK(LK)+G(1) 4012 CALL D1D1HM(T(1)).A(LA).XK(LK).Q2). 4012 CALL D1D1HM(T(1)).A(LA).XK(LK).Q2).</pre>	<pre>4015 01 = 0.0 4017 CALL D1D1wM(CON(14).A(LA).XK(LK).02) 60 T0 4998 4020 CALL D1D1wM(CON(14).A(LA).XK(LK).01) 4022 J2 = J2+1 LA = FLD(2.14.NS02(J2)) LK = FLD(22.14.NS02(J2)) LK = FLD(22.14.NS02(J2)) 60 T0 4017 4025 01 = XK(LK)*XK(LA) 4025 01 = XK(LK)*XK(LA) 4022 CALL D1D1WM(CON(14).A(LA).XK(LK).01)</pre>	J2 J2 J241 LA = FLD(5,17,NS02(J2)) LK = FLD(25,17,NS02(J2)) 02 = XK(LK)*XK(LA) 035 CALL D1D1WM(CON(14),A(LA),XK(LK).01) 4037 J2 = J2+1 LA = FLD(5,17,NS02(J2)) LA = FLD(5,17,NS02(J2)) LA = FLD(5,17,NS02(J2)) LA = FLD(5,17,NS02(J2)) 4040 01 = XK(LK)*XK(LA) 4040 01 = XK(LK)*XK(LA) 4990 011 = Q1(02+0(1)) 4990 011 = Q1(02+0(1)) 4990 011 = Z(10,240(1)) 4990 011 = Z(10,240(1)) 4991 011 = Z(10,240(1)) 4991 011 = Z(10,240(1)) 4991 012 = Z(10,240(1)) 4901 01
78+ 78+ 78+ 78+ 78+	777777788		*67 *667 *667 *677 *677 *677 *677 *677	00000000000000000000000000000000000000	***************************************
00264 00265 00265 00265 00267	00271 00273 00273 00275 00275 00275 00277	00201 00205 00205 00205 00205 00205 00205 00212 00212 00212 00212 00212	00314 00315 00317 00322 00322 00322 00325 00325 00326 00326	00331 00331 00332 003334 00335 00335 00340 00340 00340	003445 003445 003445 00351 00355 003554 003554 003554 003554 003554 003554

•

and the second second

• 1

.

A - 59

		5	•	· • · · · · · · · · · · · · · · · · · ·	
					an an tha an the grant th
					5
)				
	Ĵ.				
	1120	18 18 OI	2:3:0	TEN	3 2 4
		N N N N N N N N N N N N N N N N N N N			uui 22 # * * *
					×
•					
	ື້				
	502				
	2050				
	345				
$\langle \rangle$	0,20			•	
·	5.20		(1)	62	6 9
	e 0			-K) • 6	
	a v	<u> </u>		×	
	0 300 0 30,2 6))	.62	X CK	X · C	(LK)
	08 60 To 300 25,2030,2 ,6(LG))	(LK),62 K),61))•XK(LK +1)•XK(LA+1)•X +1)•XK(7 XK (LK
)))UCTOR)) () () () () () () () () () () () ()), XK (LK), 62 XK (LK), 63)) La+1)•XK(LK A(La+1)•XK() • A (LA+1) • X A (LA+1) • XK (a(LA),XK(LK
	11)) (J1)) (J1)) (J1)) (CPE (J2))(J2))	12)) 12)) 1(12), XK(LK), 62 4(LK), 61)	12)) (12)) 4.A(LA+1).XK(LK (1).A(LA+1).XK(J2)) (J2)) (LTA),A(LA+1),XK((1),A(LA+1),XK(12)) (12)) (14),A(LA),XK(LK) (14),62)
	01(J1)) 0 65 NS01(J1)) T ION COMDUCTOR J1)) COMDUCTOR OS2(J2)) S02(J2)) S02(J2)) S02(J2)) S02(J2)) S02(J2)) (12)) S02(J2)) (12)) (12)) (14),XK(LK),6(LG)) (1A),XK(LK),61)	02(J2)) 1502(J2)) A),A(LA),XK(LK),G2 A) A(LA),XK(LK),G1)	02(J2)) 1502(J2)) A) ()/2.0 ()/TM.A(LA+1).XK(LK ().T(1),A(LA+1).XK(02(J2)) 1502(J2)) 1),T(LTA),A(LA+1),X A) A)	02(J2)) 592(J2)) A) ()/2,0 ()/14),A(LA),XK(LK) ()/14),A(LA),XK(LK)
	6, M501(J1)) 60 TO 65 .14, MS01(J1)) /LIST S01(J1)).60,00 GO TO 300 56, MS02(J2)) 7, MS02(J2)) 14, MS02(J2)) 14, MS02(J2)) 14, MS02(J2)) 10,2015,2020,2025,2030,2 (LTA))/2.0 TM,A(LA),XK(LK),6(LG)) T(1),A(LA),XK(LK),61)	7,NSQ2(J2)) 14,NSQ2(J2)) T(LTA),A(LA),XK(LK),G2 XK(LA) T(1),A(LA),XK(LK),G1)	7, NSQ2 (J2)) 14, NSQ2 (J2)) XK (LA) (LTA)) /2, 0 A (LA), TM, A (LA+1), XK (LK A (LA), T (1), A (LA+1), XK (7,NSQ2(J2)) 14,NSQ2(J2)) A(LA),T(LTA),A(LA+1),X XK(LA) XK(LA),T(1),A(LA+1),XK(7,11502(J2)) 14,11502(J2)) XK (LA) TM, CON(14), A(LA), XK (LK) TM, CON(14), A(LA), XK (LK)
IK	1 (5,16,NSO1(J1)) 00,60,T0.65 0122,14,NSO1(J1)) VAR6,LIST R RADIATION COMDUCTOR 1,NSO2(J2)) (5,17,NSO2(J2)) (5	1 (5,17,NSQ2(J2)) (22,14,NSQ2(J2)) (22,14,NSQ2(J2)) (22,14,NSQ2(J2)) (22,14,NSQ2(J2)) 98 (41) 17 11 17 11 14 11 17 11 17	1 (5;17,NSQ2(J2)) (22,14,NSQ2(J2)) LK)*XK(LA) 98 1)+T(LTA))/2,0 AWM(A(LA),TM,A(LA+1),XK(LK 99 21 21 22 22 23 24 24 24 24 24 24 24 24 24 24 24 24 24	1 (5,17,NSQ2(J2)) (22,14,NSQ2(J2)) Arm(a(LA),T(LTA),A(LA+1),X 98 LK)+XK(LA) Lk)+XK(LA) 42 42 42	(5.17,11502(J2)) (22,14,11502(J2)) (22,14,11502(J2)) 98 1)+T(LTA))/2,0 1)w(TM,CON(14),A(LA),XK(LK) 1)w(TM,CON(14),A(LA),XK(LK) 1) 1) 1) 1) 1./(1,./61+1./62)
CNOUIK	<pre>INUE J1+1 J1+1 FED(5.16,NSQ1(J1)) 6.60.00 60 T0 65 6.60.00 60 T0 65 6.610.00 60 T0 60 T0 300 FED(22:14NSQ1(J1)).60.00 60 T0 300 FED(22:14NSQ2(J2)) FED(22:14.NSQ2</pre>	J2+1 FLD(5,17,NSQ2(J2)) FLD(22,14,NSQ2(J2)) D1D1WM(T(LTA),A(LA),XK(LK),G2 0 2998 XK(LK)+XK(LA) 2017 D1D1WM(T(1),A(LA),XK(LK),G1)	J2+1 FLD(5,17,NSQ2(J2)) FLD(22,14,NSQ2(J2)) XK(LLX)*XK(LA) 0 2998 (T(1)+T(LTA))/2,0 PLYAWM(A(LA),TM,A(LA+1),XK(LK 0 2099 PLYAWM(A(LA),T(1),A(LA+1),XK(PLYAWM(A(LA),T(1),A(LA+1),XK(J2+1 FLD(5,17,NSQ2(J2)) FLD(22,14,NSQ2(J2)) PLYAKM(A(LA),T(LTA),A(LA+1),X 2992 XK(LK)+XK(LA) D 2042 PLYAWM(A(LA),T(I),A(LA+1),XK(FLD(5,17,N502(J2)) FLD(22,14,N502(J2)) XK(LK)*XK(LA) 0 2998 0 2998 0 211,M(TM,CON(14),A(LA),XK(LK) 0 2999 T(LTA) 0 2007 T(LTA) 0 2017 1 (LTA) 0 2017 1 (LTA) 1 (LT
CNOUIK	CONTINUE END 1 = J1+1 26 = FLD(5,16,N501(J1)) 17 = CD(2,14,N501(J1)) 10 CLUDE VARG-LIST 10 CLUDE VARG-LIST 10 CLUDE VARG-LIST 10 CLUDE VARG-LIST 10 COD CONDUCTOR 11 F(FLD(2,14,NS02(J2)) 21 F(FLD(2,14,NS02(J2))) 21 F(FLD(2,14,NS02(J2))) 22 F(LD(2,14,NS02(J2))) 21 F(FLD(2,14,NS02(J2))) 21 F(FLD(2,14,NS02(J2))) 21 F(FLD(2,14,NS02(J2))) 21 F(FLD(2,14,NS02(J2))) 21 F(FLD(2,14,NS02(J2))) 22 F(FLD(2,14,NS02(J2))) 21 F(FLD(2,14,NS02(J2))) 21 F(FLD(2,14,NS02(J2))) 22 F(FLD(2,14,NS02(J2))) 22 F(FLD(2,14,NS02(J2))) 23 F(FLD(2,14,NS02(J2))) 24 F(FLD(2,14,NS02(J2))) 25 F(FLD(2,14,NS02(J2))) 26 F(FLD(2,14,NS02(J2))) 27 F(FLD(2,14,NS02(J2))) 28 F(FLD(2,14,NS02(J2))) 20 F(FLD(2,14,NS02(J2))) 20 F(FLD(2,14,NS02(J2))) 20 F(FLD(2,14,NS02(J2))) 21 F(FLD(2,14,NS02(J2))) 22 F(FLD(2,14,NS02(J2))) 23 F(FLD(2,14,NS02(J2))) 24 F(FLD(2,14,NS02(J2))) 25 F(FLD(2,14,NS02(J2))) 26 F(FLD(2,14,NS02(J2))) 27 F(FLD(2,14,NS02(J2))) 28 F(FLD(2,14,NS02(J2))) 29 F(FLD(2,14,NS02(J2))) 20 F(FLD(2,14,NS02(J2))) 20 F(FLD(2,14,NS02(J2))) 21 F(FLD(2,14,NS02(J2))) 21 F(FLD(2,14,NS02(J2))) 22 F	JZ = JZ+1 A = FLD(5,17,NSQ2(J2)) K = FLD(22,14,NSQ2(J2)) K = FLD(22,14,NSQ2(J2)) C = D101204 S = TK(LK) *XK(LA),A(LA),XK(LK),62 S = XK(LK) *XK(LA) S = ZK(LK) *XK(LA) S = ZK(LK) *XK(LA) S = ZK(LK) *Z S = ZK(JZ = JZ+1 A = FLD(5:17.NSQ2(J2)) SZ = FLD(22.14.NSQ2(J2)) SZ = XK(LK)+XK(LA) SO TO 2998 FM = (T(1)+T(LTA))/2.0 SALL PLYAWM(A(LA),TM,A(LA+1),XK(LK SALL PLYAWM(A(LA),T(1),A(LA+1),XK(M = T(1) SO TO 2999 M = T(1) SO TO 2999 M = T(1) SO TO 2993 M = T(1) SO TO 2994 M = T(1) SO TO 2004 M = T(1) M = T(1)	J2 = J2+1 A = FLD(5,17,NSQ2(J2)) LA = FLD(22,14,NSQ2(J2)) CALL PLYAWM(A(LA),T(LTA),A(LA+1),X CO 2998 1 = XK(LK)+XK(LA) 21 = XK(LK)+XK(LA),T(I),A(LA+1),XK(A) 20 TO 2042 21 L PLYAWM(A(LA),T(I),A(LA+1),XK(A)) 21 L PLYAWM(A(LA),T(I),A)) 21 L PLYAWM(A(LA),XK(LA)) 21 L PLYAWM(A) 21 L PLYAWM(A)	<pre>A = FLD(5,17,1962(J2)) X = FLD(22,14,1862(J2)) X = FLD(22,14,1862(J2)) 22 = XK(LK) *XK(LA) 20 T 299 A = T(1)+T(LTA))/2.0 A = T(LTA) 20 T 299 A = T(LTA) 20 T 299 A = T(LTA) 20 T 2017 A = T(LTA) 31(L6) = 1./(1./61+1./62) </pre>
CNOUIK	00 CONTINUE END 50 J1 = J1+1 LG = FLD(5.16,NS01(J1)) LG = FLD(2.14,NS01(J1)) IFF(LG.60.0) GO TO 65 LT = FLD(22.14,NS01(J1)) INCLUDE VARG/LIST CHECK FOR RADIATION COMDUCTOR IFF(LD12.1,NS021(J2)) LA = FLD(0.5,NS02(J2)) LA = FLD(0.5,NS02(J2)) LK = FLD(0.5,NS02(J2)) LK = FLD(5.17,NS02(J2)) LK = FLD(5.17,NS02(J2)) LK = FLD(5.17,NS02(J2)) CO 0.0005,2010,2015,2020,2025,2030,2 CO 0.0005,2005,2010,2015,2020,2025,2030,2 CO 0.0005,2005,2010,2010,2015,2020,2025,2030,2 CO 0.0005,2005,2010,2010,2010,2010,2010,2025,2030,2025,2020,2025,2020,2025,2020,2025,2020,2025,2020,2025,2020,2025,2020,2025,	11 J2 = J2+1 LA = FLD(5,17,NSQ2(J2)) LK = FLD(22,14,NSQ2(J2)) CALL DIDIWM(T(LTA),A(LA),XK(LK),62 60 T0 2998 20 G1 = XK(LK)*XK(LA) 60 T0 2017 21 CALL DIDIWM(T(1),A(LA),XK(LK),61) 22 CALL DIDIWM(T(1),A(LA),XK(LK),61)	J2 = J2+1 LA = FLD(5;17,NSQ2(J2)) LA = FLD(22;14,NSQ2(J2)) G2 = XK(LK) *XK(LA) G0 T0 2998 130 TM = (T(1)+T(LTA))/2.0 132 CALL PLYAWM(A(LA),TM,A(LA+1),XK(LK G0 T0 2999 135 TM = T(1) 55 TM = T(1) 40 CALL PLYAWM(A(LA),T(1),A(LA+1),XK(KK) 40 CALL PLYAWM(A(LA),T(1),A(LA+1),XK(KK) 40 CALL PLYAWM(A(LA),T(1),A(LA+1),XK(KK) 40 CALL PLYAWM(A(LA),T(1),A(LA+1),XK(KK) 40 CALL PLYAWM(A(LA),T(1),A(LA+1),XK(KK) 40 CALL PLYAWM(A(LA),T(1),A(LA+1),XK(KK)) 40 CALL PLYAWM(A(LA),T(1),A(LA+1),XK(KK)) 40 CALL PLYAWM(A(LA),T(1),A(LA+1),XK(KK)) 40 CALL PLYAWM(A(LA),T(1),A(LA+1),XK)) 40 CALL PLYAWM(A(LA),T(1),XK)) 40 CALL PLYAWM(A(A(LA),T(1),XK)) 40 CALL PLYAWM(A(A(LA),T(1),XK)) 40 CALL PLYAWM(A(A(LA),T(1),XK)) 40 CALL PLYAWM(A(A(A),XK)) 40 CALL PLYAWM(A(A(A),XK))) 40 CALL PLYAWM(A(A),XK)) 40 CALL PLYAWM(A(A),XK)) 40 CALL PLYAWM(A(A),XK)) 40 CALL PLYAWM(A(A),XK)) 40 CALL PLYAWM(A(A),XK)) 40 CALL PLYAWM(A(A),XK)) 40 CALL PLYAWM(A) 40 CALL PLYAWM(A),XK) 40 CAL	<pre>I42 J2 = J2+1 LA = FLD(5,17,NSQ2(J2)) LA = FLD(22,14,NSQ2(J2)) CALL PLYAWM(A(LA),T(LTA),A(LA+1),X CALL PLYAWM(A(LA),T(LTA),A(LA+1),X G0 T0 2998 U45 G1 = XK(LK)+XK(LA) G0 T0 2042 50 CALL PLYAWM(A(LA),T(I),A(LA+1),XK(S0 CALL PLYAWM(A(LA),T(I),A(LA+1),XK(S)) </pre>	LA = FLD(5,17,N502(J2)) LK = FLD(22,14,NS02(J2)) G2 = XK(LK)*XK(LA) G2 TO 2998 CAL D2P1WM(TM,CON(14),A(LA),XK(LK) G0 TO 2999 G0 TO 2995 G0 TO 2995 G0 TO 2002 G0 TO 2
CNOUIK	5000 CONTINUE END 50 J1 = J1+1 LG = FLD(5,16,NSO1(J1)) 1F(LG.60,0) 60 70 65 LT = FLD(22/14,NSO1(J1)) INCLUDE VARG/LIST CHECK FOR RADIATION CONDUCTOR INCLUDE VARG/LIST CHECK FOR RADIATION CONDUCTOR INCLUDE VARG/LIST INCLUDE VARG/LI	2017 J2 = J2+1 LA = FLD(5,17,NSO2(J2)) LK = FLD(22,14,NSO2(J2)) CALL DIDIWM(T(LTA),A(LA),XK(LK),62 60 T0 2998 2020 61 = XK(LK),XX(LA) 2025 CALL DIDIWM(T(1),A(LA),XK(LK),61) 2025 CALL DIDIWM(T(1),A(LA),XK(LK),61)	J2 = J2+1 LA = FLD(5,17,NSQ2(J2)) LA = FLD(2,14,NSQ2(J2)) C2 = XK(LX)*XK(LA) G0 T0 2998 2030 TM = (T(1)+T(LTA))/2.0 2032 CALL PLYAWM(A(LA),TM,A(LA+1),XK(LK 2035 TM = T(1) 2035 TM = T(1) 2040 CALL PLYAWM(A(LA),T(1),A(LA+1),XK(2042 J2 = J2+1 LA = FLD(5,17,NSQ2(J2)) LA = FLD(22,14,NSQ2(J2)) LL = FLD(22,14,NSQ2(J2)) CALL PLYAWM(A(LA),T(LTA),A(LA+1),X 60 T0 2998 60 T0 2042 60 T0 2042 60 C 2042 2050 CALL PLYAWM(A(LA),T(I),A(LA+1),XK(LA = FLD(2,17,N502(J2)) LK = FLD(22,14,N502(J2)) G2 = XK(LK)*XK(LA) G2 = XK(LK)*XK(LA) G0 T0 2998 CALL D2D1WM(TM,CON(14),A(LA),XK(LK) G0 T0 2999 CALL D2D1WM(TM,CON(14),A(LA),XK(LK) G0 T0 2999 CALL D2D1WM(TM,CON(14),A(LA),XK(LK) CALL D2D1WM(TM,CON(14),A(LA),XK(LK) CALL D2D1WM(TM,CON(14),A(LA),XK(LK) CALL D2D1WM(TM,CON(14),A(LA),XK(LK) CALL D2D1WM(TM,CON(14),A(LA),XK(LK) CALL D2D1WM(TM,CON(14),A(LA),XK(LK) CALL D2D1WM(TM,CON(14),A(LA),XK(LK) CALL D2D1WM(TM,CON(14),A(LA),XK(LK) CALL D2D1W(TM,CON(14),A(LA),XK(LK) CALL D2D1W(TM,CON(14),XK(LK)) CALL D2D1W(TM,CON(14)
CNOUIK	<pre>9. 5000 CONTINUE 9. 5000 CONTINUE 1. LG = FLD(5.16.N501(J1)) 1. LG = FLD(5.16.N501(J1)) 1. LTA = FLD(22.14.N501(J1)) 1. NCLUDE VARGILIST 2. LTA = FLD(22.14.N501(J1)).60, TO 300 NTYPE = FLD(0.5.N502(J2)) 1. LA = FLD(5.17.NS02(J2)) 1. LA = FLD(5.17.NS02(J2)) 1. LA = FLD(5.17.NS02(J2)) 2. Solo(2005).2010.2015.2030.2 5. 2000 TM = (T(1)+T(LTA))/2.0 2007 CALL DIDIWM(TM.A(LA).XK(LK).6(LG)) 5. 2017 CALL DIDIWM(TM.A(LA).XK(LK).6(LG)) 5. 2017 CALL DIDIWM(TM.A(LA).XK(LK).6(LG)) 5. 2017 CALL DIDIWM(TM.A(LA).XK(LK).6(LG)) 5. 2017 CALL DIDIWM(TM.A(LA).XK(LK).6(LG)) 5. 2010 TM = T(1) 5. 2010 TM = T(1) 5. 2010 TM = T(1) 5. 2010 TM = T(1) 5. 2010 COTO 2007 5. 2010 TM = T(1) 5. 2010 TM = T(1) 5.</pre>	5* 2017 J2 = J2+1 5* LA = FLD(5,17,NSQ2(J2)) 5* LK = FLD(22,14,NSQ2(J2)) 5* CALL DIDIWM(T(LTA),A(LA),XK(LK),62 5* 60 T0 2998 5* 2020 61 = XK(LK)*XK(LA) 5* 2020 61 = XK(LK)*XK(LA) 5* 2025 CALL DIDIWM(T(T))A(LA),XK(LK).61) 5* 2025 CALL DIDIWM(T(T),A(LA),XK(LK).61)	5* J2 = J2+1 5* LA = FLD(5,17,NSQ2(J2)) 5* LA = FLD(2,14,NSQ2(J2)) 62 = XK(LX)*XK(LA) 5* 60 T0 2998 5* 2032 TM = (T(1)+T(LTA))/2.0 5* 2032 TM = (T(1)+T(LTA))/2.0 60 T0 2999 5* 2035 TM = T(1) 5* 2035 TM = T(1) 5* 2035 TM = T(1) 5* 2040 CALL PLYAWM(A(LA),T(1),A(LA+1),XK(LA))/2.0 5* 2040 CALL PLYAWM(A(LA),T(1),A(LA+1),XK(LA))/2.0 5* 2040 CALL PLYAWM(A(LA),T(1),A(LA+1))/2.0 5* 2040 CALL PLYAWM(A(LA),T(1))/2.0 5* 204	5. 2042 J2 = J2+1 5. LA = FLD(5,17,NSQ2(J2)) 5. LA = FLD(22,14,NSQ2(J2)) 5. CALL PLYAWM(A(LA),T(LTA),A(LA+1),X 5. GO TO 2998 5. 2050 G1 = XK(LK) *XK(LA) 5. 2050 CALL PLYAWM(A(LA),T(I),A(LA+1),XK(5. 2050 CALL PLYAWM(A(LA),T(I),A(LA+1),XK()) 5. 2050 CALL PLYAWM(A(LA),T(I),A(LA+1),XK()) 5. 2050 CALL PLYAWM(A(LA),T(I),A(LA+1),XK()) 5. 2050 CALL PLYAWM(A(LA),T(I),A(LA+1),XK()) 5. 2050 CALL PLYAWM(A(A(LA),T(I),A(LA+1),XK())) 5. 2050 CALL PLYAWM(A(A(LA),T(I),A(LA+1),XK())) 5. 2050 CALL PLYAWM(A(LA),T(I),A(LA+1),XK())) 5. 2050 CALL PLYAWM(A(A(LA),T(I),A(LA+1),XK())) 5. 2050 CALL PLYAWM(A(A(LA),T(I),A(LA+1),XK()))) 5. 2050 CALL PLYAWM(A(A(LA),T(I),A(LA+1),XK()))) 5. 2050 CALL PLYAWM(A(A(A(A),A(A+1),XK()))))) 5. 2050 CALL PLYAWM(A(A(A),A(A+1),XK()))))))))))))))))))))))))))))))))))	<pre>5. LA = FLD(5,17,NS02(J2)) 5. LK = FLD(22,14,NS02(J2)) 62 = XK(LK) *XK(LA) 5. G0 T0 2998 5. 2055 TM = T(1)+T(LTA))/2.0 6. T0 2999 5. 2060 TM = T(LTA) 5. 2060 TM = T(LTA) 5. 2060 TM = T(LTA) 5. 2065 TM = T(LTA) 5. 2065 TM = T(LTA) 5. 2096 G(LG) = 1./(1./G1+1./G2) 5. 2996 G(LG) = 1./(1./G1+1./G2)</pre>
CNOUIK	79. 5000 CONTINUE 79. 5000 CONTINUE 80. 50 J1 = J1+1 81. E.G = FLD(5.16,NSO1(J1)) 82. LTA = FLD(22.14,NSO1(J1)) 84. INCLUDE VARG/LIST 85. LTA = FLD(22.14,NSO1(J1)) 85. CHECK FOR RADIATION COMDUCTOR 85. LA = FLD(0.5,NSO2(J2)) 85. LA = FLD(0.5,NSO2(J2)) 85. LA = FLD(0.5,NSO2(J2)) 85. LA = FLD(0.5,17,NSO2(J2)) 85. COTO(2005,2010,2015,2020,2025,2030,2 85. 2005 TM = (T(1)+T(LTA))/2.0 85. 2005 TM = (T(1)+T(LTA))/2.0 85. 2005 TM = (T(1)+T(LTA))/2.0 85. 2005 CALL DIDIWM(TM.A(LA),XK(LK),6(LG)) 85. 2015 CALL DIDIWM(TM.A(LA),XK(LK),611) 85. 2015 CALL DIDIWM(T(1),A(LA),XK(LK),611) 85. 2015 CALL DIDIW(T(1),A(LA),XK(LK),611) 85. 2015 CALL DIDIW(T(1),A(LA),XK(LK),611) 85. 2015 CALL DIDIW(T(1),A(LA),XK(LK),611) 85. 2015 CALL DIDIW(T(1),A(LA),XX(LK),611) 85. 2015 CALL DIDIW(T(1),A(LA),XX(LK),611) 85. 2015 CALL DIDIW(T(1),A(LA),XX(LK),A(LA),XX(LK),A(LA),XX(L	85* 2017 J2 = J2+1 85* LA = FLD(5:17,NSQ2(J2)) 85* LK = FLD(22:14,NSQ2(J2)) 85* Call DIDIWM(T(LTA).A(LA).XK(LK).62 85* 2020 61 = XK(LK)*XK(LA) 85* 2020 61 = XK(LK)*XK(LA) 85* 2025 CALL DIDIWM(T(T).A(LA).XK(LK).61) 85* 2025 CALL DIDIWM(T(T).A(LA).XK(LK).61)	85* J2 = J241 85* LA = FLD(5)17,NSQ2(J2)) 85* LA = FLD(22,14,NSQ2(J2)) 85* 62 = XK(LK) *XK(LA) 85* 60 TO 2998 85* 2030 TM = (T(1)+T(LTA))/2,0 85* 2030 TM = (T(1)+T(LTA))/2,0 85* 2030 TM = T(1) 85* 2030 TM = T(1) 85* 2035 TM = T(1) 85* 2040 CALL PLYAWM(A(LA),T(1),A(LA+1),XK(LA	B5* 2042 J2 = J2+1 B5* LA = FLD(5,17,NSQ2(J2)) B5* LA = FLD(22,14,NSQ2(J2)) B5* CALL PLYAWM(A(LA),T(LTA),A(LA+1),X B5* 2045 G1 = XK(LK) *XK(LA) B5* 2045 G1 = XK(LK) *XK(LA) B5* 2050 CALL PLYAWM(A(LA),T(I),A(LA+1),XK(LA)) B5* 2050 CALL PLYAWM(A(A(LA),T(I),A(LA+1),XK(LA))) B5* 2050 CALL PLYAWM(A(A(A),XK(LA))) B5* 2050 CALL PLYAWM(A(A(A),XK(LA))) B5* 2050 CALL PLYAWM(A(A(A),XK(LA))) B5* 2050 CALL PLYAWM(A(A(A),XK(LA))) B5* 2050 CALL PLYAWM(A(A),XK(LA)) B5* 2050 CALL PLYAWM(A(A),XK(LA)) B5* 2050 CALL PLYAWM(A(A),XK(A)) B5* 2050 CALL PLYAWM(A(A),XK(A)) B5* 2050 CALL PLYAWM(A) B5* 2050	$\begin{array}{cccccccccccccccccccccccccccccccccccc$
CNOUIK	361 79* 5000 Contrinue 365 79* 5000 Contrinue 365 80* 50 J1 = J1+1 364 81* LG = FLD(5.16.MS01(J1)) 365 82* LTF(L6.60.0) 60 TO 65 365 82* LTF(L6.22.14.MS01(J1)) 370 85* LTF(L0E VARG/LST 371 85* LTF(L0E VARG/LST 377 85* LTF(L0E) 371 85* LTF(L0E) 377 85* CHECK FON RADIATION COMDUCTON 371 85* LA = FLD(5.17.MS02(J2)) 375 85* COTO(2005.2010.2025.2030.2 377 85* COTO(2005.2010.2015.2020.2025.2030.2 375 85* COTO(2005.2010.2015.2020.2025.2030.2 375 85* 2005.2010.2015.2020.2025.2030.2 376 85* 2005.2010.2015.2010.2025.2030.2 377 85* 2005.2010.2015.2020.2025.2030.2 377 85* 2005.2010.2010.2010.2025.2030.2 377 85* 2005.2010.2010.2010.2025.2030.2 377 85* 2005.7010.201	0.05 85* 2017 J2 J2+1 0406 85* LA FLD(5,17,NSQ2(J2)) 0407 85* LK FLD(22,14,NSQ2(J2)) 0407 85* CALL DIDIWM(T(LTA).A(LA).XK(LK).62 0410 85* CALL DIDIWM(T(LTA).A(LA).XK(LK).62 0411 85* 2020 61 XK(LK).42 0412 85* 2020 61 XK(LK).62 0413 85* 2020 61 XK(LK).62	H15 85* J2 J2+1 H415 85* La FLD(5;17,NSQ2(J2)) H417 85* La FLD(5;17,NSQ2(J2)) H417 85* CZ XK(LK)*XK(LA) H421 85* G0 T0 H421 85* 60 T0 H421 85* 2030 TM (T(1)+T(LTA))/2.0 H422 85* 2032 CALL PLYAWM(A(LA))/2.0 H428 85* 2035 TM (T(1)+T(LTA))/2.0 H429 85* 2035 TM (T(1), A(LA))/2.0 H426 85* 2035 TM T(1) H427 85* 2032 TM T(1) H427 85* 2040 CALL PLYAWM(A(LA)).T(1).A(LA+1).XK(HA	0430 85* 2042 J2 = J2+1 1431 85* 2042 J2 = FLD(5,17,NSQ2(J2)) 1432 85* LA = FLD(2,14,NSQ2(J2)) 1433 85* CALL PLYAWM(A(LA),T(LTA),A(LA+1),X 1434 85* 60 T0 2998 1435 85* 2045 61 = XK(LK)+XK(LA) 1435 85* 2050 CALL PLYAWM(A(LA),T(I),A(LA+1),XK(1437 85* 2050 CALL PLYAWM(A(LA),T(I),A(LA+1),XK(1450 70 70 70 70 70 70 70 70 70 70 70 70 70	$1+11$ 05 LA $E \ FLD(52, 17, N502(J2))$ $1+42$ 85 LK $E \ FLD(22, 14, NS02(J2))$ $1+44$ 85 62 $x \ K(LK) + x \ K(LA)$ $0+45$ 85 60 0 $0+45$ 85 2055 Tm $1+47$ 85 2055 Tm $1+47$ 85 2055 Tm $1+47$ 85 2050 Tm $1+47$ 85 2060 $1+51$ 85 2060 $1+51$ 85 2065 $1+52$ 85 2065 $1+52$ 85 2065 $1+52$ 85 2065 $1+54$ 85 2065 $1+54$ 85 2065 $1+54$ 85 2065 $1+54$ 85 2065 $1+54$ 85 2065 $1+54$ 85 2065 $1+54$ 85 2065 $1+54$ 85 2096 $6(L6)$ $1./(1./61+1./62)$

A - 60

CNOUIK	IF(FLD(3,1,NSQ1(Jl)).EQ.1) G(LG) = G1*G2 J2 = J2+1 Continue	FFFED(3,1,NSG1(J1)).E0.0) 60 TO 55 T1 = T(1)+460.0	T2 = T(LTA)+460.0 GV = G(LG)*(T1*T1+T2*T2)*(T1+T2)	60 T0 60 6v = 6(16)	OBTAIN THE O PLUS SUMMATION 6*TA TERM	0(I) = 0(I)+6V+1(LTA) Save summation of conductors	X(LE) = X(LE)+6V CHECK FOR AD-ADINING DIFEUSION NODE OR ONE⇒₩&Y COMPLICIOR	IF (LTA.GT.HND.OR.FLD(211) INSO1(J1)) .EQ. 10 65	SAVE SUMMATION OF CONDUCTORS FOR AUJUINING NOUL	X(LEA) = X(LEA)+GV 0(LTA) = 0(LTA)+GV*T(I)	CHECK FOR LAST CONDUCTOR	CONTINUE	DTI = ISTEP/(TSTEP+ISTEPO) DT2 = TSTEP0/(TSTEP+TSTEPO)	OBTAIN NEW DIFFUSION TEMPERATURES, DTMPCC AND CSGMIN.		LEH = IEH+I Calculate C/Sk Minimum	T1 = C(1)/X(LE) TE(T1 GE CVW) 60 TO 90		KON(35) ± I Compute new temperatures using calculated source terms	T1=(DT1*X(LEH)*(C(1)/TSTEP-X(LE))+0(1))/(C(1)*(1DT2)/TSTEP+DT2*	T2 = 1.0/EXP(TSTEP+X(LE)/C(1))	TI = (!I+(I.0-T2)*0(!)/X(LE)+T2*1(I)P0.5 CALCULATE THE ABSOLUTE VALUE TEMPERATURE CHANGE	T2 = AUS(T(I)-TI) Save the largest temperature change	IF(TCGM.6E.T2) 60 T0 95 TCGM = T2	KON(36) = I Store The TEWDEBATHRES		CONTINUE	CON(17) = CKM Delta = CKM+COV(4)	IF(CKM.LE.0.0) 60 T0 996 Check For First Pass	IF(PASS.6T.0.0) 60 10 115 UNDO THE TEMPERATURE CALCULATIONS		IF (PASS.61.0.0) 60 TO 15	
	2999			U U	3	60					45	e e								06	F					95	100					110	
					<u>U</u>	ုပ	¢	، ز	U		ų		``	U.		.U	, .		U	•		ų	U		ι,	ر			ų	U.			
	80 80 80 80 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	854	88 89 89	*06	÷26	*06 *16	95 *	*26	*86 *86	101*	102+	104	1.05+	107+	100*	****	112+		115+	117*	+611	121+	122+	124+	126+	128*	120*	131*	133+ 134+	135*	1.18	140+1	
	00455 00457 00460	00464	00465	00467	00410	00471	00472	00473	00475	00477	00477	00502	00204	00505	00511	00512	00513	00516	00517	002500	00521	00522	00523	00526	00527	00530	00531	00534	00536	00540	21500 21500	00547	

• ...

.

i

ę

•

.

· •																																			•	*NEW	NBN*		01##									4 - mmi - i	*NEW								
CNOUIK	PASS = 1.0	CON(1) = TPRINT						C IS THE TIME STEP USED LESS THAN THE TIME STEP CALCULATED	115 FUSTER FLEDERAD GO TO 130	r compute the time Step			120 TSTEP = 0.05+ISTEP+CON(A)/TCGM	4 20 1 0 1 0 0 1 1 1 1 1 1 1 1 1 1 1 1 1		COLUMNOT TE THE TEMPERATURE WAS TOO LANGE	- C DEFT I TELETERTENTON CONTOUR ### 100 FX/C	130 IF (TOMIC) 00 10 IZU	C SUCHE THE FAST THE STATE OF THE CONTRACT CALL CARAGE		C CHECK IC SEE IT HERE ARE ANT ANTIMMETIC NOUES	I (NUA:LE:0) 60 10 10 10 10 10 10 10 10 10 10 10 10 10	C COMPUTE ANTIHMETIC LEMPERATURES BT SUCCESSIVE POINT OVER-RELAX												TELECT TURNET TATE TO TO FOR TO FOR		LA = FLD(5,17,1502(JJ2))	LK = FLD(22,14,NSG2(J2))	60 T0 (5005,5010,5015,5020,5025,5030,5035,5040,5030), NTYPE	× 2002 G(L) = XK(LK)+O(L)	60 TO 5999		5012 CALL DIDIWM(T(L),A(LA),XK(LK),02)		5012 GL DIDIW/COH(14).4(LA).42(LK).62)		5020 CALL 01014W (CON(14) • A(LA) • XK(LK) • 01)	5022 JJ2 = JJ2+1	LA = FLD(5,17,1)SG2(JU2))	<pre>r LK = FLD(22,14,NSq2(JJ2))</pre>	60 TO 5017	• 5025 Q1 = XK(LX) *XK(LA)		0000 CALL 1101MMCC0111141441CA1C414444			
	0553 1424	10554 1434	100000 T441		10569 147	10000 10000 1001 10000		10565 1504	10566 1514	152	10570 1534	10571 1544			101 +1010 1024 +2010		10010 F034		101 9/000	10200 1021	00600 1634	10001 164	00601 1654	10603 1664	00604 167	10605 1681		0/1 1100	111 21900	271 CTON	10010 1761	10/1 N2000	10/1 17000	10620 178-	10625 178-	10627 1784	10630 1784	0631 1784	10632 1784	0653 1784	00634 1784	0635 1784	0636 178	10071 100700 1207 1207	170	10642 1784	10043 1784	10644 1784	10645 1784)0046 178	0647 1784	00650 1784	00651 1784	19/1 20000	10653 1/dv	10024 1/8/1	
	5)	- *	- •			*		-					 - •				- 1	- '								- `			-	- 4	- •		-		. ч			5	-		~	· ب			- - -					÷	÷				-	

.-

жзэ ter in the second s

. •••

A - 62

	21++ +++	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2		2330 2001 227 227	432 8	9 # # U # # #	2 2 0 UUU 8 7 4 7 4
CNGUIK	LK = FLD(22*14*NS02(JJ2)) 02 = XK(LK)*XK(LA) 60 T0 5998 035 CALL DIDIWM(COM(1+)*A(LA)*XK(LK)*G1)	037 JJZ = JJZ+1 LA = FLD(5,117,NSQ2(JJZ)) LK = FLD(22,14,NSQ2(JJZ)) G0 T0 5012 040 01 = XK(LK)•XK(LA)	60 T0 5037 990 0(L) = 01+02+0(L) 999 JJZ = JJ2+1 000 CONTINUE 1JS JJI = JJ1+1 1JS L0 = FLD(5:16,NS01(JJ1)) LTA = FLD(22,14,NS01(JJ1)) LTA = FLD(22,14,NS01(JJ1))	THECK FOR RADIATION CONDUCTOR TF(FLD(2*1,NSQ1(JJ1)).EQ.0) GO TO 4000 MTYPE = FLD(0;5,NSQ2(JJ2)) LA = FLD(0;5,NSQ2(JJ2)) LA = FLD(2;17,NSQ2(JJ2)) LA = FLD(2;14,NSQ2(JJ2)) COTO(3005;3010;3015;3015)3025,3030,3035,3049,3045,3050,3055, 005 TM = (7(1)+T(LTA))/2.0 005 TM = (7(1)+T(LTA))/2.0 007 CALL DIDIWM(TM.A(LA),XX(LK),6(LG)) 007 CALL DIDIWM(TM.A(LA),XX(LK),6(LG)) 010 TM = T(L) 010 TM = T(L)	015 Call DJUW(T(L),A(LA),XK(LK),61) 017 JJ2 = JJ2+1 LA = FLD(5,17,13G2(JJ2)) LK = FLD(22,14,NSG2(JJ2)) LK = FLD(22,14,NSG2(JJ2)) CALL DJD1WM(T(LTA),A(LA),XK(LK),62) 60 TO 3998 020 61 = XK(LK)*XK(LA) 025 CALL DJD1WM(T(L),A(LA),XK(LK),61)	JJZ = JJZ+1 LA = FLD(22:14,NSQ2(JJ2)) LK = FLD(22:14,NSQ2(JJ2)) G2 = XK(LK)•XK(LA) G3 TM = [1(L)+T(LTA))/2.0 G3 Z CALL PLYAWH(A(LA))·T%.A(LA+1)·XK(LK).G(LG)) G3 TM = T(L) 035 TM = T(L) 040 CAL PLYAWM(A(LA))·T%.A(LA+1)·XK(LK).G1)	042 JJZ = JJZ+1 LA = FLD(5,17,NSQ2(JJZ)) LK = FLD(2,14,NSQ2(JJZ)) CALL PLYAWM(A(LA),T(LTA),A(LA+1),XK(LK),G2) CALL PLYAWM(A(LA),T(LTA),A(LA+1),XK(LK),G2) 045 G1 = XK(LK) +XK(LA) 05 G0 T0 3094 05 CALL PLYAWM(A(LA),T(L),A(LA+1),XK(LK),G1) JJZ = JJZ+1
	78* 78* 78*	738	22222222222222222222222222222222222222	U * * * * * * * * * * * * * * * * * * *	1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2		+ + + + + + + + + + + + + + + + + + +
	00655 00655 00657	000662 1 000662 1 000662 1 000662 1 000662 1 000662 1 000662 1 000662 1 000662 1 000662 1 000662 1 000662 1 000	00666 00667 00667 00672 00673 100673 100673 100673 100673 100673 100673 100673 100673 100673 100673 100673 100673 1006673 1006666		00714 1 00714 1 00715 1 00721 1 00722 1 00723 1	00725 1 00726 1 00727 1 00731 1 00732 1 00732 1 00735 1 00755 1 00756 1 007755 1 007555 1 0007555 1 0007555 1 0007555 1 0007555 1 0007555 1 0007555 1 0007555 1 0007555 1 0000000000000000000000000000000000	00000000000000000000000000000000000000

.

		• · · · · ·	an a		
				an a	
-					
)					
5	3 3 N		, 8 T		
)				ы 0	
2	2	Č,		HAN CHAN	
	ורפו	1*6	FER	IANGE	
	9 (Y		DIF	ATI0 T T T T T T T T T T T T T T T T T T T	
	5	(16)	0 TO	S ME	
	A) •)	6	0 60 11+1 35 21/5 ERA1	C RE WAS	125
	5)) • V(L	() () ()	E0.0 2)*(2)*(10 1 1EMP	HETI FO FEM FEM FEM FEM	105 105
	2.0		2.11. 1.11.1	RITIC RITIC	AT1 60 110 10
	S02 (NS02 LA) CON(61+1	(Ju) 111+1 11+1 11+1 11+1 11+1 11+1 11+1 1	M AR 60 1 19)) 19)) 19)) 19) 0 19) 0 19 19) 0 19 19 19 19 19 19 19 19 19 19 19 19 19	A5 5 P Sw EP C
	TTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTT	(1.) S61	VS1000 VS100	TI) TI) TI) TI) TI) TI) TI) TI) TI) TI)	CCKU CCKU CCKU CCKU CCKU
λIK	C (5. 1 (5. 1 (1) 1 (1)	14) 11/1			
CHOL	P C C C C C C C C C C C C C C C C C C C		S S S S S S S S S S S S S S S S S S S		
1977 1	AX80540	NONCELLON NONCELLON NONCELLON		MTMAR TTAAMAN	
	22000000000000000000000000000000000000			39 <u>2 2</u>	
	20	0 0 5 60 0 0 0 0 0 0			
	******	*******	**********	- # # # # # # # # # # # # # # # # # # #	* * * * * * * * * * * * * *
				<u> </u>	
	755	765		000000000000000000000000000000000000000	400000000000000000000000000000000000000
)	00000000	000000000000000000000000000000000000000			

-

A - 64

..

.

BBS FORVAT(46H CNONIK REQUIRES SHORT PSEUDO-COMPUTE SEQUENCE) BBS FORMAT(24H CS6WIN ZERO OR NEGATIVE) BB7 FORMAT(20H TIME STEP TOO SMALL) BB8 FORMAT(18+20H LOCATIONS AVAILABLE) BB8 FORMAT(19H NO OUTPUT INTERVAL) 0 #DIAGNOSTIC* MESSAGE(S) C TRY TO EVEN THE OUTPUT INTERVALS COLL OUTCAL C TRY TO EVEN THE OUTPUT INTERVALS 190 TPHINT = TPRINT+TSUM 195 CALL OUTCAL 196 CALL OUTCAL 195 CALL OUTCAL 195 CALL OUTCAL 196 CALL OUTCAL TSTEPO = TSTEP D0 200 I = 1.NND LE = IEHI LEH = IEHHI 0 X(LEH) = X(LE) 15TEP = DELTA+0.95 CHECK FOR TIME TO PRINT IF(XON(7).EG.00118)) G0 TO 190 CHECK FOR PRINT EVERY ITERATION IF(XON(7).EG.0) G0 TO 10 END OF UNIVAC 1103 FORTRAN V COMPILATION. CNUUIK CODE RELOCATABLE 995 WRITE (6,885) 996 WRITE (6,885) 60 TO 1000 997 WRITE (6,887) 60 TO 1000 998 WRITE (6,887) 60 TO 1000 999 WRITE (6,889) 1000 60 TO 1000 CALL EXIT CALL EXIT CNOUIK END 200 U U U U 2334 261+ 262+ 265+ 260+ 2634 01115 01115 01115 01117 01117 01120 01122 01122

х.,

•
В.	COMPUTER	LISTINGS	OF	SINDA	IMPLICIT	SOLUTION	ROUTINES	
e Second Second Second Second Second	<i></i>			.4			Page	
	CNBACK	******	• • •		*******		B-2	
	CNFWBK		- • •				B-14	
	CNVARB	, 5 6 6 ,9-6 5 :	• • •	• • • • • •	• • • • • • • • • •		B-26	-

. .

. •

ar **•**

1 N 11

B - 1

-
-
~
~
~
- 22
~
-
~
_
~
ж.
1.
~
-
_
~
-
÷.
~

DIW FOR.* CINBACK.CNBACK Univac 1108 Fortran V Athena Version 131K-10D Created on 20 Aug 70 This compilation was done on 03 Jun 70 at 14:00:11

ENTRY POINT 004236 SUBROUTINE CNBACK ,

STORAGE USED (BLOCK, NAME, LENGTH)

14253	0125	10067	0000	0000													
2	5	ö	20	õ													
	•	~			000	0001	1001	000	000	000	000	000	000	1000	000	010	
	LEME	VAF			00	õ	00	00	ğ	00	00	000	000	0.0	00	000	
CODE	FCONST+1	*SIMPLE	LARRAYS	BLANK	LITLE	remp	CAP	SOURCE	COND :	5C1	202	CONST	ARRAY	-IXCON	(SPACE	DIMENS	
	-	×			-		3	23	0			Ŧ			<u></u>		
0001	0000	0002	0004	0005	0006	0001	0010	1100	0012	0013	0014	0015	0016	0017	0020	0.021	

EXTERNAL REFERENCES (BLOCK. NAME)

•	VARULI	DUTCAL	DIDIWM	PLYAWM	DZD1WM	TOPL IN	VARLEZ	EXIT	NERH25	SUUWN S	N1025	NER105	
	0022	0023	0024	0025	0026	0027	0030	0.031	0032	0033	0034	0035	

STORAGE ASSIGNMENT FOR VARIABLES (BLOCK, TYPE, RELATIVE LOCATION, NAME)

0017 7 000000 CON	0012 R 000000 G	0006 R 000000 H	0002 I 000007 J	0002 I 000024 J2	0002 I 000020 KI	0002 I 000026 LE1	0021 I 000006 LS01	WIGN 000000 I 0200	0002 1 000004 NMC	0020 I 000001 NTH	0002 R 000043 050M
0010 R 000000 C	0002 R 000014 DN	0002 R 000040 62	0002 I 000006 IE3	0002 I 000021 J1	0017 I 000000 KON	0002 I 000066 LE	0002 I 000033 LK	0021 I 000004 NCT	0021 I 000001 NWA	0014 I 000000 NSQ2	0311 R 000000 0
0002 R 000016 AN	0002 R 000015 DD	0002 R 000050 61	0002 I 000005 IE2	0002 I 000061 JJ2	0002 I 000063 KK2	0002 I 000013 LAX	0002 I 000045 LG	0021 I 000005 NAT	0002 I 000001 11N	0013 I 000500 NSQ1	NON2 R QUONOO PASS
0002 R 000017 AA	0002 R 000035 C2	0002 R 000053 GV	0002 I 000003 IE1	0002 1 000060 JJ1	0002 1 000057 KK1	0002 I 000032 LA	0002 1 000056 FE3	Daily I ODDAH I TA	0002 I 000002 NLA	0021 1 00002 NNT	0020 000002 HX
0016 R 000000 A	0002 R 000034 C1	0002 R 000044 65UM	0002 I 000025 I	LL 00001 1000		0002 I 003062 L		0021 1 000007 1 502		0021 1 000000 1 100	0002 I 000051 HIYPE

.

s.,

. . . ÷.,

. . .

والمعترين والم

.

Ś

- 1 . T

5.42

12.4

	RLXD TC6A	MOST	105	10121	10321	10676	1151	15	1/01	1951	2005L	2020L	20401	2060L	225L	250L	275L	30L	3015L	3032L	3750L	J999L	4010L	4022L	10101	5000L	5017L	5735L	5998L	6606	8.90F	8°5F	1066	1306		
	000023	110000	000162	124000	000625	003623	002172	000171	0.02402	003551	001335	001437	001540	001717	0.03641	003721	004023	000200	003107	003237	003355	003460	001042	001114	001226	001243	002604	002716	002755	002473	000000	000037	004124	004163		
	C002 R	0002 R	0001	1000	1000	1000	1000	0001	1000	0001	1000	1000	1000	1000	1000	1000	1000	1000	1000	1000	1000	1000	1000	1000	1001	1000	1000	1000	1001	1000	0000	0000	1000	1001		
				-				-	-	~	-	-	-	-	-	-	Ţ	-	-	-	-	-	-	7	-	-		-			-	-	-	-		
	RLXA	TSTEPN	××	TOTOL	TOCOL	1051	11446	1451	1656	1906	2000L	2017L	2035L	2055L	2206	2456	270L	2999L	3010L	3030L	3045L	35L	4005L	4020L	4037L	51	5015L	5030L	5636	6406	BOL	884F	889F	995L		
	000022	000012	000000	10000	000604	002121	003764	002276	002367	003546	000764	001403	001535	001671	003634	003666	004006	001755	003103	003231	003347	000221	001035	001100	001211	00150	002603	002660	002163	002407	002040	000032	000054	004155		
	0002 R	0002 8	0015 R	0001	1000	0001	1000	0001	0001	0001	0001	0001	0001	0.001	0001	1000	0.0.0	0001	0001	0001	1000	0001	1000	0001	0001	0001	0001	1000	1000	0001	0001	0000	0000	0001		
		-4																																		
	л С С	TPRIN	×	1005	1025	1045	11336	1251	160L	185L	20L	20156	2032L	2050L	2206	2400	265L	29981	3007	30256	30421	3065L	4000	40171	40351	16661	5012L	5025L	5476	6000L	751	883F	889F	1466 (1666 I	
	000027	000010	000003	000420	000560	000734	003743	002242	002354	002767	000174	001265	001512	001626	000260	003645	003754	001727	001063	003170	003310	003454	003511	001063	001175	001240	002563	002652	002127	002765	002024	000025	000050	004146	004202	
	0002 8	0002 8	0020 R	1000	1000	0001	1000	1000	1000	1000	1000	1000	1000	0001	1000	1000	0001	1000	0001	0001	0001	0001	0001	1000	0001	0001	0001	1000	1000	0001	1000	0000	0000	0001	1000	
													,																							
	002	ž	12	1000	1020	10401	11146	120L	155L	160L	19996	20101	20301	20456	2151	2376	2606	290L	3005L	3020L	3040L	306UL	40L	40151	4030L	4998L	5010L	5022L	-5040L	60L	705	. 882F	887F	593L	1 866	
ACK	000042	640000	000052	004212	000520	0.00671	003674	002231	002345	002450	000761	001362	001505	001620	003615	000314	000352	004102	003055	003162	003266	003450	000233	001062	001137	001234	002562	002635	002747	000343	001253	000017	0.0004	004140	004177	
BACK, CNB	0002 R	0002 R	0002 R	0001	1000	0001	0001	0001	0001	0001	0001	0001	0001	0001	0.0.0	0001	0001	1000	0001	0001	1000	0001	0001	1000	0001	0001	0001	0001	0001	0001	0001	0000	0000	1000	0001	
S								_																					•	•						
	10	1000	F	1001	1015	10356	1101	11636	150L	175L	1998L	2007L	20256	2042L	2005L	2326	255L	2056	3000	3017L	3035L	3055L	36661	4012L	4025L	451	5005L	5020L	5037L	29991	6656	BOLF	896F	991L	997L	
	140000	000064	00001	0.02074	000512	0.006663	002154	004027	002322	002446	000756	001342	001445	001561	001723	0.00310	003735	004052	001760	003126	003262	003421	003506	001043	001131	000237	002554	002621	002732	002762	002501	000000	000042	004132	141400	
	002 R	0002 R	1002 R	1000	1000	1001	1001	1000	1001	1001	10.01	1000	1001	1001	1000	1001	1001	1001	1001	1000	1001	1001	10.01	1001	1000	1000	1001	1001	0010	1000	1001	0000	0000	1000	1001	

٠

.

SUBROUTINE CNB/CK IMPLICIT BACKWARD DIFFERENCING EXECUTION SUBROUTINE THE LONG PSEUDO-COMPUTE SEQUENCE IS REQUIRED, SINDA FORTRAN V	ALL NODES RECEIVE A SUCCESSIVE POINT ITERATION Relaxation criteria must be specified	OVER-RELAXATION IS ALLOWED. THE DAMPENING FACTORS ARE ADDRESSABLE	INCLUDE COMMALIST COMMON /TITLE/H(1) /TEMP/T(1) /CAP/C(1) /SOURCE/0(1) /COND/6(1)	COMMON /PC1/NS01(1) /PC2/NS02(1) /KONST/K(1) /ARRAY/A(1)	COMMON: /FIXCON/KON(1) /XSPACE/NDIM•NTH•X(1)	COMMON /DIMENS/ NND/NNA/14/17/NGT.NCT.NAT.LS01.LS02	DIMENSION COM(1) + XK(1) + XK(1)	EOUIVALENCE [KON(1), CON(1)], (K(1), XK(1)), (X(1)), NX(1)]	END	INCLUDE DEFF,LIST	·★●●★★★★★★★★★★★★★★★★★★★★	CONTROL CONSTANT I CONTAINS THE NEW PROPLEM (IJAE (TIMEN)	CONTROL CONSTANT 2 CONTAINS THE TIME STEP NEED . (DTIMEU)	CONTROL CONSTANT 3 CONTAINS THE PRORLEM STOP TIME (TIMEND)	CONTROL CONSTANT 4 CONTAINS THE TIME STEP FACTOR EXPLICIT (CSGFAC)
υu	υU	U									ů	U	J	U	U U
*** •• NM	τ.υ * *	0	**	4.6	7*	*	*~	*	ř.	8	÷9	8	8	*#	*8
00101 00101 00101	00101	00101	00103	00105	00106	00102	001100	00111	00112	00113	00113	00113	00113	00113	06113

CNBACK . CNBACK

ید ^{ایر} این این این

÷.

. CC3 55 THE INPUT NUMBER OF ITERATION DO LOOPS. INTEGER (NLOOP) CC3 CONTAINS THE OUFFUSION TEAMETANUNE CHANGE ALLOWED TIME STEP CC3 CONTAINS THE MAINUM ALLONED TIME STEP CC3 CONTAINS THE MAINUM ALLONED TIME STEP CC10 CONTAINS THE MAINUM ALLONED TIME STEP CC11 CONTAINS THE MAINUM ALLONED ARITHMETIC TEMP. CHANGE CC12 CONTAINS THE PRISENT TIME OFTEN ARITHMETIC TEMP. CHANGE CC12 CONTAINS THE PRISENT TIME OFTEN ARITHMETIC TEMP. CHANGE CC13 CONTAINS THE PRISENT TIME OFTEN ARITHMETIC TEMP. CHANGE CC13 CONTAINS THE PRISENT TIME OFTEN ARITHMETIC TEMP. CHANGE CC14 CONTAINS THE PRISENT TIME OFTEN ARITHMETIC TEMP. CHANGE CC15 CONTAINS THE PRISENT TIME OFTEN ARITHMETIC TEMP. CHANGE CC15 CONTAINS THE ARITHMETIC TEMPERATIVE CHANGE CALCULATED CONTROL CONSTANT TA CONTAINS THE ARITHMETIC RELAXATION CC113 CONTAINS THE ARITHMETIC RELAXATION CONST AND CC10 CONTAINS THE ARITHMETIC RELAXATION CONST ARIT TABE CONTAINS THE ARITHMETIC RELAXATION CONST ARITES ARIT CC22 CONTAINS THE ARITHMETIC RELAXATION CONST ARITES ARIT CC22 CONTAINS THE ARITHMETIC RELAXATION CONST ARITES ARITES ARITES CONTAINS THE DIFFUSION RELAXATION CONST ARITES ARITES CC223 CONTAINS THE DIFFUSION RELAXATION CONTACT ARITES ARITES CC223 CONTAINS THE DIFFUSION RELAXATION CONST ARITES ARITES CC223 CONTAINS THE DIFFUSION RELAXATION CONTACT ARITES ARITES CC223 CONTAINS THE DIFFUSION RELAXATION CONTACT ARITES ARITES CC223 CONTAINS THE DIFFUS 666 F(CON(5).LE.0) GO TO 999 F(CON(6).LE.0.) CCN(6) = 1.E+8 F(CON(6).LE.0.) CON(6) = 1.E+8 F(CON(10).LE.0.) CON(10) = 1.0 F(CON(10).LE.0.) CON(10) = 1.0 F(CON(11).LE.0.) CON(11) = 1.0 F(CON(11).LE.0.) CON(11) = 1.0 F(CON(11).LE.0.) GO TO 998 F(CON(13).LE.0.) GO TO 998 F(CON(13).LE.0.) GO TO 998 F(CON(23).LE.0.) GO TO 998 F(CON(23).LE.0.) GO TO 999 F(CON(23).LE.0.) GO TO 991 F(CON(23 204

	а (р. 1996) Стан	•																•	-																	*										-	
CNIACK, CNBACK	U-WDIH = WDIM	CHECK FOR EXTRA LOCATIONS FOR CALCULATED NODES Terndim.lt.d) fo to 994	TPRINT = CON(13)	INITALIZE TIME SUM BETWEEN OUTPUT INTERVALS	DOES OLD TIME PLUS THE OUTPUT INTERVAL EXCEED THE STOP TIME	IF (CON(13)+CON(18).6T.CON(3)) CON(18) = COM(3)-COM(13) Dont Fycfer II	D TSTEPN = CON(22)	IF (TSTEPN.LE.CON(8)) 60 TO 20	5 TSTEPN = CON(8)	60 10 35 Does the time shu dins the time step except mitdin interval	D IF (TSUM+TSTEPN-CON(18)) 30,35,25	DONT EXCEED IT	5 TSTEPH = CON(18)-TSUM	60 10 35 Acts Time alle the Time stor synche dividity threaved	UDES TIME SUM FLUS TWO TIME STEPS EACEED VUTPOT ANTENNAL D tettstamas.ortsteph.le.ron(181) 60 to 35	APPROACH THE OUTPUT INTERVAL GRADUALLY	TSTEPN = (COULLA)+TSUN)/2.0	STORE DELTA TIME STEP IN THE CONSTANTS L CONSTANTS - TETEDN	CONCOL TOTICT	IF(PASS.GT.0.) G0 T0 40	COU(1) = TPRINT	CON(2) # 0.0	0 CON(1) = TPRINT+TSUM+TSTEPN	COMPUTE THE WEAN TIME BETWEEN ITERATIONS	5 CON(14) = (CON(1)+CON(13))/2.0	LAX = KOV(5)	DD = 1.0-DN	AN = CON(9)	AA II J.U.AAA Do the Rei Ayatton 1009	DO 240 KI = 1+LAX	KON(20) = K1	RLXA = 0.0	RLXD = 0.0	IF(K1.61.1) 60 10 110	ZERO OUT ALL SOURCE LOCATIONS AND SHIFT TEMPERATURES	DO 50 I = 1.NNAC			5 X(LEI) = T(I)	KUNILLZ' = U Cali Vähri i	CHECK THE BACKUP SWITCH	IF(KON(12).NE.D) GO TO 15	CHECK FOR FIRST PASS	LT TT ADD RETURN NO TO BUCCALL DUTCAL	PASS = 1.0	GU TO 10	
		U		J	v	ć	د	•		, 4	ر	Û	-,-	-4	د	υ		J	ر	,		di.		U					Ļ	,					ÿ				-		.U		U				-
	30+	****	*n *n	***	190 190	37#	+ + 0 m	+0+	* - +		+ + ナ ナ オ	#0#	#9#	*/*	# 4 0 0 1 4	20#	51+	52#	+ d4	+00 +00	56¢	57+	+ 6 5 5	÷09	61+	* 29 9 7 9 7	÷ † 9	654	* 00 * 1	68*	+69 +0+	71+	72*	494	+94	16+	78*	+62	* 0 0 0 0 0 0	+T0	834	84+	+ 50 • 50	87 *	88+	8 9*	
	00156	00156	00161	00161	00162	00163	00165	00166	00120	12100	00172	00172	00175	00176	9/100	00177	00201	00200	00000	00203	00202	00200	00210	00510	00211	00212	00214	00215	01210	00217	00222	00224	00225	00226	00230	00231	00236	00241	00242	00245	00242	00246	00246	00252	0.4253	00254	

•

.

€.

CNBACK , CNBACK

 $\sum_{i=1}^{n}$

.)

3330 1000 1000 1000 1000 1000 1000 1000	A A A A A A A A A A A A A A A A A A A	N N N N N N N N N N N N N N N N N N N	* NEW * NEW * NEW * NEW * NEW	3 등 년 11111년 11111년 11111년 11111년 11111 11111 11111 11111 11111 11111 1111
<pre>60 RC = 1.E+8 UJ = 0 UJ = 0 C CALCULATE FIRST PASS TEMPERATURES AND CSGNIN D0 105 I = 1.NND Include varcilisT Include varcilisT C Fold Deltatinno THE CaPacITANCES C Fold 1.1.NS01(J1+1)).E0.0) G0 T0 2000 NTYPE = FLD(0.5.NS02(J2)) LA = FLD(5.17.NS02(J2)) LA = FLD(5.17.NS02(J2)) LA = FLD(5.17.NS02(J2)) LA = FLD(5.11.NS02(J2)) LA = FLD(5.1</pre>	1012 J2 = J2+1 LA = FLD(5,17,NSQ2(J2)) LK = FLD(5,14,NSQ2(J2)) CALL DID1WM(T(1),A(LA),XK(LK),C2) GO TO 1998 1015 C1 = XK(LK)+XK(LA),XK(LK),C2) 1020 CALL DID1WM(T(1),A(LA),XK(LK),C1) 1020 CALL DID1WM(T(1),A(LA),XK(LK),C1)	LA = FLD(5,17,NSO2(J2)) LA = FLD(22,14,NSO2(J2)) C2 = XK(LK)+XK(LA) G0 T0 1998 1025 CALL PLYAWM(A(LA),T(1),A(LA+1),XK(LK),C1)) 1030 CALL PLYAWM(A(LA),T(1),A(LA+1),XK(LK),C1) 1032 J2 = J2+1	LA = FLD(5,17,MSO2(J2)) LA = FLD(22,14,MSO2(J2)) CALL PLYAWM(A(LA),T(1),A(LA+1),XK(LK).C2) GO TO 1998 1035 C1 = XK(LK)+XK(LA) GO TO 1032 1040 CALL PLYAWM(A(LA),T(1),A(LA+1).XK(LK).C1) J2 = J2+1 LA = FLD(22,14,MSO2(J2)) LA = FLD(22,14,MSO2(J2)) C2 = XK(LK)+XK(LA) C2 = XK(LA) C2 =	1045 CALL D2DIWM(T(I).CON(14).A(LA).XX(LK).C(I)) 50 TO 1999 1994 C(I) = C1+C2 1999 J2 = J2+1 2000 CONTINUE C(I) = C(I)/TSTEPN R1 = 0.0 R1 = C(I)/TSTEPN R1 = 0.0 R1 = 0.0 R1 = C(I)/TSTEPN R1 = 0.0 R1 = 0.0 R1 = C(I)/TSTEPN R1 = 0.0 R1 = 0.0 R1 = C(I)/TSTEPN R1 = 0.0 R1 = C(I)/TSTEPN R1 = 0.0 R1 = C(I)/TSTEPN R1 = 0.0 R1 = 0.0 R1 = 0.0 R1 = 0.0 R1 = C(I)/TSTEPN R1 = 0.0 R1 = C(I)/TSTEPN R1 = 0.0 R1 = C(I)/TSTEPN R1 = 0.0 R1 = 0.0 R1 = 0.0 R1 = C(I)/TSTEPN R1 = 0.0 R1 = C(I)/TSTEPN R1 = C
666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666 666	លល់លំលំលំលំលំលំលំលំលំលំ លំលំលំលំលំលំលំលំ		0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	66666666666666666666666666666666666666
002555555555555555555555555555555555555	00274	000000000000000000000000000000000000000	00315 00315 003117 003210 003223 003224 003224 003224 003226 003224 003226 003226	00000000000000000000000000000000000000

•,

• •

.

B - 6

. .

•

.

*	
U.	
4	
m.	
- 1	
0	
×	
Ü	
0	
z	
ö	
-	

\$

* * * * * *	**************************************	ア	N N N N N N N N N N N N N N N N N N N
60 T0 4999 4010 G1 = 0.0 4012 CALL D101WM(T(I),A(LA),XK(LK),02) 60 T0 4998 4015 G1 = 0.0 4017 CALL D101WM(CON(14),A(LA),XK(LK),02) 60 T0 4998 4022 J2 = J2+1 4022 J2 = J2+1 4022 J2 = J2+1 4025 G1 = XK(LK)*XK(LA) 4025 G1 = XK(LK)*XK(LA) 4025 G1 = XK(LK)*XK(LA) 4025 G1 = XK(LK)*XK(LA) 4022 J2 = J2+1	LA = FLD(5,17,HSO2(J2)) LK = FLD(22,14,NSO2(J2)) Q2 = XK(LK)+XK(LA) G0 TO 4998 4035 CALL DIDIWM(CON(14),A(LA),XK(LK),01) 4037 J2 = J2+1 LA = FLD(5,17,NSO2(J2)) LA = FLD(22,14,NSO2(J2)) LA = FLD(22,14,NSO2(J2)) LA = FLD(22,14,NSO2(J2)) 4040 01 = XK(LK)+XK(LA) 4999 01 = 01402+0(1) 4999 01 = 02+1	<pre>5000 CGNTINUE 6ND 0(1) = 0(1)+(f(1)+(f(1)+460.0) 05UM = 0(1) 6SUM = C(1) 6SUM = C(1) 6SUM = C(1) 6SUM = C(1) 6SUM = C(1) 6SUM = C(1) 10 UL VARG+(151 11 = FLD(2-14.NS01(J1)) 10 CLUUE VARG+(151 10 CLUUE VARG+(151) 10 F(FLD(2-14.NS01(J1)) 10 F(FLD(2-14.NS02(J2)) 10 F(FLD(2-14.NS02(J2))) 10 F(FLD(2-14.NS02(J2)) 10 F(FLD(2-14.NS02(J2))) 10 F(FLD(2-14.NS02(J</pre>	<pre>2010 TM = T(1) 2015 TM = T(1) 60 T0 2007 2015 CALL D1D1WM(T(1).A(LA).XK(LK).G1) 2017 J2 = J2+1 LA = FLU(5.17+NS02(J2)) LA = FLU(22:14+NS02(J2)) LA = FLU(22:14+NS02(J2)) CALL D1D1WM(T(LTA).A(LA).XK(LK).G2) CALL D1D1WM(T(LTA).A(LA).XK(LK).G2) 2020 G1 = XK(LK).*XK(LA) 2020 G1 = XK(LK).*XK(LA) 2021 CALL D1D1WM(T(1).A(LA).XK(LK).G1) 2022 CALL D1D1WM(T(1).A(LA).XK(LK).G1) LA = FLD(15.17+NS02(J2))</pre>
••••••••••••••••••••••••••••••••••••••		100************************************	107* 107* 107* 107* 107* 107* 107* 107*
00000000000000000000000000000000000000	000374	0041000044110000441100004411000044114000004411400000441140000044114000004411400000441140000044114000004414000004414000000	00000000000000000000000000000000000000

• •

.

•

•

• • ____

																														1 an 1 .	ş. 7	موجود م د		An Sin		un i		å v	si.	,,		•	×	h -			
																																				•						•	•				
~																																															
·?																																															
		•																																													
	4	NEW +-2					NFW	NEN	() 1 1 1				MLN	NEW	21 #																																
		**	•*				4	*.*	*				-		¥																																
																																											1				
	en.,																																														
						-			62)			-					()						<u>.</u>																								
			.6116		•	K), 61			CLK)			.K) , 61					1000					-	61+0						~		_						RESENT					1 1 1	160.0				
			(K (T K)			*XK (I			-1) - XF			*XX CI					(K (LK)						# (6)					i c	2	•	1+12						IF PF										
			() () + (LA+LI			ACLA			(LA+1)					(LA) -						1) 6(1					SC TOR	20		[2]*(J					04	NODE					ຼ	+00+ (y				
		(12))	2.0			A.(I)	12.51	(27)	(LTA)			A. (1)	110	(12))		c	[4] · A					(29/						COND		2++4	[+12+]				00101		THIS	001				RATUR	16SUN	0.5			
	¥.	NS02 (A)) / 2			ALLT (NS02	A) .T((A)		A) + T (A.	1005	NS02	(Y)	57119.	CONCI					1+19.				-	0.1	LION .		G) +T2	11+11			.	122		NON	101	MICO	2000		TEMPE	NN-S	TO	, ,		
	CNBA	2+14-	+T(L] ****			MIAL		20140	MCACL) * XK (MCACL	17.1	2,14) *XK (111	MUTA		,		•	101-1	NSQ1			460.0) + 460	RADIA	1000	M+G (L	re)*(19+₽	*79+W	- 19 - 61	IATIO	.) 60	++4		1/2°N	NEW	50) 	26 C 60			
	NBACK	FLD (2 XK (LK	(1(1)) (1(1)	5999	T(1) 2032	PLYAW	U2+1 51 0 / 5		PLYAW	2998 XK (LK	2042	PLYAW	1+70	FLD (2	XK (LK	2998	0201w	2999	TILTA	7 11 T A	2032		0(3,1	1420	1	+(1)1	T(LTA	E OK	R1+6(= 050	62+6(80 671 61	62+6V	020 #	= 05U	10119	N RAD		11#14	H1+T2	H1+R2	N DE		. CE . N	RI	11	
	U	ii ii Na K		60 TO	TM =	CALL	11 12	11 11	CALL	60 10 61 10	60 TO	CALL		1 II 1 II 1 II	ย เม	1 10		60 10	12 11	ו - בי נכ	100	6(10)	IF (FL		END	11 =	11 2			0SUM	11 62			65UM	OSCN S	LHE LA	DANPE	IF (R1	н н 22 р	111	- - - -	OBTAI		HI =	11	KULLON	
	1		2030		2035	2040	2042		-	2045		2050		-		100	27 7 7 7 7		2060	2045	2	2998		2999	2			-			-	J.	2			AD	5			-	;		100				
بو	•	*.*			* *		* 1			* *			* *	• *	*	'a # a	-			* •	- 	*	*	* *		*	*	υ. * ·	• •	*	*	¥ .1	F #		•	د • •	•	*	* *		•	•	÷.	* *	*	•	
	-	107	100	101	107	101	101	101	107	107	107	107		101	107	101	107	107	101	201	107	107	107	101	107	108	109		112	113	114	115	117	119	119	121	122	123	124	126	127	128	129	131	132	133	
		00447	10000000	10454	00455	00457	09460	10402	00463	00464 30465	00466	00467	02 500	0472	00473	10125	0476	17400	00500	10000	00505	00504	00505	10200	11200	00512	00513	00513	10516	11500	00520	00521	36523	00524	00525	36526	10526	00530	25000	10534	00535	00535	00536	000000	10542	00543	
i.				. U	J	، ن	<u> </u>	5.0	، <u>ب</u>		. . .		- 4	. 0		ہ ب	ن .		~ (تر. م <u>ہ</u>			-		. 0		-	<u>ب</u>		لت. .		<u> </u>			_ c	- 0	. u	، ب	ے ب	, c	ب		ء ب	- 0	ر ي .	-	

B - 8

ويتر والم

_...•

CNBACK . CNBACK

•

33040M	******	U	105 65	CONTINUE CONVERT TEMPERATURES TO RANKINE D0 65 I = 1.NNT LE1 = IE1+I T(I) = T(I)+460. X(LE1) = X(LE1)+460.
	******	U	110	IF (RC.LE.D.) 60 TO 993 60 TO 225 Now Relax The Network by Successive Point and Extrapolation JJ = JJ+1 D0 165 I = 1.NHD R1 = 0.0
	*********	Ų	115	S = 0.0 GSUM = 0(1) GSUM = C(1) U1 = J1+1 LG = FLD(22:14.NSO1(J1)) LT = FLD(22:14.NSO1(J1)) CHECK FOW RD14TION CONDUCTOR IF(FLD(3:11.NSG1(J1)).E0.0) GO TO 120
		U U	120	T = K1 +6(LG) GSUM = 05UM+6(LG) +T(LTA) ++4 GO TO 125 GSUM = 6SUM+6(LG) GSUM = 05UM+6(LG) +T(LTA) GSUM = 0SUM+6(LG) +T(LTA) GSUM = 0SUM = 0SUM+6(LG) +T(LTA) GSUM = 0SUM+6(LG) +
		υυυ	145	TZ = USUM-MC/2004 RI = RI#22+4 S = (RI#R2)2.0 OBTAIN THE NEW TEMPERATURE OBTAIN THE OLULATED TEMPERATURE DIFFERENCE ORTAIN THE CALCULATED TEMPERATURE DIFFERENCE TI = ABS(T(1)-T2) STORE THE NEW AND OLD FEMPERATURES GO TO (160,155,150), JJ LE2 = IE2+1 LE2 = IE2+1 LE2 = IE2+1
	11100000000000000000000000000000000000		155	RI = T2-T(I) X(LE2) = T(I) X(LE3) = R1/(R1-X(LE3)) X(LE3) = R1/(R1-X(LE3)) LE3 = IE3+I X(LE3) = 12-T(I) If(R120,GE,T1) GO TO 165 RLXD = I RLXD = I KW1 = I
		ູບູ່ບູ	165	CONTINUE GO TO (180,180,170), JJ GO TO (180,180,170), JJ JJ = 0 JJ = 0 DO 175 I = 1.N.3D LE3 = 1634 LE3 = 1634 SEE IF THE EXTRAPOLATION IS ALLOWABLE JF(X(LE3), GE.0.) GO TO 175

2.

merez zi

.

		3330 WUW 1 V 2 * * * *	· * * * * * *	330 330 WUI ZZ +	
С.	0. 3))+T(I)	60 TO 6000 5,5030,5035,5040,5030), NTYPE 02)	K).02) K).01) K).01)		T0 4000
CNDACK , CNBACK	LIMIT THE EXTRAPOLATION IF(X(LE3).LT10.) X(LE3) = -1 LE2 = IE2+1 LE2 = X(LE3)+X(LE2)+(1.0-X(LE CONTINUE IF(NIA.LE.0) GO TO 220 JJI = J2 JJ2 = J2 DO 230 T = 1.NNT T(1) = T(1)460.0 DO 215 T = NNANC L = T	GSUM = 0.0 IF(K1.51.2) G0 T0 6000 IF(K1.61.2) G0 T0 6000 IF(FLD(4.1.NSQ1(JJ11)).E0.0) IT = FLD(5.17.NISQ2(JJ2)) LA = FLD(5.17.NISQ2(JJ2)) LA = FLD(22.14.NISQ2(JJ2)) LA = FLD(22.14.NISQ2(JJ2)) CO T0 (5005.5010.5015.5020.502) G0 T0 (5909.5010.5015.5020.502) G1 = 0.0 G1 = 0.0	00 10 5998 Call DiDiwm(CON(14),A(LA),XK(L) Call DiDiwm(CON(14),A(LA),XK(L) 60 T0 5998 Call DiDiwm(CON(14),A(LA),XK(L) Ju2 = JJ2+1 Ju2 = J2+1 LA = FLD(52,14,NSG2(JJ2)) 60 T0 5027 61 = XK(LK)+XK(LA) 61 = XK(LK)+XK(LA) 61 = XK(LK)+XK(LA)	JJZ = JJZ+1 LA = FLD(5,17,MSG2(JJ2)) LA = FLD(22,14,MSG2(JJ2)) LA = KLD(22,14,MSG2(JJ2)) GO TO 5998 CALL DJDJW4(COT(14),A(LA),XK(L JJZ = JJ2+1 LA = FLD(5,17,TSG2(JJ2)) CA = FLD(5,17,TSG2(JJ2)) CA = FLD(5,17,TSG2(JJ2)) GO TO 5012 01 = XK(LK)+XK(LA) GO TO 5012 01 = XK(LK)+XK(LA) GO TO 5012 01 = XL(LK)+XK(LA) GO TO 5012 01 = XL(LK)+XK(LA) GO TO 5012 01 = JJ2+1 JJZ = JJZ+1 JJZ = JJZ+1 JZ = JJZ = JZ = JZ = JZ = JZ = JZ = JZ	CONTINUE END 05UM = 0(1) 0411 - 10111 LG = FLD(5.16.MSO1(JJ1)) LTA = FLD(22.14.MSO1(JJ1)) FF(K1.61.21.20 60 TO 4600 TF(FLUE VKG2.LTST TF(FLUE VKG2.LTST)).
а Э	c 175 180 230 230	5005 5010 5012	5015 5020 5020 5020 5020 5020	5035 5040 5999 5999	185
	10000000000000000000000000000000000000	<pre>N N N N N N N N N N N N N N N N N N N</pre>	* * * * * * * * * * * * * * * * * * *	* * * * * * * * * * * * * * * * * * *	20098 20000 20098 20008 200000 20000 200000 20000 20000 20000 20000 200000 2000000
<u>_</u>	$\begin{array}{c} \textbf{0} \textbf{0} \textbf{0} \textbf{0} \textbf{0} \textbf{0} \textbf{0} 0$	000110000000000000000000000000000000000	007110 007110 007111 007113 007113 007115 007115 007115 007210 007210	00722 00723 00724 00725 00725 00733 00733 00733 00733 00733 00733 00733	00740 00741 00744 007443 00745 00745 00755 00755

. .• .

en de la seconda de la seco Seconda de la seconda de la

B - 10

•

• ٠

	330 WW 1 220 224		N 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		800 100 100 100 100 100 100 100 100 100		0 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	₩ 2 00 100 100 100 100 100 100 100 100 100	
CNPACK.CNPACK	La = FLD(5,17,HSO2(JJ2)) Lk = FLD(22,14,NSO2(JJ2)) 6010(3005,3010,3015,3020,3025,3030,3035,3040,3045,3050,3055, 5 = 3060,3065); NTYPE TM = (T(L)+T(LTA))/2,0 CALL DIDIWM(TM:A(LA),XK(LK),6(LG))	СО ТО 3094 Со то 3007 Саll Didiwm(T(l),A(la),XK(lk),61)	JJZ = JJZ+1 LA = FLD(5,17,NS02(JJ2)) Lk = FLD(22,14,NS02(JJ2)) CALL DIDIWM(T(LTA),A(LA),XK(LK),62)	61 = XK(LK) *XK(LA) 60 T0 3017 62 LL DIDIWM(T(L),A(LA),XK(LK),61)	LA = FLD(5,17,NSQ2(JJ2)) LK = FLD(5,17,NSQ2(JJ2)) LK = XK(LK)*XK(LA) 62 = XK(LK)*XK(LA) 60 TO 3998	TM = (T(L)+T(LTA))/2.0 Call PLYAWM(A(LA),TM,A(LA+1),XK(LK),G(LG)) G0 TO 3999 TM = T(L) G0 TO 3032 Call PLYAWM(A(LA),T(L),A(LA+1),XK(LK),G1)	LA = FLD(5,17,NSO2(JJ2)) LK = FLD(22,14,NSQ2(JJ2)) LK = FLD(22,14,NSQ2(JJ2)) GO TO 3998 G1 = XK(LK)*XK(LA) G0 TO 3042 CALL PLYAWM(A(LA),T(L),A(LA+1),XK(LK),G1)	JJC = JJC + 1 LA = FLD(5,17,1)SQ2(JJ2)) LK = FLD(22,14,NSQ2(JJ2)) G2 = XK(LK) *XK(LA) G0 T0 3998 TM = (T(L)+T(LTA))/2,0 CALL D2D1WW(TM,CON(14),A(LA),XK(LK),G(LG)) G0 T0 3999 TM = T(LTA) G0 T0 3007 TM = T(LTA)	60 T0 3032 6(LG) = 1./(1./61+1./62) 1F(FLD(3.1.MSQ1(JJ1)).EQ.1) G(LG) = 61*62 JJ2 = JJ2+1 JJ2 = JJ2+1 CONTINUE E40 T1 = T(L1A)+460.0 T2 = T(LTA)+460.0 CHECK F0X RADIATION CONDUCTOR CHECK F0X RADIATION CONDUCTOR G(C) (11+11+T2+T2)*(T1+T2) G(C) 105
	3005 3007	3010	2102	3020		3030 3032 3035 3040 3040	3045	3055 3055 3060 3065	3999 2999 4000 4000 4000 4000 4000 4000 4
			* * * * * • • • • • • • • • • • •		8114 8114 8114 8114 8114 8114 8114 8114	2000 2011 2011 2011 2011 2011 2011 2011			0000 000 000 000 000 000 000 000 000 0
	00755 00755 00755 00755 00755	00762 00762 00763 00764	00765 00765 00767 00770	00772	001777 01000 010010	010020000000000000000000000000000000000		01021 01022 01023 01025 01025 01025 01027 01027 01027 01032	010000 010000 010000 010000 010000 010000 010000 010000 010000 010000 010000 010000 01000000

•

•

•

			n an	$\frac{1}{2} \sum_{i=1}^{n} \frac{1}{2} \sum_{i=1}^{n} \frac{1}$	
			• • • • • • • • • • • •		ana safaanan sa kalaba 200 a mara Sana aya
			. •		· · · ·
			•		
14	x			•	
)-				
		:	•		
		4			estar grand
					 The second secon
					Ì
$\sum_{i=1}^{n}$		<u>ພ</u> ຸ			
2	A. A.	LAR			14 99 19
	ERIA	100	•		E I T CHA
	CC CC .	E			TION T
	NON NO	S		C6D	TCG
	185 185 108 285 245 245	PL IN ANGE		69./1	275 11)// REL
	FOR 55 55 515 515 70 70 70 70 70 70 70 70 70 70 70 70 70	CH 10	520	10 570	AND AND
	DUC1 FTEC FTC C C C C C C C C C C C C C	CALL	_22	PN00	URES 00
ACK	+++ COR ++ COR +	65) 28)+ PERA	ND XCLE GO GO	((6)) 111 111 111 111 111 111 111 111 111	(11) TSTE Nut U Frat Frat
CNB	00170160 001700 001700 001700 00000000	TEM (0)		Z * Z + 0 * 1 (1) * 0 * 0 * 0 * 0 * 0 * 0 * 0 * 0 * 0 *	200 200 200 200 200 200 200 200 200 200
ACK					
S			250 250 1 AR 1 AR 1 AR 1 36) 1 36)	TINU TINU TINU TINU TINU TINU TINU TINU TINU TINU TINU	2260 2260 2260 2260 2260 2260 2260 2260
·	Second Se	CHARACTER CHARAC	CONCEPCED C	CONCONCENTION OF CLUCK	151 151 151 151 151 151 151 151 151 151
	195 195 215 235 235 2250 2250	5#2	250	255 260 265 265	275
		ັ ບ			ن ن
	**************************************			**************************************	* * * * * * * * * * * * * * * * * * *
	ุลงสถางสถางสถางสถางสถางสถางสถางสถางสถางสถา	៶៱៱៱៱៱៱		งงงิงสังสังสังสังสังสังสังส์	
				40000444440000000000000000000000000000	10020000000000000000000000000000000000
				000000000000000000000000000000000000000	000000000

B - 12

.

	(DELETED) (DELETED)
	335
	3 A G
	42603742 42615064
	0 -
	23:06:44 23:06:44
COMPUJE SEQUENCE)	JOSTIC* MESSAGE(S) 12 JAN 70 12 JAN 70
ILS GG TO 5 GG TO 5 SPECIFIED A RETIVE (ALE)	0 *DIAG
CHBACK.CNBACK FFIRLXA.GT.RLXD) GO TO 285 KX2 = KX1 RLXA = RLXD KX2 = KX1 RLXA = RLXD KX2 = XK1 RLXA = RLXA KX2 = XK1 RLXA = RLXA KX2 = XK1 RLXA = RLXD RX2 = XK1 RLXA = RLXD RX2 = XK1 RLXA = RLXD RX2 = XK1 RX2 = KX1 RX2 = XK2 COUI(12) = RX1 COUI(12) = RX1 TO 201 (12) = CON1 FICK FOR TIME TO PRINT TFICM.GE.CON118) GO TO 290 CHECK FOR TIME TO PRINT FICK FOR TIME TO PUT INTERVA FICK FOR TIME TO COMPL FICK FOR TIME TO PUT INTERVA FICK FOR TIME TO PUT INTERVA FICK FOR TIME TO PUT INTERVA FICK FOR TIME FOR TIME FOR TO PUT INTERVA RTTE (6, AB4) NDIM WITE (LTO FORTRAN V COMPILATION. 1109 FORTRAN V COMPILATION. ALLCCATABLE
285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 285 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 296 2	UIVAC
**************************************	K OF U
	CNHAG

~
æ
-
LL.
~
υ
-
ۍ
Ŷ
BX.
BK.
WBK.
FWBK,
JFWBK ,
NFWBK .

 \bigcirc

GIW FOR.* CNFWBK.CNFWBK UNIVAC 1108 FORTRAN V ATHENA VERSION 131K-10D CREATED ON 20 AUG 70 THIS COMPILATION WAS DONE ON 09 JUN 7D AT 14100143

SUBROUTINE CNFWBK ENTRY POINT 004271

STORAGE USED (BLOCK, NAME, LENGTH)

004307 000127 000071	000000												•
TEMP VAR	:	100000	00000	00000	00000	00000	00000	000000	00000	000001	000000	000010	
+CODE +CONST+ *SIMPLE	*ARPAYS *BLANK	TITLE	CAP	SOURCE	COND	PC1	PC2	KONST	ARRAY	FIXCON	XSPACE	DIMENS	•
0000	0004	0006	0100	0011	0012	0013	0014	0015	0016	2100	0020	0021	

•

EXTERNAL REFERENCES (BLOCK, NAME)

.

'n

VARBLI	OUTCAL	DIDIWN	PLYAWM	D2D1WM	TOPLIN	VARBLZ	EXIT	NERR25	SUQWN \$	N1025	NER105	
0022	0023	0024	0.025	0026	0027	0030	0031	0032	0033	0034	0035	

• •

2.2

STORAGE ASSIGNMENT FOR VARIABLES (BLOCK, TYPE, RELATIVE LOCATION, NAME)

000000	000000	100000	000025	000021	000027	000000	000000	000001	000000	000000
a a i	2	-			-	1 -1	-		H	œ
0017	9000	2000	0005	0005	0005	0021	0021	0021	0014	0011
υŽ	62	53I	5	NOX	Ц	Ľ	NAT	NN	NSO	PASS
000000	000041	000000	000022	000000	0000010	000034	000000	00000	000000	000000
<u>a</u>	с С		н	H	H	÷	9I	-		¢:
0010	0005	0005	0005	0017	0.005	0.002	0.021	0.002	0013	0005
zo		E2	22	K2	AX	<u>ں</u>	TAE	LA	ИТ	×
- 4 <u>0</u>	യ പ	ы П	ש כ	يد م	5	رب ن	5	2	.Z 01	2
000010	00002	00000	00000	90000	000001	00000	000020	00000	00000	00000
αœ	c	-	-		-	ĥ	H	H	-	
0002	0005	0005	0002	0005	0005	0.002	0002	0002	0021	0020
A C 2 V	GV	IEI	100	KKI	LA	LEJ	LTA	NGT	ONIN	NTYPE
0017	5400	£000	0062	0061	0033	0.000	2400	£000	00.00	0032
200	8	8	8	0	00	00	00	00	00	00
~~~	~			-		-	-	-		and And
2000	000	2000	0000	000	000	0002	000	0051	0051	0.00
	SUM		2	÷		63	502	DIM.	UN N	HL
0 Þ	5.0	۳ و	1	E E	لىر: ق ر	1   			्य :	2 
000	100	0200	03	000	0.0	105	0.00	000	00	000
000	00	00	000	000	000	000	000	00	100	00
αα	¢	•••	-		-	-	-		-	l aria
0016 0002	0005	0002	0000	0015	0000	0002	0.021	0.020	0005	0020

۰.

SENTING CON

. Berne States Strates

ч

*

B - 14

	~	<u>a</u> ,		3	٦	<u>ب</u>	2					2	ير	بر				ير	-	<u>_</u>	بہ	ي	۲	یے	ي	2	٦	ير	<u>.</u>	٦			. [.]		
	З'г,	rste	×	1001	102	10	113	1251	1601	1.851	25	201	20.32	2050	2201	2401	2651	2998	005	302	3042	306	100C	101	10 H	000	2012	502	5530	0.09	751	3588	8485	1966	1666
	52	200	05	5	63	5	2	74	90	21	2	ŧ	5	35	÷.	11	90	36	5	20	¢.	90	10 37	99	S	n at	#	2	5 E	15	47	52	20	8	5
	0000	0000	0000	1000	0002	000	037	022	1024	030	1000	013	015	015	036	036	010	017	1031	032	033	035	035	010	012	012	026	1027	1021	030	020	000	0000	042	540
	e .	<u>د م</u>	8	Ų	Ч		ų.	Ļ	Ļ	0	U	<u> </u>	0	<u> </u>	0	Ċ	-	Č	<u> </u>	<u> </u>	ੋ			Č.			<b>ب</b>	<u> </u>	<u>ب</u>	<b>ب</b>			0	9	
	005	002	020	001	100	100	100	100	001	001	001	100	001	001	001	100	001	001	100	100	100	001	100	õ	201	001	001	100	001	100	001	000	000	001	001
	00	0.0	C	0	0	0	0	0	0	0	0	0	0	e	0	0	0	C	0	0	0	0	0	0	0	Ċ	0	0	0	0	0	¢	Ö	0	0
		-																																	
		RIN		00	201	40L	206	5	5	5	66	101	305	451	5 <mark>1</mark>	90	91	5	050	201	401	601	د.	15L	30	186	10	22L	101	L	نہ	5	75	<u>ب</u>	<del>ک</del>
	52.	0 <del>P</del>	12	10	5 10	2	Ξ	112	15	18	5	20	202	202	21	5	20	50	500	30	30	30	5	0 1 0	07	49	50	50	S S G	60	50	88	88	66	5
	0030	0100	0.054	1244	0523	0674	3726	2263	2377	2502	0764	1371	1514	1627	3647	0317	0355	4134	3107	5214	5320	3502	0233	1065	142	1237	2613	2666	3000	0346	1260	100	34,00	1172	1231
	00	38	00	õ	õ	õ	00	00	00	00	õ	00	00	00	00	00	00	00	00	60	00	00	00	00	00	00	00	00	00	00	00	9	00	õ	00
	22 20	ע <u>מ</u>	02 R		H	10	-	-	1	=		10	=	E	Ξ	1	1	1	1	-	1	-	-	1	1	Ħ	1		1	-	10	g	0	Ξ	1
	000	ŝ	000	000	000	00	00	000	000	00	000	00	000	000	00	000	00	000	00	000	000	00	000	00	000	00	000	000	000	000	00	00	000	000	000
				. 1	5	2		26	;	,	F	7	<u>_</u>	2	ي		,	.,	2	2	10	Ľ	٦	<del>ر</del> ا	٦Ľ		٦ ت	5	۲	<u>ہ</u>	5	i 1.	.1.	1	.,
	200	X X	11	1001	101	103	110	116	150	1751	199	200	202	204	206	233(	2551	2851	300(	301	303	305	399t	101	402	45L	500	5021	503	599	671	8811	8861	1166	17.66
	-	21	53	26	515	566	900	101	5 t	200	191	551	54	529	32	513	167	94	67	.60	514	53	940	950	1	37	02	552	63	513	533	906	12	49	53
	0000	0000	000	0021	000	0000	0.022	0.04.0	0023	0.025	0007	0.01	100	0015	100	000	0037	0041	100	0031	0033	0034	0035	0010	100	0005	0026	0026	0.057	0030	0025	0000	0000	0.041	0042
	CC (	r œ	œ				-		_		-				. –		. —			Ŧ	. –	_				9		÷	9. T	-			-		. —
	002	2002	0000	1000	000	1000	1000	1000	1000	1000	1000	000	1000	1000	1000	1000	000	1000	1000	1000	1000	1000	1000	1000	1000	1000	1000	1000	1000	1000	1000	0000	0000	1000	1000
	Ξ.			-	Ŭ		Ű		Ŭ	Ĩ	Ŭ		Ļ	Ç		-	Ÿ			~	Ų	0			Ĩ	Č	Č	Ĩ	-		. <del>.</del>	Ĩ	<u> </u>	-	
	- 1	00 100	NDS	Ъ.	0121	0321	0736	1506	٦	705	951.	0.05L	020	10+0	0601	SC	50L	75L	Ы	0151	032L	050	1866	0101	022L	0401	0001	0171	0.35L	1966	646	BAF	95F	106	96L
	0	7 F	-	1	2	0	5	 -0	-		1	in at	6 9	2	يم 9	in N	ñ. M	ີ ເບ	8	77. - T	5	n r	т N	3 5	ч Т	1.4	9.0	ي ي	4	مت ص	9 2	0 8	7 3.	õ. 9	ۍ س
	1000	2006 0006	1000	016	00:00	063	365	0401	017	3243	0360	1134	144	0154	0172	036.7	375	0405	0200	1314	327	0.450	0351	0104	111	0123	0124	2263	0274	0.020	0252	0000	0003	34.15	0421
	23	ы ы ж ж ж	ğ	ă	ĕ	ă	õ	č	ö	5	č	č	č	õ	č	õ	ă	ŏ	č	0	ŏ	ŏ	č	õ	ŏ	ŏ	õ	2	č	õ	õ	ä	õ	5	õ
•	03	2 2	8	10	10	101	01	10	01	101	101	101	101	01	101	101	01	01	10	01	10	10	01	101	101	01	01	101	10	01	101	00	00	10	01
	8	000	0	0	0	8	8	0	0	ö	8	ŏ	8	0	č	8	8	3	00	00	0	8	ö	8	5	8	0	8	8	ö	0	8	8	8	0
															•																				
	Z :		EPN	: i	0L	5	ي ا	ي.	1	٦	2	ō	7	5L	55	0	ہے	ہے ا	391	5	5	5		56	50	572		5	20L	9	9		u.	Ľ.	_
	US0	2 2 2 2	TS	ž	101	10	101	115	145	165	190	200	201	202	205	25.1	342	270	5 à c	105	305	354	351	400	10.4	103	5	501	503	်ရှိ	644	801	894	865	306
	110	067	012	000	1111	607	153	224	330	421	600	767	412	544	700	271	720	040	764	135	263	104	221	040	103	214	150	934	111	215	141	072	032	954	207
	000	000	000	000	000	000	002	002	002	0.02	003	000	001	001	001	000	003	004	001	003	003	003	000	001	001	001	000	0.02	005	005	002	005	0.0.0	000	0.04
	.œ. (	ec pe N A		2				منبع ا	_	سر			مر			مے ر		مين	_					مىر	_	•••	_			ہے ا			~	ċ	
	0.00	000	000	001	000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	0.00	000	000	000	000	000	000	000	000

•

CNFWBK . CNF , BK

. IMPLICIT FORMARD BACKWARD DIFFERENCING EXECUTION SUBROUTINE THE LONG PSEUDO-COMPUTE SEQUENCE IS REQUIRED. SINDA FORTRAN V ALL NODES RECEIVE A SUCCESSIVE POINT ITERATION RELAANTION CATTERIA MUST BE SPECIFIED OVER-PELAANTION! IS ALLOWED. THE DAMPENING FACTORS ARE ADDRESSABLE INCLUDE COMMON /TITLE/H(1) /TEMP/T(1) /COP/C(1) /SOUACE/0(1) /COND/6(1) COMMON /TITLE/H(1) /TEMP/T(1) /CAP/C(1) /SOUACE/0(1) /COND/6(1) COMMON /FIXCON/KON(1) /PC2/NS32(1) /KONST/K(1) /ARAY/A(1) COMMON /FIXCON/KON(1) /PC2/NS32(1) /KONST/K(1) /ARAY/A(1) COMMON /PINENS/ MND.HNA.NIT.NGT.NCT.NATLLS01.LS02 DIMENSION CON(1) /CON(1)).(K(1).XK(1)).(X(1).NX(1)) EQUIVALENCE (KON(1)).CON(1)).(K(1).XK(1)).(X(1).NX(1)) END SURROUTINE CNF, RK ***O 000000 υύυυ 00101 00101 00101 00101 00101 00100 00100 00110 001113 001113 001113 001113

CNFWBK . CNFWBK

.

	(NLOOP)	(DTMPCA)	(OPEITR)	(H3WI10)	(DAMPA)	(DAMPD)	(ATMPCA)	(BACKUP)	(TIMEO)	(TINEM)	LUCOMIC.			INTERSO I	104100	ARLXCA)	(LOOPCY)	(DTIMEL)	DTIMET)	1244000		L SURAL Y	CSGRCL)	DRLXCA	IDRLXCC)	IL INECT)	PAGECT)	ARLXCC)	(LSPCS)	(ENGBAL)	BALENGY	(NOCOPY)					-WTFST)	J-VTEST)	(LAXFAC)																									
IFWBK • CNFWBK	THE INPUT NUMBER OF ITERATION DO LOOPS, INTEGER	INTAINS THE DIFFUSION TEMPERATURE CHANGE ALLOWED	MITAINS THE OUTPUT EACH ITERATION SWITCH	NITAINS THE MAXIMUM ALLOWED TIME STEP	INTAINS THE NEW ARITHMETIC TEMP. DAMPING FACTOR	ONTAINS THE NEW DIFFUSION TEMP. DAMPING FACTOR	ONTAINS THE MAXIMUM ALLOWED ARITHMETIC TEMP. CHANGE (	ONTAINS THE BACKUP SWITCH CHECKED AFTER VARIABLES (	ONTAINS THE PRESENT TIME OR PROBLEM START TIME	CHITAINS THE MEAN TIME BETWEEN AN TTERATION	CONTRACT THE DECIDING TRADEDATION THEORY OF A CHI AFEN	CONTRINS FOR DIFFORMATION CONTRICT CONTRICTION CONTRICTION	UNIAINS ANTIMMETIC TEMPERATURE CHANGE CALCULATED	AL CONSIANT IT IS RESERVED FOR THE CASG MINIMUM	IL CONSTANT LA CONTAINS THE OUTPUT INTERVAL	CONTAINS THE ARITHMETIC PELAXATION CRITERIA ALLOWED	CONTAINS THE NUMBER OF RELAXATION LOOPS USED, INTEGER (	CONTAINS THE MINIMUM ALLOWED TIME STEP	S EOR THE INPUT TIME STEP IMPLICIT			UNIAINS THE CASE RANGE ALEUVED	ONTAINS THE C/SG RANGE CALCULATED	CONTAINS THE DIFFUSION RELAXATION CRITERIA ALLOWED	ONTAINS THE DIFFUSION RELAXATION CHANGE CALCULATED	ONTAINS THE LINE COUNTER, INTEGER	ONTAINS THE PAGE COUNTER, INTEGER	ONTAINS ARITHWETIC RELAXATION CHANGE CALCULATED	S INDICATOR DETHERMAL SPCS JETHERMAL LPCS 2=GENERAL	CONTAINS THE ENERGY HALANCE OF THE SYSTEM, IN - OUT (	ONTAINS THE DESIRED ENERGY BALANCE. USER INPUT	ONTAINS THE NCCOPY SWITCH FOR MATRIX USERS	ONTAINS RELATIVE NODE NUMBER OF CSGMIN	CHITAINS RELATIVE NODE NUMBER OF DIMPCC	CNITAINS RELATIVE NODE NUMBER OF ARLYCC	ONTAINS RELATIVE NODE NUMBER OF ATMPCC	0-41-42-43 CONTAIN DIMMY INTEGER CONSTANTS (1-J-K-1	5-46-47-48 CONTAIN DUMMY FLOATING CONSTANTS (R-5-7-U	S THE QUAST-LINEARIZATION INTERVAL FOR CINDSM	S MAT HEED AT DOFCENT		1161   F A) 60 TA 009		$(10) \cdot [1 \cdot 1 \cdot $		1(10).LE.0.) CON(ID) = 1.D	(11).LE.0.) CON(11) = 1.E+B	1(3).LE.CON(13)) 60 TO 990	1(18).LE.D.) GO TO 998	GT.O.AND.CON(19),LE.0.) GO TO 997	1(22),LE.0.) GO TO 996	-6T.0.AND.CON(26).LE.0.) GO TO 995	(31).NE.1) 60 TO 991	-1.0	IND+1	WIGN	NTU	A MARTHAN AND AND A MARTHAN AND AND A MARTHAN AND AND AND AND AND AND AND AND AND A	A. TI34 NINT	TE 2 ANNUS	4.00 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1			
Ð	cc5 15	0000	CC7 CC	CCB CC	00 600	CC10 C	CC11 C	CC12 C	CC13 0					DAL NOD	CONTRO	CC19	CC20 C	CC21 C	1001				0222	cc26 C	CC27 0	CC28 C	CC29 (	CC30 C	CC31 1	CC32 (	0.033	1000	cc35 0	cc36 0	C 37 0	CC38 0		0.044-0	C C 4 9 ]	0000		TC (KON			IF CCU	IF (COF	IF (CO)	IF (CO)	IF (CO)	IF (NN/	IF (CO)	IF (NNC	IF (KON	PASS :	II NN	H A H	1E1 =	NNC #	115.0		י   	, , , , , ,		
	Ĵ	Ū	U	U	U	Ũ	о U	U	0	ب		) (	بر		0	U	U	U				, ر		U	ų	U	J	ر. ن	Ų	U	Ū	Ū	U	Ų	۔ س	۔ س	, U		J	۔ • د	,														-			-						
	÷8	# 60	#0	8#	8*	*0	8*	8*	*8	48		• ·	* ·	<b>4</b> 10	* ©	* 80	÷8	8			4 0 4	<b>*</b>	* 0	÷8	# 8	<b>#</b> 0	8	8	<b>*</b> 8	*8	*0	* •	*	ŧ.	*	* 6		*	8*	 α					*	*CT -	* 17	15+	16*	17*	18*	<b>1</b> 9*	20*	21+	22+	23+	24.	25*	264				+62	
.)	00113	00113	00113	00113	00113	00113	00113	00113	00113	0.113			01100	0113	00113	00113	00113	00113	1100			00113	21100	0113	00113	00113	00113	00113	00113	00113	00113	00113	00113	00113	00113	21100	00113	00113	00113		11100			12100	00123	00125	00127	00131	00133	00135	00137	00141	E+100	00145	00146	00147	00150	00151	00152	00153			CCT DO	•

B - 16

•

.....

. An e vys 

an an an ann an Araban Ann

-

CNFWBK . CNFWBK

•

•

NDIM = NDIM-J Check For Extra Locations For Calculated Nodes If(NDIM.LT.0) 60 TO 994	THIALIZE TIME SUM BETWEEN OUTPUT INTERVALS	DOES Does old time plus the output interval exceed the <b>stop time</b> Tercon(13)+con(18).gt.con(3)) con(18) = con(3)-con(13)	DONT EXCEED IT TSTEPN = CON(22)	IF(TSTEPN.LE.COM(8)) GO TO 20 TSTEPN = COM(3)	GO TO 35 Does the time sum plus the time step exceen output interval	IF(TSUM+TSTEPN-CON(10)) 30,35,25 Dont Exceed it	TSTEPN = CON(18)-TSUM	DOES TIME SUM PLUS TWO TIME STEPS EXCEED OUTPUT INTERVAL	IF (TSUM+2.0+15TEPN.LE.CON(18)) 60 TO 35 Approach the output Interval Gradually	ISTEPN ≈ (CON(IB)=TSUM)/2.0 STORE DELTA TIME STEP IN THE CONSTANTS	CON(2) = TSTEPH	CALCULATE THE NEW TIME TERPASS.GT.O., GO TO WO	CON11) = TPRINT	CON(2) = 0.0 20 to be	CON(1) = TPRINT+TSUM+TSTEPN	COMPUTE THE MEAN TIME RETWEEN ITERATIONS Countly = {con(1)+con(13)/2.0		DN # CON(10)	AN = CON(9)	AA == 1.0-AAA	TSTEP = TSTEPN/2.0 D0 THE RELAXATION LOOP	DO 240 K1 = $1 \cdot LAX$	KON(ZU) = KI	RLXA = 0.0	RLXD = 0.0 TFIK1.6T.11 G0 T0 110		ZERO OUT ALL SOURCE LOCATIONS AND SHIFT TEMPERATURES	DO 50 L = 1.FNC D(1) = 0.0	D0 55 I = 1.NNT	LEI = JEI+I / X(LE1) = T(I)	KON(12) = 0	CALL VARBLI Pusce the marking cwitch	IF (KUN(12), NE, D) GO TO 15	CHECK FOR FIRST PASS IF(PASS.GE.0.1 GO TO 60	CALL DUTCAL PASS = 1.0
	ų	n	2	15		50	SS.	. ب. ، ا	00	- 0	35	<b>v</b> -	Ŭ		ç t	i de la como	2		-						-			50	;	55	}	-		•	
U	IJ	U	ų.	• ,	U	. u	).	U	Ű	ų	ı .	U				U,					U	•					u					Ļ	<b>.</b>	U ·	
0-1 N	* * 1 2 = 4	10	**	**	* * N m	# #	**		* *	* *	*	# # # 4	÷	*		**	*	*. 	* * ± 10	*9	* *	*6	* *	*	* *		م	* *	*6	* *	*	* *	• • •	40	8 <b>*</b>
וניו ניו ניו	יימי מ	א מי ל	מימינ	ার র	* *	33	च व	. <b></b>	⇒ i0	ហេល	140	លជ	η Ω	νņ u	o jin	ω.	فرد	÷C \	0.0	¢	0.0			-		- 1-	~ 1			a) a	901	αa	0 00	¢0 ¢C	0.00
00156	00161	00162	00165	00166	00171	00172	00175	00176	C0177	00201	00202	00202	00505	00206	00210	00210	00212	00213	00215	C0216	00217	00220	00224	00225	00226	00231	00231	00235	C0237	00243	00245	00246	00247	00247	00254

-

. **.** 

.

	3330 UUU 1 222 * * * *	A NO A UUI UU UUN UU * * * *	830 330 UUI UUI XX 24 X* 24 X* 4 X* 4 X* 4 X* 4 X* 4 X* 4 X* 4 X*	220, 2220, UUI UUUI VZ4 ZZZ *** ***
CNFWBK • CNFWBK	<pre>60 T0 10 60 T0 10 60 RC = 1.E+8 7 C = 1.E+8 60 C C C C C C C C C C C C C C C C C C C</pre>	1012 J2 = J241 LA = FLD(5,17,1502(J2)) LK = FLD(25,14,NS02(J2)) CALL DID1WH(T(1),A(LA),XK(LK),C2) GO TO 1998 1015 C1 = XK(LK)+XK(LA) GO TO 1012 1020 CALL DID1WH(T(1),A(LA),XK(LK),C1) LA = FLD(5,17,NS02(J2))	LA = FLD(5.17,NS02(J2)) LA = FLD(2.11,NS02(J2)) C2 = XK(LX)*XK(LA) 60 T0 1998 1025 CalL PLYAMM(a(LA),T(1),A(LA+1),XK(LK).C(1)) 1030 G0 1999 1032 J2 = J2+1 LA = FLD(5.17,NS02(J2)) LA = FLD(	LA = FLD(5,17,NS02(J2)) LK = FLD(22,14,NS02(J2)) C2 = XK(LK)*XK(LA) 60 T0 1998 1045 CALL D2DIWM(T(1),CON(14),A(LA),XK(LK).C(T)) 60 T0 1999 1990 C(1) = C1+C2 1999 J2 = J2+1 2000 CONTINUE C(1) = C(1)/TSTEP R1 = 0.0 C(1) = C(1)/TSTEP R1 =
	<b>6 6 6 6 6 6 6 7 6 7 7 7 7 7 7 7 7 7 7 7</b>	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	96 96 96 96 96 96 96 96 96 96 96 96 96 9
	0005222 005222 005222 0052222 0052222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 005222 00522 00522 00522 00522 00522 00522 00522 00522 00522 00520 00522 00520 00520 00520 00520 00520 00520 00520 00520 00520 00520 00520 00520 00520 00520 00500000000	20000000000000000000000000000000000000	00000 000110 0001110 0001114 0001114 0001114 0001116 0001116 0001116 0001122 0001122 0001122 0001122 0001122 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 0001124 00001124 00001124 0000000000	00000000000000000000000000000000000000

* .**•** ...

B - 18

CNFWBK . CNFWBK

.

.

. .

....

00353	101#	4005	0(1) = XK(LK)+0(1)	
10000	101	4010		
00356	101+	4012	Call DiDIWM(T(I),A(LA),XK(LK),02)	
00357	101*		60 T0 4998	
0036U	101+	4012	01 = 0.0 Call Didiwm(Con(14),A(LA),XK(LK),02)	
00362	101*		60 T0 4998	
00363	101+	4020	CALL DIDIWM(CON(14) . A (LA) . XK(LK) . 01)	
00364	+101	4022		Part -
00200	*101		LA = FLUID/1///DUZ/UZ/1/ IX = FLUID2//4.NSG0//P/	
00367	+101		60 TO 4017	2 - + +
00370	101*	4025	01 = XK(LK)+XK(LA)	
00371	+101		60 TO 4022	
00372	101+	4030	CALL DIDIWM(CON(14) A(LA) XK(LK) OI)	
00374			UZ = UZ+1   A = F  D15, 17, 15602(J01)	MUN#
00375	+101		LK = FLD(22,14,NSQ2(JZ))	*NEW
00376	101*		02 = XK(LK) +XK(LA)	**-5
00377	101+		GO TO 4998	
00100	101	4035	CALL DIDIWM(CON(14)+A(LA)+XK(LK)+G1)	
00402	101+		TA = FLD(5,17,NSQ2(J2))	MEN
00403	+101	•	LK = FLD(22,14,NSQ2(J2))	*NEW
00000	101+		G0 T0 4012	2-**
00405	101+	4040	01 = XK(LK)*XK(L4)	
0.0 + 0.6	101+		G0 T0 4037	
20400	101+	4998	G(I) = G1+02+G(I)	
01410	*101	555		
11+00		nnnc		
21200	#101#		ENU 0(1) = 2.0±0(1)+C(1)±(T(1)+&60.0) .	
011100				
00415	104+		65UM = C(1)	
00416	105+	70	1+10 = 10	
00417	1064		LG = FLD(5+16+NSQ1(J1))	
00420	107+		LTA = FLD(22,14,NS01(J1))	
12400	# 20 U		LIAE Z LIA+IEA Incline VABG.ITSY	
00423	1094		TE(ELD(2,1,NSO1(J1)),EQ.D) 60 TO 3000	
00425	109+		NTYPE = FLD(0.5.NSO2(J2))	*NEW
00426	109+		LA = FLD(5,17,11502(J2))	*NEW
00427	*60 C		LK = FLO(22,14,NSQ2(J2))	AUN*
00430	*601	4 41	6010(2005/2010/2012)/2020/2025/2020/2025/2040/2042/2042/2042	
00400	1001	2006	D CODECODE DISTURDED	
00432	109+	2007	CALL DIDIWM(TM+A(LA),XK(LK),6(LG))	
00433	109+		60 TO 2999	
00434	+60T	2010	(I) I I I I	
00435	109*		G0 T0 2007	
004.36	*501	2012	CALL UIUIWM()(([]))A(LA))XK(LK))GI) 15 = 12+1	
00000	1001		06 - 5 - 015, 17, 15, 150 - 19 )	#NFW
00441	+601		LK = FLD(22,14,NSQ2(J2))	MBN#
00442	109+		CALL_DIDIMM(T(LTA);A(LA);XK(LK);G2)	2 + + +
0 d d d d d d d d d d d d d d d d d d d	100*	0000		
24400	109+	1909		
00446	109+	2025	CALL DIDIWM(T(I), A(LA), XK(LK), G1)	

÷

.

÷

	0 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	2 2 0 WW 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	■ ■ Q ■ U = Z = Z = Z = Z = Z = Z = Z = Z
- CNFWBK , CNFWBK	J2 = J2+1 LA = FLD(5.17.NSQ2(J2)) LA = FLD(22.14.NSQ2(J2)) G2 = XK(LK)+XK(LA) G2 TM = (T(1)+T(LTA))/2.0 2035 TM = (T(1)+T(LA))/2.0 2035 TM = T(1) 2035 TM = T(1) 2035 CALL PLYAWM(A(LA),TM,A(LA+1),XK(LK),61) 2040 CALL PLYAWM(A(LA),T(1),A(LA+1),XK(LK),61) 2040 CALL PLYAWM(A(LA),T(1),A(LA+1),XK(LK),61)	Z042 JZ = JZ+1 LA = FLD(5,17,1502(JZ)) LK = FLD(22,14,NS02(JZ)) CALL PLYAWM(A(LA),T(LTA),A(LA+1),XK(LK),G2) 60 T0 2998 2045 G1 = XK(LK)+XK(LA) 2050 CALL PLYAWM(A(LA),T(1),A(LA+1),XK(LK),G1) 2050 CALL PLYAWM(A(LA),T(1),A(LA+1),XK(LK),G1)	JZ = 12+1 LA = FLD(5,17+)SO2(J21) LX = FLD(5,17+)SO2(J21) CZ = XX(LK)=XX(LA) 60 0 299 205 TM = 7(1/1)+T(LTA)/2.0 206 TM = 7(1/1) 206 TM = 7(1/1) 207 2032 2099 JZ = J2+1 3000 CONTINUE 1 = 7(1)+460.0 7 = 7(1)+100.0 1 = 7(1)+100.0 1 = 7(1)+100.0 1 = 7(1)+100.0 2 = 100.110 = 0.0 2 = 0.01 +0.0 1 = 7(1)+100.0 2 = 0.01 +0.0 2 = 0
	• • • • • • • • • • • • • • • • • • •	**************************************	11111111111111111111111111111111111111
ċ	0004450 0004550 0004551 0004552 0004552 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 0004555 00045555 00045555 00045555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 00055555 00055555 00055555 00055555 00055555 00055555 00055555 000555555	00465 00464 00465 00465 00467 00470	0004774 004474 0004774 0004774 000500 000501 000500 0005113 000500 0005113 000500 0005113 000500 0005113 000500 0005113 000500 0005113 000500 0005113 000500 0005113 000500 0005113 000500 0005113 000500 000500 0005113 000500 000500 0005113 000500 0005113 000500 000500 0005113 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000500 000000

)

## B - 20

•

;•

×	
2.4	
- FE	
- ¥	
WB	
H.	
<u> </u>	

00543	オナロー		R1 = C(1)/62	
00544	#001 #96		27 1X1 • VE • KC / 60 10 103	
00547	137+		Kon(35) = I	-
00550	138*	105	CONTINUE	
00550	*>01	5	CONVENT FEMPERATORS JO RANNING -	
00555	111+			
00556	142+	2	T(I) = T(I)+460. Vite1) - Vite11_Ake0	
00561	1404	6	AILEL A AILELY 4400	
00562	145*		IF(RC.LE.0.) 60°TO 993	
00564	146*	,	G0 T0 225	
00564	147*	0 - -	NOW RELAX THE NETWORK BY SUCCESSIVE POINT AND EXTRAPOLATION	
00506	149#			
00571	150*		R1 = 0.0	
00572	151+		S = 0.0	
00573	152*			
00575	154+	115		
00576	155*		LG = FLD(5,16,NSQ1(J1))	
00577	156*		LTA = [LD(22.14.NSQ1(J1)]	
00577	157*	IJ	CHECK FOR RADIATION CONDUCTOR	
00000	1591			
00603	160*		osum = osum+6(L6)∗T(LTA)**4	
0.0604	161+		60 TO 125	
00605	162*	120	65UM = 65UM+6(L6)	
00000	163+		OSUM = QSUM+6(LC) + f(LTA)	
00606	164#	Ű	CHECK FOR LAST CONDUCTOR	
0.0607	165*	125 r	IF(NSG1(UI).6T.0) GU TO II5 Dampen radiation on This Node is dresent	
00011	167*	,	TE(R1.LE.0.) 60 TO 145	
00613	168+		R2 H R1+T(1)++4	
00614	169*		T2 = (QSUM+R2)/6SUM	
00615	170+		R1 = R1*T2**4	5
00616	171+	Ļ	5 5 (K1+K2)/2.0 00171N TUS NEW TEADEDS	Ζ.
01000	173+	145	T2 = DN*((0SUN-5)/6SUN)+DD*T(1)	•
00617	174+	J	OBTAIN THE CALCULATED TEMPERATURE DIFFERENCE	
00620	175*		TI = AUS(T(I)=T2) crone the new and a remorphylice	
02000	177+	د	JIONE THE NEW AND YEAR CONES 60 TO (160-155-150); JU	
00622	178*	150	LE2 = IE2+I	
00623	179*		LE3 = IE3+I	
00624	180*		R1 = T2-F(1)	
02000	+ 10 1			
07000	1834	•	ANTESS - MIXINI-ANTESS	
00630	1.84 +	155	LE3 = 1E3+1	
00631	105*		X(LE3) = T2-T(1)	
00032	186*	160	T(I) Z T2 Tr(A) V) CE T1 CA TA 12E	
010.35	144			
00036	199*	•		
00h37	<b>*06</b> T	165	CONTINUE	
14000	191+	Ĺ	GU TO (180,180,170), JJ Deregons, itnear Extradol Atton on the Erran Filmorian Filmve	
00642	193*	170	LLN VIN LINCAN LATING VERTIONS VIT THE THINKS TAPPED VIT VITE LINCAN TAPPED	

~	and the second se																																													
																10 e ##	ı									ADV+	01 <b>i</b> +	<b>.</b>							*NEW	A U A	***									
CNFWBK , CNFWBK	$\begin{array}{llllllllllllllllllllllllllllllllllll$	C SEE IF THE EXTRAPOLATION IS ALLOWABLE Tervites.ge.n.) 60 to 175	C LIMIT THE EXTRAPOLATION IF(X(LE3).LT10.) X(LE3) = -10.	LE2 = IE2+1 T/1 = V/1 E3/1+V/1 E3/1+10 O-V/1 E3/1+T/11	175 CONTINUE	180 IF(NHA-LE-0) 60 TO 220		DO 230 I = 1.NNT	230 T(I) = T(I)=460.0		L - 1 IF(K1.6T.2) 60 TO 6000	INCLUDE VR02/LIST	IF(FLD(4,1,NSQ1(JJ1+1)),EQ.0) 60 TO 6000			CO TO (5005) 5010 5015 5026 5025 5030 5035 5040 5030) NTYPE	2002 0(L) = XK(LK)+0(L)		SULO GI = U.U SAT2 CALL DIDIWM(TTL).A(LA).YK(LX).02)		5015 G1 = 0+0	5017 CALL DIDIWM(CON(14), A(LA), XK(LK), 02)		5020 CALL DIDI#M(CON(14)+A(LA)+XK(LK)+GI) 6032 112 - 1124		LK = FLD(22+14,NSQ2(JJ2))	G0 T0 5017	5025 01 = XK(LK) *XK(LA)	5030 CALL DIDIWIA(CON(14),A(LA),XK(LK),01)	JU2 = 4J2+1	LA = FLD(5+17+NS02(JU2))	DZ = XK(LK) *XK(LA)	GO TO 5998	5035 CALL DIDIWM(CON(14),A(LA),XK(LK),Q1) 5037 -172 = 14241	LA = FLD(5,17,4502(JU2))	LK = FLD(22,14,NSQ2(JU2))		30440 WI # ANILNJ*ANILAJ 60 TO 5037	5998 G(L) = 01+02+0(L)	1+2-D = 7-D = 2-D		65UM = 0.0	02UM # 0(1)	LG = FLD(5,16,NS01(JJ1))	LTA = FLD(22,14,NS01(JJ1))	
	194+	196#	198*	200	205#	203+	205#	2064	207*	1000	210+	211+	211+	***		211	211+	511*	*112	211*	211+	\$113	211	2114	211*	211*	211+	\$112 *112	211+	211*	211*	211+	211*	211+	211+	211+	211*	\$11×	211*	211*	2114	212*	213*	215*	216*	
$\geq$	00643	00646	00647	00653	00655	00657	00662	00663	00666	0/000	9000 HL	00676	00677	10/00	20200	00204	00705	00706	10/00	00711	00712	61700	00714	00715	1200	00720	00721	00722	00724	00725	00726	00730	00731	00732	00734	00735	00736	00740	00741	00742	00744	00745	00746	00750	00751	

ng ternés vezen di sub cur annés, mili man cu di su nya cur a Nya ternés vezen di sub cur annés, mili man cu di su nya cur a su

•

×
Ű,
3
<u>L</u>
z
ų.
3
₩.
Ξ.
Ž
5

;

• 10

		- <b>:</b> •			
			N R N LULI N N N		
<pre>IF(K1.6T.2) 60 T0 4000 INCLUDE VRG2.LIST IF(FLD(2.11.NSQ1(JJ1)).E0.0) 60 T0 4000 NTYPE = FLD(0.5.NSQ2(JJ2)) LA = FLD(25.14.NSQ2(JJ2)) LA = FLD(25.14.NSQ2(JJ2)) LA = FLD(25.14.NSQ2(JJ2)) 60T0(3005.3010.3015.3020.3025.3030.3035.3040.3045.3050.3055. 3007 CALL DIDIWM(TW.A(LA).XK(LK).6(LG)) 60 T0 3999 3010 TM = T(L) 60 T0 3999 3010 TM = T(L)</pre>	3015 CALL UIDIWM(I(L),A(LA),XN(LK),61) LA = FLU(5,17,NSQ2(JJ2)) LK = FLD(22,14,NSQ2(JJ2)) CALL DIDIWM(T(LTA),A(LA),XK(LK),62) 60 T0 3994 3026 61 = XK(LK)+XK(LA) 5025 CALL DIDIWM(T(L),A(LA),XK(LK),61)	JJZ = JJZ+1 LA = FLD(5.17.NS02(JJ2)) LK = FLD(2.14.NS02(JJ2)) G2 = XK(LK)+XK(LA) G0 T0 3998 3032 CAL PLYAUM(A(LA))/2.0 3032 CAL PLYAUM(A(LA))/7.0 5035 TM = T(L) 5035 TM = T(L) 5040 CAL PLYAUM(A(LA).T(L).A(LA+1).XK(LK).6(LG)) 5040 CAL PLYAUM(A(LA).T(L).A(LA+1).XK(LK).61)	JUL 2 FLD(5,17,NSQ2(JJ2)) LK = FLD(5,14,NSQ2(JJ2)) CALL PLYAWM(A(LA),T(LTA),A(LA+1),XK(LK),G2) GO TO 3998 3045 G1 = XK(LK)*XK(LA) GO TO 3642 3050 CALL PLYAWM(A(LA),T(L),A(LA+1),XK(LK),G1)	JJZ = JJZ+1 LA = FLD(5.17,MS02(JJZ)) LK = FLD(22,14,MS02(JJZ)) G2 = XK(LK)+XK(LA) G0 TO 3993 3055 TM = (T1L)+T(LTA))/2.0 CALL D2D1WM(TM,CON(14).A(LA),XK(LK).G(LG)) G0 TO 3999 3060 TM = T(LTA) 5055 TM = T(LTA) 60 TO 307 505 TM = T(LTA)	3998 G(LG) = 1./(1./G1+1./G2) IF(FLD(3.11.NSQ1(JJ1)).EQ.1) G(LG) = G1*G2 3999 JJ2 = JJ2+1 4000 Continue END TI = T(1)+460.0 T2 = T(LTA)+460.0
					2209 2218 2218 2218 220 220
00755 00755 00755 00760 00760 00762 00765 007654 007654 007664 007664	000110 001712 001712 001715 00170 00176 0100 0100 0100 0100 0100 01		01015 01015 01020 01021 01022	01024 01027 01027 01027 01031 01033 01033 01033 01033 01035 01035 01035 01035 01035 01035 01035 01035 01035 01035 0105 010	010445

		a se a ser a s La ser a s
2		
$\bigcirc$	DR Go To 190 1412) Ation Criteria Was Met 25 Ation Criteria Was Met 15 Met Vas Met 45 Kin 65 Were Too Large	TCGD 75 0/TCGA 10 FARENHEIT
CNF WBK . CNF WBK	<pre>C CHECK FOR RADIATION CONDUCT FF(FLD(3,1,NSG1(JJJ1),EG.0) 6V = 6(LG) +(T1+T1+T2+T2)+(T) 6V = 6(LG) 195 6SUM = 6SUM+6V+T2 6V = 6(LG) 195 6SUM = 6SUM+6V+T2 C CHECK FOR LAST CONDUCTOR F(NSG1(JJ1),6(T,0) 6O TO 18 C CLUATE THE WFERATU T = ABS(T2-T1) T(1) = T2-460.0 F(RLXA.6E.T1) 60 TO 215 RLXA = T T = ABS(T2-T1) T(1) = T2-460.0 F(RLXA.6E.T1) 60 TO 215 RLXA = T KK2 = T KK2 = T KK2 = T KK2 = T KK2 = T KK2 = T C CONTINUE C SE IF THE ARITHMETIC RELAX F(C) 235 T = 1,NNT C SE IF THE ARITHMETIC RELAX F(C) 235 T = 1,NNT C SE IF THE ARITHMETIC RELAX F(C) 235 T = 1,NNT C SE IF THE CONTINUE C CONTINUE C CONTINUE C CONTINUE C CONTINUE C CONTINUE C C CONT(26)) 60 TO 240 C C C C CONT(26)) 60 TO 240 C C C C C C C C C C C C C C C C C C C</pre>	D0 250 1 = 1.NUE LE = 1E1+1 = C(1) = TEFEP T1 = ABS(T(1)-X(LE)) TCGD = T1 TCGD = T1 XON(36) = 1 XON(36) = 1 TF(TCGD.LE.CON(6)) GO TO 26 TF(TCGD.LE.CON(6)) GO TO 27 TF(TCGD.LE.CON(1)) GO TO 27 TCGA = T1 T1 = ABS(T(1)-X(LE)) TCGA = T1 T1 = ABS(T(1)-X(LE)) T1 = ABS(T(1)-X(LE)) T1 = ABS(T(1)-X(LE)) T2 = 151+1 T1 = ABS(T(1)-X(LE)) T2 = 151+1 T2 =
	************************************	8 0 0 0 0 0 0 0 0 0 0 0 0 0
$\sim$	00000000000000000000000000000000000000	

B - 24 19 - a

...

•

.

٠.

.

		and the second se
3 2 4		
CMFWBK , CNFWBK	281         C         STORE         THE FEMERATURE         AND         RELAKATION         CHANGES           283         CONIL3) = TEGO         TEGO         CONIL3) = TEGO         CONIL3) = TEGO           284         FUXA, ELKACO         FIRLAN, FIRLANO         GO TO 285         CONIL3) = TEGO           284         FUXA = RKI         FUXO         GO TO 285         CONIL3) = CON         CON           285         FUXA = RKI         FUXO         FUX         FUXO         FUXO         FUXO           285         FUXA = RKI         FUXO         FUXO         FUXO         FUXO         FUXO           285         CONIL37) = KUZ         FUXO         FUXO         FUXO         FUXO         FUXO           285         CONIL37         FUXO         FUXO         FUXO         FUXO         FUXO           285         CONIL37         FUXO         FUXO         FUXO         FUXO         FUXO           285         CONIL37         FUXO         FUXO         FUXO         FUXO         FUXO           285         FUXO         FUXO         FUXO         FUXO         FUXO         FUXO           285         FUXO         FUXO         FUXO         FUXO         FUX	
		)     

۰.

CNVARB, CNVAPB

.

GIW FOR.+ CNVARB-CNVARB UNIVAC 1108 FORTRAN V ATHENA VERSION 131K-10D CREATED GN 20 AUG 70 THIS COMPILATION WAS DONE ON 09 JUN 70 AT 14:00:51

SUBROUTINE CNVARB ENTRY POINT 004272

STORAGE USED (BLOCK, NAME, LENGTH)

...

000072	000000	000000												
VAR		-	100000	100000	100000	100000	00000	00000	100000	000001	100000	1000000	000003	010000
*SINPLE	*ARHAYS	*BLANK	TITLE	TEMP	CAP	SOURCE	CONU	PC1	PC2	KONST	ARHAY	FIXCON	XSPACE	DIMENS
0002	000	0002	0006	0001	0010	1100	0012	£100	1100	0.015	0016	0017	0020	0021
	0002 *SIMPLE VAR 000072	0002 *SINPLE VAR 000072 0004 *ARHAYS 000000	0002 *510715 1127 000123 0002 *5107LE VAR 000072 0004 *ARHAYS 000000 0005 *8LANK 000000	0002 *5UNJE VAR 000072 0004 *ARMAYS 000072 0004 *ARMAYS 000000 0005 *BLANK 00000 0006 TITLE 000001	0002 *5107151757 000072 0002 *51076E VAR 000072 0004 *ARKAYS 000000 0005 *8LANK 00000 0006 TITLE 000001 0007 TEMP 000001	0002 *51MPLE VAR 00012 0004 *ARMAYS 000000 0005 *BLANK 000000 0006 TITLE 000001 0001 TAMP 000001 0010 CAP	0002 *5107LE VAR 00002 0004 *ARKAYS 00000 0005 *BLANK 00000 0006 TITLE 000001 0007 TEMP 000001 0011 SOUNCE 000001	0002 *SIMPLE VAR 00012 0005 *BLANK 00000 0005 TITLE 000001 0006 TITLE 000001 0010 TEMP 000001 0011 SOUNCE 000001 0011 SOUNCE 000001 0012 CONU	0002 *510755 2500 0005 *510755 2500 0005 *8LANK 00001 0006 TITLE 000001 0010 CAP 000001 0010 CAP 000001 0011 CONCE 000001 0011 CONCE 000001 0013 PC1 000001	0002 *51MPLE VAR 00012 0004 *ARMAYS 000000 0005 *BLANK 000000 0005 TITLE 000001 0001 TEMP 000001 0010 CAP 000001 0011 SOUNCE 000001 0012 CONU 000001 0012 PC1 000001 0014 PC2 000001	0002 *5INPLE VAR 00072 0004 *ARKAYS 00000 0005 *BLANK 000000 0006 TITLE 000001 0001 CAP 000001 0011 SOUNCE 000001 0012 CONU 000001 0013 PC1 000001 0014 PC2 000001 0015 KONST 000001	0002 *5107LE VAR 00072 0004 *ARKAYS 00000 0005 *BLANK 000000 0007 TITLE 000001 0007 CAP 000001 0011 SOUNCE 000001 0012 CONU 000001 0012 PC1 000001 0014 PC2 000001 0014 PC2 000001 0015 AKKAY 000001	0002 *51075 551075 0005 *51075 5510 0005 *8LANK 00000 0006 TITLE 000001 0010 56000 00001 0011 5600 00001 0012 600001 0013 PC1 000001 0013 PC1 000001 0015 80057 000001 0015 80057 000001 0015 80057 000001 0015 81807 000001	0002 *51MPLE VAR 00012 0005 *51MPLE VAR 00012 0006 TITLE 000001 0001 CAP 000001 0011 SOUNCE 000001 0012 CONU 000001 0012 PC1 000001 0014 PC2 000001 0015 PC1 000001 0015 PC1 000001 0015 PC1 000001 0010 PC2 000001 0010 PC2 000001 0011 PC2 000001 0012 PC2 000001 0010 PC2 000001 000001 000001 000001 000001 000001 000001 000001 000001 000001 000001 000001 000001 000001 000001 000001 000001 000001 0000001 000001 0000001 000001 0000001 000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 0000001 00000000

EXTERNAL REFERENCES (BLOCK - NAME)

.

VARBLI	0101WM PLYAWM D201WM TOPLIN	EXIT NERR25 NWDU5 NIO25 NER105
0022 0023	0024 0025 0025	0031 0032 0033 0034

÷

STORAGE ASSIGNMENT FOR VARIABLES (BLOCK, TYPE, RELATIVE LOCATION, NAME)

+

0002 R 00015 BETAO 0002 R 00015 DD 0002 R 000046 G1 0002 I 000064 JJ2 0002 I 000064 JJ2 0002 I 000064 JJ2 0002 I 000013 LAX 0022 I 000005 HAT 0022 I 000005 HAT 0022 I 000005 HAT 0022 I 000000 PASS
0002 R 000054 BETAN 0002 R 000036 C2 0002 R 000094 C2 0002 I 000004 IE1 0002 I 000063 JJI 0002 I 000063 JJI 0002 I 000063 LA 0002 I 000063 LA 0002 I 000063 LA 0002 I 000063 LA 0002 I 000001 LE3 0002 I 000002 NNT
0002 R 000016 AN 0002 R 000055 C1 0002 I 000057 G5UM 0002 I 000026 I 0002 I 000031 JJ 0002 I 000031 JJ 0002 I 000031 JJ 0002 I 000031 JJ 0002 I 0000037 LE2 0021 I 000007 LE2 0021 I 000007 LE2 0021 I 000007 LE2
0002 R 000017 AA 0017 R 000000 CON 0012 R 000000 G 0002 I 000007 J 0002 I 000027 J 0002 I 000027 J2 0002 I 000021 K1 0002 I 000027 LE1 0020 I 000000 HDIM 0020 I 000000 HDIM
0016 N 000000 A 0010 R 000000 C 0002 R 000014 DN 0002 I 000004 DN 0017 I 000006 IE3 0017 I 000006 IE3 0012 I 000001 KoN 0012 I 000001 LE 0021 I 000001 NHA

.

n Na sang tanggan kanang karang sa pang sa basa pang

,

21 FER 71

. B - 26

	-	_	÷.,																	-														
	ЧЧ И И И И И И И	100	205	5	200	ដ	2	ď	56	2	200	181	ಕ	80	0	٦	25	205	ş	50		12	30	196		221			<u>_</u>	5	1	ที่		
κ. N	<b>F</b> .F	Ĩ	-	-	-			-	-	2	N.	Ċ,	~	N.	č.	N.	n	5	0 0	5		÷.	37 - 1 27 - 1 27 - 1	3- ) 3- )		ີດ ເດີ	ត	ŝ	2	ŝ	i a	6- ) 6- )	5	
0020	0010000	1246	521	270	3726	224	2374	2205	1770	37	1516	6.31	3647	325	1361	1134	101	3214	5326	2205	233	01		124	519	2002		351	1264	100	100	117	1231	
00	õõ	00	õõ	ö	é	õ	00	00	ĕ	ē	5	00	8	000	00	00	00	00	ŝ	00	ē	8	8	00	00	ë	00	00	00	000	õ	ē	200	
84 64 64 64	02 02 04 04	-			-	-	-		-	_		_								 	-	-		_	_			-		ä	0		_	
000	0000	000	600	000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	00	000	000	000	000	000	000	000	000	000	000	000	
	<b>-</b>	Jor.	1150	351	2	676	20	75L	1996	0.71	025L	<u>12</u>	1651	336	55L	35 <b>L</b>	2001	1710	351	1550	1666	121	125L	<u>ب</u>	100 <b>1</b>	205	371	1666	16	L.	365	H	۲ ۲	
02	57 00	Ца	ž	ୁ ଲ	Ξ	긑		а. П	5	ດ. ຄ	3	2 2	ດ. +	ີ. ເ	3 2	ы т	202	ñ	ň t	ที่ ค	6 0	37 37	3		ທີ່ ສ	ິດ. 1	ທີ. ຄະ	ະດີ ດາ	6	8	8	ŏ. 	ő	
900	100	207	052	167	550	106	535	250(	176	135	145	157	173	0310	376	110	177	316	331	345	354	202		22.0	500	202	276	301	253	000	100	116	25	
000	80	00	00	00	8	ŝ	00	C C	č	S	60	00	ő	ö	00	00	00	8	20	8	č	00	8	0	ê	ë,	00	8	00	00	00	0	00	
R R N N	C C	-		-	<del></del>	-	-	-	-	***	<b>,</b>			-		-			-	-	-1		-	-	-		–	:	1	0	c			
000	000	000	000	000	000	000	000	000	000	000	000	0.00	000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	000	
																					• .													
				1	ŝ	5					r	1												,	1		1	•						
	SUN SUN	5	0121	0.321	073(	150	51	701	956	0051	0201	010	0.00	251	50L	756	Ľ,	0151	0321	020	99AI	010	0221	040	000	017	0351	9981	646	80F	855	<b>1</b> 06	966	
0.4	アー		9 9		ŝ	9 9			- - -	ā v	Q.	~	2 0	οi m	Ω Μ	ິດ: ເວ	6 0	ř) H	1	ē N	10. N	37. M	न ज	4	ي م	ۍ ح	e Q	ية م	9 9	9.0	1 8	6 9	è С	
004	006	016	046	063	365	401	117	243	360	134	145	155	173	367	375	405	020	314	327	3:10	351	105	112	123	125	253	274	300	252	000	003	415	421	
88	000	8	00	00	2	00	3	8	00	00	8	00	00	00	00	00	00	00	00	00	ĉ	00	8	8	00	8	00	00	00	00	0	8	00	
	200	17	7	Ę		1	H	H	1	ä	-	Ħ	=	-	-	-	Ξ	H	-	-	=	=	-	Ħ	=			. 10	-	õ	0		-	
000	000	00	000	000	000	000	000	00	000	000	000	000	000	000	00	00	000	000	8	00	ŝ	000	000	õ	00	8	000	00	õ	00	õ	00	00	
																					• .													
	Z	:	_							_	د.		_				_		ر		-	بہ	بہ	ب										
NS N	CGA	i X	010	030	05L	151	451	65L	30L	000	017	035	055	216	451	701	666	010	030	045	5	005	020	037	<u>.</u>	015	030	556	9.9.9	٦ ۲	1946	19 B	956	
0 # 9 #	20		-	Т	-	។ ភ	म 9	61	0	3	ಷ ಸ	9 9	2	4	0 2	5	4 9	ы С	10 10	е 1	<b>.</b>	а 9	1.4	त्र २	ັດ ວ	n n	0	ê S	9 -	ଜୁନ	8	80. 11	6	
2000	000	0000	9450	1900	215	1220	0232	1241	360	107	141	154	1170	027	372	14.04	1176	313	326	1340	0023	1104	111	1122	015	1263	1271	216	1244	2050	2000	000	1420	
88	66	20	8	0	0	ö	5	00	0	2	2	č	č	0	5	5	5	0	ö	2	Ξ	2	8	5	5	č	8	8	ă	ö	ă	8	8	
200	200	, in	10	01	01	10	01	5	10	5	5	.10	10	10	10	10	01	10	01	10	5	01	10	5	10	5	01	01	10	01	00	00	61	
000	000	8	00	00	00	00	00	00	00	8	00	00	00	8	00	Ö	00	00	00	00	ŝ	00	8	8	8	8	8	00	00	8	00	0.0	00	
	G	Ļ	ي	ي.	<u>ب</u>	Q					_	<u> </u>	Ľ				ي	ي.	يہ.	ب	ي.	ب	<b>ب</b> ہ	<u>ب</u>	ي	ي	ب.		ي					
ĽXA ĽXA	1 CTC	5	1005	1025	1045	[13]	1251	160L	1851	SOL	2015	2032	0503	2201	1012	2051	966.0	50.07	30.25	5042	3065	1001	1017	1035	5661	5012	5025	5446	5000	751	3035	38.8F	1466	1666
00	000	20	51	67	5	23	56	03	21	74 2	76	53	5	99	17	06.4	0	5	22	Ņ	90	- 17	74 1	90	51 4	13	22	32	5	1	25	50 4	00	27
000	000	000	100	0.05(	200	037	022	0241	030.	100	013	015.0	016.	0360	036	040	017	031	032	033	035	035	010	012	012	026	027	021	0.30	020	000	000	042	042
00	200	o co	õ	Ó	0	Ċ	0	Ö	õ	Ó	õ	õ	ő	õ	ö	Ô	õ	Ō.	Ő.	0	0	0	Ö	Ċ	Ö	Ó	0	õ	Ö	0	0	Ö	0	Ó
110	200	50	10	10	5	5	10	10	10	10	Ū,	01	Ģ	0	10	10	10	5	5	10	5	5	5	01	5	5	10	10	0	10	00	00	10	5
80	000	200	8	00	8	8	00	00	00	00	8	00	00	00	00	8	00	00	00	00	3	8	00	00	00	8	00	00	00	8	8	00	ខ	8

CNVARB, CNVARB

.

. SUBROUTINE CNVARB IMPLICIT FORWARD-BACKWARD DIFFERENCING EXECUTION SUBROUTINE IMPLICIT FORWARD-BACKWARD DIFFERENCING EXECUTION SUBROUTINE WITH AN INTERNALLY CALCULATED BETA WEIGHTING FACTOR THE LONG PSEUDO-COMPUTE SEGUENCE IS REGUTAED. SINDA FORTRAN V ALL NODES RECEIVE A SUCCESSIVE POINT ITERATION RELAXATION CRIFFIA MUST RE SPECIFIED OVER-HELAXATION IS ALLOWED. THE DAMPENING FACTORS ARE ADDRESSABLE INCLUDE COMM.LIST COMMON /TITLE/H(1) /TEMP/T(1) /CAP/C(1) /SOURCE/G(1) /COND/G(1) COMMON /TITLE/H(1) /PC2/NS02(1) /KONST/K(1) /ARRAY/A(1) COMMON /FIXCON/KON(1) /YSPACE/NDIM.NTH.X(1) /SOURCE/G(1) /COND/G(1) COMMON /FIXCON/KON(1) /XSPACE/NDIM.NTH.X(1)) COMMON /FIXCON/KON(1) /XSPACE/NDIM.NTH.X(1)) COMMON /FIXCON/KON(1) /XSPACE/NDIM.NTH.X(1)) COMMON /FIXCON/KON(1) /XSACF/NDIM.NTH.X(1)) COMMON /FIXCON/KON(1) /XSACF/NDIM.NTH.X(1)) COMMON /FIXCON/KON(1) /XCM(1)).(K(1).XK(1)).(X(1).NX(1)) THOLUDE DEFFILIST ********* CONTROL CONSTANT DEFINITIONS AND NAMES CONTROL CONSTANT 1 CONTAINS THE NEW PROBLEM TIME CONTROL CONSTANT 2 CONTAINS THE TIME STEP USED END C • + + + 0000000 Ú Ú 

,

																						,									•												-						
	(TIMEND)	INI DOP)	(DTMPCA)	(DPCITR)	(HIWER)	(DAMPA)	(DAMPD)	(ATMPCA)	TINCOUT	(TIMEM)	(DTMPCC)	(ATMPCC)	(CSGMIII)	(OUTPUT)	(AKLXCA)	(DTINEL)	(DTIMEI)	(CSGMAX)	(CSGHAL)	(CSGRCL)		(LINECT)	(PAGECT)	( ARL XCC )	(LSPCS)	(ENGUAL)	( NOCOPY )				-UTEST)	U-VTEST)	(LAXFAC)																
	LEM STOP TIME	DO LOODS, INTEGER	RE CHANGE ALLOWED	ON SWITCH	E STEP	. DAMPING FACTOR	. DAMPING FACTOR	ITHRETIC TEMP. CHANGE	KEU AFTEN VANJABLES Suith find find	URLEM STARI FINE An Tecartan	NR TERMITON	CHANGE CALCULATED	THE C/56 WININUM	PUT INTERVAL	ION CRITERIA ALLOWED	UN LOUPS USEUFINICOCK Me sted	ICIT	•		TED	ON CHITERIA ALLONEU On Change Calcin aten		6ER	CHANGE CALCULATED	THERMAL LPCS.2=GENERAL	THE SYSTEM, IN - UUI	ANCE / USEN JNFOI MATRIX USERS	F CSGMIN	F DTMPCC	F ARLXCC	F ALMFCC FD FONCTANTS <b>(1.1.1.4</b> )	ING CONSTANTS (R-S+T-	RVAL FOR CINDSM									1.66	995	,					
œ	3 CONTAINS THE PROB	JULI SALA DE L'ERALINO	DIFFUSION TEMPERATU	OUTPUT EACH ITERATI	MAXIMUM ALLOWED TIM	NEW ARITHMETIC TEMP	E NEW DIFFUSION TEMP	E VAXINUM ALLOWED AR	E BACKUP SWITCH CHEC	E PRESENT THE OR PRICE	E MEAN TIME OF WEEN	ITHMETIC TEMPERATURE	17 IS RESERVED FOR	18 CONTAINS THE OUT	E ARITHMETIC RELAXAT	E FIUMBER OF RELAXALI F MINIMIM ALLOWED TI	INPUT TIME STEP IMPL	E C/SG MAXIMUM	E C/SG RANGE ALLOWED	E C/SG RANGE CALCULA	E DIFFUSION RELAXATI	F LINE COUNTER INTE	E PAGE COUNTER, INTER	ITHINETIC RELAXATION	R. OTHERMAL SPCS. 1=	E ENERGY BALANCE OF	E UESIKEU EREKGI BAL F ROCOPY SWITCH FOR	LATIVE NODE NUMBER OF	LATIVE NODE NUMBER OF	LATIVE NODE NUMBER O	LAIIVE NODE NUMMER U	CONTAIN DUMMY FLOAT	I-LINEARIZATION INTE	AT PRESENT	60 TO 889	CON(6) = 1.E+B	$CON(8) = 1 \cdot E + 8$		) CON(11) = 1.F+8	(13)) 60 70 990	) 60 T0 998	CON(19).LE.0.) 60 TO ) 60 TO 904	CON(26) LE.0.) 60 TO	166 10 991					
CNVARB . CNVAR	CONTROL CONSTANT	CONTRUE CONSTANT	TOTAL THE THE THE	CC7 CONTAINS THE	CCB CONTAINS THE	CC9 CONTAINS THE	CC10 CONTAINS TH	CC11 CONTAINS TH	CCL2 CONTAINS TH	CCIS CONTAINS TH	TT SULATION ALING TH	CC16 CONTAINS AR	CONTRCL CONSTANT	CONTRUL CONSTANT	CC19 CONTAINS TH	CC20 CCNIAINS IN	CC22 IS FOR THE	CC23 CONTAINS TH	CC24 CONTAINS TH	CC25 CONTAINS TH	CC26 CONTAINS TH	CC28 CONTAINS TH	CC29 CONTAINS TH	CC30 CONTAINS AR	CC31 IS INDICATO	CC32 CONTAINS TH	CCGG CONTAINS IN	CC35 CONTAINS RE	CC36 CONTAINS RE	CC37 CONTAINS PE	CC38 CONTAINS PE	84-64-94-54-4433	CC49 IS THE QUAS	CC50 IS NOT USED	ENU TE(KON(S)   F.O)	IF (CON(6) .LE.0.)	IF (CON(8) . LE . 0.)	IF(CGN(9).LE.0.)	TF (CON(11) - FE-0-	IF (CON (3) .LE.CON	IF (CON(IB) .LE.U.	IF (NNA.61.0.AND.	IF [NND.GT.0.AND.	IF (KON (31) . NE . 1)	PASS = -1.0	NLA = NUIM NN - NIID+1	NNC = MID+NNA	IEI = WTH TEO = WILLEWIT	IE3 = IE2+NND
	ų,	<b>.</b>	ي ر	e S	U	U	U	Ú (	0	0	<u>ب</u> د	<u>ي</u> د	, c	J	U i	υı	ي ر	ن ر	U	U	U L	ي ر در	ງ.ບ 	U	U I	0.0	U C	<u>ن</u> ر	ن ز	Q (	0	ن ر	ο U	U															
		* * *		ő	÷.	<b>*</b>	*6	*0	<b>*</b> i	# ·	* * 5 g	4 4 N 0	*6	*6	<b>.</b>	# 1 0 0	• <b>*</b>	***	*6	*0	* : 0 0	+ + ^ 0	÷ 6		<b>*</b>	ð 6	* 1 5 0	• •	6	6	* (	* * 7 0	*6	<b>*</b>	**		12*	• •	***	* 9 • •	17*	*81 19*	1 A A A A A A A A A A A A A A A A A A A	21*	*20 55	* 1 N 0	5 C1	264	284
	00113	C1100	21100	01100	00113	00113	00113	00113	61100	00113	\$T100	CT100	00113	00113	00113	00113		00113	00113	00113	00113		00113	00113	00113	00113	51100	00113	00113	00113	00113	21100	00113	00113	00114	00117	00121	00123	22100	00131	00133	00135	00107	00143	00145	00149	00150	14100	00153

•

.

•

B - 28

•; 

•,,

CNVARB + CNVARB

•

<b>.</b>
,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,

.

.

.

-

	•			
	MU MU MU MU MU MU MU MU MU MU MU MU MU M	* NEW * NEW * NEW * * * * * * *	N N N N N N N N N N N N N N N N N N N	* NEW * NEW * NEW
	MIN 1940,1045), NTYPE		•	
	CONTRIBUTION AND THE CSG THE CAPACITANCES 1+1).50.0) 60 TO 2000 502(J2)) 2(J2)) 02(J2)) 1015(J2)) 1015(J2) 1015(J2) 1015(J2) 1015(J2) 1035(J2))	2(J2)) 02(J2)) A(LA),XK(LK),C2) A(LA),XK(LK),C2) 22(J2)) 02(J2)) 7(1),A(LA+1),XK(LK),C(1) 7(1),A(LA+1),XK(LK),C1)	7 (1), A (LA+1), XK (LK), C2) 7 (1), A (LA+1), XK (LK), C1) 2 (J2)) 2 (J2)) 2 (J2)) CON(14), A (LA), XK (LK), C(1)	1+1)) <b>.E0.0) GO TO 5000</b> 502(J2)) 2(J2)) 02(J2))
CNVARB, CNVARB	CALL OUTCAL PASS = 1.0 60 TO 10 80 TO 10 80 C = 1.6+8 80 CALCULATE THE OLD 10 100 I = 1.100 10 100 I = 1.100 17 F(FU01/11.05010) 17 F(FU01/11.05010) 17 F(FU01/11.05010) 17 F(FU01/11.05010) 17 F(FU01/11.05010) 17 F(FU01/11.05010) 10 C = 100000000000000000000000000000000	LA = FLD(5,17,150 LK = FLD(22,14,NS) CALL D1()1WM(T(T)) GO TU 1998 1015 C1 = XK(LK) $*XK(LA)$ 1020 CALL D101WM(T(T)). LA = FLD(5,17,NS0 LA = FLD(22,14,NS) LA = FLD(22,14,NS) LA = FLD(22,14,NS) LA = FLD(22,14,NS) LA = FLD(23,17,NS0) 1025 CALL PLYAWM(A(LA)) 1030 CALL PLYAWM(A(LA)) 1030 CALL PLYAWM(A(LA)) 1032 LA = J2+1 1032 LA = J2+1 1032 LA = J2+1 1032 LA = FLD(5,17,NS0)	LK = FLD(22:14,NS) GALL PLYAWM(A(LA) GALL PLYAWM(A(LA) GALL PLYAWM(A(LA) J2 = J2:1 LA = FLD(2:14,NS) LA = FLD(2:14,NS) LK = FLD(2:14,NS) LA = FLD(2:14,NS) LS = J2:1 1993 C(1) = C(1) T C(1) T C(1) T C(1) = C(1) T C(1) T C(	IF (FLD(4,1,NS01(4) NTYPE = FLD(0,5,N LA = FLD(5,17,NS0 LK = FLD(22,14,NS
	C C	20000000000000000000000000000000000000	20098344444444444444444444444444444444444	
	000025555 00025555 00025555 00025555 00025555 00025555 00025555 00025555 0002555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 00055555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 0005555 00055555 00055555 00055555 00055555 00055555 00055555 00055555 00055555 00055555 00055555 00055555 00055555 00055555 00055555 00055555 00055555 00055555 00055555 00055555 00055555 00055555 00055555 000555555	00000000000000000000000000000000000000	00000000000000000000000000000000000000	00344 00344 00347 00350

B - 30

•

•

	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	330 330 UUI UUI VZ 22 *** ***	2 2 8 2 2 8 2 2 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	2 8 8 2 8 2 8 2 8 2 8 2 8 2 8 2 8 2 8 2
CAVARBICAVARD 60 T0 (4005.4010.4015.4020.4025.4030.4035.4040.4030). NTYPE 4005 0(1) = XK(LK)+A(1) 60 T0 499 4010 01 = 0.0 4012 CALL DIDIWM(T(1).A(LA).XK(LK).02) 60 T0 4998 0.0.0	4017       CALL       D1D1wm(CON114).A(LA).XK(LK).02)         60       T0       499         4020       CALL       D11wm(CON144).A(LA).XK(LK).01)         4022       J2       - J2+1         LA       FLD15.17.NS02(J2))       LA         LA       FLD12.14.NS02(J2))       LA         60       T0       4017         4025       01       214.NS02(J2))         4030       CALL       D1D1WM(CON144).A(LA).XK(LK).01)	J2 = J2+1 J2 = FED(5,17,MSQ2(J2)) LK = FED(5,17,MSQ2(J2)) LK = FED(22,14,MSQ2(J2)) Q2 = XK(LK)*XK(LA) Q0 TO 4998 4035 CALL DIDIWM(COU(14),A(LA),XK(LK),Q1) 4035 CALL DIDIWM(COU(14),XK(LK),Q1) 4035 CALL DIDIWM(COU(14),XK(LK),Q1) 4035 CALL DIDIWM(COU(14),XK(LK),Q1) 4035 CALL DIDIW(COU(14),XK(LK),Q1) 4035 CALL DIDIW(COU(14),XK(LK),Q1) 4035 CALL DIDIW(COU(14),XK(LK),Q1) 4035 CALL DIDIW(COU(14),XK(LK),Q1) 4035 CALL DIDIW(COU(14),XK(LK),Q1) 4035 CALL DIDIW(COU(14),XK(LK),Q1) 4055 CALL DIDIW(COU(14),XK(LK),Q1) 4055 CALL DIDIW(COU(14),XK(LK),Q1) 4055 CALL DIDIW(COU(14),XK(LK),Q1) 4055 CALL DIDIW(COU(14),XK(LK),Q1) 4055 CALL DIDIW(COU(14),XK(LK),XK(LK),Q1) 4055 CALL DIDIW(COU(14),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK(LK),XK	<pre>4040 0! = XK(LK)*XK(LA) 60 T0 4037 4999 0(1) = 01+02+0(1) 4999 0(1) = 01+02+0(1) 4999 0 = 00 4010 = 0.0 60 C01/TNUE ENC 60 0.0 70 J1 = J1+1 LG = FLD(5,16,0501/J1) LG = FLD(5,16,0501/J1) LG = FLD(22,14,0S01/J1) LG = FL</pre>	<b>2005</b> TM = (T(1)+T(L/A))/2.0 2007 CALL DIDI#M(TM.A(LA).XK(LK).6(LS)) 60 T0 2999 2010 TM = T(1) 2015 CALL UIDI#M(T(1).A(LA).XK(LK).61) 2017 J2 = J2+1 LA = FLD(25.17.NSO2(J2)) LA = FLD(25.14.NSO2(J2)) LA = FLD(25.14.NSO2(J2)) LA = FLD(25.14.NSO2(J2)) LA = FLD(25.14.NSO2(J2)) LA = FLD(25.14.NSO2(J2)) 2010 D1 2093 2010 2010 2017 2020 G1 = XK(LK).KK(LA).XK(LK).62) 2020 G1 = XK(LK).KK(LA) 2020 G1 = XK(LK).KK(LA) 2020 G1 = Z(1).WW(T(1).A(LA).XK(LK).61) 2020 CALL UIDIWW(T(1).A(LA).XK(LK).61)
				1007
00354 00355 00355 003554 003554 003555	00360 00361 00365 00365 00365 00365 00365 00370	00372 00373 00374 00377 00377 00400 00402 00402	00000000000000000000000000000000000000	00000000000000000000000000000000000000

÷

•

•

CNVARB . CNVARB

¥ -

B - 31

. •

			ara 1997 - Santa S	an a	n en
				- - -	
<u>)</u> .					
and the second					
24 34 Q	*3	FQ 330			
					, i
)		~			
	(re)) 61)	61) (10)	29		
	1910	(K (LK	•19. 9	75. CRIA	
	× × ×	+1).) . XK(		12) (12) (4) (4) (11) (11)	ų
	+1) •1	A(LA- LA+1	) 6(1 C ^T OR		) 1.0 Ankii
		TA)•.	62) 62)	ABIL 7 11 4 4 4	3 0 10 10 1 10 1
2512	TW: TW: TW: TW: TW: TW: TW: TW: TW: TW:	•T(L •T(L •T(L •T(L •)		1)), ++72* V*(T V*(T V*(T V*(T 0 10	P 0 99 NETA KES
VARB NSQ K(LA	(LA)	(LA) (LA) (LA) (LA) (LA) K(LA) K(LA)	-/61 01(J 1460 A)+4	2)+6 2)+6 2)+6 NIMU 01 NIMU	TSTE 60 1 00/1 • 0) • 1) NT VI NT VI
9 CI 4	M (A M (A M (A M (A	24 X X X X X X X X X X X X X X X X X X X	A A A A A A A A A A A A A A A A A A A	T C C C C C C C C C C C C C C C C C C C	
VAR LD:	1222299	20112220122012000000000000000000000000		10 1 1 C 0 0 1 C 0 C 0 C 0 C 0 C 0 C 0 C	
					ONTINO ONTINO CRETAO ONVEN
2222	20020000000000000000000000000000000000	223885552582825	3 8 8 88 (TRESCENCE	E S S	2004848001
i	50 50 50 50 50 50	50 50 50	C 336 50 50	, u	а 0
				5 + + + + + + + + + + + + + + + + + + +	2 + + + + + + + + + + + + + + + + + + +
				,	~~~~
0000					

.

•

. *

**B** - 32

•

. .

.

5 450 •

CNVARB , CNVARB

•

11000000000000000000000000000000000000	44444444444444444444444444444444444444	<u>.</u> . <u>.</u> . <u>.</u>	65 105 115 115 125 125 150 150 155 155	T(1) = T(1)+460. NON KELAX THE NETWORK BY SUCCESSIVE POINT AND EXTRAPOLATION UJ = 0. UJ = 0.1.1. TE (1)-07.1.1 60 T0 110 TE (1)-07.1.1 60 T0 110 TE (1)-07.1.1 60 T0 110 TE (1)-07.1.1 60 T0 110 TE = 1E21 0.1.2.1.1 60 T0 110 TE = 1E21 0.1.2.1.1 60 T0 110 0.1.2.1.1.1 60 T0 110 0.1.2.1.1.1 10 TE = 1E015,16,MSQ1(JJ)) 0.1.2.1.1.1 10 1.1.2.1.1.1 10 1.1.2.1.1.1 10 1.1.2.1.1.1 10 1.1.1.1.1.1 10 1.1.1.1.1.1 10 1.1.1.1.1.1.1 10 1.1.1.1.1.1 10 1.1.1.1.1.1 10 1.1.1.1.1.1 10 1.1.1.1.1.1 10 1.1.1.1.1.1 10 1.1.1.1.1.1.1 10 1.1.1.1.1.1 10 1.1.1.1.1 10 1.1.1.1.1 10 1.1.1.1.1 10 1.1.1.1.1 10 1.1.1.1.1 10 1.1.1.1.1 10 1.1.1.1.1 10 1.1.1.1 10 1.1.1 10
00633 00634 00637 00637 00637 00637 00637 00641 00641	1 4 5 9 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	U	165	RLXD = T1 KK1 = 1 CONTINUE GO T0 (180,180,170), JJ FERFORM LINEAR EXTRAPOLATION ON THE ERROR FUNCTION CURVE JJ = 0 If (koh(34),NC.n) 60 T0 180 U0 175 I = 1.0140 U0 175 I = 1.0140
00646	192+	Ų		SEE IF THE EXTRAPOLATION IS ALLOWABLE

•

.

60
άĽ.
>
Z
U
8
Ba
ARB
VARB
VARB
NVARB
CNVARB

•

and a second secon

 $\mathcal{I}$ 

	23 A 1997 1997 1997 1997 1997 1997 1997 199	BBQ BB BDUI DUU N2+ V2 ** *	N 1 *
<pre>Ff(x(LE3).6E.0.) G0 T0 175 LIMIT THE EXTRAPOLATION Ff(x(LE3).LT10.) X(LE3) = -10. LE2 = IE2+I T(1) = X(LE3).x(LE2)+(1.0-X(LE3))*T(I) T(1) = X(LE3).x(LE2)+(1.0-X(LE3))*T(I) F(NNA.LE.0) G0 T0 220 JJJ = J1 JJZ = J2 JJJ = J1 JJZ = J2 JJZ = J2 JZ = J2 J</pre>	Q(L) = XK(LK)+Q(L)         G0 T0 5999         Q1 = 0.0         CALL DIDIWM(T(L),A(LA),XK(LK),02)         G0 T0 5998         Q1 = 0.0         G1 = 0.0         G2 T0 5998         CALL DIDIWM(CON(14),A(LA),XK(LK),Q2)         JJ2 = JU2+1         JJ2 = JU2+1         LA = FLD(5,17,1502(JJ2))         LA = FLD(22,14,NSQ2(JJ2))         LA = FLD(52,14,NSQ2(JJ2))	01 = XK(LK)*XK(LA) 60 TU 5022 CALL DIDIWM(CON(14).A(LA).XK(LK).01) JJ2 = JJ2+1 LA = FLD(5.17.NS02(JJ2)) LA = FLD(22.14.NS02(JJ2)) 02 = XK(LK)•XK(LA) 02 = XK(LK)•X(LA) 03 TO 5998 CALL DIDIWM(CON(14).A(LA).XK(LK).01) JJ2 = JJ2+1 LA = FLD(22.14.NS02(JJ2)) LA = FLD(22.14.NS02(JJ2))	G0 T0 5012 01 = XK(LK)*XK(LA) G0 T0 5037 12 = JJ2+1 JJ2 = JJ2+1 CONTINUE END 05UM = 0(1) L = 1 JJ1 = JJ1+1 L = 1 L = 2,11+11561(JJ1) F(K1.6T.2) 60 T0 4000 INCLUUE VR62.LIST CHECK FOR PADIATION CONDUCTOR
C 1175 230 230	5015 5012 5015 5015 5027 5020 5022	5025 5030 5035 5035	5040 5998 5999 5999 5999 5999 5999 5000 5000
1444 1644 1644 1644 1644 1644 1644 1644	2022 2022 2022 2022 2022 2022 2022 202	207** 207** 207** 207** 207**	2074 2074 2074 2014 2015 2008 2115 2115 2115 2115 2115 2115 2115 211
00000000000000000000000000000000000000	00705 00705 007104 007114 007114 007115 007715 007715 007715 007715 007715	00723 00724 00724 00725 00725 00725 00733 00733 00733 00733	00735 00737 00741 00741 00741 00744 00744 00744 00750 00750 00752

**B - 3**4

8
Ω.
<u> </u>
2
Ζ.
Q.
8
œ
<u>ح</u>
~
0

•

•

3330 3330 3332 3332 3332 3332 3332 3332	334 UUI 22* **	330 UUI ZZ# ##	330 UUI VZ* **	330 99 97 97 97 97 97 97 97 97 97 97 97 97	
<pre>IF(FLD(2*1.NSQ1(JJ1)).EQ.0) GO TO 4000 NTYPE = FLD(0.5.NSQ2(JJ2)) LA = FLD(5*17.NSQ2(JJ2)) LK = FLD(22.14.NSQ2(JJ2)) GOTO(3005.3010.3015.3020.3025.3030.3035.3040.3045.3050.3055. 5 3060.3065.3010.3015.3020.3025.3030.3035.3040.3045.3050.3055. 5 TM = (T(L)+T(LTA))/2.0 7 TM = (T(L)+T(LTA))/2.0 6 TO 3999 6 TO 3999 6 TO 3007 6 OT 3007 6 OT 3007</pre>	UJZ = JUZ+17/NSQ2(JJZ)) LA = FLD(22/14/NSQ2(JJZ)) LK = FLD(22/14/NSQ2(JJ2)) CALL DIDIMM(T(LTA).A(LA).XK(LK).62) GO TO 3998 GO TO 3917 GO TO 3017 CALL DIDIMM(T(L).A(LA).XK(LK).61)	ULZ = JJZ+1 LA = FLD(5+17+NSQ2(JJ2)) LK = FLD(25+14+NSQ2(JJ2)) G2 = XK(LK)+XK(LA) G0 T0 3999 TM = T(L)+T(LA)+TM.A(LA+1)+XK(LK).6(LG)) G0 T0 3999 TM = T(L) G0 T0 3032 CALL PLYAWM(A(LA)+T(L)A(LA+1)+XK(LK).6(L))	ULZ = JUZ+1 LA = FLD(5+17+NSQ2(JJ2)) LK = FLD(22-14+NSQ2(JJ2)) CALL PLYAWM(A(LA)+T(LTA)+A(LA+1)+XK(LK)+G2) GO TO 3998 60 TO 3942 60 TO 3042 1 CALL PLYAMM(A(LA)+T(L)+A(LA+1)+XK(LK)+G1) JUZ = JUZ+1	LA = FLD(5,17,HSQ2(JJ2)) LK = FLD(22,14,NSQ2(JJ2)) G2 = XK(LK)*XK(LA) G3 T0 3996 T = (T(L)+T(LTA))/2.0 CALL C22LW(TN,CON(14),A(LA),XK(LK),G(LG)) G0 T0 3999 T = T(LTA) G0 T0 3032 G0 T0 3032	<pre>6 (LG) = 1./(1./G1+1./G2) 1 F(FLD(3.1.NSQ1(JJ1)).EQ.1) G(LG) = G1+G2 1 J2 = J2+1 continue continue fun t1 = T(1)+4.60.0 T1 = T(L1)+4.60.0 F2 = T(LA1)+4.60.0 F2 = T(LA1)+4.60.0 F2 = F(LA1)+4.60.0 F2 = F(LA1)+4.60.0 F2 = F(LA1)+4.51.0 F2 = F(LA1)+7.51.51.0 F2 = F(LA1)+7.51.51.5 F2 = F(LA1)+7.51.51.0 F2 = F(LA1)+7.51.51.0 F2 = F(LA1)+7.51.51.0 F2 = F(FA1)+7.51.51.0 F2 = F(FA1)+7.51.0 F2 = F(FA1)+7.</pre>
3006	3025	3030 3040 3040	30 00 30 00 30 00	3055 3066 3066	9996 9999 99994
<b>8</b> 88888888888888888888888888888888888					
0007652 007557 007557 007552 007552 007552 007554 007554 007555 007554 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 007555 0007555 0007555 0007555 0007555 0007555 0007555 0007555 0007555 0007555 00000000	00771 00773 00773 00773 00775 00775 00775		01014 01015 01015 01017 01020 01021 01022 01022	01025 01025 01027 01027 01035 01035 01035 01035	010441 010441 010441 0104441 0104441 0104441 0104441 0106441

.

÷

.

•

• •

3			
Č	DUCTOR TO 185 ERATURE 1 15	RELAXATION CRITERIA WAS MET TO 225 ELAXATION CRITERIA WAS MET TO 245 TOPLIN CHANGES WERE TOO LARGE CHANGES WERE TOO LARGE ON(6)/TCGD TO 265 ON(6)/TCGD TO 275 ON(11)/TCGA AGK TO FARENHEIT	AND RELAXATION CHANGES
CNVARB , CNVARB	<b>190 GV = G(LG)</b> <b>195 GSUM = GSUM+6V+T2</b> <b>195 GSUM = GSUM+6V+T2</b> <b>CHECK FON NEGATIVE CON</b> <b>16 (NSO1(JJ).6T.0) GO</b> <b>17 (NSO1(JJ).6T.0) GO</b> <b>17 (NSO1(JJ).6T.0) GO</b> <b>17 (NSO1(JJ).6T.0) GO</b> <b>17 (NSO1(JJ).6T.0) GO</b> <b>17 (NSO1(JJ).65.0) GO</b> <b>1</b>	<b>235</b> F(1) = T(1)+460. F(1) = T(1)+460. F(1) = T(1)+460. F(1) = T(1)+460. F(1) = T(1)+460. F(1) = T(1)+460. F(1) = F(1) = 0.0 ZEC IF (RLX).LE.CON(20)) GO ZED IF (KON(7).EG.05) GO TO ZALL OUTCAL COLL OUTCAL COLUCERT TEMPERATURE COLL OUTCAL COLUCERT TEMPERATURE COLL OUTCAL COLUCERT TEMPERATURE COLUCERT TEMPERATURE COLUCERT TEMPERATURE COLUCERT TEMPERATURE COLUCERT TEMPERATURE COLUCERT TEMPERATURE COLUCERT TEMPERATURE COLUCERT TEMPERATURE COLUCERT TEMPERATURES COLUCERT TEMPERATURES COLUCERT TEMPERATURES	CON(15) = 1117-700 STORE THE TEMPERATURE CON(15) = TCGD CON(16) = TCGA
	U U		Û,
	<pre>************************************</pre>	**************************************	2794
:			01171
.)			

.• 1

· · · · ·

*

B - 36
CNVARB CNVARB

CON(27) = RLXD If(RLXA.6T.RLXD) GO TO 285 KK2 = KK1	RLXA = RLXD	285 KON(37) = KK2 CON(30) = RLXA	CON(12) = 0.0	CALL VARBL2	L CHECK THE DACKOF 2415CH	C ADVANCE TIME	CON(13) = CON(1)	CURCK FOR TIME TO PRINT	IF (TSUM. GE.CON(18)) GO TO 290	C CHECK FOR PRINT EVERY ITERATION	IF(KON(7).NE.0) CALL OUTCAL	C TDY TO FUEN THE DUTPUT INTERVALS	290 TPRIMT = TPRINT+TSUM	CALL OUTCAL	C IS TIME GREATER THAN ENU COMPUTE TIME		NDIM = MLA	RETURN	990 WRITE (6,880)	60 10 1000 DD1 4017776_681)		993 WRITE(6.883)		994 WK(IEKOF884J NULM) Go to long	995 WRITE(6,885)	60 10 1000	996 WRITE(6,886)	60 10 1000 997 WRITE (6,857)	60 TO 1000	998 WRITE(6,888)	60 T0 1000 000 WDTTE/6.4001	1000 CALL OUTCAL	CALLEXIT	BBU FORMAT (29H TRANSIENT TIME NOT SPECIFIED)	BBI FORMAT (4.5H CNVARB REQUIRES LONG PSEUDO-COMPUTE SEQUENCE)	885 FORMATICAN RELAVITON UNICATA NOT MELT 885 FORMATIC24H CSGWIN ZERO OR NEGATIVED	884 FORMAT(18,204 LOCATIONS AVAILABLE)	BUS FORMAT(10H NO DRLXCA)	BBG FORMAT(IOH NO DTIMEI)	BB7 FORMATCION NU AMERCA) Ama Formatcion no cutput interval)	BAG FURMAT (GH NO NLOOP)	END	UNIVAC 1108 FURTRAN V COMPILATION. 0 +DIAGNOSTIC+ MESSAGE(S)	SYMBOLIC	JE NELOCAIABLE
				<u>н</u> .		*									н.								*				<u>н</u> .									* *		1		<b>x</b> - 4			01-10		õ
281-2	283	285	2861	287	289.4	2903	162	202	567	2951	5962	1600	2995	300		200	100	305	305	202	000	3101	115	110	3140	315	910	318	319.	320	321	2220	3241	325	326	120	329	330	100	555	) 3 ) M	235	0~13	AKB	AKB
1175	1201	1202	1204	1205	1206	1206	1210	11211	1212	1212	12214	01210	1117	1220	1220	1221	11224	1225	11226	02210	1233	11234	1236	1221	1243	1245	11246	1221	1253	1254	11256	1261	1262	11263	1264	11266	1267	1270	11271	1272	11274	01275		ž	CN

• 7145

•

C.

· ·

Page

٠

÷.,

CINDSS	••••••	C-2
CINDSL	• • • • • • • • • • • • • • • • • • • •	C-11
CINDSM	••••••	C-20

ŝ
ŝ
0
Z
H
Û
S
ŝ
0
ź
-
U.
_

DIW FOR.* CINDSS.CINDSS UNIVAC 1108 FORTRAN V ATHENA VERSIOH 131K-10D CREATED ON 20 AUG 70 THIS COMPILATION WAS DONE ON 09 JUN 70 AT 14:00:06

SUBROUTINE CINDSS ENTRY POINT 003062

STORAGE USED (BLOCK, NAME, LENGTH)

003077	MP 000106	AR 000045	000000	000000	00001	10000	00001	10000	00001	0001	10000	10000	10000	00001	00003	00010	
*CODE	*CONST+TE	*SINPLE V	* APRAYS	*BLANK	TITLE 0	TEMP 0	CAP 0	SOUNCE 0	COND 0	PCI 0	PC2 0	KONST 0	ARRAY 0	FIXCON 0	XSPACE 0	DIMENS 0	
0001	0000	0002	000	0002	0006	0001	0010	0011	0012	0013	4100	0015	001.6	0017	00500	0021	•

EXTERNAL REFERENCES (BLOCK, NAME)

C - 2

•

VARELI	DIDIWM	DZD14M	NONLIN	VARUL2	EXIT	NERR25	NWDUS	N102\$	NER105
0022	0024	0026	0027	0030	0031	0032	0.0.33	10034	0035

STORAGE ASSIGNMENT FOR VARIABLES (BLOCK. TYPE. RELATIVE LOCATION. NAME)

0002 R 00014 DM 0002 R 000030 62 0002 I 000034 L/2 0002 I 000034 L/1 0002 I 000034 L/1 0002 I 000004 N/4 0002 R 000002 N/4 0002 R 000042 N/ 0002 R 000042 N/ 0002 R 000042 N/

0002 R 000015 DD 0002 R 000027 61	0002 I 000036 JJI	0017 I 000000 KON	0002 I 000016 LE	0021 1 000007 LS02	0021 I 000003 NGT	0021 1 000000 NND	0002 I 000017 NTYPE	0011 R 00000 G	0002 R 000013 RLXA	0002 R 000032 T2
0017 R 000000 CON 0002 R 000033 GV	0002 I 000003 IE	0015 I 000000 K	0002 I 000005 LAX	0024 I 000006 LSOI	0020 I 000000 NDIM	0002 I 000002 NNC	0020 I 000001 WTH	0002 R 000000 PASS	0002 R 000023 02	0002 R 000031 T1
0010 R 000000 C 0002 R 000040 65UM	0002 I 000007 I	0002 I 000011 J2	0002 I 000020 LA	0002 I 000021 LK	0021 I 000004 NCT	0021 1 00000 NNA	0014 I 000000 NSO2	0002 I 000042 N2	0002 R 000022 01	0002 R 000026 TM
0016 R 000000 A 0012 R 000000 6	0006 R 000000 H	0002 I 000010 JI	0002 I 000041 L	0002 I 000024 LG	0021 I 000005 NAT	0002 I 000001 NN	0013 I 000000 1500	COU2 I 000035 NI	0002 R 000043 00UT	0007 R 005000 T

:1 FC8 71

Ś

n. Naria

	0002543 1151 000145 1616 000254 20151 000774 20151 0011077 20151 0011077 20151 0011077 30051 0012015 2116 0022104 30210 0022554 40151 0022554 40151 0022554 40151 0022554 40151 002551 6576 001502 50751 001502 50751 0000000000000000000000000000000000	
	<b>6000000000000000000000000000000000000</b>	
	002620 155L 002620 155L 000651 2010L 0002556 1998L 0002503 2036 001212 2055L 001212 2055L 001223 4012L 002235 5000L 001431 4345 001431 4345 001431 4345 001431 4345 001431 4345 001713 6001L 001713 6001L 001713 6001L 001713 6001L	REAL COD COD COD COD COD COD COD COD COD COD
		COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6(1) COND/6
CINDSS, CIN7SS	0001         000104         10           0001         000125         15           0001         000125         15           0001         000125         15           0001         000125         15           0001         000125         0001           0001         000125         0001           0001         000126         205           0001         001024         205           0001         001027         2001           001160         2055         0001         00127           0001         001160         2055         0001         00127           0001         001160         2055         0001         00127         2040           0001         001216         2094         0001         01214         2991           0001         002215         3010         0001         01216         2091           0001         002215         3045         0001         01216         2091           0001         002215         3045         0001         01216         2091           0001         002215         0001         01265         2091           0011 <td< td=""><td>UGROUTINE CINESS TEADY STATE EXECUTION SUBPOUTINE FOR SINDA FE SHORT PSEUDO-COMPUTE SEQUENCE IS REQUIRED FF SHORT PSEUDO-COMPUTE SEQUENCE IS REQUIRED FF USION NODES RECEIVE A BLOCK ITERATION VER-RELAXATION IS ALLOWED. THE DAMPING FACTORS AND VER-RELAXATION IS ALLOWED. THE DAMPING FACTORS AND VER-RELAXATION IS ALLOWED. THE DAMPING FACTORS AND VERLUDE COMM.LIST OWMON /FITLE/NIL) /CZP/SQL1) /SOURCE/01) / OWMON /FITLE/NIL) /TEMP/TIL) /CAP/CIL) /SOURCE/01) / OWMON /FITLE/NIL) /TEMP/TIL) /CAP/CIL) /SOURCE/01) / OWMON /FITCON/KONIL) /YSPACE/NDIM.NTH.X(1) OMMON /FITCON/KONIL) /YSPACE/NDIM.NTH.X(1) OMMON /FITCON/KONIL) /XSPACE/NDIM.NTH.X(1) OMMON /FITCON/KONIL) /XSPACE/NDIM.NTH.X(1) OMMON /FITCON/KONIL) /XSPACE/NDIM.NTH.X(1) OMMON /FITCON/KONIL) /XSPACE/NDIM.NTH.X(1) OMMON /FITCON/KONIL) /XSPACE/NDIM.NTH.X(1) OMMON /FITCON/KONIL) /XSPACE/NDIM.NTH.X(1) NCUDE DEFF.LIST OMMON /FITCON/KONIL) /XSPACE/NDIM.NTH.X(1) NCUDE DEFF.LIST OMTROL CONSTANT DEFINITIONS AND NAMES ATTATA ONTROL CONSTANT 2 CONTAINS THE TIME STEP USED ONTROL CONSTANT 2 CONTAINS THE TIME STEP OF ONES ONTROL CONSTANT 2 CONTAINS THE TIME STEP OF OF ONES ONTROL CONSTANT 3 CONTAINS THE NEW DAMPING FACTOR CONTAINS THE MAXIMUM ALLOWED ATTIME FACTOR CONTAINS THE MAXIMUM ALLOWED ATTIME START CONTAINS THE NAXIMUM ALLOWED ANTH. TIME CONTAINS THE PRESENT TIME OF ONTAINS THE PRESENT TIME OF ONTAINS THE PACTOR CONTAINS THE PACKUP SWITCH TEMPERATURE CHANGE CALCULA CONTAINS THE PACKUP SWITCH CHECKED ATTOR CONTAINS THE PACKUP SWITCH CHANGE CALCULA CONTAINS THE PACKUP SWITCH CHANGE CALCULA</td></td<>	UGROUTINE CINESS TEADY STATE EXECUTION SUBPOUTINE FOR SINDA FE SHORT PSEUDO-COMPUTE SEQUENCE IS REQUIRED FF SHORT PSEUDO-COMPUTE SEQUENCE IS REQUIRED FF USION NODES RECEIVE A BLOCK ITERATION VER-RELAXATION IS ALLOWED. THE DAMPING FACTORS AND VER-RELAXATION IS ALLOWED. THE DAMPING FACTORS AND VER-RELAXATION IS ALLOWED. THE DAMPING FACTORS AND VERLUDE COMM.LIST OWMON /FITLE/NIL) /CZP/SQL1) /SOURCE/01) / OWMON /FITLE/NIL) /TEMP/TIL) /CAP/CIL) /SOURCE/01) / OWMON /FITLE/NIL) /TEMP/TIL) /CAP/CIL) /SOURCE/01) / OWMON /FITCON/KONIL) /YSPACE/NDIM.NTH.X(1) OMMON /FITCON/KONIL) /YSPACE/NDIM.NTH.X(1) OMMON /FITCON/KONIL) /XSPACE/NDIM.NTH.X(1) OMMON /FITCON/KONIL) /XSPACE/NDIM.NTH.X(1) OMMON /FITCON/KONIL) /XSPACE/NDIM.NTH.X(1) OMMON /FITCON/KONIL) /XSPACE/NDIM.NTH.X(1) OMMON /FITCON/KONIL) /XSPACE/NDIM.NTH.X(1) OMMON /FITCON/KONIL) /XSPACE/NDIM.NTH.X(1) NCUDE DEFF.LIST OMMON /FITCON/KONIL) /XSPACE/NDIM.NTH.X(1) NCUDE DEFF.LIST OMTROL CONSTANT DEFINITIONS AND NAMES ATTATA ONTROL CONSTANT 2 CONTAINS THE TIME STEP USED ONTROL CONSTANT 2 CONTAINS THE TIME STEP OF ONES ONTROL CONSTANT 2 CONTAINS THE TIME STEP OF OF ONES ONTROL CONSTANT 3 CONTAINS THE NEW DAMPING FACTOR CONTAINS THE MAXIMUM ALLOWED ATTIME FACTOR CONTAINS THE MAXIMUM ALLOWED ATTIME START CONTAINS THE NAXIMUM ALLOWED ANTH. TIME CONTAINS THE PRESENT TIME OF ONTAINS THE PRESENT TIME OF ONTAINS THE PACTOR CONTAINS THE PACKUP SWITCH TEMPERATURE CHANGE CALCULA CONTAINS THE PACKUP SWITCH CHECKED ATTOR CONTAINS THE PACKUP SWITCH CHANGE CALCULA CONTAINS THE PACKUP SWITCH CHANGE CALCULA
	00 XX 334 1651 10 XX 10 XX 10 2000 10 2001 10 2001	
<b>N</b>	80000000000000000000000000000000000000	

•

C - 3

۰.

.

CONTROL CONSTANT IS CONTAINS THE OUTPUT INTERVAL COUPEND CONTAINS THE ARTHMETIC RELAXATION CRITERIA ALLOWED (ARLXCA) CC29 CONTAINS THE MINIMUM ALLOWED TIME STEP CC22 CONTAINS THE NUMBER OF RELAXATION LOOPS USED.INTEGER (LOOPCT) CC22 CONTAINS THE C/SG MAXIMUM CC22 CONTAINS THE C/SG MAXIMUM CC23 CONTAINS THE C/SG MAXIFUME CC23 CONTAINS THE LIFE C/SG MAXIFUME CC23 CONTAINS THE LIFE C/SG MAXIFUME CC23 CONTAINS THE LIFE COUNTER. INTEGER CC23 CONTAINS THE LIFE CONTER. INTEGER CC23 CONTAINS THE DIFFUSION RELAXATION CHANGE CALCULATED CC23 CONTAINS THE DIFFUSION RELAXATION CHANGE CALCULATED CC23 CONTAINS THE PAGE COUNTER. INTEGER CC23 CONTAINS THE PAGE COUNTER. INTEGER CC23 CONTAINS THE PAGE COUNTER. INTEGER CC23 CONTAINS THE PAGE CONTER. INTEGER CC23 CONTAINS THE PAGE CO IF(KON(5).LE.U) GO TO 999 IF(CON(9).LE.0.) CON(9) = 1.0 IF(CON(10).LE.0.) CON(10) = 1.0 IF(NNJ.GT.0.AND.CON(10).LE.0.) GO TO 998 IF(NND.GT.0.AND.CON(29).LE.0.) GO TO 997 IF(KON(31).NE.0) GO TO 994 60 T0 15 10 CON(1) = CON(13)+CON(18) 1 F(CON(1) = CON(13)+CON(18)) CON(14) = CON(1)+CON(13) CON(14) = CON(1)+CON(13) CONPUTE STEADY STATE TEMPERATURES LAX = KON(5) DO 120 K1 = 1.LAX KON(20) = K1 ZERO OUT ALL SOURCE LOCATIONS DO 20 I = 1.NNC NLA = IJDIM NTH=NTI+HIJD NDIM=NDIM-NND  $IF(NDIM_LT_0) GO TO 996$  CON(1) = CON(13) CON(2) = 0.025 CALL VARULI IF(PASS.6£.0.)50 TO 2 CALL OUTCAL PASS = 1.0 CON(14) = CON(13) 0=(1)0 2 15 50 52 U Q *** 204 25* 26* ****** *s1 *91 18* 22* 53# 54* 37. 36+ 39. *0.1 00145 00146 00151 00152 00152 00153 00157 00157 00157 00163 00165 00166 00170 90144 00171

.

2

• .

CINDSS, CINDSS

•

		•	0 10 75	
J2 = 1	RLXD = 0.0	RLXA = 0.0	IF (NND.LE.0) GC	DN = CON(10)
*0*	***	45+	46+	47+

DD = 1.0-DN	C ZERO OUT EXTRA LOCATIONS	DO 30 I = 1.NND	LETIE+I	AN YII FIED. D
			÷	
#8 <b>#</b>	#6#	50#	51*	50¥
	1	N	ທ	4

	NODES
	DIFFUSION
	H
	S
	ITERATION
	BLOCK
30 ALLE	DO A
	U
+ 10	53+

D0 70 I = 1 NND	
54 *	1

	计算法 计算法
1+31=37	
55*	

= J2+1 = J2+1 NTINUE D	1998 J2 = J2+1 1999 J2 = J2+1 2000 Continue END	56* 1998 J2 = J2+1 56* 1999 J2 = J2+1 56* 2000 Continue 56* Elud					
= J2+1 = J2+1 NTINUE D	1998 J2 = J2+1 1999 J2 = J2+1 2000 Continue EIJD	56* 1998 J2 = J2+1 56* 1999 J2 = J2+1 56* 2000 CONTINUE 56* 2000 CONTINUE	**********	•			
0.010.00	2000 2000 2000 2000 2000	564 1999 US 564 2000 CC 564 2000 CC		2 = J2+1	2+1 = J2+1	JUTINUE	Q

부	3330 10111 2224 444	0 X X WU 1 N X * * *	330 101 102 24 44 45	330 WWI Z 2* * * *
J2 = 1 RLXD = 0.0 RLXD = 0.0 RLXD = 0.0 IF(NND.LE.0) 60 T0 T5 DN = CON(10) DD = 1.0-DM ZER0 OUT EXTRA LOCATIONS DD 30 I = 1.0ND DD 31 I = 1.0ND LE=IE+1 30 X(LE)=0.0 D0 A BLOCK ITEHATION ON THE DIFFUSION NODES D0 A BLOCK ITEHATION ON THE DIFFUSION NOTES D0 A BLOCK ITEHATION ON THE DIFFUSION NOTES D0 A BLOCK ITEHATION ON	1999 J2 = J2+1 2000 CONTINUE END INCLUDE VAR0.LIST INCLUDE VAR0.LIST IF(FLD(4.1.NSQ1(J1+1)).EQ.0) GO TO 5000 NTYPE = FLD(0.5.NSQ2(J2)) LA = FLD(22.14.NSQ2(J2)) LA = FLD(22.14.NSQ2(J2)) LA = FLD(22.14.NSQ2(J2)) GO TO (4005.4010.4015.4020.4025.4030.4035.4040.4030). NTYPE 4005 Q(1) = XK(LK)+Q(1) 4012 CALL D1D1WM(T(1).A(LA).XK(LK).Q2) 4012 CALL D1D1WM(T(1).A(LA).XK(LK).Q2)	<pre>60 T0 4998 4015 01 = 0.0 4017 CALL DID1WM(CON(14).4(LA).xK(LK).02) 4020 CALL DID1WM(CON(14).4(LA).xK(LK).02) 4022 J2 = J2+1 4022 J2 = J2+1 LA = FLD(22.14.NSQ2(J2)) LA = FLD(22.14.NSQ2(J2)) LA = FLD(22.14.NSQ2(J2)) 4025 01 = XK(LK)*XK(LA).XK(LK).01) 4030 CALL D1D1WM(CON(14).4(LA).XK(LK).01)</pre>	UZ = UZ:1 LA = FLD(5,17,NSQ2(J2)) LK = FLD(22,14,NSQ2(J2)) Q2 = XK(LK)+XK(LA) Q2 = XK(LK)+XK(LA) Q35 CALL DID14M(CON(14),A(LA),XK(LK),01) 4035 CALL DID14M(CON(14),A(LA),XK(LK),01)	LA = FLD(5,17,NS02(J2)) LK = FLD(22,14,NS02(J2)) 60 T0 4012 4040 01 = XK(LK)*XK(LA) 60 T0 4037 60 T0 4037 4998 A(1) = 01+02+0(1) 4999 JZ = J2+1 5000 CONTINUE
Ŭ Ŭ				
# # # # # # # # # # # # # # # # # # #			8477 847 847 847 847 847 847 847 847 847	
00173 00176 00176 00176 00200 00200 00200 00200 00200 00205 00205 00205 00205 00213 00213 00215 00215 00215 00215 00215 00215 00215 00215 00215 00215 00215 00215 00215 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 00200 000000	4465443324436666666666666666666666666666	<ul> <li>Construction</li> <li>Const</li></ul>	00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 00255 000255 00025 00255 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00025 00000000	00265 00265 00266 00265 00265 002671 00271

60 T0 4022 U2 = 12+1 U2 = 12+1 L4 = FLD(5,17,NS02(J2)) LK = FLD(22,14,NS02(J2)) LK = FLD(22,14,NS02(J2)) 02 = xK(LK) *XK(LA) 02 = xK(LK) *XK(LA) 037 U2 = 12+1 LA = FLD(5,17,NS02(J2)) LA = FLD(5,17,NS02(J2)) LA = FLD(5,17,NS02(J2)) LA = FLD(5,17,NS02(J2)) LA = FLD(22,14,NS02(J2)) LA = FLD(22,14,NS02(J2)) U2 = 0,4037 U2 = 0,4

•

÷

•

00273

Ξ.
U
۰.
S
S
0
2
-
U

...

 $\bigcirc$ 

n an an tao Marina an Anna an Anna An Anna an Anna an Anna

				*NEW	MUN+		L						MUN+						*NEW								M3N+							* L L A										
CIHDSS - CINDSS	LG = FLD(5,16,NSO1(J1)) If(L6,E0,0) GU TO 50	LTA = FLD(22,14,NSQ1(J1)) TACHIDE VADE-175T	C - CHECK FOR RADIATION CONDUCTOR	IF(FLD(2.1.NSG1(J1)).E0.0) 60 TO 3000 NTYPE # FLD(0.5.NS02(J2))	LA = FLD(5,17,HSO2(J2))	LK = FLU(22014,1202,021) GOTO(2005,2010,2015,2020,2025,2030,2035,2040,2045,2050,2053,	<pre>\$ 2060.2065) * NTYPE</pre>	2005 TM = (T(I)+T(LTA))/2.0	ZUUT CALL UJUIWM(IM*A(LA/*AK(LK)*G(LG)) Go to 2999	2010 TM = T(1)	GO TO 2007	ZUIS CALL UIUIWM(I(I)+A(LA)+XK(LK)+61) 2017 (2 = /2+1	LA = FLD(5,17,11502(J2))	LK = FLD(22,14,NSQ2(J2))	CALL UIDIWM(!(LIA).A(LA).AKILK).627 Go to 299A	2020 61 = XH (LK) + XK (LA)		2023 CALE ULUIWANI (11/14) LANARAY 1011 UP 11 U2+1	LA = FLD(5,17,NS02(J2))	LK = FLD(22+14+NSQ2(J2))	62 = AN(EN/*AN(EA) 60 T0 2998	2030 TM = (f(I)+T(LTA))/2.0	2032 CALL PLYAWM(A(LA),TM,A(LA+1),XK(LK),G(LG))	2035 1M = 1(1)	GO TO 2032	2040 UALL PLYAWM(A(LA),11(1),A(LA+1),XK(LK),61) 2042 UP = U2+1	LA = FLD(5,17,NS02(J2))	LK Z FLD(22,14,NS02(J2)) Colt BLVANDALAN, TTTTTAN AATLAVRIEN, G2,	CALL FLIARMIAILAILAILAILAILAILAILAILAILAILAILAILAIL	2045 61 = XK (LK) *XK (LA)	60 T0 2042 2050 Fail Pi VAWWAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAA		LA = FLD(5,17,11502(J2))	LK # FLD(22,14,NSQ2(J2)) 63 - YK/(Y/*YK/12)	60 TO 2998	2055 TM = (T(I)+T(LTA))/2.0	CALL DZDIWM(TM,CON(14),A(LA),XK(LK),G(LG))	60 TO 2999		2065 TM = T(LTA)	GO TO 2032	2996 6(L6) = 1./(1./61+1./62) reserved as the street of th	1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.	3000 CONTINUE
,	59 <b>*</b>	614	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	63#	*		63#	÷.9	* * 0 0	63#	÷	+	63#	63 <b>#</b>	60# 60#	63.	*29 *	*09	63#	#D9		63+	63+	+ n9	*09	60#	*£9	63#		***	63 <b>*</b>	- 20 + 20	63*	63# ***	63*	63+	63 <b>+</b>	* 99 * 19	# 0 9 # 6 9	÷.29	63+	63+	***	63+
	00275	00200	10200	00.304	00305	00307	00307	00310	11000	00313	00314	21500	00317	01320	00321	00323	00324	00326	00327	00330	10000	00333	40000 4000	00336	00337	00.540	00342	00343 11111	00345	00346	003470	00351	00352	00.553	00355	00356	00357	00360	10200	00363	00364	00365	01370	00371

a star a barren er en ser Ser en ser en

C - 6

•

						A C A A A A A A A A A A A A A A A A A A
CINDSS, CINDSS	END IF (FLD(3,1,NSQ1(J1)).EQ.0) GO TO 40 T1 = T(1)+460.0 T2 = T(LTA)+460.0 GV = G(LG)+(T1+T2+T2)+(T1+T2) GV = GLG)+(T1+T2+T2)+(T1+T2)	<pre>6 = 6(LG) 5 X(LE) = 5(LE)+6V 6 (1) = 0(1)+6V*T(LTA) CHECK FOR ADJOINING DIFFUSION NODE, WATCH FOR ONE WAY CONDUCTOR IF (LTA.6T.NND.OR.FLD(21.11.NS01(J1)).E0.1) 60 TO 50 IF (LTA.6T.1A) CLEEI = IE+LTA CLEEI = 2 X(LE1)+6V CLEEI = 2 X(LE1)+6V CLEEI = 0(LTA)+6V*T(1) 0(LTA) = 0(LTA)+6V*T(1) CLECK FOR LAST CONDUCTOR TO THIS NODE CLECK FOR LAST CONDUCTOR TO THIS NODE</pre>	T T T T T T T T T T T T T T T T T T T	CON(27) = RLXD IF(NMA.LE.0) 60 T0 115 DN = CON(9) DD = 1.0-DM JJ1 = J1 JJ2 = J2 D0 A SUCCESSIVE POINT ITERATION ON THE ARITHMETIC NODES D0 A SUCCESSIVE POINT ITERATION ON THE ARITHMETIC NODES D0 A SUCCESSIVE POINT ITERATION ON THE ARITHMETIC NODES D0 A SUCCESSIVE POINT ITERATION ON THE ARITHMETIC NODES TO 110 T = NN.NNC D0 110 T = NN.NNC C = T INCLUDE VR02.LIST INCLUDE VR02.LIST INCLUDE VR02.LIST INCLUDE VR02.LIST INCLUDE SO (JJ1+11).EQ.0) 60 TO 6000 L = T INCLUDE VR02.LIST INCLUDE	60 T0 (5005,5010,5015,5020,5025,5030,5035,5040,5030), NTYPE 0 (L) = xk(Lk)+G(L) 60 T0 5999 61 T = 0.0 61 = 0.0 61 T 0101WM(T(L),A(LA),XK(LK),02) 61 = 0.0 7 CALL D101WM(CON(14),A(LA),XK(LK),02) 7 CALL D101WM(CON(14),A(LA),XK(LK),02) 7 CALL D101WM(CON(14),A(LA),XK(LK),02)	<pre>2 JJZ = JJZ+1 LA = FLD(5.17.41502(JJZ)) LK = FLD(22.14.4052(JJZ)) 60 TO '5017 60 TO 5022 JZALL ELD1WM(COTT(14).A(LA).XK(LK).01) JJZ = JJZ+1 LA = FLD(5:417.41502(JJZ))</pre>
		τα υ Ο Ο	י ט ט ט ני ט ט	v r	501 501 501 501 501	502 502 503
	000000 00000 0000 0000	4 4 4 7 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	→ → → → + + + + + + + + + + + + +	99999999999999999999999999999999999999	**********	*********
	00372 00375 00375 00376 00376		10000000000000000000000000000000000000	00000000000000000000000000000000000000	00000000000000000000000000000000000000	000464 00465 00465 00464 00464 00464 00464

.**.**,

		NJ 8														#NEW #NEW		D1++										<b>1</b>				* NF W	* NEW	2-++									2====				*NEW	- <b>m</b> .
CINDSS, CINDSS	LK = FLD(22,14, NS02(JJ2))	GZ Z XK(LK)*XK(LA) GD TO 5998	5035 CALL DIDIWM(COM(14) A(LA) XK(LK) 01)	BUST UVE = 54541 LA = FLD(5,17,NS02(JJ2))	LK = FLD(22,14,NSQ2(JU2))	5040 01 = XK(LK)*XK(LA)	60 T0 5037 5008 b(1) = 01+02+6(1)		6000 CONTINUE	END Contractions of the contraction of the contract		LTA = FLO(22,14,NSO1(JJ1))	INCLUDE VRG2.LIST	C CHECK FOR RADIATION COMPUCTOR		NITTE FLU(U/D/NSGZ/UUZ)		GOTO (3005, 3010, 3015, 3020, 3025, 3030, 3035, 3040, 3045, 3050, 3055,	\$ 3060;3065); NTYPE	2005 TM = (f(L)+([L)+([L])/2.0	COC CALL DIDIENT FRANKEN	3010 TM = T(L)	60 TO 3007	3015 CALL DIDIWM(T(L),A(LA),XK(LK),GI)	3017 JU2 = JU2+1		CALL DIDIWM(T(LTA),A(LA),XK(LK),62)	G0 T0 3998	3020 G1 = XK(LK)*XK(LA)	60 T0 3017	3025 CALL DIDIWH(T(L),A(LA),XK(LK),61)		LK = FLD(22+14+NSQ2(JU2))	G2 = XK (LK) + XK (LA)		3032 CALL PLYAWIA(A(A),TM,A(LA+),XK(LK),6(LG))	60 T0 3999	3035 TM = T(L)		3040 CALL FILMMARICATTICLIAR LATIVANICATOR	LA = FLD(5,17,4)502(JJ2))	LN = FLD(22+14+NS02(JJ2))	CALL FLYAWM(A(LA)/J(LIA)/A(LA+I)/XK(LK)/GZ) Gn tn 3998	3045 G1 = XK (LK) *XK (LA)		JUDU CALL FLIANWIA (LA) ( (L) (ALLATI) (XN (LATI) (XN (L) (G))	LA = FLD(5,17,11502(JJ2))	
	*66	*66	*66	*66	*66	*66	#66	*66	<b>*66</b>	<b>#66</b>	+101	102+	103+	104*	* 101	***	1044	104+	104*	++01	101	104*	104*	104*	104	****	101	104*	104*	104+	104*	***	104+	**01	* 101	104	104+	104+	***	104*	104*	104*	104*	104 +	101*	* 10 *	* + + 0 1	, ,
	00471	2400	00474	00476	00477	00501	00502	00504	00505	00200	00210	00511	00512	00512	00513	00210	00517	06520	02400	00521	005230	00524	00525	00526	00527	00000	00532	00533	00534	00535	00536	0.0540	00541	00542	00543	00545	00246	00547	002200	00552	00553	00.554	00555	00557	00260	00552	06563	

 $\mathbf{i}$ 

. • .

and and a second sec Second second

.

•.

C - 8

S
ŝ
<u> </u>
Z
0
S.
\$
0
z
U.

LK = FLD(22,14,MSQ2(JJ2)) G2 = XK(LK)+XK(LA) G0 T0 3998 3055 TM = (T(L)+T(LTA))/2.0 CALL D2D1WM(TM,CON(14),A(LA),XK(LK),6(LG)) COLL 3999 3060 TM = T(LTA) G0 T0 3077 G0 T0 3077	3065 TM = T((TA) 60 T0 3032 3996 6(L6) = 1./(1./61+1./62) 16(FLD(3.1.NSu1(JJ1)).EQ.1) 6(L6) = 61*62 3999 JJ2 = JJ2+1 4000 Cuntinue	<pre>FYU FYU T1 = T(1)+460.0 T2 = T(1)+460.0 F2 = T(LTA)+460.0 GV = G(LG)*(T1*T1+T2*T2)*(T1+T2) GV = G(LG)*(T1*T1+T2*T2)*(T1+T2) GV = G(LG)*(T1*T1+T2*T2)*(T1+T2) GO T0 90 GO T0 90 GO T0 90 GO T0 90 GO T0 = G(LG) GO T0 = G(LG) GO T0 = G(LG) GO T0 = G(LG) GO T0 90 GO T0 = G(LG) GO T0 90 GO T0 = G(LG) GO T0 90 GO T0 90 GO T0 90 T1 = ABS(T(1)+D1+90+00T)/GSUM T1 = ABS(T(1)+T2) C 5T0NE THE NEW TEMPERATURES</pre>	<pre>F(ALL = 15 F(ALX = 55.71) GO TO 110 RLX = 71 N2 = 1 N2 = 1 N</pre>	125 KON(37) = N2 IF(RLXA.6T.RLXD) GO TO 155 IF(RLXA.6T.RLXD) GO TO 155 CON(37) = RLXD KON(37) = RLXD KON(	<pre>div = div+d(1)     141 = div+d(1)     Lia = FL0(22,14,NSQ1(J1))     If(LTa.LE.NNC) G0 T0 175     LG = FL0(5,16,1501(J1))     If(FL0(3,1,NSQ1(J1)).EQ.0) G0 T0 170     T1 = T(1)+460.0     T2 = T(LTA)+460.0     QOUT = 00UT+6(LG)*(T1**4-T2**4)</pre>
***************************************	******			13324	114654444444444444444444444444444444444
00564 00565 00565 00565 00570 00572 00572	00574 00576 00576 00576 00501	00620 006514 006510 006514 006514 006514 006514 006514 006514 006514	00624 00624 00624 00624 00624 00633 00633 00633 00643 00644	000645 000651 000651 000651 000651 000651 000651 000651 000651 000651 000651 000651	00662 00662 00664 00664 00667 00667 00673

•

.

•

*NE%

i

Э.,

è

CINDSS.CINDSS	G0 T0 175	170 GOUT = GOUT+6(L6)*([1)-[(LTA)]	C CHECK FOR LAST CONDUCTOR TO THIS NUCL 176 TETNEOTITITET DI 60 TO 166		CON(32) = ABS(GIN-GOUT)	CALL VARBL2	CON(13) = CON(1)	CALL OUTCAL	WRITE(6+882)	WRITE(6,883) KON(20),CON(32)	KON(28) = KON(28)+2	IF(CON(3).6T.CON(1)*1.000001) 60 T0 10		NDIM = NLA	RE TOKN	994 WRITE(6+884)	GO TO 1000	996 WRITE(6+886) NDIM	GO TO 1000	997 WRITE(6/887)	60 10 1000	998 WRITE(6,888)	60 TO 1000	999 WKITE(6,889)	1000 CALL OUTCAL	LALL EXI Ded from titu v	BB3 FORMAT(10H LOOPCT = 16,10H ENGBAL = E12.5)	BB4 FURMAT (46H CINESS REQUIRES SHORT PSEUDO-COMPLITE SEGUENCE)	885 FORMAT(35H ITERATION COUNT EXCEEDED. NLOOP = . 110)	B46 FORMAT(18,20H LOCATIONS AVAILABLE) .	887 FORMATTIDH NO DRLXCA)	BBB FORMATIIOH NO ARLXCA)	689 FORMAT(14H NO LOOP COUNT)	END	UNIVAC 1108 FORTRAN V COMPILATION. 0 *DIAGNOSTIC* MESSAGE(S)	SYMBOLIC DDE REIDCATABLE	
	150+	151+	1524	154+	155*	156*	157+	158*	159*	160+	101+	162*	1004	104 *	165#	166*	167*	166*	169*	170*	171*	172*	173*	174*	175*	* 1 0 *	178*	179*	180*	181*	1.82*	183+	1.84 +	185*			1
	00674	00675	2/900	00700	00702	00703	00704	00705	00706	00110	00714	00715	11/00	00720	00721	00722	00724	00725	00730	00731	00733	00734	00736	00737	14200	24/00	00744	00745	00746	00747	00750	00751	00752	00753			

ي من المراجع : الميوم في يور ها الارام ماريخ :

÷,

• 1

u≜u Seenaar je

. The state of stranger standard and set have go

,•

and and a second se Second second

	L FFB 71 13:59:58, 71							0002 R 00035 GN 0002 I 00035 GN 0015 I 000035 GN 0022 I 000000 K 0022 I 000000 K 0022 I 000000 K 0022 R 000000 NJ 0002 R 000001 NJ 0002 R 000001 NJ 0002 R 000005 92 0002 R 000030 TM
	~							0002 R 000011 DD 0002 I 000031 G1 0002 I 000031 G1 0002 I 000013 J2 0002 I 000022 L4 0002 I 0000022 L4 0021 I 0000023 L4 0021 I 000001 NAA 0014 I 000000 NSO2 0002 R 000024 01 0007 R 000000 J
	LEATED ON 20 AUG 70	•			•	•	RELATIVE LOCATION NAME)	0017 R 000000 CON 0002 R 000035 GV 0002 I 000003 IE1 0002 I 000012 J1 0002 I 000014 L6 0021 I 000026 L6 0021 I 000026 L6 0021 I 000001 NN 0013 I 000100 11501 0002 R 000100 11501 0002 R 000146 0007
CINDSLICINISL	.•CINOSL HENA VERSION: 131K-10D CR E 04 09 JUN 70 AT 13:59:58	ENTRY POINT 003163	VAME . LENGTH)	CMP 003200 AR 000110 AR 0000050 0000001 000001 000001 000001 000001 000001 000001 000001 000001 000001 000001 000001 000001 000001 000001 000001 000001 000001 000001 000001 000001 000000			R VARIARLES (BLOCK, TYPE,	0010 R 000000 C 0002 R 000011 I 0002 I 000011 I 0002 I 000013 JJ2 0002 I 000013 JJ2 0002 I 000013 LE2 0002 I 000027 LF2 0002 I 000002 NNT 0022 I 000002 NNT 0022 R 000014 RLXD 0002 R 000014 RLXD
	GIW FOR+* CINDS UNIVAC 1108 FORTRAN V AT THIS COMPILATION WAS DONE	SUBROUTINE CINDSL	STORAGE USED (BLUCK, P	0001       *COUE         0000       *SIMPLE         0005       *SIMPLE         0005       *ARHAYS         0007       TITLE         0007       TEMP         0011       SOUKCE         0012       CONU         0013       PC1         0014       PC2         0015       PC1         0016       ARHAY         0017       FIXCON         0017       FIXCON         0017       FIXCON         0017       CONU         0017       FIXCON         0017       FIXCON         0017       SOUC         0017       CONU         0017       CONU         0017       FIXCON         0017       CONU         0017       CONU         0017       CONU         0017       CONU	I LAGAN GOUN	0022 0UTCAL 0023 0UTCAL 0025 PLYAWM 0026 0201WM 0027 NONLIN 0031 EXIT 0033 NWUU5 0033 NWUU5 0035 NERK25 0035 NERL25	STORAGE ASSIGNMENT FO	0016 R 000000 A 0012 R 000000 G 0005 I 000000 H 0002 I 000042 JJI 0017 I 000042 JJI 0021 I 000042 LE1 0021 I 000043 N57 0021 I 000040 N40 0021 I 000040 N40 0021 R 000400 C

•.

•

.

.

1.

-

<i>.</i>	11111111111111111111111111111111111111	
	1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1	
	000000000000000000000000000000000000	
		V RESSABLE ESSABLE (COTIME (CTIME (CTIME (CTIME (DTIME (DTIME (DTIME (DTIME (DTIME (DTIME (DTIME (DTIME (DTIME
	14400 1440 1440 1440 1440 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401 1401	FORTRAN ATION ARE ADDR ARE ADDR RRAY/A(1) 32 32 32 32 32 34 34 5 44 5 44 5 44 5 4
	<pre>&gt;&gt; 0000 000000000000000000000000000000</pre>	SINDA SINDA JULTER JINT ITER FACTORS FACTORS FACTORS FACTORS FACTOR FACTOL SULTCH AND HAME AND HAME FEP VSED CHANGE CHANGE AMPTING F AMPTING F AMP
•		CLAR FOR CESSICE FOR CESSICE FOR CESSICE FOR COMPLICATION COMPLIANT COMPLIANT COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING COMPLETING
	11111111111111111111111111111111111111	N SUBROUT TE SEQUED TVE A SUC LOWED, TV SUBROUT SUBROUT A SUC A SU
N. SL	8 8 9 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	CLUST SCORPLOR CLUST SCORPLO CLUST
INDSL.CI	<b>8111111111111111111111111111111111111</b>	UTINE CUL NUTINE
G	<b>71</b> 100 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11300 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 11000 110000 110000 110000 11000000	SUARO SUARO STEAD ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITHU ANITH
	00000000000000000000000000000000000000	

•

٠.

CINDSL. CINDSL

. Antonio de la construcción de la co

. •

•

.

and the second se and and the state of the second

CINDSL+CINDSL	CC16 CONTAINS ARITHMETIC TEMPERATURE CHANGE CALCULATED (ATMPCC)	CONTROL CONSTANT IT IS RESERVED FOR THE C/SG MINIMUM (CSGMIN)	CONTROL CONSTANT 16 CONTAINS THE OUTPUT INTERVAL (OUTPUT)	CC19 CONTAINS THE ARTTHMETIC RELAXATION CRITERIA ALLOWED (ARENCA)	CCC2 CONTAINS THE NUMBER OF RELAXIION LOUIS OSENIIN'EGER (CONCUL)	CC21 FC TONTAINS THE MINIOUN ACCOVED THE STEP OF TO ACCOVED THE STEP	CC22 ANTONIA THE AVEN INFLICT	CC23 CONTAINS THE C/S6 MAXIMUM	CC24 CONTAINS THE C/SG RANGE ALLOWED	CC25 CONTAINS THE C/SG RANGE CALCULATED (CSGRCL)	CC26 CONTAINS THE DIFFUSION RELAXATION CRITERIA ALLOWED (DRLYCA)	CC27 CONTAINS THE DIFFUSION RELAXATION CHANGE CALCULATED (DELYCC)		CC29 CONTAINS THE PAGE COUNTER'S INTEGER CONTAINS ANALYSIN (PAGECI)	CC30 CONTAINS ANTIFFUELIC RELAXATION CHANGE CACOULATED (APLACC)	COST DE INCONTATING THE FREEDEN RATES AND THE MARE FREEDEN AND AND AND AND AND AND AND AND AND AN	COUNTRY THE DESIRED ENERGY RALANCE OF TRUDY (BALENCE) COUNTRY (BALENCE)	CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR	CC35 CONTAINS RELATIVE NODE NUMBER OF CSGMIN	CC36 CONTAINS RELATIVE NODE NUMBER OF DIMPCC	CC37 CONTAINS RELATIVE NODE NUMBER OF ARLXCC	CC38 CONTAINS RELATIVE NODE NUMBER OF ATMPCC	CCOSCHOLHEI-FRANC CONTAIN DUMMI INTEGER CORDIANIS (I-FALFMEST) Attribuertister Contain Diumy Elonating Constant, Chistanic (FALFMERT)	CC4+++2-46-++++4+46 COUNTRY COUNT FLOATING COLUMNING (XX-2-++2-49 FLOATING COLUMNING COUNTRY COUNT	CC50 IS NOT USED AT PRESENT		IF(KON(5).LE.0) 60 T0 999.	IF (CON(9).LE.0.) COM(9) = 1.0	I CONTUDITE: UNIT CONTUDITE INU	IFINNAROTAU-AND.CONTEXTAL CONTENT OF TO 997	IF (KOW(31) •NE.1) GO TO 994	PASS = -1.0			161 + 1111 162 = 161 +NNC	NLA = ROIM	JU = 2*NNC	NTH = NTH+JJ	COLTA I NICALTA	CON(1) = CON(13)	CON(2) = 0.0	CON(14) = CON(13)		IF(CON(1)-CON(3).61.0.) CON(1) = CON(3)	CON(14) = (CON(1)+CON(13))/2.0	CON(2) # CON(1)-CON(13)	COMPUTE STEADT STATE TEMPENALORES	CINDA = XONDA	00 145 K1 = 1,LAX	1+PL = U.	$KG(1(20) \equiv K]$	DU 15 1 # 1+NNC	
	U.	υ	U	U I	ບ່	י ר	5	0	U I	U I	U	U			ų	ې ر	<u>ں د</u>	) c	u U	U	U.	U	ں ر	ي د	u u	•																				5	و			1.	. •	ر	
	<b>#</b> 0	8 8	<b>8</b>	*	* : 0 :		0	# 0	¥ Ø	# 00 /	*	<b>*</b>	* *			0 0			• # 0 40	<b>*</b> 0	*8	.# 4 60 9	# 4 0 a			8*	<b>*</b> 6	*01	***	1.1	**	15*	16+	* 4		20*	21*	55 57	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	50 1 1 1 1 1 1 1 1	26*	27*	100	100	31*	12	# 0 0 1	* * * ¥ * 1	90 <b>*</b>	37+	*80 10		
	21100	00113	00113	00113	51100	CT 100	CT100	21100	00113	00113	00113	00113	51100	C1100	21100	CT100	21100	51100	00113	00113	00113	00113		1100	00113	00114	00115	11100	12100	52100	00127	00131	00132	22100	00135	00136	00137	00140		5±100	00145	00146	101100	00151	00153	00154	49100	22100	00157	00162	00163	00164	•
																																		•																			

•

· •

-

÷

ж. "**•** 

n. ■ •	N-1 1-1 1-1 1-1 1-1 1-1 1-1 1-1 1-1 1-1	2013년 11년년 1 22 년 11년 1 11년 1 1 11년 1 11년 1 11 11 11 11 11 11 11 11 11 11 11 11 1	BBQ BBQ UUI UUI UUI 200 *** ***	жа Ши ж ж ж ж ж ж ж ж ж ж ж ж ж ж ж к к к к
15 0(1) = 0.0 CALL VARBLI 17(PASS.6E.0.) 60 T0 20 CALL OUTCAL PASS = 1.0 20 J1 = 0 20 J1 = 0 21 = 1.0 RLXD = 0.0 RLXD = 0.0 RLXD = 0.0 PN = CON(10) DN = CON(10) DN = CON(10) DD = 1.0-DN DD = 1.0-DN	CONTRACTOR CONTRACTIST INCLUDE DUMC.LIST INCLUDE DUMC.LIST IF(FLD(1:1:MSQ1(J1+1)).EQ.0) 60 TO 2000 NTYPE = FLD(0:5:NSC2(J2)) GO TO (1999.1998.1998.1998.1998.1998.1998.1998	INCLUDE VAR0.LIST INCLUDE VAR0.LIST IF(FLD(4,1.NSQ1(J)+1)).EQ.0) 60 TO 5000 NTYPE = FLD(0.5.NSQ2(J2)) LA = FLD(5.17.NSQ2(J2)) LA = FLD(22.14.NSQ2(J2)) LA = FLD(22.14.NSQ2(J2)) CO TO (4005.4010,4015,4020,4025,4030,4035,4030). NTYPE 4005 0(1) = X4(LK)+G(1) 4005 0(1) = X4(LK)+G(1) 4010 01 = 0.0 4010 01 = 0.0 4015 01 04998 4015 01 04998 4015 01 04998 4015 01 04998 4020 01 D11WM(CON(14).A(LA).XK(LK).02) 4020 01 D11WM(CON(14).A(LA).XK(LK).02) 4020 01 04998 4020 01 04998 4010 01 01 0488 4010 01 01 0488 4010 01 01 0488 4010 01 04888 4010 04888 4010 04888 4010 04888 4010 04888 4010 04888 4010 048888 4010 048888 4010 048888 4010 0488888 4010 04888888 4010 04888888 4010 048888888 4010 04888888888888888888888888888888888	4025       01       7       4017         4025       01       7       4025       14.0502         4025       01       2       2       14.0502         4030       Call       0127       14.0502       14.010         4030       Call       10101WM       14.010       14.010         12       J2+1       J2       J2+1       14.0502       14.010         12       J2+1       10101WM       14.0502       14.010       14.010         12       J2+1       14.0502       14.010       14.010       14.010         12       J2+1       14.0502       14.010       14.010       14.010       14.010         13       14.0502       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010       14.010	4000 CA       - 5 - 5 - 5 - 5 - 5 - 5 - 5 - 5 - 5 - 5
######################################	,	************************		
001474 001774 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 001175 0001175 0001175 0001175 0001175 0001175 0001175 0001175 0001175 0001175 0001175 0001175 0001175 0001175 00010000000000		00223 00224 00224 00224 00224 00224 00224 00224 00224 00224 00224 00224 00224 00224 00224 00224 00224 00224 00224 00224 00224 00224 00224 00224 00224 00224 00224 00224 00224 00224 00224 00224 00224 00224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 002224 0002224 002224 0002224 0002224 0002224 0002224 0002224 0002224 0002224 0002224 0002224 0002224 0002224 0002224 0002224 0002224 0002224 0002224 0002224 0002224 0002224 0002224 0002224 0002224 0002224 0002224 0002224 0002224 0002224 0002224 0002224 000224 000224 000224 000224 000224 000224 000224 000224 000224 000224 000224 000224 000224 000224 000224 000224 000224 000224 000224 000224 000224 000224 000224 000224 000224 000224 000224 000224 0000224 000224 0000224 0000224 0000224 0000224 0000224 0000224 0000224 0000224 0000224 0000224 0000224 0000224 0000224 0000224 00000224 00000000	0025544440002244400022444000022550000022500000000	002264

**C - 1**4

•

· . . . •

333 333 34 34 34 34 34 34 34 34 34 34 34	NU 19 19 19 19 19 19 19 19 19 19 19 19 19	NEN NEN * *	* NEW * NEW * * 12	N N N N N N N N N N N N N N N N N N N
CINDSL.CINDSL END 25 JI = J1+1 LG = FLD(5,16,NSO1(J1)) LTA = FLD(22,14,NSO1(J1)) LTA = FLD(22,14,NSO1(J1)) LTA = FLD(22,14,NSO1(J1)) C CHECK FOR RADIATION CONDUCTOR TF(FLD(2,1,NSO2(J2)) NTYPE = FLD(0,5,NSO2(J2)) LA = FLD(22,14,NSO2(J2)) LA = FLD(22,14,NSO2(J2)) C CHECK FOR RADIATION CONDUCTOR C CHECN COLD 2005,2010,2015,2020,2025,2030,2035,2040,2045,2050,2055, C COLD DIDWW(TM:A(LA):XK(LK),6(LG)) C C C C C C C C C C C C C C C C C C C	2010 0 = 1041 LA = FLD(5,17,4)SQ2(J2)) LK = FLD(22,14,NSG2(J2)) CALL DID1WM(T(LTA),A(LA),XK(LK),62) GO TO 2998 2020 G1 = XK(LK)*XK(LA) 2025 CALL DID1WM(T(1),A(LA),XK(LK),G1) 2025 CALL DID1WM(T(1),A(LA),XK(LK),G1)	LA = FLD(5,17,MSQ2(J2)) LK = FLD(22,14,NSQ2(J2)) G2 = XK(LK)*XK(LA) G3 T0 2998 2030 TM = (T(1)+T(LTA))/2.0 2035 CALL PLYAWM(A(LA),TM,A(LA+1),XK(LK),G(LG)) G0 T0 2999 2035 TM = T(1) 2040 CALL PLYAMM(A(LA),T(1),A(LA+1),XK(LK),G1) 2040 CALL PLYAMM(A(LA),T(1),A(LA+1),XK(LK),G1)	LA = FLN(5,17,NSQ2(J2)) LK = FLD(22,14,NSQ2(J2)) LK = FLD(22,14,NSQ2(J2)) LK = FLD(22,14,NSQ2(J2)) CAL PLYAWM(a(LA),T(LTA),A(LA+1),XK(LK),G2) 2045 G1 = XK(LK) *XK(LA) 2042 G1 = XK(LA) 2050 CAL PLYAWM(a(LA),T(1),A(LA+1),XK(LK),G1) 2050 CAL PLYAWM(a(LA),T(1),A(LA+1),XK(LK),G1)	LA = FLOTS.17.4502(J2)) LA = FLOT2.14.4502(J2)) G2 = XK(LK)*XK(LA) G2 = XK(LX)*XK(LA) G4 = (T(1)+T(LTA))/2.0 C4L D2D1WM(TM.CON(14),A(LA),XK(LK),G(LG)) G0 T0 2999 2060 TM = T(LTA) C0 T0 2097 2065 TM = T(LTA) C0 T0 2032 2994 G(LG) = 1./(1./G1+1./G2) FF(FLD(3.1.NS0!(J1)).EG.1) G(LG) = G1*G2 2999 J2 = J2+1
, , , , , , , , , , , , , , , , , , ,	* * * * * * * * * * > > > > > > > > > >			* * * * * * * * * * * * * * * * * * *
00022747374 0002274757477777777777777777777777777777		000000000000000000000000000000000000000		

.

•

٠

6 × 2 1

and the second second

-
z
_
<b>U</b>
•
_
3
Ô.
z
-
U.

1 ì

00365	62#	3000	CONTINUE	
00367	*29		IF(FLD(3,1,NS01(J1)),EQ.0) 60 TO 30	•
00371	* 79		$T_1 = T(1) + 460.0$	
00372	65*		T2 = T(LTA)+460.0	
00373	<b>\$</b> 99		6A = 6(FC)*(1]*1]+12*12)*(1]+12	
100274	*/9			
2/200	+ 0 4	0 K		· · · · · · · · · · · · · · · · · · ·
00377	+02	2	011) # 011)+6V*T(LTA)	
00377	11+	U	CHECK FOR LAST CONDUCTOR TO THIS NODE	-
00400	72*	I	IF(NSQ1(J1).6T,0) 60 TO 25	
00402	*0^		T2 = DD*T([)+DN*Q(])/GSUM	
00402	* + + + + + +	U.	OBTAIN THE CALCULATED TEMPERATURE DIFFERENCE	
20400	101	.c	IL = 400/1/1/-1/-2/ Store The New Temperatures and Extradol Attan Factors	
10100		2	31.001 - 101 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114 - 114	
00402	78+	55	VI TOTOSTOVICE TO TOT	
00400	+62	I,	LE2 7 IE2+I	
00402	<b>80</b> *		RI = T2-T(I)	
00410	81*		X(LE1) = T(1)	
1100	* + N N		X(LE2) = RI/(R1=X(LE2))	
27500		÷0		
00414	*198	2	X(LE2) = T2+T(1)	
00415	86*	65	$T(1) = T_2$	
00415	87*	U	SAVE THE MAXIMUM DIFFUSION RELAXATION CHANGE	
00416	88*		IF(RLXD.6T.T1) GO TO 70	•
00420	#68 #68		RLXD = 71	•
		40		
22400	*16	5	CONTRACT - BLYC	
12100	+ 10 + 10		LEANNELLE-D) 60 TO 130	
00427	* 16	75	DN = CON(9)	
00430	*56	2	NO-0.1 = 1.0-DN	
15+00	<b>*</b> 96			
200400	*	ç	ULD I UL Di currente datat ticattan da aditumette ucare	
00433	* 0 0 0	ڊ	DO 125 I E NNANC	
00436	100+		65UM = 0.0	
0.0437	101+			
01+00	102*		INCLUDE VRQ2.LIST	
14400	102*		IF(FLD(4,1,NSQ1(JJ1+1)),EQ.0) 60 TO 6000	
04400	102*			
111100	104			#NE#
00445	102*		CA - FLATSAILANSUSSANSA - CONSTRUCTION - CONSTRUCTI	
24400	102*	5005	0(T) # XX(CK)+0(T)	
00450	102+		60 TO 5999	
00451	102*	5010	Q1 = 0.0	
00452	102+	5012	CALL DIDIWM(T(L),A(LA),XK(LK),02)	
00400	102+	4015		
00455	102*	5017	CALL DIDIWM (COUTLY) A (LA) XK (LK) 02)	
00456	102*	•	60 TO 5998	
00457	102+	5020	CALL DIDIUMICON(14) A (LA) XK (LK) OI)	
00400	1024	2700	JUK = JUK 1 1 A = FID15.17.15021.41211	ANEW.
00+62	102+		LK = FLD(22+14+NSQ2(JJ2))	NUN+
00463	102+		G0 T0 5017	01 = + + +

•

**.** .

.

.

C - 16

.

· .

																															•																		
			Ma	A	N			μ	MJ	Ņ														)							MJ	×.	N				МIJ	33	N I I I							R.	M	N	
			<b>Z</b> *	<b>Z</b>	*			ž	*	¥												21	2.4								ž	*	#				1	1		•		- 1				2	ž	*	
25 01 # XK(LK)*XK(LA)	GU TU BUZZ BA CALL DIDIWM(CON(14)+A(LA)+YK(LK)+Q1)		LA = FLD(5,17,NSQ2(JU2))	LK = FLD(22+14+NSQ2(JJ2))	02 = XK(LK) + XK(LA)			LA = FLU(5,17,1502(JU2))	LK = FLD(22,14,NS02(JJ2))	G0 T0 5012	040 01 = XK(LK)+XK(LA)		198 G(L) = GI+G2+G(L)					INCLUDE VRG2+LIST	CHECK FOR RADIATION CONDUCTOR	IF(FLD(2,1, NSG1(JJ1)), EQ. 0) 60 70 4000	NTYPE = FLD(0,5)NSO2(JJ2))		「K = FLU(22)はいい34(10/21) GOTO12105、3015、3015、1000、305、3030、3035、4000、3045、3050、3055、		<pre>% 2/60/2003/1 Miles 105 TM = (T(L)+T(LTA))/2.0</pre>	107 CALL DIDIWM(TM,A(LA),XK(LK),G(LG)) .	GO TO 3999	110 TM = 7(L)	60 10 3UU7 115 CALL DIDIWM(TIL).A(LA).XK(IK).61)		LA = FLD(5,11,NSQ2(JU2))	LK = FLD(22,14,NSQ2(JJ2))	CALL DIDIMM(T(LTA),A(LA),XK(LK),G2)			25 CALL UIDIWM(T(L),A(LA),XK(LK),61)	ULZ = ULZ+1	LA E FLU(5) L'NSQC/UUE) 	CN = 1.0.422.4.4.4.5.4.6.6.6.6.6.6.6.6.6.6.6.6.6.6.6	60 TO 3998	130 TM = (T(L)+T(LTA))/2.0	J32 CALL PLYAWM(A(LA),TM,A(LA+1),XK(LK),6(LG))	60 TO 3999	60 TO 3032	140 CALL PLYAWM(A(LA).T(L).A(LA+1).XK(LK).61)		LK = FLD(22,14,NS02(JJ2))	CALL PLYAWM(A(LA),T(LTA),A(LA+1),XX(LK),G2)	60 TO 3998
ŝ	×.	2				ų	ភ ភា	•			ភ	ì	ົດໄ	́л`	õ				ပ						ที	ที	i	ค	7	ที่ที่	•				5.	ň	,				ñ	ค	, F	<b>.</b>	ñ	ñ	•		
32.	2	12	25	32*	* (	+ '+ N C		*2	12*	12+	*2	* 2 2 5	*			1	+	*0	17+	12*	**	*	* *		1	+20		*	* *	*	17+	17+	***		**	12*	*	* *	+ +	**	17+	***	**	17.	*21	**	*	17.*	14
-				1		ي ر س		-	-	10	-			-1-		 		13	1	2	10		-			10	10	3			2	10				10	23			1	10	-		19	10	ن د ج اب		10	7
00464	19400 99100	00467	00470	12400	00472	72 #00	00475	00476	00477	00500	00501	00502	20200	+0200	20200	00510	0.0511	00512	00512	00513	00515	00516	005300		00521	00522	00523	00524	00525	00527	00530	00531	00532	15000	00535	00536	00537	01000	00542	00543	0.0544	00545	00546	00550	00551	00552	00554	00555	00556

CINDSL, CINDSL

<u>ا</u>

...

•

j.

•

. .

	CINDSL, CINDSL	3045 61 = XK(LK) +XK(LA)	60 10 JU42 Angr rait di yawwiati atati ati ati. Ykriki. Git		La = FLD(5,17,NS02(J2))	LK H FLU(22:14:NSG2(JU2)) G2 H XK(LK)+XK(LÅ)	60 TO 3998	3055 TM = (T(L)+T(LTA))/2.0 2011 Decimiter Convision 2012/2012 Decimiter Convision	CALE DEULAMI IN CONTLATERINATAN LAN LAN PURCHI Co to 3000	3060 TM = T(LTA)	60 TO 3007	3065 TM = 1(LTA) GO TO 3032	3998 G(LG) = 1./(1./61+1./62)	IF (FLD(3,1,NSQ1(JUI),EQ.1) 6(LG) = 61+62	3999 JJZ = JJZ+1 4000 CONTINUE		IF(FLD(3.1.NS01(JJ1)).EQ.0) GO TO 85	12 ± 1/1/1400.0 T2 ± T(1TA)+4660.0	GV = 6(L6) + (T1+T2+T2) + (T1+T2)	G0 T0 90	85 6V = 6(L6) 90 pt1) = 0(1)+6V±7(LTA)	6SUM = 6SUM+6V	CHECK FOR LAST CONDUCTOR TO THIS NODE	IF (NSW1(JUI)*61*U) 60 10 80 T2 # DD+T(I)+DN+6(I)/65UM	$T_1 = ABS(T(1) - T2)$	STORE THE NEW TEMPERATURES AND EXTRAPOLATION FACTORS	$10 \ Le1 = 1 \ le1 + 1$	LE2 = IE2+I		X(LE2) = I(1) X(LE2) = R1/(R1-X(LE2))	60 TO 120	AID LEZ = 4EZ+L X(LE2) = T2=T(T)	120 T(1) = 72	IF(RLXA.6T.T1) GO TO 125		125 CONTINUE	CONCOUNT REXA Sec 16 the delayation rditedia add wet	130 IF (RLXA.LE, CON(19).AND.RLXD.LE.CON(26)) 60 TO 150	IF(JJ.LE.2) GO TO 140	CUL 1 0 135 1 4 1944		SEE IF THE EXTRAPOLATION CRITERIA ARE MET	<pre>// IF(X(LE2).6E.0.) GO TO 135 refx()F2).170.) Y(1F2) = _8.</pre>		T(I) = X(LE2)+X(LE1)+(1.0-X(LE2))+T(I)	JUNUL CONTINUE	_
4	، به مسور ۶		÷.			* *			* *	, F #	*	* *	• •	*	* *			; F 4	*		* *			* *		* 1	e .#	*		• •		* *	*	* •	• *	°.	* *	- 	*	* *	- 4		* 4	. 4	* 1	*	
		101	101	101	101	107	107	101		107	101	107	107	107	107	107	108		111	112	11.1	115	110	118	119	120	121	123		126	127	1291	130			30	0.1 1	137	138	140.		5 4 7	0 H T	145	140		
$\sum_{i=1}^{n}$		00557	00200	00562	00563	00565	00566	00567	0/200	00572	00573	00574	00576	00577	00501	00603	00000	00607	00610	00611	00613	00614	00614	21900	00620	00620	00622	00623	00624	00626	00627	00631	00632	00633	00636	00637	14000	00642	00044	00646	00652	00052	00653	00657	00660	10000	

**/EW **/EW

-

.

× ...

## C - 18

•

CINDSL,CINCSL 144 IF(KON(7).NE.0) CALL OUTCAL 144 CONTINUE WRITE(6,882) WRITE(6,882) LAX WRITE(6,885) LAX 150 KON(37) = N2 150 KON(37) = N2 150 KON(37) = N2	CON(30) = $RLXD$ CON(37) = N1 CON(37) = 0.0 CON(37) = 0	175       F(NSOI(J).67.0) 60 TO 165         195       CONTINE         195       CONTINE         COLL VARUL2       AB5(aIN-GOUT)         CON(13)       CON(1)         CALL VARUL2       CON(1)         CALL OUTCAL       WITE(6.083)         WRITE(6.083)       KON(20).CON(32)         WRITE(6.083)       KON(20).S         MITH       EL1         NDIM       MAITE(6.084)         GO TO 1000       GO TO 5         GO TO 1000       GO TO 5         GO TO 1000       GO TO 1000	997 WRITE(6.887) 60 T0 1000 999 WRITE(6.888) 60 T0 1000 999 WRITE(6.888) 60 TO LOUD 60 CALL OUTCAL 1000 CALL OUTCAL CALL EXIT 882 FORMAT(10H LOOPCT = 16.10H ENGBAL = E12.5) 883 FORMAT(10H LOOPCT = 16.10H ENGBAL = E12.5) 883 FORMAT(10H LOOPCT = 16.10H ENGBAL = E12.5) 884 FORMAT(10H LOOPCT = 16.10H ENGBAL = E12.5) 885 FORMAT(10H LOOPCT = 16.10H ENGBAL = E12.5) 886 FORMAT(10H LOOPCT = 16.10H ENGBAL = 17.10H ENGBAL = 110)
15224	1555 1556 1556 1556 1557 1657 1657 1657		10000000000000000000000000000000000000
00663 00665 00665 000667 00074 00074 00074 00074 00074	000111640000000000000000000000000000000	00017335 0007335 00073335 00073335 00073335 00073335 0007335 0007335 0007335 0007335 000735 000735 000735 000735 000735 0000735 000735 0000000000	

•

٠

•

·••

•••

C - 19

•

.•

•

•

CINDSM

LAI FOR'+ CINDSM UNIVAC 1108 FORTRAN V ATHENA VERSION 131K-10D CREATED ON 20 AUG 70 THIS COMPILATION WAS DONE ON 09 JUN 70 AT 22:16:55

SUBROUTINE CINDSM ENTRY POINT 001175

STORAGE USED (BLOCK, NAME, LENGTH)

001211 000067 000044	000000		یند. مع مراجع
K MP	000001		000000 000000 000000
+CODE +CONST+T +SIMPLE	+ARRAYS +BLANK TITLE TEMP CAP	SOURCE COND PC1 PC2 KONST	ARRAY FIXCON XSPACE DIMENS
1000	0002 0005 0005 0005	0011 0012 0013 0014 0015	0016 0017 0020 0021

EXTE

•

		•						
NAME)								
(BLOCK,	•							
ERENCES	NONLIN	OUTCAL	VARBL2	EXIT	NERR25	SUDWN SUDWN	N1025	NER105
ERNAL REF	0022	0023	0024	0025	0026	0027	0030	1500

٠ . STORAGE ASSIGNMENT FOR VARIABLES (BLOCK, TYPE, RELATIVE LOCATION, NAME)

...

і. Яс. А.,

•

• •

÷

. مۇرۇرۇر

2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 200 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2000 2	3416
000001110000013	000603
00000000000000000000000000000000000000	1000
0 10 10 10 10 10 10 10 10 10 10 10 10 10	206
00000000000000000000000000000000000000	000507 3
2 7 11 1 1 1 1 1 1 2 2 2 2 2 2 2 2 2 2 2	1000
42084 C 2 1984 X	
000000 CO 000001 IE 000001 IE 000001 IE 000001 IE 000001 IE 000001 IE 000001 IE 000001 IE 000001 II 000001 II 000001 II 000001 II 000001 II 000000 II 0000000 II 000000 II 000000 II 000000 II 0000000 II 0000000 II 00000000	00162 30
0017 R 0002 R 00022 R 00022 R 00022 R 00022 R 00022 R 00015 R 00015 R 00015 R 00010 R 00010 R 00010 R 00010 R 00010 R 00010 R 00010 R 00000 R 0000 R 00000 R 00000 R 0000 R 000 R 0000 R 000	0001 0
	16
00000 C 00005 C 00005 C 00055 C 00055 C 00055 C 00052 C 00052 C 00552	0332 25
10000000000000000000000000000000000000	001 00
<u>N</u>	
11111111111111111111111111111111111111	251
	000157
00000000000000000000000000000000000000	1000

21 DEC 70

C - 20

•

÷	÷	•		
	S (NOCOPY) S (1-J-K-L-MTEST) S (1-J-K-L-MTEST) TS (R-S-T-U-VTEST) NDSM (LAXFAC)			ARIZE RADIATION
• ••••	THE NOCOPY SWITCH FOR MATRIX DSER RELATIVE NODE NUMBER OF CSGMIN RELATIVE NODE NUMBER OF DTMPCC RELATIVE NODE NUMBER OF ARLXCC RELATIVE NODE NUMBER OF ARLXCC 43 CONTAIN DUMMY INTEGR CONSTANT 48 CONTAIN DUMMY FLOATING CONSTANT 48 CONTAIN DUMMY FLOATING CONSTANT 48 CONTAIN DUMMY FLOATING CONSTANT 48 CONTAIN DUMMY FLOATING CONSTANT 40 CONTAIN CONTAUT 40 CONTAIN DUMMY FLOATING CONSTANT 40 CONTAIN CONTANT 40 CONTAIN DUMMY FLOATING CONSTANT 40 CONTAIN CONTAUT 40 CONTAIN CONTAUT 40 CONTAIN CONTAUT 40 C	(0) CON(9) = 1.0 (0.0) CON(10) = 1.0 (1) GO TO 998 (0) GO TO 996 (0) GO TO 995 (0) TO 995	3) 13) 3)+CON(18) 3)+CON(18) (1)+CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON(13) )-CON	HEARITIES 1 GO TO 25 1 SAVE ALL TEMPERATURES AND LINE
CINDSM	CC34 CONTAINS CC35 CONTAINS CC35 CONTAINS CC37 CONTAINS CC37 CONTAINS CC39-40-41AINS CC39-40-47AINS CC49 IS THE 00 CC49 IS THE 00 CC49 IS NOT US	IF (CON(9).LE. IF (CON(10).LE. IF (CON(33).LE IF (CON(33).L	IEST = IEET = IE	UDELXXX       UDELXXX         XXXDUM       0.001         EVALUAME       0.001         EVALUAME       0.001         RELAX       XXX         RELAX       XXX         DAMP       - CONLIN         IF(LPASS.61AH)       - 1         CALL       OUTCAL         IF(LPASS.61AH)       - 1         CALL       OUTCAL         SOUM       ALL         OIN       0         OIN       0         OIN       0         IF(LS       - 1         IF(LPASS.61AH)       - 1
	••••••••••••••••••••••••••••••••••••••	10111111111111111111111111111111111111	L	C NWJWWWWWWWC WWJTTTTTTTTTTTTTTTTTTTTTTTTTT
3°'	<b>5000000000000000000000000000000000000</b>	00117		

den for en den de de service en la recente de la desta de la de Novembre de la desta de la d Novembre de la desta de la d

C - 22

•

N 464 .

CINDSM

SWITCH CONDUCTOR VALUES WITH FOURTH EXTRA ARRAY IN X DO 55 I = 1,NGT Le4 = IE4+I SAVE THE MAXIMUM TEMPERATURE CHANGE AND LOCATION IF (RLXD.6T.TI) 60 TO 85 RLXD = TI NI = I C CHECK FOR RADIATION CONDUCTOR LF4 = IE4+16 C CHECK FOR RADIATION CONDUCTOR LT7 = FLD(22,14,NS01(J1)).56,0) 60 T0 40 LT1 = T(1)+460.0 T1 = T(1)+460.0 T2 = T(LT4) = 6(L6) t0 T(LE4) = 6(L6) t0 T(L64) = 6( JI = J1+1 LG = FLD(5,16,NSQ1(J1)) 6(I) = 6SUM (1) = 6SUM (1) = 6SUM X(LE3) = T(I) 32 u U u ت u 60<del>+</del> 62+ 63# 64# 66* +69 744 85# 68+ 134 *05 101+ 102+ 65# 72# 81# 82# 84* 91* 92# *26 # 116 95* *001 105+106+107+ 111* 113* 83* 87.* **#**96 *66 104* 108* 109* 110+ 1.12* 114+ 88 97. 98* 115# 116 117* 118+ 1194 00210 00211 00212 00213 00213 00301 00302 00303 00303 00256 00260 00261 00261 00262 00264 00264 00265 00266 00266 00270 00271 00272 00273 00275 00276 00277 00200 00306 40200

12 CUNITIONE THE MUM TEMPERATURE CHANGE OVER THE GUASI-INTERVAL EXCEEDS IF THE MULT RELAX. REDUCE CRITERIA AND PERFORM MORE ITERATIONS IF(GSUM.LE.RELAX) 60 TO 130 REDUCE LAXFAG TO THE DIFFERENCE RETWEEN NLOOP AND LOOPCT. SO THAT NLOOP REMAINS THE MAXIMUM NUMMER OF ITERATIONS POSSIBLE LAXFAG = KON(5)-KON(20)
12B LAXFAG = KON(5)-KON(20)
12B LAXFAG = KON(5)-KON(20)
12B LAXFAG = KON(5)-KON(20)
13D N2
13D N2
13D N2
13D N2
141
13J = JI+I CONTINUE SEE IF THE RELAXATION CRITERIA WAS MET IF(JJLE.2) G0 T0 95 IF(JJLE.2) G0 T0 95 JJ = 0 D0 90 I = 1.NNC LE1 = IE1+I LE2 = IE2+I SEE IF THE EXTRAPOLATION CRITERIA ARE MET IF(X(LE2).GE.0.0.0R*ABS(X(LE2)*T(I)-X(LE1))).GE.RLXD) G0 T0 90 T(I) = X(LE2)*X(LE1)+(1,0-X(LE2))*T(I) CONTINUE CHECK TO SEE IF MAXIMUM NUMBER OF ITERATIONS HAS BEEN EXCEEDED IF(KON(20).GE.KON(5)) 60 TO 130 Has not been exceeded, reduce relax to 0.001 100 CONTINUE 105 KON(20) = KON(20)+KK 2 STORE CONDUCTANCE VALUES BACK IN THE G ARRAY 20 110 I = 1,NGT 10 G(I) = X(LE4) C HECK IF THE INITIAL NLAX ITERATIONS HAVE BEEN PERFORMED 110 G(I) = X(LE4) C HECK IF THE INITIAL NLAX ITERATIONS HAVE BEEN PERFORMED 110 G(I) = X(LE4) C HECK IF THE INITIAL NLAX ITERATIONS HAVE BEEN PERFORMED 110 G(I) = 1,NC 2 APPLY DAMPING FACTOR AND REDUCE RELAX BY ONLY 0.005/NLAX 115 T(I) = 1,NNC 115 T(I) = 1,NNC 115 T(I) = 1,NNC C AFTER THE NLAX INITIAL ITERATIONS, REDUCE RELAX BY ONLY 0.005/NLAX C AFTER THE NLAX INITIAL ITERATIONS, REDUCE RELAX BY 0.001 115 T(I) = DAMPING C AFTER THE NLAX INITIAL ITERATIONS, REDUCE RELAX D 0.001 115 T(I) = DAMPING FACTOR AND REDUCE RELAX BY ONLY 0.005/NLAX C AFTER THE NLAX INITIAL ITERATIONS, REDUCE RELAX D 0.001 GSUM = 0.0 OBTAIN THE MAXIMUM TEMPERATURE CHANGE DO 125 I = 1.NNC LE3 = I3.H LE3 = I53.H IF(qoUT = ABS(T(1)-X(LE3)) IF(qoUT = GSUM = QOUT 90 CONTINUE 95 CON(30) = RLXD KON(37) = N1 IF(KON(7).NE.0) CALL OUTCAL XXX = XXXDUM 6SUM = 0.0 CINDSM CONTINUE CONTINUE 0 ii II Ň Ž 50 125 128 130 120 135 Ų a U υ U.U v U u U υü Ú 151+ キコロ 135+ 136# 1334 146# 147* 161+ 162* 163# 164+ 165+ 168* +691 121+ 11* 132* +001 *071 *** 145* *611 155* 156+ 157* 159* 160* 167* 171+ 173+ 148+ 150# 153+ 154+ 158+ 166* 173# * 7 2 3 175* 176* 177* 72# 003210 00316 00317 00317 00322 003223 003224 003224 003224 00333 00400 00405 0.0327 00403 10100

÷

ч. л Т CON(32) = ABS(nJH-00UT) GO TO (160,155), N2 IF(CON(32),LE.CON(33)) GO TO 160 NLOOP HAS NUT BEEN EXCEEDED, MAXIMUM TEMPERATURE CHANGE IS LESS NLOOP HAS NUT BEEN EXCEEDED, MAXIMUM TEMPERATURE CHANGE IS LESS NLOOP HAS NUT BEEN EXCEEDED, MAXIMUM TEMPERATURE CHANGE IS LESS NLOOP HAS NUT BEEN EXCEEDED, MAXIMUM TEMPERATURE CHANGE IS LESS NLOOP HAS NUT BEEN EXCEEDED, MAXIMUM TEMPERATURE CHANGE IS LESS NLOOP HAS NUT TO RELAXATION CRITERIA BUT ENGBAL IS GREATER THAN THAN OR EVEL TO RELAXATION CRITERIA BUT ENGBAL IS GREATER THAN SXXXDUM = XXXDUM/5.0 XXX = XXXDUM/5.0 CO TO 128 EITHER NLOOP HAS BEEN EXCEEDED OR ENGBAL IS LESS THAN OR EGUAL TO BALENG. IN EITHER CASE PRINT LOOPCT AND ENGBAL. INCREMENT TIME AND PROCEEC WITH THE PROBLEM. (0 WRITE(6.084) KON(20).CON(32) THE PROBLEM. CON(28) = KON(20).CON(32) KON(27) = KLXD KON(37) = KLXD KON(37) = NLXD KON(37) = NLXD KON(37) = CON(1) CALL VARBL2 CON(13) = CON(1)*1.000001) GO TO 5 IF(CON(3).GT.CON(1)*1.000001) GO TO 5 NTH = IE1 LTA = FLD(22.14.NS31(J1)) C CHECK FOR BOUNDATY NODE IF(LTA.LE.NNC) 60 T0 145 LG = FLD(5,15.NS01(J1)) LG = FLD(5,15.NS01(J1)) C CHECK FOR RADIATION COMDUCTOR IF(FLD(3.1.NS01(J1)).EQ.0) G0 T0 140 T1 = T(17)+460.0 T2 = T(LTA)+460.0 T2 = T(LTA)+40.0 T2 = T(LTA)+70.0 T2 = T(LTA)+70.0T2 = T(LTA)+70.0 T2 = T(LTA)+70.0T2 = 
 BBU
 FORMAT(10H0LOOPCT = 16.10H ENGBAL = E12.5)
 BBS
 FORMAT(10H NO LAXFAC)
 BBS
 FORMAT(10H NO NLOOP)
 FORMAT(10H NO NLOOP)< WRITE(6,885) NDIM 60 T0 1000 60 WRITE(6,896) 60 T0 1000 997 WRITE(6,887) 998 WRITE(6,887) 998 WNIE(6,889) 60 TO 1000 999 WRITE(6,889) 1000 CALL OUTCAL NDIM = NLA CINDSM CALL EXIT RE TURN 0 Z Z Z 966 155 160 366 υ U U U U Ų Ú 000 **** 50000 50000 205+ 884 894 1904 191* ******** 204* 214* 228* 2344 235* 180+ 00413 00413 00414 00416 00421 00436 00441 00445 00450 00452 00453 00455 00455 00455 09400 00463 00466 00472 00475 00503 00200 01100 04400 01100 00200 00502 00505 00506 00501 00504

END OF UNIVAC 1108 FONTRAN V COMPILATION. 0 *DIAGNOSTICS MESSAGE(S)